INFORMATION TO USERS

This manuscript has been reproduced from the microfilm master. UMI films the text directly from the original or copy submitted. Thus, some thesis and dissertation copies are in typewriter face, while others may be from any type of computer printer.

The quality of this reproduction is dependent upon the quality of the copy submitted. Broken or indistinct print, colored or poor quality illustrations and photographs, print bleedthrough, substandard margins, and improper alignment can adversely affect reproduction.

In the unlikely event that the author did not send UMI a complete manuscript and there are missing pages, these will be noted. Also, if unauthorized copyright material had to be removed, a note will indicate the deletion.

Oversize materials (e.g., maps, drawings, charts) are reproduced by sectioning the original, beginning at the upper left-hand corner and continuing from left to right in equal sections with small overlaps.

Photographs included in the original manuscript have been reproduced xerographically in this copy. Higher quality $6^{\circ} \times 9^{\circ}$ black and white photographic prints are available for any photographs or illustrations appearing in this copy for an additional charge. Contact UMI directly to order.

ProQuest Information and Learning 300 North Zeeb Road, Ann Arbor, MI 48106-1346 USA 800-521-0600

UMI®

NOTE TO USERS

This reproduction is the best copy available.

UMI

MULTIMODAL, ASYNCHRONOUSLY SHARED SLIDE SHOWS AS A DESIGN STRATEGY FOR ENGINEERING DISTRIBUTED WORK IN THE NATIONAL AIRSPACE SYSTEM

DISSERTATION

Presented in Partial Fulfillment of the Requirements for

the Degree Doctor of Philosophy in the Graduate

School of the Ohio State University

By

Roger James Chapman, B.S, M.S.

The Ohio State University 2002

Dissertation Committee:

Professor Philip J. Smith

Professor Charles E. Billings

Professor Nadine Sarter

pproved by Adviser

Department of Industrial & Systems Engineering Graduate Program

UMI Number: 3039456

UMI®

UMI Microform 3039456

Copyright 2002 by ProQuest Information and Learning Company. All rights reserved. This microform edition is protected against unauthorized copying under Title 17, United States Code.

> ProQuest Information and Learning Company 300 North Zeeb Road P.O. Box 1346 Ann Arbor, MI 48106-1346

Copyright by

Roger James Chapman

2002

ABSTRACT

This research focused on the design and evaluation of a multimodal asynchronous communications tool that uses an annotated slide show structure for messages, to support collaborative analysis of post-operations in the National Airspace System (NAS). Voice-based asynchronous annotations communicated between writers working collaboratively have been found to support more extensive discussion of problems in draft documents and to be preferred over text based discussion for some types of feedback. People point naturally when discussing the content of a shared image, and voice synchronized with pointing in asynchronous annotation systems has been found to be more efficient in scheduling tasks, than voice-only, or text only communication. Synchronized voice and pointing has also been shown to focus attention and improve retention of information in multimedia presentation systems. This research involved investigating how different communication modes interact with problem solving and the discussion of distributed information, with a view to improving collaborative post operations analysis within and between the Federal Aviation Administration (FAA) and the major airlines in the United States. More specifically how synchronized voice and pointing annotation over asynchronously shared slide shows composed of post operations graphical and tabular data differs in its effect compared to text based annotation, as collections of flights ranked low by standard performance metrics are discussed by FAA and airline representatives.

ü

A part-task simulation study had 36 AOC dispatchers, chief dispatchers or ATC coordinators from Northwest Airlines communicate with 36 ARTCC (Air Route Traffic Control Center) traffic managers at eight different Centers about performance issues for flights between 9 different city pairs. Each Northwest Airlines participant communicated with one FAA participant, in a one-to-one pairing. A separate slide show consisting of screen captures showing post operations data was created for each city pair. For each slide show, two Northwest Airlines participants were asked to annotate the slides using text, pen marks or arrows, and two were asked to annotate the slides using synchronized voice and pointing, pen marks and arrows. For each slide show, four traffic managers were asked to respond to the annotated message using the same annotation tools the sender used.

The results from this research show that:

- The communications mode failed to show an effect on the number and type of ideas generated for improving performance.
- 2. The most common strategy proposed was a change of route.
- Almost all the dispatchers' ideas were responded to by traffic managers in both modes: 27 out of 27 for text mode participants and 27 out of 29 for voice and pointing mode participants.
- 4. The dispatchers' combined problem solving and message creation time was shorter (ANOVA, $\alpha < 0.05$) when working in the voice and pointing mode than the text based mode, as was the dispatcher-traffic manager pairs' total combined problem solving and message creation time (ANOVA, $\alpha < 0.01$). The average task time was 18 minutes 1 second for the voice and pointing mode dispatchers and 28 minutes for the text mode dispatchers (a difference of 9 minutes 59 seconds or 55%). The average dispatcher-traffic manager pair task time was 35

üi

minutes and 41 seconds for the voice and pointing mode and 53 minutes and 22 seconds for the text mode. This is a difference of 17 minutes 41 seconds or 50%.

- 5. In both communication modes the dispatchers normally did not comment on the time of day, time of year, or the number of flights involved on the first slide they made a comment, even though that information was always in the image. The city pair involved was also only mentioned by 5 out of 18 dispatchers working in the text mode and 11 out of 18 dispatchers working in the voice and pointing mode. There was a difference between the two modes in this regard with more dispatchers mentioning each of these categories in the voice and pointing mode (although only the difference between the number mentioning the city pair and time of year were statistically significant at $\alpha < 0.05$). This is significant because this information contributes towards diagnosing the cause of the flights per performance, and some participants indicated that by stating it was difficult to solve the problem without some of this information.
- 6. 18/18 (100%) of the dispatchers in the voice and pointing mode made deictic gestures during the creation of their messages. 17/18 (94%) of the traffic managers responding in this mode also made deictic gestures.
- 7. 15/18 (83%) of the dispatchers and 9/18 (50%) of the traffic managers working in the voice and pointing mode still created static annotations to the objects or locations they referenced with deictic gestures.
- 9/15 (60%) of the dispatchers and 9/9 (100%) of the traffic managers working in the voice and pointing mode created static annotations while speaking, as opposed to before or after speaking.
- Dispatchers and traffic managers considered the system they used to be usable and valuable, but dispatchers using the text mode rated the usability and utility

of the system higher on average than dispatchers using the voice and pointing mode, whereas for traffic managers the reverse was true.

- 10. The slide image provided a context for comments, but the slide show structure used created a loss of temporal context and comments regarding multiple slides and slides other than the one the comment was placed on meant a slide could not be treated as a disjoint entity. These problems were addressed with a redesign idea to support these contexts.
- 11. The comments and annotations made provide a vocabulary of syntax and semantics for a more task specific version of the system. Such a system would be tailored for practitioners to work with in this domain, but should be implemented through configuration options to maintain the integrity of the system as a domain free system.

ACKNOWLEDGMENTS

I would like to express my thanks to my adviser, Dr. Philip J. Smith, for his guidance throughout my graduate work at Ohio State. I would also like to thank my other committee members, Dr. Charles Billings and Dr. Nadine Sarter, and also Dr. David Woods who has always been willing to share his insight and knowledge of human factors with me. Northwest Airlines and the Federal Aviation Administration made this research possible with their participation. In particular, I would like to thank Charles Bailey, Traffic Management Officer for the New York ARTCC, for his willingness to discuss the National Airspace System with me and for his encouragement for my work. Finally, I would like to acknowledge the support of my family and friends for their faith, and patience.

1

VITA

March 22, 1962	Born, Baldock, England
1986	B. S., Computer Science summa cum laude Eastern Kentucky University Richmond, Kentucky
1987	M.S., Computer Systems Engineering University of Edinburgh Edinburgh, Scotland
1988 - 1990	Instructor of Computer Science Transylvania University Lexington, Kentucky
1990 - 1995	Associate Professor of Computer Science Rocky Mountain College Billings, Montana
1993	Faculty Fellowship Los Alamos National Laboratory Los Alamos, New Mexico
1994, 1995	Faculty Fellowship Oak Ridge National Laboratory Oak Ridge, Tennessee
1996 - 2000	Graduate Research Associate Industrial & Systems Engineering The Ohio State University
2000 - 2001	Assistant Professor of Computer Science University of Hawaii, Hilo

PUBLICATIONS

1. Chapman, R. J., Smith, P. J., Billings, C. E., McCoy C. E., and Obradovich J. H. (2001). Collaborative Constraint Propagation as a Planning Strategy in the National Airspace System. *Proceedings of the 2001 Annual Meeting of the IEEE Society on Systems, Man and Cybernetics*. Tucson, AZ, 3484-3490.

2. Smith, P. J., Billings, C., Chapman, R. J., Obradovich J. H., McCoy, C. E. (2001). Information Management to Support Distributed Decision-Making in the National Airspace System. *Proceedings of the Human Factors and Ergonomics Society* 45th Annual Meeting, Minneapolis, MN, 54-57.

3. Chapman, R. J., Smith, P.J., Klopfenstein, M., Jezerinac, J., Obradovich, J., McCoy, C.E., and Orasanu, J. (2000). Supporting collaboration in the National Airspace System with multimodal, asynchronous communications. *Proceedings of the 2000 Conference on Human Performance, Situation Awareness and Automation: User-Centered Design for the New Millennium.* Savannah, GA, 324-329.

4. Chapman, R. J., Smith, P.J., Klopfenstein, M., Jezerinac, J., Obradovich, J., and McCoy, C.E. (2000). CSLANT: An asynchronous communication tool to support distributed work in the National Airspace System. *Proceedings of the 2000 Annual Meeting of the IEEE Society on Systems, Man and Cybernetics*. Nashville, TN, 1069-1074.

5. Smith, P. J., Billings, C., Chapman, R. J., Obradovich, J., McCoy, E. and Orasanu, J. (2000). Alternative Architectures for Distributed Cooperative Problem-solving in the National Airspace System. Proceedings of *Human Interaction with Complex Systems (HICS)* 2000, Urbana-Champaign, IL, 203-207. IEEE Computer Society Press.

6. Smith, P. J., Billings, C., Chapman, R. J., Obradovich, J., McCoy, E. and Orasanu, J. (2000). Alternative "Rules of the Game" for the National Airspace System. *Proceedings of the Fifth Conference on Naturalistic Decision Making*. Tammsvik, Sweden.

7. Obradovich, J. H., Smith, P. J., Denning, R., Chapman, R. J., Billings, C., McCoy, E. and Woods, D. (1998). Cooperative problem-solving challenges for the movement of aircraft on the ground. *Proceedings of the Human Factors and Ergonomics Society* 42 *Annual Meeting*, 57-61.

FIELD OF STUDY

Major Field: Industrial & Systems Engineering

Concentration: Cognitive Systems Engineering

TABLE OF CONTENTS

Page
ABSTRACTii
ACKNOWLEDGMENTSvi
VITAvii
FIELD OF STUDYix
LIST OF FIGURES
LIST OF TABLES
CHAPTER 11
INTRODUCTION1
Document Overview2
CHAPTER 2
COMPUTER SUPPORTED COOPERATIVE AND DISTRIBUTED WORK
Introduction3
The Time Space Matrix4
Task and Environment Dimensions7
An Application-Level Technology8
Perspectives
Distributed Systems and Distributed Work20
CHAPTER 3
TEXT-BASED TO MULTIMODAL-BASED ASYNCHRONOUS COMMUNICATIONS
Asynchronous distributed interaction22
Non-computer supported asynchronous communications

Computer supported asynchronous communications	23
Electronic Mail	24
Multimodal Asynchronous Messaging and Annotation	26
Messages and Annotations	26
Voice, Text and Graphics	27
Deictic Gesturing	36
Drawing Tools	43
Popular Software with Features for Annotation	44
Microsoft Office 2000	44
E-mail systems with html formatted content	51
RealNetwork's RealSlideshow	52
CHAPTER 4	55
COLLABORATIVE ANALYSIS OF NATIONAL AIRSPACE SYSTEM POST	
OPERATIONS DATA AS A TEST BED	55
OPERATIONS DATA AS A TEST BED	55 55
OPERATIONS DATA AS A TEST BED Introduction The National Airspace System	55 55 55
OPERATIONS DATA AS A TEST BED Introduction The National Airspace System Agents	55 55 55 56
OPERATIONS DATA AS A TEST BED Introduction The National Airspace System Agents Goals	55 55 55 56 59
OPERATIONS DATA AS A TEST BED Introduction The National Airspace System Agents Goals Matching Data with Control	55 55 55 56 59 59
OPERATIONS DATA AS A TEST BED Introduction The National Airspace System Agents Goals Matching Data with Control Communications	55 55 56 59 59 61
OPERATIONS DATA AS A TEST BED Introduction The National Airspace System Agents Goals Matching Data with Control Communications The Post Operations Evaluation Tool	55 55 56 59 61 62
OPERATIONS DATA AS A TEST BED Introduction The National Airspace System Agents Goals Matching Data with Control Communications The Post Operations Evaluation Tool	55 55 56 59 59 61 62 62
OPERATIONS DATA AS A TEST BED Introduction The National Airspace System Agents Goals Matching Data with Control. Communications The Post Operations Evaluation Tool Introduction Conducting a Search.	55 55 56 59 61 62 62 63
OPERATIONS DATA AS A TEST BED Introduction The National Airspace System Agents Goals Matching Data with Control Communications The Post Operations Evaluation Tool Introduction Conducting a Search Data Mining	55 55 56 59 61 62 62 63 63
OPERATIONS DATA AS A TEST BED Introduction The National Airspace System Agents Goals Matching Data with Control Communications The Post Operations Evaluation Tool Introduction Conducting a Search Data Mining POET Results	55 55 56 59 61 62 62 62 63 67 72

CHAPTER 5	86
THE DESIGN OF THE COLLABORATIVE SLIDE ANOTATION TOOL (C-SLANT)	86
Introduction	86
Requirements	88
Interface Design Features	91
The Slide/Annotation Manager Window	93
Annotation Tools	94
Synchronized speech and pointing	94
Text boxes	98
Ink and highlighter freehand pens	99
Arrow markers	100
Abstraction from the file system	100
Slide Sequencing Window	100
Architecture	101
Implementation	102
Software Development Environment	102
Audio Format	103
Compression	103
Summary	104
CHAPTER 6	105
EXPERIMENTAL PROCEDURE	105
Subjects	105
Experimental Design	106
Scenarios used	109
Procedure	115
Data collection	121

Data Analysis	
CHAPTER 7	
RESULTS AND DISCUSSION	
Introduction	123
Discourse and Problem Solving Analysis	
Dispatcher task times	125
Dispatcher situation assessments	135
Dispatcher direct questions	139
Dispatcher ideas for improving performance	141
Traffic manager task times	145
Traffic manager responses to the dispatchers' performance questions a	nd
ideas	153
Other information exchanged	157
Analysis of Annotation Modes	158
Speech and pointing	158
Text comments	173
Pen marks	175
Arrows	177
Annotation mode interactions	178
Annotation of POET windows	180
Annotation of Search Summaries	180
Annotation of Scatter Charts	
Annotation of Tables	
Annotation of Maps	183
Within Slide Window Management and Navigation	187
Between Slide Navigation	190

Analysis of Implementation Issues	193
File size and transmission time	194
Questionnaire Results	196
Participants Profile	197
Participants use of E-mail	198
Perceived usability and value of C-SLANT	200
Participant suggested enhancements	206
Traffic Managers' critique of dispatcher messages	208
Structured Interview with the ZNY TMO	209
Discussion of Results	213
Comparison between the text and voice and pointing modes	213
Aviation context issues	214
Further Enhancements to C-SLANT	216
CHAPTER 8	240
SUMMARY AND CONCLUSIONS	240
Introduction	240
Summary of results	240
Problem solving summary	240
Context referencing summary	247
Conclusions	252
Research context summary	252
Application of this research	256
LIST OF REFERENCES	263
Appendix A. SLIDE SHOW SCENARIOS	270
Appendix A. SLIDE SHOW SCENARIOS	270 293

Appendix D. DISPATHCHERS M	ENTIONING BASIC FLIGHT CLASSIFYING DATA	300
Appendix E. DISPATHCHER QU	ESTIONS	302
Appendix F. IDEAS FOR IMPROV	/ING PERFORMANCE	306
Appendix G. DISPATCHERS' PRO	OPOSED ROUTE DESCRIPTIONS	323

LIST OF FIGURES

<u>Figure</u> Page
Figure 2.1 Two dimensions of the groupware spectrum (Ellis, Gibbs, and Rein, 1991)7
Figure 2.2 A Winograd-Flores Structure for Action-Oriented Conversations based on linguistic
acts14
Figure 3.1 Asynchronous communication methods in order of development (Daly-Jones et al.,
1997)
Figure 3.2 The architecture of Paivio's dual-coding theory (Ware, 2000)
Figure 3.3 Designing connections between text and annotations (Faraday & Sutcliffe, 1998)34
Figure 3.4 MRAS "Add New Annotation" dialog box (Bargeron, Gupta, Grudin & Sanocki, 1999).
Figure 3.5 Actions for each session (Bly, 1988)
Figure 3.6 Freestyle image desktop; handwriting and voice-annotation tool (Francik, 1996)
Figure 3.7 A comment in Microsoft Word
Figure 3.8 Comment management in Microsoft Word
Figure 3.9 A reference to an object in Microsoft Word made explicit
Figure 3.10 A comment in Microsoft Excel
Figure 3.11 An html formatted E-mail message
Figure 3.12 A RealSlideshow slide show53
Figure 3.13 An audio caption in RealSlideshow
Figure 4.1 5,050 aircraft simultaneously operating in the NAS
Figure 4.2 A concept of the traditional air traffic management operational process (Smith, et al.,
1999)

63
66
68
70
72
73
74
76
77
78
79
30
81
82
83
87
92
93
94
95
96
97
98
9 8
99
.01

Figure 5.12 C-SLANT asynchronous communications via E-mail	.102
Figure 6.1 Enroute Centers	.106
Figure 6.2 Equipment setup	.122
Figure 7.1 Dispatcher task time distribution	.127
Figure 7.2 Normal probability plot of the residuals for dispatcher task time	.130
Figure 7.3 Residuals versus the fitted values for dispatcher task times	.131
Figure 7.4 Main effects plot for dispatcher task times	.133
Figure 7.5 Interaction plot for dispatcher task times	.134
Figure 7.6 Loss of time of day information	.136
Figure 7.7 Traffic manager task time distribution	146
Figure 7.8 Dispatcher + traffic manager task times	.148
Figure 7.9 Normal probability plot of the residuals for dispatcher + traffic manager task times	.150
Figure 7.10 Residuals versus the fitted values for dispatcher + traffic manager task times	151
Figure 7.11 Main effects plot for dispatcher + traffic manager task times	152
Figure 7.12 Interaction plot for dispatcher + traffic manager task times	152
Figure 7.13 Example of a verbal error during pointing	161
Figure 7.14 Dispatcher recording lengths	167
Figure 7.15 Dispatcher slide show recording lengths	169
Figure 7.16 Dispatcher and traffic manager slide show recording lengths	170
Figure 7.17 Text boxes left covering significant data	174
Figure 7.18 Example of confusion between controls in the image and C-SLANT	188
Figure 7.19 Poor arrangement of player windows	189
Figure 7.20 C-SLANT's recordings window	218
Figure 7.21 The SoundBrowser window (Degan, et. al., 1992)	226
Figure 7.22 Example of the demonstrated need for "labels"	228
Figure 7.23 Example of the need for more conhisticated arrow annotation tools	231

Figure 7.24 C-SLANT asynchronous communications via the Web	238
Figure 7.25 Java based applications (adapted from Slack, 2000)	239
Figure 8.1 Main effect of the mode on dispatcher time	245
Figure 8.2 Main effect of the mode on dispatcher+traffic manager time	246
Figure 8.3 Factors influencing system acceptability (Neilson, 1993)	256

LIST OF TABLES

Table	<u>Page</u>
Table 2.1 Groupware time space matrix (Ellis, Gibbs, and Rein, 1991)	4
Table 2.2 Costs and benefits of synchronous and asynchronous communications (Daly-J	ones et al.,
1997)	5
Table 3.1 Basic asynchronous communications methods	28
Table 4.1 Changes in the distribution of control and data for traffic flow management in	the NAS
(adapted from Smith, et al., 2000)	60
Table 5.1 Requirements for C-SLANT	90
Table 5.2 Slide / annotation manager window buttons	93
Table 5.3 Basic implementation summary of C-SLANT	104
Table 6.1 Experimental design	108
Table 6.2 Scenarios summary	109
Table 6.3 Experimental procedure for each participant pair	115
Table 7.1 Dispatcher task time (in seconds) summary	126
Table 7.2 ANOVA for dispatcher task times	129
Table 7.3 ANOVA for dispatcher task times (with ZNY data removed)	132
Table 7.4 Number of dispatchers mentioning certain flight classifying data	135
Table 7.5 Number of direct questions asked by each of the 36 dispatchers	140
Table 7.6 Number of dispatcher ideas for performance improvement	142
Table 7.7 Classification of dispatcher ideas regarding performance improvement	142
Table 7.8 Count of dispatchers' proposed routes	143
Table 7.9 Count of dispatchers using pen or arrow marks with a proposed route	143

Table 7.55 Frequency of E-mail use	198
Table 7.34 Frequency of sending attachments	199
Table 7.35 Type of attachments sent	199
Table 7.36 Frequency of audio attachments	200
Table 7.37 Perceived usability of C-SLANT	200
Table 7.38 Perceived utility of C-SLANT	200
Table 7.39 Potential flight or airspace attributes for POET 1.0.	215
Table 7.40 Resign ideas to improve discourse structure	216
Table 7.41 Redesign ideas to improve problem solving	219
Table 7.42 Summary of C-SLANT based on Grice's Maxims of Relevance (1975)	220
Table 7.43 Summary C-SLANT based on guidelines by Wojahn, Neuwirth and Bullock (1998)	223
Table 7.44 Redesign ideas for improving voice and pointing based communications in C-SLA	NT
	224
Table 7.45 Redesign ideas for improving text based communications in C-SLANT	228
Table 7.46 Redesign ideas for improving visible annotations	
	230
Table 7.47 Redesign ideas for single slide operations	230 232
Table 7.47 Redesign ideas for single slide operations Table 7.48 Redesign ideas for slide navigation	230 232 233
Table 7.47 Redesign ideas for single slide operations Table 7.48 Redesign ideas for slide navigation Table 7.49 Resign ideas for message management	230 232 233 234
Table 7.47 Redesign ideas for single slide operations Table 7.48 Redesign ideas for slide navigation Table 7.49 Resign ideas for message management Table 7.50 Redesign ideas for low-level computer efficiency	230 232 233 234 236
Table 7.47 Redesign ideas for single slide operations Table 7.48 Redesign ideas for slide navigation Table 7.49 Resign ideas for message management Table 7.50 Redesign ideas for low-level computer efficiency Table 8.1 Research context summary	230 232 233 234 236 253
Table 7.47 Redesign ideas for single slide operations Table 7.48 Redesign ideas for slide navigation Table 7.49 Resign ideas for message management Table 7.50 Redesign ideas for low-level computer efficiency Table 8.1 Research context summary Appendix A.1 ZID: DTW to RDU Slide 1	230 232 233 234 236 253 270
Table 7.47 Redesign ideas for single slide operations Table 7.48 Redesign ideas for slide navigation. Table 7.49 Resign ideas for message management Table 7.50 Redesign ideas for low-level computer efficiency. Table 8.1 Research context summary. Appendix A.1 ZID: DTW to RDU Slide 1 Appendix A.2 ZID: DTW to RDU Slide 2	230 232 233 234 236 253 270 271
Table 7.47 Redesign ideas for single slide operations Table 7.48 Redesign ideas for slide navigation. Table 7.49 Resign ideas for message management Table 7.50 Redesign ideas for low-level computer efficiency. Table 8.1 Research context summary. Appendix A.1 ZID: DTW to RDU Slide 1 Appendix A.2 ZID: DTW to RDU Slide 2 Appendix A.3 ZID: DTW to RDU Slide 3	230 232 233 234 236 253 270 271 271
Table 7.47 Redesign ideas for single slide operations Table 7.48 Redesign ideas for slide navigation. Table 7.49 Resign ideas for message management Table 7.50 Redesign ideas for low-level computer efficiency. Table 8.1 Research context summary. Appendix A.1 ZID: DTW to RDU Slide 1 Appendix A.2 ZID: DTW to RDU Slide 2 Appendix A.3 ZID: DTW to RDU Slide 3 Appendix A.4 ZID: DTW to RDU Slide 4	230 232 233 234 236 253 270 271 271 271
Table 7.47 Redesign ideas for single slide operations Table 7.48 Redesign ideas for slide navigation Table 7.49 Resign ideas for message management Table 7.50 Redesign ideas for low-level computer efficiency Table 8.1 Research context summary Appendix A.1 ZID: DTW to RDU Slide 1 Appendix A.2 ZID: DTW to RDU Slide 2 Appendix A.3 ZID: DTW to RDU Slide 3 Appendix A.4 ZID: DTW to RDU Slide 4 Appendix A.5 ZID: DTW to RDU Slide 5	230 232 233 234 236 253 270 271 271 272 272

Appendix A.7 ZTL: DTW to ATL Slide 2	
Appendix A.8 ZTL: DTW to ATL Slide 3	
Appendix A.9 ZTL: DTW to ATL Slide 4	
Appendix A.10 ZTL: DTW to ATL Slide 5	
Appendix A.11 ZTL: DTW to ATL Slide 6	275
Appendix A.12 ZTL: DTW to ATL Slide 7	276
Appendix A.13 ZFW: MSP to DFW Slide 1	276
Appendix A.14 ZFW: MSP to DFW Slide 2	277
Appendix A.15 ZBW: DTW to BOS Slide 1	277
Appendix A.16 ZBW: DTW to BOS Slide 2	278
Appendix A.17 ZBW: DTW to BOS Slide 3	278
Appendix A.18 ZBW: DTW to BOS Slide 4	279
Appendix A.19 ZBW: DTW to BOS Slide 5	279
Appendix A.20 ZBW: DTW to BOS Slide 6	
Appendix A.21 ZDC: MEM to EWR Slide 1	
Appendix A.22 ZDC: MEM to EWR Slide 2	
Appendix A.23 ZDC: MEM to EWR Slide 3	
Appendix A.24 ZDC: MEM to EWR Slide 4	
Appendix A.25 ZDC: MEM to EWR Slide 5	
Appendix A.26 ZDC: MEM to EWR Slide 6	
Appendix A.27 ZMP: MDW to MSP Slide 1	
Appendix A.28 ZMP: MDW to MSP Slide 2	
Appendix A.29 ZMP: MDW to MSP Slide 3	
Appendix A.30 ZMP: MDW to MSP Slide 4	
Appendix A.31 ZMP: OMA to MSP Slide 1	
Appendix A.32 ZMP: OMA to MSP Slide 2	

Appendix A.33 ZMP: OMA to MSP Slide 3	286
Appendix A.34 ZNY: DTW to EWR Slide 1	
Appendix A.35 ZNY: DTW to EWR Slide 2	
Appendix A.36 ZNY: DTW to EWR Slide 3	
Appendix A.37 ZNY: DTW to EWR Slide 4	
Appendix A.38 ZNY: DTW to EWR Slide 5	289
Appendix A.39 ZNY: DTW to EWR Slide 6	289
Appendix A.40 ZOB: MSP to EWR Slide 1	290
Appendix A.41 ZOB: MSP to EWR Slide 2	290
Appendix A.42 ZOB: MSP to EWR Slide 3	291
Appendix A.43 ZOB: MSP to EWR Slide 4	291
Appendix A.44 ZOB: MSP to EWR Slide 5	292
Appendix C.1 Training Slide 1	295
Appendix C.2 Training Slide 2	296
Appendix C.3 Training Slide 3	296
Appendix C.4 Training Slide 4	297
Appendix C.5 Training Slide 5	297
Appendix C.6 Training Slide 6	298
Appendix C.7 Training Slide 7	298
Appendix C.8 Training Slide 8	299

xxiv

CHAPTER 1

INTRODUCTION

In recent years there have been a number of efforts under the Collaborative Decision Making (CDM) program (Metron, 2001) that have demonstrated the benefits of disseminating both knowledge and goals between organizations in the National Airspace System (NAS¹), and of collaboratively building decision support tools to assist in the interpretation of the large interrelated dataset produced. This research involved the development and evaluation of the Collaborative SLide ANnotation Tool (C-SLANT), an asynchronous communications tool that was developed to support discussion of data from the CDM product called the Post Operations Evaluation Tool (POET). It is suggested here that slide show based, multi-modal, messaging systems are valuable in this distributed work domain for asynchronous problem solving during post operations analysis, and in other situations where static images contain valuable information and there are knowledge/belief gaps between collaborators, and potential ambiguities in referencing the images.

As discussed in more detail in chapter 2, Ellis, Gibbs and Rein (1991) state that there are at least five key perspectives on groupware: distributed systems, communications, human-

¹ The FAA defines the NAS as "The common network of U.S. airspace; air navigation facilities, equipment and services, airports or landing areas" (http://www.fly.gov/Glossary_of_Terms/acronyms.html)

computer interaction, artificial intelligence, and social theory. The research presented here focuses on the first three perspectives.

Document Overview

Chapter 2 of this document provides a brief description of distributed work and Computer Supported Cooperative Work (CSCW) research, to help put the research presented here in context.

Chapter 3 describes previous research on multimodal asynchronous communication and annotation systems.

Chapter 4 describes the NAS, in general; distributed work and communications between those on the ground controlling aircraft, and managing airline operations; and the Post Operations Evaluation Tool.

Chapter 5 describes the design rationale for C-SLANT.

Chapter 6 describes the experimental procedures that were used to study the effect of using this multimodal communications tool for collaborative problem solving involving airline dispatchers and FAA traffic managers.

Chapter 7 analyses the results of the data collected and presents the redesign of some C-SLANT features.

Chapter 8 summarizes the significance of this research and implications for designing multimodal asynchronous communications in general.

2

CHAPTER 2

COMPUTER SUPPORTED COOPERATIVE AND DISTRIBUTED WORK

Introduction

The phrase "computer-supported cooperative work" was coined by Irene Greif and Paul Cashman in 1984 in the call for an invitation-only workshop on CSCW held in Cambridge, Massachusetts (Bannon and Schmidt, 1991). Bannon and Schmidt report that, in a panel discussion at the Conference on Computer-Supported Cooperative Work in 1988, Irene Grief commented that they coined the phrase partly as a shorthand way of referring to a set of concerns about supporting multiple individuals working together with computer systems. Seeking a more precise description, Baecker (1993) defines Computer-Supported Cooperative Work (CSCW) and groupware as follows:

Computer-supported cooperative work (Greif, 1988; Galegher, Kraut, and Egido, 1990; Greenberg, 1991) is computer-assisted coordinated activity such as problem solving and communication carried out by a group of collaborating individuals. The multi-user software supporting CSCW systems is known as *groupware*, although the latter term is sometimes broadened to incorporate the styles and practices in group process and dynamics that are essential for any collaborative activity to succeed, whether or not it is supported by computer. CSCW may also be viewed as the emerging scientific discipline that guides the thoughtful and appropriate design and development of groupware (Greenberg, 1991).

The Time Space Matrix



Table 2.1 Groupware time space matrix (Ellis, Gibbs, and Rein, 1991)

Most taxonomy of CSCW technologies distinguishes them in terms of their abilities to bridge time and to bridge space (Baecker, 1993). Table 2.1, is a typical time space matrix, for which Ellis, Gibbs and Rein (1991) provide the following examples:

Same Place, Same Time:	Meeting room
Same Place, Different Times:	Physical bulletin board
Different Places, Same Time:	Real-time document editor
Different Places, Different Times:	Electronic mail

Daly-Jones et al., (1997) attempt to summarize the costs and benefits of synchronous and

asynchronous systems, as shown in Table 2.2.

		Synchronous (e.g., telephone)	Asynchronous (e.g., voicemail)
Costs	To Initiator (owner of task)	Frequent failed connections waste time.	Slower pace of interaction makes composition harder work. Slower response may delay task closure. Have to re-instate context on receipt of reply
	To Recipient	Interrupts current task, have to re-instate context after interruption	Slower pace of interaction makes generating common ground harder work.
Benefits	To Initiator (owner of task)	If connection successful: Composition easier; Faster task closure; Context available on reply.	Failed connections not generally a problem
	To Recipient	Fast pace of interaction facilitates generation of common ground.	Fewer interruptions as when to deal with message is under recipient's control. More time to think about response.

 Table 2.2 Costs and benefits of synchronous and asynchronous communications (Daly-Jones et al., 1997)

In reality there are few purely synchronous and purely asynchronous systems. Similarly, while Ellis et al., indicate, "Groupware that specifically supports simultaneous activity is called real-time groupware; otherwise, it is non-real-time groupware", the definition would be more accurate if the wording was "attempts to support...". Some attempts at synchronous implementations have actually foundered because the synchronization desired was not achieved. Cognoter (Tartar, Foster and Bobrow, 1991), for instance, is an example because of delays in updating information shared between multiple workstations and synchronizing speech with that information. This system is discussed in more detail later in this chapter.

Systems are classified by <u>typical use</u>, because systems designed for asynchronous communications can actually be used relatively synchronously and vice-versa. For instance, two individuals may actually exchange a number of short E-mail messages with very small intervals of time between them, each person waiting for a response from the other between messages. Chat programs, normally classified as a synchronous communications systems, may similarly be adapted to more asynchronous use, if the system does not require the sender and recipient of messages to actually be online at the same time (e.g. Yahoo Messenger). In general, the synchronization required depends on characteristics such as the balance of information sharing, the urgency of the communication and feedback, how much time is needed to gather information for the next communication, and how acceptable it is to interrupt other tasks being performed by potential communications. There are also significant differences between both asynchronous and synchronous systems due to other factors that are part of the system, such as the nature of the information being communicated itself, and the knowledge, tasks, and goals of the communicating agents themselves.

Table 2.2 is perhaps more usefully viewed as no more than a set of characteristics often found with the synchronous and asynchronous aspects of a communications system. This way, all the costs can be considered as potential negatives and all the benefits as potential positives in a system being developed. Thus, for instance, the pace of interaction, support for establishing and re-instating context, the effort involved in establishing a common ground, and the impact on other work requirements are all issues regardless whether or not the system is labeled as synchronous or asynchronous.
Task and Environment Dimensions



Figure 2.1 Two dimensions of the groupware spectrum (Ellis, Gibbs, and Rein, 1991)

Ellis, Gibbs and Rein (1991) define groupware as:

"Computer-based systems that support groups of people engaged in a common task (or goal) and that provide an interface to a shared environment." (pg. 10).

Figure 2.1 shows that they consider systems to vary in the extent to which they can be considered to be groupware, and this can be determined by how much the system supports working towards a common goal and working in a shared environment. Thus, with scales like this it is possible to indicate to what extent an application can be considered groupware, but more important is whether the application supports the intended use. For instance, a shared environment is more important if that environment is relevant to the work. If the agents involved actually need to work with distinct data and only need to communicate the results of processing that data, all agents don't need to see all the data. Similarly, they may very appropriately have distinct tasks, so the right selection of groupware features is more important than the number of them.

An Application-Level Technology

Ellis et al present a second taxonomy based on common application functionality that demonstrates the breadth of the groupware domain:

Message Systems

The computer-based message system is the most common type of groupware, which in its simplest form just supports the asynchronous exchange of textual messages between groups of users. E-Mail (Electronic Mail) and Newsgroups or Bulletin Board Systems are particularly common. E-Mail systems will be discussed in more detail in Chapter 4.

• Multiuser Editors

Multiuser editors to support the joint composition and editing of a document can be useful to divide work and share ideas, but it means there must be some coordination of that work. If the work is simple enough it can be divided into disjoint subunits and worked on independently by different authors. Normally, however, it is not that simple and their has to be a system for passing control of a document, or portions of it, from one author to another while keeping track of who has done what and what the constraints are on each author for each session. Software may help in the process by "locking" resources and providing read or write access, conceptually the same as is done with database systems when they are shared between multiple users. Tracking information may also be maintained within the document itself, as is done with Microsoft Word. That system is discussed further in chapter 4.

Group Decision Support Systems and Electronic Meeting Rooms

DeSanctis and Gallupe (1985) define a Group Decision Support System (GDSS) as "an interactive computer-based system that facilitates the solution of unstructured problems by a set of decision makers working together as a group". Kraemer and King (1988) state that the key concept behind the GDSS is to increase the speed at which decisions are reached when decision makers attended meetings without reducing, and, it is hoped, enhancing, the quality

of resulting decisions. Huber (1984) suggests the "losses" of productivity in group decision making occur because discussions are dominated by certain individuals, low-status members defer to high status members, group pressures lead to conformity of thought, miscommunication among members is common, and insufficient time is spent in problem exploration and generation of alternatives. Huber further suggests that GDSSs can help alleviate these problems by providing a personal computer for each participant, a public display screen for all, computing and communication capabilities that allow for accessing databases and communicating with the group leader and the public display, and software for word processing, data access and management, graphics, and "controls" to permit communications with others or the group. Ellis et al., indicate there are GDSS aids for decision structuring, such as alternative ranking and voting tools, and for idea generation or issue analysis. In their survey of GDSSs, Kraemer and King identify benefits of and barriers to successful GDSSs. They describe "affective" benefits from enlivened meetings involving organized interactions for everyone with the system and a sense of group cohesion during the group learning processes (as the system is taught) and assigned tasks. The technology is perceived as neutral and the meetings are normally very structured and professional. There is an agenda focusing on the issues and decisions and consensus are normally achieved relatively quickly. They also point out the benefits of basic information visualization tools, shared access to information retrieval systems, and modeling and simulation tools. The voting and confidence measurement tools reveal where there is consensus or not and help to measure it. However, Kraemer and King also point out that the successes are not always a function of the technology but the processes themselves, that faster decisions with more consensus are not necessary better decisions, that the process tends to use the prescriptive decision making model, and that many commercial systems have been abandoned. In general, they are positive about the value of GDSSs for helping with parts of the group

decision making process, but are less positive about the potential of complete package systems to be used for a variety of group decision making tasks.

Computer Conferencing

Ellis et al., describe three ways that computers are used in conferencing:

Real-Time Computer Conferencing

Here the purpose is to share computer applications synchronously. It can be done by adapting the use of an application designed for single users and organizing some turn-taking process, so that everyone has a common view but one person is in control at a time, as is done with Microsoft Netmeeting for instance, or by using applications specifically developed for multiple users where each user has their own context. The Colab (Stefik et al., 1987a; Stefik et al., 1987b; Tatar et al., 1991) is an example of the latter approach. Stefik et al., (1987b) state that WYSIWIS (What You See Is What I See) is a foundational abstraction for multiuser interfaces that expresses many of the characteristics of a chalkboard in face-to-face meetings. They also state that in its strictest interpretation, it means that everyone can also see the same written information and also see where anyone else is pointing. After their initial experiences developing Colab as an experimental meeting room at Xerox PARC, in which computers support collaborative processes in face-to-face meetings. Stefik et al., concluded that WYSIWIS is crucial, yet too inflexible when strictly enforced. The Colab was designed for small working groups of two to six people using personal computers connected over a local-area network. Two of its meeting tools were: (1) Boardnoter that most closely imitates the informal functionality of a chalkboard with pointers for each user, pens, an eraser, and pointer icon image that can be placed on the screen, analogous to magnetic arrows; and (2)

Cognoter, a more structured meeting tool for organizing ideas for a presentation, where the meeting process is organized into a sequence of stages: brainstorming, ordering and grouping, evaluation, and outline generation. For both tools they suggested benefits from relaxing the WYSIWIS constraint, including the following:

<u>Issue</u>: The WYSIWIS display of cursors for multiple users is unacceptably distracting.

Design solution: Support local pointing and telepointing.

<u>Issue</u>: Small grain-size updating allows small grain-size collaboration, but is computationally expensive.

<u>Design solution</u>: Broadcast changes to information when the user indicates completion or after a reasonable time interval.

<u>Issue</u>: Teleselections can be confusing if they are displayed in the same way as local selections. Confusion is especially likely in the middle of local operations that require their own selections.

<u>Design solution</u>: Modify the display of teleselected objects so that they are distinguishable from objects locally selected, for example, by using different shades of gray.

<u>Issue</u>: Meetings often need multiple chalkboards, but there is room to show only one on a computer display.

<u>Design solution</u>: Boardnoter provides multiple visible boards by arranging shrunken versions of them as icons in a stampsheet below the chalk tray. <u>Issue</u>: The screen can be crowded with windows used mainly by other participants.

<u>Design solution</u>: Allow participants to select independently which windows are at full scale.

<u>Issue</u>: When windows are shrunk to the miniature scale of stamps, it is no longer possible to assess quickly the locus of activity or which information has changed since the windows were last viewed at full scale.

<u>Design solution</u>: Unlike conventional icons, stampsheet icons should actively indicate when information is changing.

<u>Issue</u>: Identifying recent changes is also important for full-size windows, when participants refocus on a window.

Design solution: Provide a facility for highlighting recent changes.

<u>Issue</u>: Subgroups need to cause other group members to attend to a particular item without interfering with the activities of other subgroups.

<u>Design solution</u>: Provide a way of using the telepointer to show only within subgroup boundaries.

<u>Issue</u>: Although stampsheets reduce the contention, public and private windows still compete for display space. When a participant adds a new public window to a subgroup, it can occlude windows of other participants.

<u>Design solution</u>: Participants can control the placement of all windows on their displays. New public windows may be repositioned after they appear. A number of problems occurred with Cognoter that were associated with the an inaccurate model of conversation and these will be discussed in the section on conversation-oriented models.

Computer Teleconferencing

Teleconferencing can simply involve telephone conferencing or video and audio conferencing. Computer technology can be useful for transferring and managing such communications and this type of technology can help provide some basic presence for individuals communicating at a distance. Ware (2000) reports that it is now generally accepted that in most situations the audio link is more important than the visual one.

• Desktop Conferencing

Desktop conferencing uses the workstation as the conference interface, but it also runs applications shared by the participants. Modern desktop conferencing systems support multiple video windows per workstation. This allows display of dynamic views of information, and dynamic video images of participants.

Intelligent Agents

Nonhumans participating in an electronic meeting are a special case of intelligent agents. In general, intelligent agents are responsible for a specific set of tasks, and the user interface makes their actions resemble those of other users.

Coordination Systems

Ellis et al., indicate that coordination systems can be categorized by one of the four types of models they embrace: form, procedure, conversation, or communication structure oriented.

o Form-oriented models

Form-oriented models typically focus on the routing of documents (forms) in organizational procedures. These systems address coordination by explicitly modeling organizational activity as fixed processes (Lochovsky et al., 1988).

• Procedure-oriented models

Procedure-oriented models view organizational procedures as programmable processes; hence the phrase "process programming" (Balzer et al., 1989). This approach was first applied to coordination problems in the software process domain and takes the view that software process descriptions should be thought of and implemented as software.

• Conversation-oriented models

Conversation-oriented models are based on the observation that people coordinate their activities via their conversation (Flores, et al., 1988; Woo 1990). The underlying theoretical basis for many systems embracing the conversation model is speech act theory (Searle, 1969). For example, The Coordinator (Flores, et al., 1988) is based on a set of speech acts (i.e. requests, promises, etc.) and contains a model of legal conversational moves (e.g. a request has to be issued before a promise can be made). As users make conversational moves, typically through electronic mail, the system tracks their requests and commitments.



Figure 2.2 A Winograd-Flores Structure for Action-Oriented Conversations based on linguistic acts (http://www.arch.usyd.edu.au/kcdc/journal/vol3/dcnet/avarena/)

Figure 2.2 is an example that shows how this approach can be considered to model different types of conversation with state-transition-diagrams. Tartar et al., (1991) indicate that the initial observations of Cognoter were hampered by the ability of people who were very familiar with the performance characteristics of the software to compensate for its problems, because when they evaluated it with "real" participants the system was so bad that half the participants stopped using the system altogether and ultimately threatened to walk out. One of the problems identified was described as ambiguous deictic referencing. Because users had different displays, delays in updating screens, and anonymous actions, a change could be made on a user's screen not only causing them to ask "who did that", but also for those asked to not know what "that" was.

Tarter et al., also considered that they unintentionally used an inaccurate model for the conversation they tried to support: "We claim that many of the serious problems in Cognoter stem from a culturally prevalent, easy-to-make assumption that communication consists of bits of verbal or textual material passed whole from one person to the next." (pg. 206). They concluded that the efficiency they gained by avoiding "production blocking" through parallel typing was lost in the increased amount of work that had to be done to maintain communication based on the parcel-post model.

They concluded that a better model would be an interactive model of conversation, such as that offered by the field of psycholinguistics (Clark, et al., 1986; 1989).

Tarter et al., also note the distinction between conversation and literary communication made by Clark and Wilkes-Gibbs (1986) who describe *mutual responsibility* versus *distant responsibility* respectively:

"The participants in a conversation try to establish, roughly by the initiation of each new contribution, the mutual belief that the listeners have understood what the speaker meant in the last utterance to a criterion sufficient for current purposes (p. 33)."

"The speaker or writer tries to make sure, roughly by the initiation of each new contribution, that the addressee should have been able to understand his meaning in the last utterance to a criterion sufficient for current purposes. (p. 35)."

Thus, in the interactive model, both participants are active, even when only one is actually speaking. People nod, complete or reshape one another's phrases, and say "uh-huh" (Duncan 1973). From the speaker's perspective, each conversational move involves not only its own contents but also a projection of what the next move will be. No response from the listener may normally be assumed to mean understanding and acceptance, but if a try-marker (Sacks & Schegloff, 1979) is used where the tone indicates a request for judgment about acceptability, no response indicates trouble such as the other person not listening. Listeners have the ability to make statements explicitly or implicitly which add to accept, reject, question or modify what the speaker has just said. In the interactive model, this listener response is a crucial part of conversation. In fact, Clark and Schaefer (1989) have gone so far as to advance the notion that the basic unit of conversation consists of two parts, a presentation and an acceptance phase. Together the two constitute a contribution. Since, as mentioned above, nonresponse is in fact a statement, the ability to perform the acceptance portion actively is crucial.

Tatar et al., point out that the writing actions are neither appendages to the verbal conversation nor independent of it. Writing and talking is intricately bound together in a similar fashion to the way a statement and its response are bound together. However, the coordination issues are even more complex because a person can write while talk is occurring. There are therefore more kinds of moves to be made. Tatar et al., go on to point out that the same is true for other media too:

"The study of traditional representational media contains ample evidence that people use the basic 'interactive' conversational paradigm. It is highly structured and dependent on both time and context. While there may be interesting

.

16

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

modifications in response to different representational media, we have every reason to believe that the same resources are available for projecting the next move, for making and obtaining listener response, and for mid-course correction." (pg 197).

Tatar et al., were willing to admit that the absence of coordination between speech and writing led to a highly frustrating experience. They also note that they failed to achieve shared references to objects as Clark and Wilkes-Gibbs (1986) have noted is necessary:

"We had hoped that the differences between the displays at Cognoter stations would be transparent to the users or at least be accepted without too much difficulty. However, in fact people were not looking at the same surface (separate screens). This meant that they lost most of the gaze and gesture information and their base-level knowledge of what was being talked about was less than with other representational media." (pg. 202). In addition, because the windows could be controlled separately indexical descriptions such as "It's in the upper left" do not work under these circumstances.

Communication structure-oriented models

Communication structure-oriented models describe organization activities in terms of role relationships. For example, in the ITT approach (Holt, 1988), a person's electronic work environment is composed of a set of centers, where each center represents a function for which the person is responsible. Within centers are roles that perform the work and objects that form the work materials for carrying out the function of the center. Centers and roles have connections to other centers and roles, and the role scripts of the interacting roles govern the behavior of the connections.

Ellis et al., point out that there is often merging of these functionalities as, for instance, intelligent message systems can and have been used for coordination and desktop conferencing systems can and have been used for group editing. However, some systems can be classified by a primary emphasis and intent, and this in turn, may depend upon the perspectives of the system designers.

Perspectives

Ellis et al., state that there are at least five key disciplines for perspectives for successful groupware:

• Distributed Systems Perspective

Because their users are often distributed in time and/or space, many multiuser systems are naturally considered to be *distributed* systems. The distributed systems perspective explores and emphasizes this decentralization of data and control. Essentially, this type of system infers global system properties and maintains consistency of the global state by observing and manipulating local parameters. The investigation of efficient algorithms for distributed databases is a major research area in distributed systems theory. Some of these research results are applicable to groupware systems. For example, implementing electronic mail systems evokes complex distributed systems issues related to robustness: recipients should be able to receive messages even when the mail server is unavailable.

Communications Perspective

According to Ellis et al., one of the commonly posed challenges of groupware to communications technology is how to make distributed interactions as effective as faceto-face interactions. Perhaps the correct view of this challenge is that a remote interaction, supported by appropriate technology, presents an alternative medium. While this will not replace face-to-face communications, it may actually be preferable in some situations for some groups because certain difficulties, inconveniences, and breakdowns can be eliminated or minimized. For example, distributed interactions allow participants to access other relevant information, either via the computer or in a book on the shelf, without interrupting the interaction flow. This is analogous to findings on the use of telephone, electronic mail, and other technologies. While none of these replace face-to-face interaction, each has a niche where it is a unique and useful mode of communication. The challenge, then, is to apply appropriate technological combinations to the classes of interactions that will benefit the most from the new medium.

Human-Computer Interaction Perspective

This perspective emphasizes the importance of the user interface in computer systems. Ellis et al., described Human-computer interaction as a multidisciplinary field, relying on the diverse skills of graphics and industrial designers, computer graphics experts (who study display technologies, input devices, and interaction techniques), and cognitive scientists (who study human cognitive, perceptual, and motor skills). Ellis et al., note that groupware challenges human-computer interaction researchers to consider factors such as group dynamics and organizational structure, which are not normally considered relevant to user interface design.

Artificial Intelligence Perspective

With the emphasis on theories of intelligent behavior, this perspective seeks to develop techniques and technologies for imbuing machines with human-like attributes. The artificial intelligence (AI) approach is usually heuristic or augmentative, allowing information to accrue through user-machine interaction rather than being initially complete and structured. Ellis et al., claim that this approach blends well with groupware's requirements. For example, groupware designed for use by different groups must be flexible and accommodate a variety of team behaviors and tasks.

Social Theory Perspective

This perspective emphasizes social theory, or sociology, in the design of groupware systems. Systems designed from this perspective embody the principles and explanations derived from sociological research. The developers of Quilt (Fish et al, 1988), for example, conducted systematic research on the social aspects of writing, and from this research they derived the requirements from their collaborative editing environment. As a result, Quilt assigns document access rights according to interactions between users' social roles, the nature of the information, and the stage of the writing project.

Distributed Systems and Distributed Work

According to Rasmussen et al., (1991), "a system is characterized by distributed decision making to the extent that it lacks a centralized control agent, or decision maker". They also note that as the complexity increases, it becomes impossible to embody the required control structure in a single decision unit. Group decision-making also involves multiple decision makers working together. However, in group decision making there is an attempt to achieve consensus and each member is capable of understanding the problem as a whole, whereas in distributed decision making each member has a model of a limited part of the problem, so the problem becomes coordinating their efforts.

Decker (1987) provides a review of distributed problem-solving techniques in AI systems that can also be applied to human and human-automation systems. The taxonomy has four dimensions:

- 1. The level of problem decomposition (granularity);
- 2. The distribution of expertise;
- 3. The methods for achieving distributed control;

4. The process of communication.

Rasmussen et al., note that most distributed systems attempt to solve course grained problems and that the distribution of expertise involves a range of expertise rather than highly specialized people. They also note that the decomposition can often not be decided *a priori*. Thus, an appropriate decomposition must be decided, which requires communication, and for complex systems that are dynamic over time the decomposition may have to be periodically revisited. Rasmussen et al., note that the level of cooperation between agents varies in different systems, but generally one agent is willing to adjust its goal to accommodate the needs of other decision makers. Again, this requires communication, or an accurate model of what the other decision makers' goals are. The latter approach can be a problem if the model is not accurate, or changes without communication between decision makers. One decision maker's goals may also change because of different situational awareness, so that even if another decision maker has an accurate model of the first decision maker's reasoning, it won't predict the goal correctly because the correct model of reasoning is applied to an inaccurate model of the perceived environment for the first decision maker.

Rasmussen et al., describe communication as the cement of the organization, and the greater the need for coordination and cooperation, the greater the necessity for communication. Decker describes systems whose structure changes during problem solving to take advantage of new opportunities as having a dynamically opportunistic organization. At the same time Decker notes that communications in AI systems normally takes more processing than computation and the same is often true between humans. Thus, the cost of communications must be considered when organizing a group of distributed decision makers.

CHAPTER 3

TEXT-BASED TO MULTIMODAL-BASED ASYNCHRONOUS COMMUNICATIONS

Asynchronous distributed interaction

Asynchronous communications systems are said to "transcend the limitations of time and space" (Baecker, 1993), allowing communications among groups of physically dispersed individuals, where the communication does not take place "at the same time". Although the focus of the research presented in this dissertation is related to computer supported asynchronous communications, that context involves characteristics also found in non-computer based communications, so it is worth considering what has been learned from experience with both non-computer supported asynchronous communications.

Non-computer supported asynchronous communications

Mail systems are a popular example of asynchronous communications. Their value was appreciated long before electronic forms were created, with systems evolving through the generations to provide increasingly rapid, inexpensive, and ubiquitous asynchronous communication. A mail messaging system supports the communication of a message composed of (1) information sufficient for the Message Transfer Agent (MTA) to deliver the message and (2) content, which actually contains that which the sender wishes to communicate to the recipient (Chapman, 1987). Content on paper may involve printed text, handwriting or images. If the message is actually more a parcel than a letter, then additional utility beyond that of a pure communication may be added by including other materials (e.g. a gift). With the invention of audio tape recorders, the sender was given the option of sending an audio message asynchronously through the mail system in the form of an audio tape, that could supplement the normal text based message, or perhaps occasionally be the entire content. Similarly, videotapes can be sent if moving pictures, in addition to audio are desired. By attaching audio tape recorders to telephones, people are provided with answer phone systems where they are able to create entirely audio-based asynchronous communications – with voice mail being in some ways just a more "intelligent" technologically advanced version of this. Facsimile systems are another popular example of asynchronous communications systems, which are useful when paper materials need to be shared quickly and some degradation in the quality of the content is acceptable.

Computer supported asynchronous communications

Computer supported asynchronous communications are possible without specifically designed electronic messaging systems because people can be directed to computer stored information in other ways. Thus, ever since it has been possible to put computer files in the file space accessible by a computer user it has been possible to make that file the content of a computer based communication for specific users, as long as the recipients are somehow told they should take a look at those files. For instance, people adapt File Transfer Protocol (FTP) programs and World Wide Web sites to make them messaging systems, by posting files on the respective server and telling others the address to use to retrieve the files with a suitable client program. Sometimes the file is analogous to the content of a letter and other times it may be considered to be more analogous to the content of a parcel, having utility distinct from a pure communication (e.g. a program).

Electronic Mail

Messaging Handling Systems (MHSs) add value because they not only provide abstraction from computer files to messages and protocol based file transfer processes to "send message" and "check mail" commands, but also because they support the user in managing the collections of messages that are accumulated as messages are received and sent. Even so, as Email has become increasingly popular the challenge of avoiding E-mail overload (Whittaker and Sidner, 1996) for the user has also increased.

Older E-mail systems were consistent with older computer interfaces in general – they were text based and their interfaces didn't support abstractions we consider basic today, such as WIMP (windows, icons, mouse, and pull-down menus) and direct manipulation (Zielgler and Fahnrick, 1988; Shneiderman, 1998). For instance, in 1991 Sproull and Kiesler described the Email systems of that time as having the following key characteristics:

- E-mail is asynchronous
- E-mail is fast
- E-mail is text-based
- E-mail can be addressed to multiple receivers
- E-mail has a built-in external memory
- The external memory can be processed by computer.

E-mail users must often also consider the implemented bandwidth and any file size restrictions of the networks involved between sender and recipient. A major reason the E-mail systems mentioned by Sproull and Kiesler were fast is that they were text-based. Even a user connecting to a network with an old 9600 bits per second modem could send 1200 characters per second with an ASCII (American Standard Code for Information Interchange) based E-mail system, so most text messages in a few seconds could be sent to or received from an E-mail server. As people start to send more alternative media files, that are much more greedy in terms of storage requirements, the issue today is increasingly not if people have access to E-mail, but how confined they are by the speed and file size restrictions involved in that access. File size restrictions and bandwidth are two reasons why people have adapted their messaging behavior to use FTP (File Transfer Protocol) programs (that don't normally impose file size limits) and the Web (which now increasingly provides streaming options, lowering the time between starting file transfer and file viewing when video or audio is involved.

Today's E-mail systems are more sophisticated, but the majority of messages still only have text as their content. This may be sufficient to convey the message the sender wishes the recipient to receive, but sometimes asynchronous communications may be usefully enriched with additional formats and doing so should not return the user to a low-level of abstraction from the details of the machine. For most E-mail systems that is the case, as the user has to use separate programs to create disjoint portions of the message, such as audio or video, and then navigate through the computer's file system to locate and attach them to the E-mail message as separate files.

One way text based E-mail messages are used differently from their regular mail counterparts is the extent to which a sender's text is often reused by the recipient in creating a response. This is such a recognized practice that E-mail programs try to explicitly support the respondent intermixing, but keeping distinct, responses to particular portions of the text from the original message. In some senses the reply a person gets back is an annotated version of the original message. When the user includes images or audio with a message, as separate files, annotating a response in a similar fashion is not normally possible.

Neuwirth et. al. (1998) studied the effect of the external representation in a problem solving task involving E-mail exchanges. They first conducted an observational study where the task given to participants was to act as a system administrator and try to track down E-mail "spoofers". Participants needed to communicate with others acting as system administrators for other systems that the spoofers might be sending messages from. The participants needed to keep track of the information in a number of messages over time and some started to use paperbased representations of the state of their progress as they searched for culprits. In an experiment that followed this study participants were asked to look at the current state of a system administrator's representation of the problem-solving task - half when the representation was a number of E-mail messages and half where the E-mail messages were supplemented by an electronic representation similar to those previously created on paper. The electronic representation was created using a program that could build a table of data from filters placed on incoming and outgoing messages as well as user-entered data. They found that those with the additional task specific representation could more quickly answer questions about the current state of the problem-solving task. This research demonstrates that E-mail systems should not be studied or designed in isolation from context., because these representations were intrinsically tied to the task.

Multimodal Asynchronous Messaging and Annotation

Messages and Annotations

Annotation: "A critical or explanatory note."

Comment: "A note in explanation, expansion, or criticism of a passage of a book, article, or the like; an annotation."

(Webster's Encyclopedic Unabridged Dictionary of the English Language) An annotation or comment is a communication regarding something that already exists, which is intended to add in some way to what is referenced. It can be created for oneself or for others to see. Messages often contain "annotations or comments" because they reference both what exists outside the message itself, the contents of previous messages, and the contents within the current message.

Asynchronous Multimodal Systems



Figure 3.1 Asynchronous communication methods in order of development (Daly-Jones et al., 1997)

When both human auditory and visual modalities are involved, systems are said to be 'multimodal' (Daly-Jones et al., 1997). Figure 3.1 shows Daly-Jones et al.'s breakdown of popular asynchronous communications methods in terms of which of these two modalities they supported in 1997, and the fact that none of the methods at that time supported multimodal messaging. However, written text is not the only form of visual information, and the representation of information can be significant at both the perceptual and cognitive levels.

Voice, Text and Graphics

One basic dimension, which might usefully supplement the summary in Figure 3.1 is whether or not the communication method supports the transmission of graphical images as well as just text or audio.

Table 3.1 is an adaptation of Figure 3.1, with this added dimension. None of the basic asynchronous communication methods in Figure 3.1 that support audio also support graphics, but most that support text also support graphics (although the mixing of the two may require using a separate application for E-mail depending on the E-mail program).

	Speech	Text	Graphics
Verbal messages left with co-workers	•		-
Answerphone	~		
Voicemail	~		
Notes, Letters		~	~
Fax		~	~
E-mail		~	-
Pager		~	

 Table 3.1 Basic asynchronous communications methods

 support for speech, text and graphics

The dual coding theory of Paivio (1986), shown in Figure 3.2, proposes that that there are fundamentally different types of information stored in working memory; he calls them *imagens* and *logogens*. Imagens denote the mental representations of visual information, while logogens denote the mental representations of language information.



Figure 3.2 The architecture of Paivio's dual-coding theory (Ware, 2000)

The division of real-world visual information into just two categories: text and graphics is also certainly a gross oversimplification. Complex information can be presented in a myriad of forms.

"To envision information – and what bright and splendid visions can result – is to work at the intersection of image, word, number, art. The instruments are those of writing and typography of managing large data sets and statistical analysis, of line and layout and color."

Tufte's Introduction (1990)

Semiotics is the study of signs and symbols. Three basic forms of representation identified by those working in this area are prepositional reference for linguistic or digital forms; iconic reference for objects that resemble that which they represent; and analogical where the representation captures the structure and behavior of that which is represented (e.g. a map).

Norman (1988) makes the point that the same image can represent different things to people of different cultures (e.g. "arrival time" normally means the time the wheels are first on

the ground as a plane lands to FAA employees, but the same term mean the time in at the gate to people working for the airlines) and also that meaning is derived from experience with interfaces.

The representation effect (Kotovsky, Hayes and Simon 1985; Zhang & Norman, 1994) of a particular visual image also refers to more than just the effect of a graphical image versus one composed of text:

The *representation effect* refers to the phenomenon that different isomorphic representations of a common formal structure can cause dramatically different cognitive behaviors. One obvious example is the presentation of numbers (for cognitive analyses, see Nickerson, 1988; Norman, 1993; Zhang 1992; Zhang & Norman, 1993). We are all aware that Arabic numerals are more efficient than Roman numerals for multiplication (e.g. 73x27 is easier than LXXIII x XXVII), even though both types of numerals represent the same entities – numbers. (Zhang & Norman, 1994).

Wojahn et. al. (1998), looked for effects from the location of comments in a collaborative writing task. They compared three conditions: *Split-Screen Interface*, where comments were placed in a horizontal frame below the main text; *Interlinear Interface*: where the comments were placed between text in the main window of the document itself; and *Aligned Interface*: where comments were placed in a vertical frame to the right of the main text and aligned horizontally. They found that subjects communicated about more problems in the Interlinear and Aligned conditions, without a significant difference in time on task, than subjects in the Split-Screen condition.

A fundamental characteristic of speech compared to more visual forms of communication, such as text or graphics images) is that the listener is freer to use other modalities, such as vision, while processing the speech input (Streeter, 1988). Most research comparing spoken with written communication has been concerned with synchronous communication (Daly-Jones et al., 1997). Perhaps the classic study being conducted by Chapanis (1975) who found that speech based communications during problem solving involved many

more words per minute than text and that the tasks were solved faster when speech was used. Neuwirth, et al. (1994), however did conduct a study of spoken versus written asynchronous communications. They compared the nature and quantity of voice and written comments produced in each mode, when reviewers gave feedback to writers. They found:

- Reviewers used more words in voice than text mode during the same time period, but that the same number of annotations was made on average. The additional words were attributed in part to providing more reasons why the reviewers thought something was a problem and for polite language that mitigated the problem.
- 2. Evaluations of reviewers were less positive when reviewers produced written annotations than spoken.
- 3. Comments about low-level mechanics (e.g. spelling or grammar) were preferred in text. Overall, the findings support the utility of voice modality for supporting collaborative writing activities – although it should be noted that while people can generally speak faster than they type, people also generally process typed text faster than voice recordings (Streeter, 1988).

In eye tracking and recall testing studies, Faraday and Sutcliffe (1997, 1998) investigated attention and comprehension by users interacting with multimedia presentations. In one study (1997) research students observed a commercially produced CD-ROM, 'The Etiology of Cancer' while wearing eye-tracking devices, then had their domain knowledge recall measured. The presentation showed DNA repair by photoreactivation, during which objects, labels and arrows would be revealed from a previously hidden state and some objects would move across the screen. From this study, a set of guidelines for controlling attention in multimedia presentations was produced:

Use object motion to control attention and viewing order;

(Participants' attention was drawn to motion and fixations tracked the moving objects path)

• Use animation with care;

(Multiple simultaneous animations from moving objects or revealed objects and labels, or too rapid motion, sometimes caused attention to unintended areas)

- Gradually reveal labels, symbols and objects, to control viewing order;
 (Revealing hidden objects, labels or arrows, when focus on them was desired, was effective in drawing attention)
- Reveal related elements in a sequence;
 (Simultaneous animation sometimes caused attention to objects or labels in an undesired order)
- Reveal important information to emphasise it;
 (Static objects and labels received less attention than those revealed or in motion)
- Ensure that an object and its label appear together to improve identification; (Fixation moved back and forth between an object and its label)
- Use speech to reinforce information presentation, reveal objects and labels when cued in the speech track;

(Participants fixated on verbally referenced objects and labels)

- Allow reading time after cueing a label;
- Use symbols to direct attention to specific objects and locations;

(The arrow symbol shifted fixations to that which the arrow pointed to)

Speech information should reinforce image;

(Propositions given only in the image or animation without speech cueing were poorly recalled)

• Captions or labels may be useful in re-inforcing the speech track;

(Propositions given in speech which were reinforced by labels were well recalled)

- If the speech information is complex or important then present it concurrently in a caption; (Propositions given only in the speech track were generally poorly recalled)
- Cue animations with speech;

(Animated objects, which were cued by the speech track, were well recalled)

• Show the effect of actions.

(Complex actions shown in animations, without showing the resulting effect, were poorly recalled)

In another study (1998) they investigated how people fixate back and forth between text and that which it refers to in an image. In contrast to pointing, which generally precedes related speech (Oviatt, 1997), Faraday and Sutcliffe reported that people would read a sentence, and then look for the reference in an accompanying diagram. At times they found people not able to move smoothly to that referenced, so they developed a system where users can click on a button after the associated text to cause that referred to in the image to be highlighted or animated. One of their examples is shown in

Figure 3.3. This helped control the users' attention as intended. They also recommended placing any label for an object as close to the object referenced as possible, which is consistent with Gestalt principles (Ware, 2000).



Figure 3.3 Designing connections between text and annotations (Faraday & Sutcliffe, 1998)

The Microsoft Research Annotation System (MRAS) supports streaming video on the web and personalized annotations by those who view it. MRAS allows a video to be paused for note taking, with a contextual link to the current position in the video when that is done. Figure 3.4 demonstrates that viewers can create audio annotations in addition to text and share those annotations with others. Bargeron et al (1999) conducted two studies of its use. In the first they studied personal note taking of users watching a video using MRAS and using traditional pen and paper methods. All participants used both methods – half using MRAS first and half paper. They found when users used the MRAS system they took longer, pausing or rewinding the video. They also found that MRAS-first participants took paper notes differently from paper-first participants suggesting the MRAS system caused a shift from the traditional way of taking notes during a presentation. The MRAS system was preferred to paper notes and the point was made that its value would increase over time as the video would be more difficult to remember and a quick review of noteworthy parts would be useful.

Add New Annotation	×
Target	
URL file:////elisar5/time.compression.studies/esoloway	512.ast
• 000:06:47	000:07:45
Annotation	
Annotation Set TOC	
Email To jereny@smallware.com	
Summary. Why bother with Web-based Collaboration?	
	Tarras at the set
Annotation Type C Text: C Audio	OK

Figure 3.4 MRAS "Add New Annotation" dialog box (Bargeron, Gupta, Grudin & Sanocki, 1999).

In a second study participants added their notes to a shared folder. Each participant worked in one of three annotating conditions: Text-only, Audio-only or Text-and-audio. When replies were made in the text-and-audio condition participants were as likely to use text to reply to audio annotations, as they were to use audio. Participants generally felt it took more effort to listen to audio than to read text. One participant in the text-and-audio condition was frustrated with the speed of audio annotations, saying that "I read much faster than I or others talk. I wanted to expedite a few of the slow talkers." Another participant pointed out that "it was easy to read the text [in the preview pane] as the video was running. By the time I stopped to listen to the audio (annotations) I lost the flow of information from the video." This research is an example where audio annotations did not work as well as text, in contrast to some situations mentioned in the research cited earlier. In summary, it would appear that it is more difficult to listen to an audio annotation while also watching a related video with its own sound than it is to listen to audio annotation while looking at a related still image. This does not seem counter intuitive.

Deictic Gesturing

Tang conducted an ethnographic study of eight short small-team design sessions (Tang 1989; Tang 1991; Tang & Leifer 1988), where each team was given problems to solve and large sheets of paper as a shared work surface. Tang built a descriptive framework to help organize the study of work surface activity, where every user activity was categorized according to what action and function it accomplished, as listed below (Tang 1989):

- Listing: produces alpha-numeric notes that are spatially independent of the drawing;
- *Drawing:* produces graphical objects, typically a 2-dimensional sketch with textual annotations that are attached to the graphic;
- Gesturing: is a purposeful body movement that communicates specific information e.g. pointing to an existing drawing.

Gesturing played a prominent role in all work surface actions (~35% of all actions). Gestures were used to enact ideas, signal turn-taking and to focus the attention of the group (~58% of the gestures were to focus attention).

Bly (1988) studied two-person design sessions in three different settings: face-to-face, geographically separated with an audio/video link, and a telephone only connection. The study tasks themselves involved no new graphic design, but rather extensions of existing user interface mechanisms. Three action types were identified: a draw action was a graphic mark on paper; a write action was a text or alphanumeric mark on the paper; and a gesture was a motion specifically related to an existing mark on the paper. According to Bly, "Gestures allowed the designers to act out a sequence of user actions illustrating a mechanism and to point to a drawing cluster as reference or emphasis to objects and ideas." The division of actions into these three categories is shown in Figure 3.5. There was a higher percentage of gestures in this study than

Tang's for the Face-to-Face condition, and the decline from the Face-to-Face session to the Media Link session may have been partly because the visual information reportedly lagged behind the audio therefore making it difficult to send the image of a gesture synchronized with verbal comments.



	Face-to-Face		Media Link		Telephone	
Draw	93	29%	50	31%	164	43%
Write	35	11%	61	38%	173	46%
Gesture	197	61%	50	31%	43	11%
Total:	325		181		380	

Figure 3.5 Actions for each session (Bly, 1988)

"In human communication theory, a gesture that links the subject of a spoken sentence with a visual reference is known as a *deictic gesture*, or simply *deixis*." (p. 324, Ware, 2000). This type of gesture was found to occur frequently in Tang's and Bly's studies and in others involving human-human communications (Francik 1996; Daly-Jones et al., 1997). It is a natural form of communication and one that makes referencing objects, locations, and sequences of locations and objects, more efficient and less ambiguous.

Tang and Bly call for research to investigate how to develop tools to support gestured activity. Software supporting gesturing with a pointer on a two-dimensional image has fewer degrees of freedom than a real hand in 3 dimensional spaces, but it does nonetheless support deictic gestures to that image. Faraday and Sutcliffe's research on users' interaction with multimedia involving animated objects seems relevant here as supporting gestures through a pointer results in an animated object, although the reference is to that pointed to and not the moving object itself. Four of there observations seem particular relevant:

- Participants' attention was drawn to motion and fixations tracked the moving objects path;
- 2. Static objects received less attention than those in motion;
- 3. The arrow symbol shifted fixations to that which the arrow pointed to.
- Propositions given only in the image or animation without speech cueing were poorly recalled, but animated objects, which were cued by the speech track, were well recalled.



Figure 3.6 Freestyle image desktop; handwriting and voice-annotation tool (Francik, 1996)

Freestyle (Francik and Akagi, 1989; Francik et al., 1991; Francik 1996), was developed by Wang Laboratories in 1988, but it never became widely used and is now no longer available (Francik, 1996). It was an ambitious attempt at a particularly natural virtual working desktop with:

- Thumbnails of actual file contents to represent files on the desktop, rather than a categorical icon and a filename;
- A stapler object to group collections of files together;
- A stylus (see Figure 3.6) specifically engineered to be suited for writing and drawing, rather than CAD applications (which was what styluses were used for at that time).
 This pen also had a physical eraser operated like the eraser on the top of a pencil.
- A recorder that captured synchronized handwriting, or other stylus cursor movement.

The extensive use of metaphors lead to the marketing slogan, "You already know how to

use it". It was intended to be a very paper-like environment², implemented on the then widely available 286 PC, but using an operating environment separate from Microsoft Windows. Combined with image capture, scanners, fax interface cards, electronic mail, and high-resolution graphics, it was also a multimedia communications system that drew a lot of attention from the computer- human interaction academics when it was made public:

When I read that Wang was demoing a product called Freestyle, I have to admit that I was somewhat underwhelmed. I mean, who needs to see yet another drawing program. Word of mouth during the conference had it, however, that Freestyle was a hot design worth seeing, so during a boring lecture I decided to give it a try. It turned out that Wang Freestyle did indeed represent interesting interface innovations and that it was not related to the popular graphics program Freehand with which I had probably confused it subconsciously. (http://www.useit.com/papers/tripreports/chi89.html)

Technologies or CAT group (Francik, 1996).

² The development team's view of mice was reflected in its name, the Computer Annotation

The Freestyle system was intended to support spontaneous communication rather than structured workflows – as the name implied. The most frequently used addresses or distribution lists could be turned into "mail slot" icons residing directly on the desktop, so that after creating a message, sending it was a simple task.

Although its users liked it, it took time before they were able to envision how to use the technology to meet real organizational needs. It tended to be first used for very structured processes, for signature processes and electronic distribution of scanned-in paper documents although the designers considered it best suited for problem solving, planning, and design. There was also the tendency for small workgroups to be setup with the system in organizations, to keep the cost down, making it difficult to achieve a critical mass of users. The cost of the system was reduced by supporting mouse controlled annotation and display on lower-resolution monitors, however, the new desktop interface may have just appeared at a difficult time and with insufficient resources: 'The last thing the world needs is another user interface', said Stewart Alsop 2d, editor of the PC Letter newsletter, referring to Wang's attempt to enter a field recently stirred up by OS/2 Presentation Manager, NextStep, NewWave, Metaphor, Windows, Deskmate, The Mac Desktop [sic], mice, and other technologies designed to make it easier to use computers" (Lewis, 1988). "As impressive as Freestyle is, the system that finally brings its wonders to the masses will probably be Freestyle-inspired rather than Freestyle itself. Wang's sorry financial condition may preclude it from devoting the resources necessary to establish Freestyle's new technology in the PC marketplace" (Bonner, 1990). Francik, herself adds, "In casual conversation, more than one person has dubbed Freestyle 'the Xerox Star of pen-based computing.' There is both flattery and irony in that comparison." (Francik, 1996).

In a more recent attempt at a particularly natural environment involving both multimedia and multimodal messaging, Ricoh Innovations Inc, has developed a touch screen

based pointing system called TouchVerse. In a presentation given at Stanford University³, Greg Wolff highlighted the disembodiment of attachments that occurs with E-mail messages when a person replies to a message, and the fact the voice-mail has no visual context. TouchVerse allows the users to drag documents onto a blank message and then create audio messages on the same screen that have a scroll bar with automatically generated lines from the scroll bar (positioned based on time) to any referenced document, if the person refers to that while speaking.

Lotus Screencam⁴ is a software application, developed by Lotus Development Corporation, which records the user's computer screen image (including any visible pointer) and any audio produced by the user at the same time to form a single "video" that can be used as an asynchronous communication. Although it can be used to capture the dynamics of many different annotation modes (such as audio, text and pen based), it quickly produces very large files and provides a very basic file management environment. This is perhaps an application most suited for demonstrating how to use a software application, where the underlying image changes often, and the visual dynamics are particularly valuable.

In a study involving the use of Lotus Screencam (Daly-Jones et al., 1997), 'managersecretary' pairs were asked to complete an asynchronous appointment-scheduling task and an equipment-booking task in three independent conditions. *Fax-only* involved using Microsoft

⁴ There are two versions of Lotus Screencam – a Windows 95 release and a Windows NT4x release. The Windows 95 version does not record on all systems running Windows 98 and there are no plans to create a version for Windows 2000 (http://www.lotus.com/products/screencam.nsf)

³ The presentation "In support of multimedia conversations" presented 2/11/2000 is available via streaming video at:

http://murl.microsoft.com/videos/stanford/CS547b/000211_OnDemand_100_100K_320x240.ht

Paintbrush; *Voicefax* involved using Lotus Screencam with a Paintbrush image; and *Voice-only* involved just audio. They found that for both sending and receiving users rated voicefax the most useful, then fax-only, then voice-only. In each condition 30 minutes was allocated to complete the task. Users took the same amount of time to complete the tasks in each condition, but fewer messages were sent with voicefax.

There are also currently a number of multimodal human-computer interactive systems being developed that support potentially highly expressive input such as hand gestures, eye gaze, and head and body movements, and annotations (Oviatt, 1999, Oviatt et. al. 1997; Hardock, et. al. 1993; Koons & Sparrell, 1994; Kullberg, 1995). This research is related to multimodal computer supported asynchronous collaboration between humans, but there are important differences. Multimodal human computer interaction is often concerned with machine learning of human commands - including complex gesturing (McNeill, 1992) synchronized with speech (Bourguet & Ando 1998). The ultimate goal is to make computer agents understand fully embodied human communications as well as humans do. In computer supported asynchronous communications however, it is another human who will interpret the message sent. The goal is to provide a representation of the communication that is sufficiently rich for the foreseen collaborative tasks. Thus, many of the issues in human-computer multimodal research are not relevant here, but general observations such as the fact that in one study 95% to 100% of users preferred to interact multimodally when they were free to use either speech or pen input in a spatial domain, but that users typically intermix unimodal and multimodal expressions (Oviatt, 19995), are interesting assuming they translate to computer supported multimodal human to human communication.

⁵ A corresponding presentation at StandfordUniversity is available via streaming video at: http://murl.microsoft.com/videos/stanford/CS547/990226_OnDemand_100_100K_320x240.ht m
The research presented here does not consider annotations in 3-dimensional space, however, there has been research in that area, including the development of a voice-based annotations system for annotations in immersive environments (Harman et al., 1996).

Drawing Tools

In addition to highlighting the significance of gesturing, Bly (1988) also noted (1) participants expressed frustration when not sharing the same writing/drawing space and (2) the frequent rapid switching between annotation marks when drawing and writing (stating that few, if any, existing computer supported sketching tools allow such rapid transitions). Greenberg et al., (1992) developed a multi-user sketchpad (called Sketchpad), that is a synchronous communications system, but the issues raised are also relevant for asynchronous communications systems. Following the guidance of Tang and Bly, Sketchpad was developed with the intention of supporting tele-presence (Egido 1988) and to make switching between annotation modes as effortless as possible. Thus, their design criteria include the following:

- Since gestures must be seen in order to convey information, all cursors within a work surface are always visible to all participants. Cursors are also made prominent on the display by their larger than normal size.
- Cursors are unique, each identifying the person it belongs to by labeling it with the user's name. In addition (and more subtly) each cursor is orientated at different angles.
- Cursor movements appear with no apparent delay on all displays, which means that they remain synchronized with verbal communication.
- The simplicity of GroupSketch allows it to have a nearly modeless interface. When no
 mouse buttons are depressed, the cursor is in the pointing gestural state. Drawing occurs
 as long as the left button is depressed, which also turns the cursor into a pen. Typing
 immediately inserts text at the current cursor location. The cursor image changes to the

43

pen, and automatically reverts back to the hand-shaped cursor after a reasonable pause in typing is detected.

In asynchronous communications a clear voice and deictic gesturing would seem to have the potential to provide a degree of tele-presence too; user specific pointers could still be useful when saving the recordings of multiple users; and the ease of changing annotation modes is also relevant. Both situations need to support WYHIWIS (What you hear is what I said), and while asynchronous communications may not be a WYSIWIS (What you see is what I see) situation⁶, they can perhaps still be WYSIWID (What you see is what I did) if dynamic activities such as gesturing are captured.

Popular Software with Features for Annotation

Some asynchronous systems support richer interaction by allowing the integration of original content and annotations in a variety of forms such as graphics, text, audio and video, so that, for instance, an image can be seen with comments about it without having to consciously work with multiple files and programs. In some cases these systems involve programs originally intended for other purposes, but where the user has adaptively created a messaging system. In other cases, they are systems specifically designed to create this type of environment.

Microsoft Office 2000

Microsoft Office is a popular package of commonly used applications – the most recent version of the package being Microsoft Office 2000. Three of its applications that can be used to create content for a message, which can be annotated by sender and recipient to create a message richer than text, are Powerpoint, Word and Excel.

⁶ Unless WYSIWIS can also be interpreted as what you see is what I saw.

Microsoft Powerpoint

Powerpoint is most often used as a presentation program rather than as part of a messaging system. However, if a person wishes to receive a response to information placed on a series of slides the slide show can be transmitted as an E-mail attachment and the recipient asked for feedback in the form of annotations on the slides. The original content can include text, images, graphical annotations, audio and video. Thus, for instance, an image could be a map with arrows drawing the recipient's attention to particular points on the map, text providing additional information about what is at those points, and an audio voice over providing a personalized greeting, additional information, and questions about the information shown on the image supplemented by knowledge from the sender. However, Powerpoint was not designed for messaging, so users using it that way to a large extent have to create their own grammar for ordering and showing who created what annotations. Microsoft Office 2000 does allow the user to place a "comment" on a slide resembling a "sticky note" and associate a person with that comment, but otherwise there is little support for tracking annotations about the contents of a slide and the slide show in general⁷. The slide show structure itself is, however, simple and the inherent decomposition provides a means for attaching context to particular annotations on a slide.

Microsoft Word

Word is normally used for creating largely text-based documents, but with figures sometimes included. Although classified as a word processor it actually has many of the features

⁷ Microsoft Powerpoint does contain a built-in link to the program Microsoft Netmeeting for synchronous discussion about a slide show. There is apparently also support for asynchronous discussion, in what the documentation describes as resembling the structure of a newsgroup, if a group of collaborators are all connected to a server with Microsoft Office Server Extensions.

that were once only associated with desktop publishing programs. The built in annotation features are clearly designed for people collaborating to produce a Word document as the end product of their work. Of course, not all communications are concerned with producing an increasingly refined artifact that is part of the communication.

Figure 3.7 shows how a comment can be created in a separate window from the main text. A comment can be in the form of text or audio. What is created is really a "hypercomment" so that pointing at text which has a comment linked to it causes that comment to appear in a popup window⁸. Figure 3.8 shows support for navigating through a series of comments, and controlling changes to the document.

⁸ Wojahn et al's (1998) *Split-Screen Interface* is essentially the same as in Microsoft Word except the comment did not popup when the mouse pointer was passed over the text that had a comment attached to it.



Figure 3.7 A comment in Microsoft Word



Figure 3.8 Comment management in Microsoft Word

While Word is quite sophisticated in placing annotations within and between text and handling issues of control between users, it does not support very well placing an annotation at an x, y position on an image or notational pen marks over that image suggesting possible changes to the image or the world it represents. Figure 3.9 demonstrates that an arrow can be placed pointing to part of an image, but controlling the placement of text or pen marks overlaid on an image is a difficult formatting process. Figure 3.9 actually not only involves a marking on the image, but between the text and the image, making for very little momentum loss as the reader moves between the two. Put the cursor in some line of code and then choose Build menu/Start Debug/Run To Cursor. The program will execute up to that line of code. Or set a breakpoint in your code by putting the cursor there and then using the toggle button on the toolbar (the little hand), then choose Build/Debug/Go.

EXERG 4 Microsoft Visual C++ - (EXERO-4 CPP)	
C for faith the least front faith Tone Wrann, Link	
(Globals) 💌 (Al global members) 🗾 🌢 main	

Figure 3.9 A reference to an object in Microsoft Word made explicit

Microsoft Excel

Excel also supports hypercomments, but they are attached to cells rather than chunks of text. Figure 3.10 shows a comment attached to a column header. Again Excel supports navigating through the sequence of comments created, but it does not directly support audio based comments⁹.

⁹ Audio files may be attached to a cell as an "object", as may other file types.

£7 Mill 10	out	<1 P.1	on git in	d.,								
	FOR S	yen.	Insert #F	anat e	Tools Data	Window	Help		1	213.3	717-07-12-35	
		3 6	B	b 6	07	5.20			rial	- 10 Ti	. <i>L</i> L F	
U U	3			Q1								
A A	部日数	KC-2	15 DIS	REA	SEP 3	12:2603	R	EC STARS	SE ST.		SAVE AND	N SEX
ាក												
27										_	Room Chann	
3¥ 98.4	Dep	Anty	Conte	S.V/T	空:08.之	St.WR	Sta	Bin Presen	tation Hultim	edia 01	On the following	scale. Q7
*4£ • .1.	DTW	EWR	ZNY	V	Y	₹Y=t 25	5 Y 7 / 3	Y	EN.		2 please indicate I	how ily d
5 2	DTW	EWR	ZNY	V.	N.	N	- N	N	N-		3 easy/difficult yo	u found ver
6.3	DTW.	EWR	ZNY	$\mathbf{T}_{\mathbf{r}}$	N	-N	2 . N. ~ 2	N. 🥏	N N		1. Verv easv:	Ver
1-1-1-1	DIW	EWR	ZNY	T 🔛	NPS	Y	N	ZQX-			2. Easy;	il y a p
186 5	MSP	EWR	ZOB	V	N	N	N	N	N		3. Neutral;	ija ž
139 <u>1</u> 5	MSP	EWR	ZOB	Ň	N	Y	N	N	N	:	3 4. Difficult;	ily 🧃
	MSP	EWH		Ţ	N	Y	N	N	N		5. Very Caricus.	Uy 🏹
	MSP	EWH	ZOB	اد ع د	N Number 1995	Y Vistan 2015	Y Contractor	N	Y	يد بديد بديد	1 5	3 Daily
12 9	DIW	RUU	-2D.→-	Y	NA .	NA :		NA -	NA		2	TNA
LIJ TIU	UIW	RUU		Y.	1.1		N		N-			J Laily
14-11		RUU		(N	N .	N S	N	N N		2	4 Month
		RUU			N.S.	1. 171.04	<u>Nor</u>	<u>Na an</u>	A CONTRACTOR OF A CONTRACT			C. J Dally
	DTW			S.	Y	Y	N	N	Ŷ		2 4	4 Ually 🖂
403 15	DTH	AIL		Ŷ	. .	v						ائند ≰ م/رد م
	DTW			+	N	Y V	N	N	Ŷ		2 4	4 N/A
20117	MOD		21L 75%	- 1.50	N States and a	T V V V V V V	N Statistics	T.	¥ 8-∞	، د. ۵۰ وژی کلیه	I 4 ಗ್ರಾಮ್ ಭಾಗದ ಕ್ಷಕ್ರಪ್ ತ್ಯಾ	LUany -
21 19	MOD	DEM		N				N SAL				A 2 EL
27 10	MCD	DEM	TEN	-		-17		and the second				
22 20	MCD		7514	- T								3 James
24 21	DTW	BOS	7014	ieliaide. V	Linford	N	<u>تسالي منعن</u> N	M	V North Contraction			A Mooly
35 2		BOS	7RW	v	N	N V	N	IN N	T		2 4 2 A	A Dareivi
26 27	ntw	BOS	7RW	Ť		v.	V	N	N V		· ·	5 Daily
77 24	ntw	BOS	7RW	Ť	lv l	Ŷ	N	N	v			A Daily *
	A . 60	1			L		1	14 			ي. مان المارية مان المانية مانية الم	
Callun			hu Dee			L						
CLERING	يحسدوه	التباد الي		u Ujā		in the second	213725	C. S. S. Sandara and C.	leas-theat tell is		A CALL AND A	COLLECT CARE

Figure 3.10 A comment in Microsoft Excel

Microsoft Paint

There are a large number of graphics programs that allow annotation of a single image. If providing an annotated "picture" significantly enhances the context of a message these programs can be useful to create a more fully descriptive message. They are not multimodal, but they provide a means to create external memory annotations that can be referred to with audio and text communications. Microsoft Paint is a program that has a modest number of annotation options. The fact that this type of program is considered fundamental is demonstrated by its inclusion in the Microsoft Windows operating system. E-mail systems with html formatted content

The Hypertext Markup Language (HTML) is the basic language used to create web pages, but a number of E-mail programs now provide it as an option to format the content of a message. E-mail messages in this case have the same potential for formatting as web pages. They may then, for instance, include text and images in the content of a message, as shown in Figure 3.11. Additionally, an audio or video recording could be part of the message with hypertext/"hyperimage" links to those recordings. They don't support annotations tools such as a freehand pen or arrows.



Figure 3.11 An html formatted E-mail message

51

RealNetwork's RealSlideshow

RealSlideshow is another application centered on the idea of creating a presentation as a series of slides. It is however different from Microsoft Powerpoint in several ways:

- 1. It focuses much more on synchronizing voice and a series of images.
- 2. Figure 3.12 shows the main window where the user can place a series of images and voice-overs within the very visual temporal frame of reference.
- As shown in Figure 3.13 it assumes the images are already created although it does allow captions to be assigned to each image, rather like subtitles can be used in a movie.
- 4. Saving the slide show creates a streaming media slide show, which is particularly useful if the recipient(s) can not access the Internet with a fast connection.

It is unfortunate that only a preview of the image is seen as a voice-over is created for it. Adapting RealSlideshow for use as a messaging system is possible, but it is particularly difficult to try to reuse the content of a sender's message.

1 ^m Realblidesh	ow Untitled 1							
Ele. Edit Vew	Presentation Help					ľ	isit RSCentr	1
lmages	lina Searan							
download	-				· · · · · · · · · · · · · · · · · · ·			
Voice			<u> </u>					
Music								
Timeline	0:10	0:20	0:30	0:40	0:50	1:00	1:10	1:2
C C								
Project	et i shire i s		1.10.4					ं्र
Propertie	IS Layout.			Generate	2. P	lay 🛄 🍰 🖉	J. Send	
1 STORE 414 5								
and the second second second		SUMPLY ADDALA	v.v	1	بر بن وتيجر الاست	ببه الم فرور و و	a state of the second	1. 2. 14

Figure 3.12 A RealSlideshow slide show

,



Figure 3.13 An audio caption in RealSlideshow

CHAPTER 4

COLLABORATIVE ANALYSIS OF NATIONAL AIRSPACE SYSTEM POST OPERATIONS DATA AS A TEST BED

Introduction

Decker (1987) describes a taxonomy of problems for distributed systems that has four dimensions: (1) the level of decomposition (granularity); (2) the distribution of expertise; (3) the methods for achieving distributed control; and (4) the process of communication. In this chapter the NAS is overviewed with particular attention to the nature and significance of communications in the system, and its relationship to the other key factors that Decker describes. The Post Operations Evaluation Tool is also described in this chapter, with particular attention given to the information it contains that can be shared between the airlines and FAA; its representation; and the impact of that representation on multimodal systems that might be used for communications. The type of information not contained in POET that might also predictably be discussed is also described.

The National Airspace System

The NAS is certainly a large, complex system. One dimension of this is the amount of traffic that is actually involved. There are from 4,000 to 6,000 aircraft operating in the NAS during peak periods, which equates to approximately 50,000 aircraft operations per day (http://www.fly.faa.gov/Information/information.html). Figure 4.1 shows approximately 5,000 aircraft operating in the system at one time, although the aircraft are obviously not drawn to scale



Figure 4.1 5,050 aircraft simultaneously operating in the NAS (http://www.fly.faa.gov/Information/information.html)

Agents

The FAA defines the NAS as "The common network of U.S. airspace; air navigation facilities, equipment and services, airports or landing areas" (www.fly.faa.gov/ Glossary_of_Terms/acronyms.html). While the components mentioned are part of the NAS, the definition is misleading in that it doesn't mention the people or their organization that are a critical part of the system. The tasks they perform are essential and the design of these tasks and the environment in which they are to be performed is a critical design within the system. The FAA and the major airlines both have a major impact on how the NAS is used. Some of their key strategic and tactical decision makers concerned with traffic operations and management are as follows:

Airline personnel

- 1. Pilots and other flight crew members, responsible for individual flights;
- 2. Dispatchers, within the Airline Operation Centers (AOCs), who are jointly responsible with flight crews for flight safety;
- Air Traffic Control Coordinators (ATCCs) who are responsible for air traffic control liaison on a daily operational level.

FAA personnel

- 1. Air Traffic Control System Command Center (ATCSCC) personnel, who have primary responsibility for ensuring a safe and efficient flow of air traffic through the system. They have primary responsibility for national strategic air traffic management;
- 2. Traffic Management Unit (TMU) personnel at enroute Air Route Traffic Control Centers (ARTCCs) and TRACONS;
- Air Traffic Controllers (ATCs) in these enroute Centers and in Terminal Area Traffic Control and Airport Control facilities. These people are responsible for tactical air traffic control.



Figure 4.2 A concept of the traditional air traffic management operational process (Smith, et al., 1999)

Figure 4.2 shows what Smith et al., (1999) refer to as the traditional air traffic management operational process, because of the hierarchical nature of the control and communication channels between strategic and tactical decision makers. As will be discussed, the actual channels can vary depending on the task.

Information Technology

As a cognitive system, the NAS can be considered to involve both human and machine cognitive agents, and the interactions that occur between all agents are often critical. There is also diversity in the purpose for which technology exists in the NAS (Billings 1996), but here the focus will primarily be on technologies intended to provide information and support communications related to flight management. As Figure 4.2 shows, these information systems can serve a number of purposes and their autonomy can vary too – some passively providing situation awareness data for human decision makers to use; some assisting in planning and scheduling by maintaining constraints; some more actively implementing decision making procedures; and some providing computer mediated communications.

Goals

The NAS is complex in part because of the number of individual agents and objects in the system, but there is also diversity in the tasks being performed and the goals in performing those tasks. At an organizational level there are some differences because of the different goals of the respective organizations. The FAA is interested in the efficient use of the airspace, and part of that philosophy is captured by the fact it refers to the airlines as "the customers". At the same time it must also act as both a policing organization and a refereeing organization, as it is concerned with both safety and fair competition between its customers. The airlines are also interested in efficient use of the airspace, but so they can provide a profit making, competitive, quality service to the customers they carry on their flights. Both the FAA and airlines are also of course concerned with safety.

Matching Data with Control

As shown in Table 4.1, historically, traffic flow management has been a function under the control of the FAA, with traffic managers at various facilities making decisions about what routes could be flown by flights scheduled by the airlines. The first major change arose in 1992, with a shift from "management by direction" to "management by permission" (Smith, et al., 2000). This allowed the airlines to request non-preferred routes (routes for flights that differed from the FAA-assigned preferred routes). An airline could send a message via teletype to the ATCSCC requesting an alternative route for a particular flight. A specialist would then evaluate this request, checking with traffic managers at the involved regional ARTCCs, and approve or disapprove it.

59

1. Management by Direction:						
FAA traffic managers simply inform an airline regarding the route that can be used by a						
particular flight	particular flight					
2. Management by Permission (1992-1995):						
A default flight plan is assigned by the FAA, w	hich can be revised if the Airline Operations					
Center requests an alternative and receives per	mission from FAA traffic management staff					
Benefit from change Cost from change						
 Increased understanding by users of 	 Increases in the time, and therefore 					
provider and system constraints from	personnel, required to accomplish these					
one-on-one interactions between TMUs	interactions concerning requests for					
and AOCs	exceptions to the preferred route structures;					
 Increased understanding by providers of 	• The increased flexibility afforded to airspace					
user economic and operational needs	users, while considerable, was still felt to be					
	inadequate by air carriers.					
3. Management by Exception (1995-present):						
An Airline Operations Center can simply file th	ne flight plan that it desires, subject to some basic					
constraints, for a given flight						
Benefit from change	Cost from change					
 Greater flexibility for users to 	Significant additional information must be					
accommodate economic and other	considered by dispatchers if AOCs are					
business objectives	going to plan effectively around known or					
 Potentially, an ATM system that is more 	predicted ATC constraints					
directly responsive to user requirements • Less information and knowledge exchange						
and needs, as the users now have a way	occurs between TMUs and AOCs, because					
to express their preferences for routes	users do not have to interact with traffic					
very clearly (by actually filing those	managers prior to executing their plans.					
routes)						

Table 4.1 Changes in the distribution of control and data for traffic flow management in the NAS(adapted from Smith, et al., 2000)

Management by permission gave the airlines a means to try to improve their efficiency as a routine mechanism by which they could request more economical flight plans for their aircraft. It also caused routine interactions to occur between airline ATC Coordinators, ATCSCC traffic specialists and TMU staff, giving each group a broader understanding of the factors considered by the other group, resulting in more effective and efficient interactions. The primary weakness of this paradigm was that it was manpower-intensive and was thought by the airlines to be excessively conservative at times in the approval of requests for alternative routes.

Although the management by permission architecture was viewed as a significant improvement, in 1995, its perceived limitations were sufficient to motivate a revised program

based on "management by exception". This new program, known as the expanded National Route Program (NRP), allowed the airlines, subject to certain constraints, to file the routes that they preferred for each flight. FAA traffic managers would then monitor conditions, watching for situations (such as severe weather) when the program had to be canceled temporarily in particular portions of the country. This architectural change significantly altered the allocation of control, requiring dispatchers to consider factors (such as the prediction of air traffic bottlenecks) that in the past had been handled largely by FAA traffic managers, if the dispatchers wanted to file effective routes.

As these approaches demonstrate, there are some standard questions that can be asked when considering the distribution of control and data:

- Does the person or group in control of a situation have the data or knowledge to support an effective decision?
- 2. Does the person or group with the data or knowledge have control to make a decision?
- 3. Are there interactions when the data and control don't reside with the same person or group?

Communications

Communications in the NAS have evolved to produce diversity. For instance, the ATCSCC provides a regular Strategic Planning Team (SPT) teleconference to allow the Command Center, ARTCCs, airport tower and airline representatives to interact in real-time and discuss plans to respond appropriately to forecast weather and other events that will impact traffic. Increasingly the ATCSCC is using the Web to broadcast procedures as well as to provide realtime data such as traffic advisories (Federal Aviation Administration, 2001). The Collaborative Convective Forecast Product (Aviation Weather Center, 2001) is used to issue 6 forecasts a day after chat discussions between the FAA and airline's weather experts. ACARS is used to transmit text-based messages to and from flight crews. Email is used to transmit messages and documents within organizational networks and on the Internet. The impact of the communications method on a particular task in a particular context is important.

Two general questions that can be asked with regards to communications are:

- 1. Does the system reliably and efficiently support the transfer of relevant and meaningful data?
- 2. Is the information and data exchange synchronized with the collaborators' tasks?

The Post Operations Evaluation Tool

Introduction

The FAA's Collaborative Decision Making (CDM) program (Metron, Inc., 2001) has been developing a collection of tools that help groups in the NAS collaborate to reduce congestion and improve NAS efficiency. One such tool is the Post Operations Evaluation Tool (POET). It allows users to review an individual flight or collections of flights and display flight details and performance metrics including departure, en route, and arrival delays and filed versus actually flown flight tracks. The evaluation tool's server is installed at the Air Traffic Control System Command Center (ATCSCC) with access to archived FAA Enhanced Traffic Management System (ETMS) data that is updated on a daily basis. The Post Operations Evaluation Tool has the capability to integrate FAA data with airline provided flight data (such as the predicted vs. actual fuel consumption, or taxi in and out times) to give a more complete picture of what is happening in the NAS. Thus, because its information comes from data obtained from both the FAA and the airlines, it can provide information to both groups, and because it allows flights to be aggregated it can also be used to reveal patterns that might otherwise not be detected.

Conducting a Search



Figure 4.3 The search builder window

Figure 4.3 shows the Search Builder window, which is used to create the initial query of the database. The Search Builder window is automatically opened when POET is started. Building a search involves selecting or entering items in the fields under the Selection Criteria, Flight Grouping, or Data Mining tabs. The Run Search button at the bottom of the Search Builder window is for carrying-out the search. At any time the Cancel button may be used to cancel the search building and return to the previous screen. Clicking the Reset button returns all Search Builder fields to their default states. A search summary representing the current state of the Search Builder tabs appears at the top of the Search Builder window. As entries in the different fields are changed the search summary is also changed automatically.

The same search summary is displayed on many other POET screens to identify the current search. The **Selection Criteria** tab allows a search to be focused by restricting the flights returned to those that have a particular set of characteristics. In the example shown in figure x, the flights returned are restricted to those flying from PHX to DFW, departing between 6/9/98 and 6/16/98 that were filed through UKW3, the northwest cornerpost for DFW. More specifically, each criterion is defined as follows:

- The Minimum Number of Flight Instances field specifies how many instances of a flight must match the conditions specified in the Search Builder window for that flight to be included in the results. The default number is 1, but this default may be changed by selecting <u>F</u>ile from the Menu Bar, then selecting Poet Properties..., then selecting General Options, and finally typing the new number in the **Default Minimum Number of Flight Instances** field. (In POET a flight is defined as a collection of flight instances that occur day after day, such as a month of flights from PHX to DFW scheduled to depart at 1437Z.)
- The Airlines field is used to specify the airlines to be included in the search. More than one can be selected.
- The **Departure and Arrival Airport** fields are used to specify the departure and arrival airports to be included in the search, respectively.
- The Aircraft Type field is used to specify the aircraft types to be included in the search. More than one type can be selected.
- The NRP field is used to specify whether flights that were filed under the North American Route Program are to be included in the search or not. The value for this field can be All, NRP Only, or non-NRP Only.
- The Departure Date Range fields are used to specify a block of consecutive departure days to be used in the search. A matching flight instance must have departed on or after the From date and before or on the To date. The form for both the From and the To field is mm/dd/yy (e.g. 4/1/98, 04/01/98).

64

- The Scheduled Departure Time(Z) fields are used to restrict matching flight instances to those that have a scheduled departure time on or after the From time and before or on the To time. In both cases Zulu times are used.
- The Scheduled Arrival Time(Z) fields are used to restrict matching flight instances to those that have a scheduled arrival time on or after the From time and before or on the To time. In both cases Zulu times are used.

The Filed Route Segment(s) field defines a comma separated list of route segments that must match the route filed. Examples:

TXOTXO included in routeTXO.UKW3TXO.UKW3 included in routeJEN, UKW3JEN or UKW3 included in route

• The Flight Number(s) field defines a comma separated list of flight numbers. A matching flight instance's flight number must be one of these.



Figure 4.4 The flight-grouping tab

The **Flight Grouping** tab shown in Figure 4.4 is used to decide how flight instances found by the search will be grouped in the Flight Table. With the default columns (Airline, Departure Airport, Arrival Airport, Departure Time, and Aircraft Type) selected, groups are produced that each have a unique combination of values for these criteria, and where there is at least one flight instance for the group found by the search. If one of these criteria is removed as a column in the table, flight instances will no longer be grouped separately by that criterion.

The **Columns** frame is used to make changes to the list of selected columns and the **Flight Table** below it shows how the Flight Table will appear with the current settings. Double clicking on a field in either the **Available** or the **Selected** list will move that field to the other list. Alternatively, after clicking once on a field in the Available or Selected list the right arrow or left arrow button, respectively, may be

used to move that field to the other list. To move all fields at once the double right arrow or double left arrow button can be used. The down arrow and up arrow buttons may be used to change the order of the selected fields by highlighting a field and using the appropriate arrow. Note: the last four columns shown in the Flight Table above, (Performance Data, Actual, Planned, and Difference), are actually fixed as these fields are always used in flight tables.

Data Mining

Additionally, there is built-in a collection of data mining tools to assist the user in recognizing patterns and trends within the data. Some of the patterns currently recognized include *circular airborne holding*¹⁰, arrival fix swaps, and flown routes that differ significantly from the routes filed.

Clicking on the **Explore Data Mining** button in the Flight Table window produces the Data Mining environment for the flight instances included in the selected row of the Flight Table when the button is used.

¹⁰ The holding pattern is actually more "race track" shaped.



Figure 4.5 The data mining environment

The data mining environment includes a Flight Display Panel and a Algorithms Selection Panel, as shown in Figure 4.5. The display panel shows the tree structure of Flight IDs (FIDs) and a 2-D map view of flight routes. The Algorithm List shows all the algorithms available. A algorithm is selected by clicking on it. The Options panel displays the available options associated with the current algorithm. Below the Algorithm List and Options panel a brief description of the algorithm is displayed in the Description Text Box. Below this are a number of buttons for running the algorithms and managing the results. Below this the Current Algorithm field displays the algorithm or algorithms that have been run and for which results are displayed in the Flight Display Panel. Algorithms may be run sequentially. When this is done, the Data Miner continuously clusters the

results of prior algorithms. The Current Algorithm displays the sequence of algorithms that have been run,

separating each with a colon (:).

Data Mining Algorithms

Spatial Clustering of Filed Routes

Groups routes based on the similarity of their filed flight plans. Similarity is determined using the Spatial Similarity Algorithm (SSA). Filed flight plan information is contained in the RT message portion of the POET database.

Options: Track Variance

Spatial Clustering of Flown Routes

Groups routes based on the similarity of their flown flight paths. Similarity is determined using the SSA. Flown flight path information is contained in the TZ message portion of the POET database. *Options*: Track Variance

Compared Filed to Flown Routes

Compares a flown flight path to a filed flight path and flags the flown path as either similar or dissimilar. The SSA is used to generate a variance, which is compared to the user specified threshold, and perform the grouping.

Options: Track Variance

Compared Filed to Flown Routes, Corrected Origin and Dest

Since TZ messages are generated while the aircraft is actually in the air, the origin and destination point may not be contained in the flown flight path. This may lead to erroneous results when comparing the flown and filed flights. This algorithm is virtually identical to Compared Filed to Flown Routes with the exception that the origin and destination points are inserted at the beginning and end if not found in the TZ message.

Options: Track Variance

OOOI Time Delay Clustering

OOOI is an acronym for Out of Gate, Off the ground, On the ground, Into the gate. In each instance the algorithm calculates the difference or delay, in minutes, between the filed plan and the actual event. A single parameter (Delta Time) is specified as a threshold. Routes are then identified as falling under the threshold delay or equal to/greater than the threshold delay. The OOOI options are mutually exclusive choices which can be chained together by running them sequentially. *Options*: Out Time On Time Off Time In Time Delta Time

Compare Cruise Route

Analyzes amendment messages for chosen routes to determine whether any changes occurred in the aircraft's flight plan. It compares the FZ flight plan message against the AF messages and returns as soon as one of the following determinations is made: (a) the flight was changed on the ground, (b) the flight was changed in air, (c) the flight was diverted, or (d) the flight was not changed. This analysis occurs outside the SID and STAR portions of flight, i.e. for the cruise portion only. *No Options*

Length Comparison of Filed to Flown

This is a straightforward algorithm that compares the sum of segment lengths of the filed route and the flown route. It creates three groups: Longer, Equivalent and Shorter. A route is Longer if its filed length is (1 + Longer%) percent longer than the Flown route length. A route is Shorter if its Filed route length is (Shorter%) percent shorter than the Flown route length. Otherwise, the route is marked Equivalent. Options: Longer % Shorter %

Efficiency Determination Based on Delay and Fuel Burn

One measure of efficiency as applied to aircraft routing is airborne delay and fuel burned. A simple measure of efficiency using airborne delay and fuel burned is to define a straight line function with percent more fuel burned and delay in minutes. The vertical axis (fuel delta) is the difference between actual and planned fuel burn, while the horizontal axis (airborne delay) is the difference between actual and planned flight time. The line connecting the threshold fuel burn (Fuel Burned %) and delay (Delay (mins)) values marks the boundary between inefficient and efficient flights. All points that fall above the line are deemed inefficient while all points below are deemed efficient.

Options: Delay (mins) Fuel Burned %

Airborne Holding



Figure 4.6 Portions of a holding pattern (Stern, J. M. 1998)

Airborne holding occurs when a flight is put into a temporary race-track shaped flight pattern, as shown in Figure 4.6, in order to achieve better spacing, wait for weather conditions to improve, or absorb a delayed approach into an airport. Holding normally occurs relative to previously established holding fixes that appear on aviation charts. Because instances of holding are not normally captured as such elsewhere in the NAS, an algorithm needed to be developed for POET to automatically detect holding based on the flight path actually flown. The algorithm applied to each individual flight uses a time ordered sequence of x-y coordinate position values, time value, and Center value. The algorithm is essentially as follows:

- Approximate the path flown between individual "radar hits" with straight lines and consider one line crossing another as potential evidence of a flown holding pattern, with the intersection close to the holding fix.
- 2. In order to avoid false alarms the following was also required to consider holding to have occurred:
 - a. The intersection is ruled out if it occurs within 30 miles of the origin and 10 miles
 of the destination, as a flight path may cross itself during arrival or departure.
 - b. The time of radar hits are specified in minutes. Because the merging of data from radar in different Centers' regions was found to generate false alarms close to the point where a flight path leaves one Center region and enters another, some points were removed as a simple method for "cleaning" the input stream of flight path portions and speeds that would be very unlikely or impossible.
 Specifically, if the time sorted list of points produces a point from one Center, then from another, and then from the first Center again, the second Center's point is filtered out. When both the next points are from a different Center than the current one, the next point is only accepted if it is at least 3 minutes later than the current one.
 - c. The area created within the "loop" detected by the algorithm has to be at least 10 square miles.

Because this is a data-mining algorithm (i.e. applied to a very large number of records in a database), efficiency of the algorithm is important to reduce run-time. Thus, in the future there might be some benefit in developing a list of holding fixes for each city pair to simplify the search algorithm. This might also improve the accuracy of the algorithm, as it is currently conservative in detecting holding. At present a short hold can easily be missed if the sampling of points does not happen to generate an intersection when those points are connected. A "spike" in the flight path close to a known holding fix might be sufficient to accurately detect holding.

POET Results



Figure 4.7 Repeated pattern across flight instances

The ability to see multiple instances of flights with a common set of attributes over an extended period of time is useful for finding what is systematically happening to those flights. Figure 4.7, for instance, shows quite clearly a pattern of deviations to the south from the planned route for a particular flight over a three-month period. The ability to plot performance metrics over time can reveal clusters of flights in a particular time period, and also indicate how an average performance value is found in a representation much more meaningful to the average user than, for instance, an indication of mean and standard deviation. The tabular data can also be sorted by performance metric values to allow the more extreme flights groups to be investigated first as a simple searching heuristic.

	Window Help					
E Sector Forward Wor		Open Sav	Note			
Contraction of the second seco	ccess): Minimisferf	1-1- Deles -16/19	V1998 To 06/16/1	SR: DepartTime (000/1o2400:	
Tetal Flatta (Anim Department)	ntwe Aispert Anival / 9	ispert.Departur		, 10		
Performance Metrics	Min Avg Ma Difference(A-P) Difference	nce(A-P)	1			
Off Time (mins) On Time (mins)	5.0 7.0	_63.0 63.5				
The strangest reading and Designation	en er an andere	A States (States			4/26/99	449 EM

Figure 4.8 The performance summary table

When a successful search is complete a performance **Summary Table** appears, such as shown in Figure 4.8. The search summary in the **Search** field is the same as the corresponding Search Builder window. The **Total Flights** field contains the number of flights that were returned by the search. The **Total Instances** field is the number of instances of these flights.

The Air Time is the time between the takeoff and landing. The Off Time is the takeoff time. The On Time is the landing time. The Minimum Average Difference is the minimum average difference between the actual and predicted performance parameter values of the flights found. The Maximum

Average Difference is similar to the Minimum Average Difference, except that it is the maximum value rather than the minimum that is shown. Thus, if there were two flights found in a search and one had an average air time difference for its flight instances of 3.1 mins and the other 5.7 mins, the Minimum Average Difference on the Air Time row would display 3.1 and the Maximum Average Difference 5.7. (Note for the air time the difference between the actual and planned is shown as a percentage of the planned.)

Double clicking on a row in the table produces a more detailed table focusing on the respective performance data parameter and automatically opens the **Displays** window with the **Scatter Chart** and **Bar Chart**. In general, in POET, double clicking on any flight table row will produce more detail on the information shown in that row.

Edit: View - Gio- Addine: Window Help Statu: - Forcard: Watspace Now Open Statu: - Forcard: Watspace Now Open Statu: - Forcard: Watspace All: FAX: Data Only/Access:: Minimum Fight: Statu: - Forcard: Departure: All: Departure: All: PHX: Departure: Anive: All: PHX: DPAV: 1913 MD80 2 All: PHX: DPAV: 1913 MD80 4 All: PHX: DPAV: 1913 MD80 4 All: PHX: DPAV: 1913 MD80 4 A	POET									_ 6
Back Proved Workspace Now Open Seve Note Seve Now Open Seve Note Note Seve Now Open Seve Note Seve Now Open Seve Note Seve Particle Seve Nove Seve Seve Particle Seve Seve Seve Seve Seve Seve Particle Seve <	Edit View	io Add Ins	Window H	n				(11) (11) (11) (11) (11) (11) (11) (11)		
Construct of common Plant Commo	Back - Forver	and Works	pace			Note				
Active Condition State Condition of the Condition of	Storage F. L. 1986	Southition y	Eliophi 1							
Fighter Date Maining Departure Annual Departure Annual Departure Accual Acc	ecetion:Seach1	sch: Summer	Fights1							
Explore Data Mining. Departure Arrine Arrival Arrive Arrow Departure Arrow Accual Type Number of Instances Performance Metrics Planned Accual (Actual Planned) Adl BBAS DFW 2027 MD80 4 Arri me (mins) 1035 122.0 12.5 11.42 A Adl BBAS DFW 1913 MD80 2 Air ime (mins) 111.5 122.0 12.5 11.42 A Adl BBAS DFW 1913 MD80 2 Air ime (mins) 111.5 122.0 12.5 11.42 A Adl BBAS DFW 1913 MD80 4 Air ime (mins) 105.8 114.8 3.0 8.52 Add FIRE DFW 120.0 16.5 8.5 7.52 Scatter Chaet Bar Chaet 4 Air ime (mins) 108.0 116.5 8.5 7.52 Scatter Chaet Bar Chaet 4 Air ime (mins) 108.0 16.5 8.5 7.52 Scatter Chaet Bar Chaet 4 Air im	eerch: FAA D Deale	ata Onty(Ac LRIX: AmAp	Cores): Minda CDFW: see:	etaferfit 1;	Deles: 06/09/1	998 To 06/16/1998: Depa	tTime:0001	o 2400.		
Arrivel Arrivel Arrivel Arrivel Arrivel Arrivel Arrivel Arrivel Arrivel Arrivel Arrivel Arrivel Arrivel DFW Departure Time (Z) Arrivel Type Number of Instances Performance Metrics Planned Actual (A	Explore Date	a Mining								
AL PhX DFW 2027 MD80 4 AirTime (mins) 109.5 122.0 12.5 11.4% AL PhX DFW 1913 MD80 2 AirTime (mins) 111.5 122.5 11.0 9.93 AL PhX DFW 1440 MD80 4 AirTime (mins) 105.8 114.8 9.0 8.52 AL PhX DFW 2205 MD80 4 AirTime (mins) 108.0 116.5 8.5 7.92 AL PhX DFW 2205 MD80 4 AirTime (mins) 108.0 116.5 8.5 7.92 Cortch1 Schter Onet Bar Chest Bar Chest V//// V///	Aitine Airport	e Anival Airport	Departure Time (Z)	Aircraft Type	Number of Instances	Performance Metrics	Planned	Actual	Difference Actual - Planned	
All time (mins) DEW 2205 MD80 Aut time (mins) DEW 116.5 8.5 7.92 control 1 och Summeriv: ElightettetDisplaye Aut time (mins) 108.0 116.5 8.5 7.92 control 1 och Summeriv: ElightettetDisplaye Aut time (mins) 108.0 116.5 8.5 7.92 Scatter Chait Bar Chait Aut time (mins) 108.0 116.5 8.5 7.92 Scatter Chait Bar Chait Aut time (mins) 108.0 116.5 8.5 7.92 Scatter Chait Bar Chait Aut time (mins) 108.0 116.5 8.5 7.92 Scatter Chait Bar Chait Aut time (mins) 108.0 116.5 8.5 7.92 State	AL - PHX	DFW DFW	2027	MD80 MD80 MD80	4	AirTime (mins) AirTime (mins)	109.5 111.5	122.0 122.5	125 11.0	11.4 % ▲ 9.9%
Scattel tech Summery: Flight=1: Display: Scatter Chat Bar Chat Str Asis 518.0 19	AL PHX	DFW	2205	MD80	4	Air time (nims) Air Time (mins)	108.0	116.5	5.0 8.5	8.54 7.9 2
Scatter Chart Bar Chart X/Y Avii: 518.0 518.0 160.0 150.0 140.0 120.0 100.0 100.0 100.0 54004/5/1998 06/4 5/1998 06/15/19	earch1 ±ch; Su	mmary: Fligt	hts1: Displa	ys						
XY Ase \$16.0 16.0 16.0 12.0 10.0 10.0 2400/5/1998 06/15/1988 06/15/1988 06/15/1988 06/15/1988 06/15/1988 06/15/1988 06/15/1988 06/15/1988 06/15/1988 06/15/198 06/15/1988 06/15/1988 06/15/1988 06/15/	Scatter Chart	Bar Cha	• <u> </u>							
160 5140 120 100 510	X/Y Aves									
2140 120 2100 2100 2100 2100 20015/1998 06/15/1988 06/100000000000000000000000000000000000	E 160									
E-120 	414.0									
200/5/1998 06/15/19	₹120 ₹100									
	06/15/19	9806/15/1998	06/15/19980	6/15/199806/	5/1998 06/15/1	998 06/1 5/1 998 08/1 S/1 998 1	06/15/1990 06/	5/199806	15/1998 06/15/19	9806/16/19
	STREET ANDERS	1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1		. A 1. 1. 1993	1.4.4.4.6.4.7		3 - 12 - F	123522		377 - 1

Figure 4.9 Flight level details

After double clicking on a row in the Summary Table, a **Flight Table** focusing on the respective performance data parameter is produced. In the example shown in Figure 4.9 Air Time (mins) was selected from the Summary Table, thus the **Planned** and **Actual** fields represent average planned and actual air times. Each row in this table is a flight and the number of instances of each flight are shown in its **Number of Instances** column. By default, the rows are sorted in descending order by the **Difference** field. The **Airline, Departure Airport, Arrival Airport, Departure Time,** and **Aircraft Type** fields appearing in the table are the default fields that may be changed by using the Flight Grouping tab in the Search Builder window.

At the end of a row there will sometimes be one of three special symbols. A \blacktriangle means that there is at least one filed route for the flight that has a statistically significant high average difference (actual planned) for the instances of that filed route. Similarly, a \checkmark means there is at least one filed route that has a low average difference. A \blacksquare means there is at least one flight route that has a significantly high average difference and at least one flight route that has a significantly low average difference. The Filed Route Table reveals specifically, which filed routes where statistically significant. Double clicking on a row in the Flight Table produces the Filed Route Table for that rows flight.

Text Let: View Go Addine Windows Let:	ne co
Back Wolconce New Donn	
🕄 Transfer och Sammary Flaght (* 1997) 1997 - Marine State (* 1997)	
Construct Construct Construction Statements (Fightal U.W.C. Statements Construction Statements Constru	12.0
Structure FAA Date UnigAccess): Minimater at 1 - Date: 05/05/1598 To 05/15/1588; Depart Line: 0001 to 2400-144	
Reflet PK Antel Drv.	
	1997 - K.
Explore Data Mining.	1. 2.
Departure Arrivel Departure Arrivel Trans 7 Ton	
ALL PROTECTION OF THE PROTECTI	x x
AAL PHX 1913 MD907 2 ArTime (mins) 111.5 122.5 11.0 9.9	z :
A A A A A A A A A A A A A A A A A A A	Z i
	े न से हैं।
Search1 sch: Summary: Elights1: Displaye	
	1000
Scatter Chart	A sure of the
	1 2 1
	21 E S
120	s 84
	<u></u>
2/2-7/00	
	김태명

Figure 4.10 The scatter chart window

A Scatter Chart, such as shown in Figure 4.10, appears automatically with the Flight Table when a performance parameter is selected from the Summary Table. The Scatter Chart presents the performance parameter data for each flight in a graphical form that helps in the detection and visualization of relationships between parameters and unusual values. As a different row is highlighted in the table the Scatter Chart is changed to reflect the new flight's data. Each dot in the Scatter Chart represents one flight instance. If that dot is double clicked, the Flight Instance Window for that flight instance appears.

E LIFE D								a d
	Go Addins Window	Self and the post of		A CANADA		S States		C TO AL P
				D	(z, p, q)	the states	Lan Sa Torr	
Stacks Store	Workspace	New 3	Several Pre	Note			130.762	
📬 Search1 och	r Summary Flights)		·			· · · · ·		E
Landing Seach	IncheSummery Flores				(1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1			12 M
Spector Frant			05/09/1998 10	IS/IS/ISSR Depart		2400-4	1.1.1	
	HIT XYA MARINI NAME		Sec 5				2.25	E.J.
		6			1000			
Explore Da	ta Mining.							-
		Anna B			Time		1-40-54	<u></u>
Anne Anne A	Asport	Type 2 time	lances CPer	omance Metrics	Planned	Actual Actua	Planned	的工作
AAL PHX	DFW 1347	MD80	2	AirTime (mins)	101.5	116.0	14.5	14.3%
ALSERIK		MDBD	2	AirTime (mins)	111.5	122.5	11.0	9.92
MALL S PHOC	DFW: 1440-	MD90		AirTime (mins)	105.8	114.8	9.0	8.5%
			• জনিকাৰ্ডিনেটা জনিকাৰ্ডিনেটা	Arline (mns)	106.0	116.5	8.5 	7.97
lin an blir blir b	and the second second second							
A REAL PROPERTY AND A REAL	ERCONTRA DE LA CAR			Alex Connection	10000	- 		10 A 10 A 10
Scatter Chart	Ber Chart							15
XX Aut		NALSH American Providence			1. A.		2.977	
100 H 100		Server Start Star		Martin States	ير ويو الرائد	2		
					algen .			
		1.15 1.2.3			5.00			1. A
								2
	1			175- 1842 EV				
10.0 T	0 10.8-10.8 To 11.5-	124 To 137		140.To 148	156 T	164	17210	180
and the special and and	وجريان سرور ماي حريز وتوري فالمع	in the second second second		and a survey of the		A/24/99	11-59	
and a second	A consideration of the second second second	The second se	HARD THE REAL PROPERTY OF		MARCH STREET, STRE		Carlo Balling	and the second second

Figure 4.11 The bar chart

A **Bar Chart**, such as shown in Figure 4.11, describing the instances of a flight can be produced by clicking on the "Bar Chart" label at the top of the Bar Chart tab. The Bar Chart's y-axis is always the number of occurrences in the performance parameter of the values shown on the x-axis. Thus in the example shown in figure x, for instance, there were two instances of the flight departing at 2027 that took between 10 and 10.8 minutes longer than planned to fly from PHX to DFW.



Figure 4.12 The chart builder window

Clicking the X/Y Axis button on either the Scatter or Bar Chart produces the Chart Builder window shown in Figure 4.12. This can be used to change the parameters used on the x and y-axis for the Scatter Chart and the y-axis for the Bar Chart. In both cases the y-axis is the Planned, Actual or Difference (Actual - Planned) for a performance parameter. The same options are available for the x-axis of the Scatter Chart, plus the Date option. For the Bar Chart the x-axis is always the number of occurrences.
Second State Description Description Description Description Second State New Open Second State Note: Note: Second State Second State Note: Note: Note: Note: Second State Second State Second State Note: Note: Note: Second State Second State Second State Note: Second State Note: Second State Field Route Number of Instances Performance Metric Performance Metric Performance Metric Actual Optimics Field Route 1 Artime (mins) 113.0 123.0 10.0 883		CONTING OF
Forward Workspace New Open Save Note Save LL & A commune of Legit Legit Route Save Note Save LL & A commune of Legit Legit Route Save Save FAA: Date: Object Commune of Legit Legit Route Faa: Faa: Commune of Legit Faa: Number of Instances Performance Metrics Fied Route Instances Performance Metrics Fied Route 1 Artime (mins) 113.0		200
Support Line (Support Line (Support) Line (Support Line (Support	Back Montage Wolkspace	िईउल
FAA: Dute: Obj/Accessel Lindental and E. J. Date: US (S/138) (c/D/156) (SS: Descat Fire: C/D/152) (c/D/157) Factor: Date Conjugation of E. J. Date: US (S/138) (c/D/157) (c	R Service March Soundhary (Elegent S. Faber) Rouger (
FAA: Date:/ Only/Accesse) Historial and Fail Date:/ OD/Accesse) Historial and Fail Definition Definition Definition Definition Definition The provide of the second secon		1.100
Explore Data Merror Number of Instances Performance Metrics Planned Actual Difference instance Field Route 1 ArTime (mins) 113.0 123.0 10.0 8.82	Neth-1 FAV Date (Indeferrent Malanda Farit 10 date (16/19/1984) (205/16/1984) (Depart Fare (1004) (2/2015)	
Englister Data Mining Number of Instances Law Contraction Difference Difference <thdiference< th=""> Difference <thdiference< th="" th<=""><td></td><td></td></thdiference<></thdiference<>		
Field Route Number of Instances Performance Metrics Planned Actual Otherance Field Route Instances Performance Metrics Planned Actual (Actual - Planned) Actual	Explore Data Minnya	
Proceedings and the second sec	Number of	
PRAEDISCH1:CIE-12:EIE-150UNC:IEN3.0EV4:22:51 1 AirTime (mins) 113.0 123.0 10.0 8.82		
	HIGHENDEDHICTER/SEEP/SOUNCIEN/S	1 2
		555
TAA Tiel Inst	Actual Book	New F
TAA Tied Book		a straight
Accord Books (Marcold Sector) (Marcold S		10
		TA IN
		*
		Marshare.

Figure 4.13 The filed route table and the flight route map

The Filed Route Table, shown in Figure 4.13, is produced when a particular flight is selected from the Flight Table. Each row represents a different filed route. The route itself appears in the first column and the number of flight instances with that filed route for this flight is shown in the Number of Instances column. The performance columns are the same as in the Flight Table. The routes actually flown for a filed route can be produced by double-clicking on a row in the Filed Route Table. The Flight Route Map, also shown in Figure 4.13, is displayed automatically with the Filed Route Table when a flight is selected from the Flight Table. At that point, only the filed routes will be displayed on the map. The one currently highlighted in the Filed Route Table will be drawn as a series of connected lines that are a thicker green than the other filed routes.



Figure 4.14 The expanded filed route table

If a flight route is double-clicked in the Filed Route Table the individual flight instances for that particular flight also appear to produce an expanded Filed Route Table, as demonstrated in Figure 4.14. More than one flight route can be expanded in this way and instances can be hidden again by simple double-clicking on the instances' filed route. The **Actual Route** flown for each instance displayed in the Filed Route Table is shown with black lines in the Flight Route Map. These lines are thicker if they are for the route currently highlighted in the Filed Route Table.



Figure 4.15 A single flight instance is highlighted

If an individual flight instance is highlighted in the Flight Route Table the respective route flown in the Flight Route Map is redrawn with a blue line and all other routes flown are shown with thin black lines. If a route flown is clicked on in the Flight Route Map the corresponding instance in the Flight Route Table becomes highlighted, as shown in Figure 4.15. More than one instance can be selected in either the Flight Route Table or Flight Route Map by holding the Ctrl-key and clicking on multiple instances. Double clicking on a flight instance in the Flight Route Table produces the Flight Instance Window for that flight.



Figure 4.16 The flight instance window

The Flight Instance Window, shown in Figure 4.16, focuses the search results on a single flight instance. All the available data for this flight instance is displayed as demonstrated. The Flight Instance Window is produced by either clicking on a flight instance entry in an expanded Filed Route Table or by clicking on a flight instance's dot in a Scatter Chart.

Using POET Results



Figure 4.17 Influence diagram for POET performance metrics

The Post Operations Evaluation Tool provides individuals with more data and useful representations for aggregated data, but this system reveals *what* happened to flights and the *why* must be inferred from the data. The influence diagram in Figure 4.17 demonstrates some of the factors that can influence a flight and thus the performance metrics shown in POET. A person analyzing a set of performance metrics may apply backward or forward chaining reasoning to determine what could have produced the results he or she is concerned with, but one reason the process shown is an oversimplification is because it doesn't represent the decision making process that the Centers use and the interactions between traffic flows.

The knowledge necessary to determine actions to improve performance for the flights POET helps identify is sometimes distributed. Communications may therefore be necessary among FAA traffic managers at the ATCSCC and enroute centers, and personnel at AOCs (Airline Operations Centers) to learn what actions might actually be viable. The graphical and tabular nature of much of the information presented in the Post Operations Evaluation Tool means that discussion of its results must support discussion of data formatted that way. In addition, when proposing actions to improve the performance of flights identified with this tool an environment is necessary that accommodates expression of those actions.

Common forms of communication between these organizations are E-mail, facsimile messages, voice-mail, and regular mail for asynchronous communication, and telephone (including multi-party telecom) for synchronous communications. While there are clearly some advantages to real-time communications, these interactions often need to be asynchronous in order to be practical. The efficiency and effectiveness of available asynchronous communication methods is therefore important for post operations analysis.

In their analysis of cooperative problem solving for flight planning of commercial aircraft, Smith, et. al. (1995) point out the benefits of the knowledge sharing that takes place between the AOCs, enroute centers, and the ATCSCC by telephone and teletype. They also state that three goals could potentially improve cooperation even further:

- Cooperative technology should help facilitate information exchange between individuals at very diverse locations and with diverse skills, goals and backgrounds.
- Cooperative technology should use intelligence in ways which aid human decision makers without sacrificing their goals and needs.
- Cooperative technology should have the capacity to help individuals build interpersonal relationships with the people that they will work and coordinate with.

In addressing the first goal the value of potentially sharing displays is discussed, including the following quote from a participant in their study: "A phone call's time consuming. A picture is worth a thousand words." Given that different organizations will often have different information or the same information represented differently, the ability to reference the same displays while their significance is discussed is useful in trying to form shared mental models.

84

The second goal is consistent with the approach taken with the Post Operations Evaluation Tool. This tool reduces problem solving search processes by number crunching to sort and aggregate large amounts of data within flexible categories determined by <u>the user</u>, and <u>the</u> <u>user</u> selectively applying heuristic data mining algorithms to further avoid what might otherwise be a relatively blind search.

In discussing the third goal, it was suggested that telephone conversations would be better for building interpersonal relationships than typed messages.

CHAPTER 5

THE DESIGN OF THE COLLABORATIVE SLIDE ANOTATION TOOL (C-SLANT)

Introduction

The Post Operations Evaluation Tool supports post operations analysis, but has no builtin mechanism for communicating about the results produced. Creating such a mechanism was considered because it would allow queries, data structures, and other context specific information from POET itself to be transmitted, and thus allow the recipient (or the recipient's software) to use that information to interact further with POET. However, this (1) would have complicated the development of POET; and (2) would have removed the ability to create messages on computers not running POET. A separate program called the Collaborative SLide ANnotation Tool (C-SLANT) was therefore developed to provide enhanced collaborative analysis considering the working environment of those who need to look at the results of an analysis and to use their knowledge to provide input. C-SLANT was developed to support asynchronous discussion of Post Operations Evaluation Tool results, supplemented by related data such as weather, after currently available asynchronous communications methods (normally E-mail and facsimile) were ruled out because they were not considered sufficiently efficient and effective.

Figure 5.1 shows a typical sequence of events to create a C-SLANT message and for a recipient to respond to that message. Alternatively, a C-SLANT message can be posted on a website and viewed on demand, by anyone with access to the site and a copy of C-SLANT.



Event	Description
	Using C-SLANT the context of a message is formed by taking a series of computer
L .	screen snapshots and/or by loading previously created graphics files
3	The images are annotated with synchronized voice and pointing; pen marks; text-based
2	notes, and arrow markers
3	The slideshow is saved
4	The annotated message is added to an E-mail message as an attachment
5	The message is sent through the user's E-mail system
6	The annotated message is loaded into C-SLANT via the recipient's E-mail system
7	The recipient views and listens to the message
0	The recipient possibly interacts with other software to gather screen captures to be part
0	of the response
9	The recipient creates an annotated response
10	The recipient saves the message with both sets of annotations
11	The annotated message is added to an E-mail message as an attachment
12	The message is sent through the recipient's E-mail system

Figure 5.1 A typical C-SLANT event sequence

Two constraints imposed by C-SLANT are that a message must be built on top of one or more images and annotated via the C-SLANT annotation tools (or by visual annotations placed on the image through another application before it is captured or imported). Otherwise, the message is relatively unstructured in that there are not any "fields" that must be completed by the user.

Requirements

The following requirements were placed on the design of C-SLANT, given the context of the intended use, described in the previous chapter:

• To not be over intrusive;

Both the FAA and airlines' personnel experience fluctuations in workload as weather severity varies, traffic density varies, and other events occur causing spontaneous problem solving to be required. Sometimes individuals have sufficient information about the procedures employed and the current conditions to estimate when not to initiate a synchronous communication (for instance, by telephone) but this can not always be the case. In addition, because the traffic manager will normally be in control of the situation, the dispatcher (or more likely ATC Coordinator) has added incentive to request an action at a time when the workload of the traffic manager is relatively low. In real-time dynamic situations the potential benefit will override the cost of communicating at a bad time for the traffic manager, but when the topics for discussion are relatively strategic, rather than tactical, the temporal flexibility means this is not necessary.

• To support annotation of Post Operations Evaluation Tool screens and related data from other applications that may not be available to all parties communicating;

It was anticipated that the Post Operations Evaluation Tool would be made available to each of the enroute centers as well as the command center, and that possibly the major airlines would have access to it, but this was not certain. The Post Operations Evaluation Tool does not have weather data, which would be very valuable when evaluating what happened on individual days, so that information might be one example of what could be combined with POET data. If an analysis was focused on a particular airport, then additional information about that airport might likewise come from another source (such as a CTAS display). It general, because these are separate organizations, they have different information and information represented differently in their respective software applications. Thus, the ability to easily take a snapshot from any software application was highly desirable.

• To be easy to learn and use;

Although computer systems are widely used within the FAA and the airlines, it could not be assumed that users would be highly computer literate in general. In addition, the frequency with which it could be used was not clear, so it would need to be as intuitive to use as possible.

• To support a core set of common annotation techniques;

Programs with features for annotating vary in exactly what options they provide, but they normally include the ability to type and draw, underline or highlight. They also normally do so in more than one color. Supporting these features would allow transfer of knowledge from other applications and also support basic annotation methods.

• To support synchronized voice and pointing annotations;

Assuming a person is not intimidated by having his or her voice recorded, the ability to quickly create an annotated message over computer screen snapshots seemed much more efficient and effective than trying to use traditional software with annotation features or fax machines. Because employees of the FAA and airlines often have their voices recorded it was assumed this group would be comfortable having their voices recorded.

• To only require inexpensive and easily obtainable technology.

In order to support installation of the system with minimal equipment and labor requirements, and to allow a critical mass of users to develop quickly, it was necessary to keep the equipment needs to a minimum. In general, it was also highly desirable to require no special equipment, so that there would not be any unpredicted side effects to current FAA and airline applications if C-SLANT hardware was added to those systems.

89

Requirements	Comments
To not be over intrusive	Asynchronous messages that can be sent as an E-mail attachment or posted on a website.
To support annotation of Post Operations Evaluation Tool screens and related data from other applications that may not be available to all parties communicating	 A set of screen images may be captured to allow the information in the Post Operation Evaluation Tool's graphical and tabulated data displays (or other software's displays) to be the context of a message. Visual annotation options were designed to contrast with these display backgrounds
To be easy to learn and use	 A slide show structure was used to organize these images, as the concept was considered to be simple and ubiquitous, even if the user is not familiar with the most commonly used software implementation of it - namely, Microsoft PowerPoint. No attempt was made to provide extensive slide formatting functionality as in Microsoft PowerPoint. While it could be
	 used to create one-way communications, C-SLANT was not designed to be a rich presentation system, but instead a simple, but effective messaging system. A small set of interface options, that are predicted to serve the novice-users requirements are visible at all times, and more advanced and less frequently used configuration options are available through pull-down menus.
To support a core set of common annotation techniques	 Pen marks support common annotations such as underlining, highlighting, circling, drawing an object, or a path The arrow marker has been shown to shift fixations to what the arrow points to (Faraday & Sutcliffe, 1998) and this complements the pen, whose markings could be confused with routes if used on the route maps. Text boxes allow short comments or extended narratives as the text box window will scroll automatically when necessary
To support synchronized voice and pointing annotations	 A recorder is included that has the same functionality as an audio-only recorder, except it will also record the current x-y position of a special pointer for deictic gestures as that audio is recorded A subject can be saved for each recording to label that recording

Table 5.1 Requirements for C-SLANT

(Continued)

Table 5.1 (Continued)

To only require inexpensive and easily obtainable technology	 Support for a very authentic virtual reality environment was not considered because of the goal of an inexpensive system involving easily obtainable technology and because it was not clear such a high level of virtual "presence" in a 3 dimensional environment is necessary for these tasks. One might also question the social practicality of wearing, for instance, a head mounted display and data glove in this context. C-SLANT is a Microsoft Windows application requiring a sound card and microphone, but no other specialized hardware.

Interface Design Features

Figure 5.2 shows a background image from a Post Operations Evaluation Tool screen capture; the Slide/Annotation Manager window, which contains the majority of the interface options in C-SLANT; the deictic gesturing pointer which is always visible when the point is above the captured image; the voice and pointer recorder with a 7 second recording playing; a single text box window; and an arrow marker.



Figure 5.2 C-SLANT's slide editing mode

The Slide/Annotation Manager Window

••			
Flex Edit N	liews Option		
		175 200	
Goto Laplure			ide Hecordings

Figure 5.3 The slide / annotation manager window

The slide / annotation manager window shown in Figure 5.3 can be moved anywhere on the screen and its location is saved when a slide is saved, allowing it to be positioned so as to cover the least important data or an unused part of the screen.

	Move to the previous slide
	Move to the next slide
	The recorder allows the user to make an voice annotation and also have any pointing saved and synchronized with the audio so that when playing back the recording the mouse pointer movements are seen as they were made during the recording
	The text box allows a text comment to be placed above a slide
	The arrow marker allows arrows to be placed on a slide to mark particular points on the slide
	The pen allows freehand markings to be placed on a slide
	This button opens the slide sorter to allow the order of the slides to be changed
	This button opens the user's default E-mail program with the current slideshow as an attachment
Goto Capture	This button switches C-SLANT to the screen capture mode
	A color palette is provided in C-SLANT to allow the user to change the active pen color, arrow color, or text color
Hide Recordings	This button allows the recordings window to be displayed or hidden

Table 5.2 Slide / annotation manager window buttons

Figure 5.2 shows C-SLANT in its slide-editing mode. Clicking the *Goto Capture* button switches C-SLANT to its screen capture mode, where the small moveable window shown in Figure 5.4 appears above the user's desktop or open application(s). Entire screens or individual windows can be captured using the *Capture* button to produce a new slide. The *Goto Editor* button puts C-SLANT back in the slide editor mode.

1.	t i		
Ele (Options	<u>. Нер</u>	
	Carl		Goto
	Capital		Editor

Figure 5.4 Screen capture window

The same effect can actually be achieved by (1) using the PrintScrn key to take a screen capture (or the Alt-PrintScrn to capture the current window) and copy it to the clipboard, followed by (2) issuing the Paste command in C-SLANT. More experience PC users would know this "trick", but it could not be assumed for less experienced users.

Annotation Tools

Synchronized speech and pointing

After selecting the recorder button in the Slide,'Annotation Management window the window shown in Figure 5.5 appears. After clicking on the record button, C-SLANT records any audio input while simultaneously recording any movement of the pointer relative to the underlying image. Because the only visible dynamic to be saved is the pointer's position, little additional storage beyond the size of the image itself is required to capture this effect. Very long

audio recordings however can still cause a file size problem for low bandwidth communications when the message is sent as an E-mail attachment.

0 mins 9 secs			
			د - محمد به ما اینام به ما است. از 12- المحمد - المحمد المحم
	ind:-EHe	cord	Save Clear

Figure 5.5 C-SLANT's Recorder

A pointer for deictic gestures

Mode errors are possible when annotating over screen images that appear to have interactive components such as buttons, but actually are just portions of the picture. A pointer consisting of a hand holding a "stick" (shown in Figure 5.2) was therefore used. It affords (Norman, 1988) pointing more than button pushing, so it provides mode feedback. The metaphor of a person using a stick to point at a chalkboard also is appropriate. It appears when pointing over the background image, and changes to the normal Window's pointer when moved over an interactive object, such as a button. When the user selects a pen or an arrow marker, the pointer changes to a pen or arrow pointer. When the user moves the pointer into a text box the pointer changes to a standard text-editing pointer.

Providing a temporal-positional frame of reference for deictic gestures

A slider is provided as part of the recorder in C-SLANT so that during playback the user can see the current position in the recording and move forward or back in the recording by sliding the position indicator. A side-effect of synchronizing the pointing with speech is that the pointer's positions are retraced during sliding, so that the user has a frame of reference in the image and not just the normal temporal frame of reference provided by audio only recorders.

The player window

When the user saves a recording on a slide a player window, such as shown in Figure 5.6, is created that can optionally be given a subject (as a visual static description for the dynamics of the audio and pointing) and moved to a position close to that the recording references and/or where it covers the least or no information in the background image¹¹.

Saturday	Jul 14 20	01 19 1	b E
Play F	Stop	brite	0 mins 0 secs
From: Rog		Length	0 mins 46 secs
	a aa <u>ar</u> a		
Subject			

Figure 5.6 The C-SLANT player window

Message chunking

C-SLANT supports the creation of message chunks in the sense that it supports attaching annotation to context framing individual background images. Multiple speech and pointing recordings can be created for a single image where further sub-chunking is required, shown in Figure 5.7, as can multiple text boxes, line drawings, and arrow markers.

¹¹ The player window has been replaced by a multiple recordings composite window (described in chapter 7) in the current version of C-SLANT, but separate player windows for each recording were used in the experiment described in chapter 6.



Figure 5.7 Verbal/deictic referencing abstraction hierarchy grounding asynchronous communications

In Clark and Brennan's (1991) discussion of grounding in communication, they point out how two people working together at the same place and time benefit from the ability of the speaker to check that the person they are talking to seems to understand what is said very shortly after it is said, "They [speakers] can present a difficult utterance in installments and check for understanding after each installment". This is clearly not fully replicated here, because the speaker would likely not proceed to the next installment if the last one did not appear to be understood. However, at least the potential to package and save a "collection of installments" that can be separately responded to allow the speaker to get feedback directly tied to a particular installment if that was something the respondent didn't understand. The analogy to a speaker accepting questions during a presentation versus taking comments and questions at the end of a presentation seems appropriate here. Clark and Brennan also state that the most basic form of positive evidence a speaker gets that they are being understood comes from what appears to be continued attention from the other participants in the communication. In the environment provided by C-SLANT the speaker works in a quasi-joint activity (Clark, 1996) proceeding "as if coordinating actions with another agent". If the speaker is more confident that the recipient's attention will be correctly focused the speaker is perhaps less concerned about not receiving immediate evidence of continued attention.

Text boxes

Text boxes support visible sentences on a slide, which can be quickly scanned or read. These also require very little storage compared to their audio equivalent. Used poorly, they may cover important parts of an image and there is the potential for comments to be more ambiguous when referring to an image if there is no direct link from the text to that referenced. However, they do provide an external memory that might, for instance, be used to note key points or data related to a more extensive audio recording. The background color of the text boxes is light yellow so that they resemble "sticky notes". The font color can be changed by the user to provide a basic means of distinguishing selected text, such as who wrote what.

			×
This	looks inter	esting.	
J			. <u></u>

Figure 5.8 Text box with no scroll bar



Figure 5.9 Text box with a scroll bar

Figure 5.8 and Figure 5.9 show that the scroll bar is automatically generated when the

text is too long to fit into the visible text area.

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.



Figure 5.10 Text abstraction hierarchy

Just as multiple recordings can be associated with an individual slide, so too can multiple text boxes. This allows for sub-chunking again, as shown in Figure 5.10, but also for recipients to share a text box and place a response close to the original comment.

Ink and highlighter freehand pens

Annotations on paper are commonly performed with a pen or highlighter so these features are natural for the user. Given the Post Operations Evaluation Tool often has a lot of data packed fairly densely; the highlighter was considered essential to avoid covering data. The pen and highlighter were also specifically expected to be used to propose and highlight routes.

Arrow markers

Particular x, y positions on a screen can be marked by placing arrow markers on them as shown in Figure 5.2, where a particular point on an individual flown route is marked. In general, the arrow marker is a static pointer compared to the dynamic pointing that occurs during synchronized speech and mouse pointer recording. It also differs in that it stays constantly visible above the image unless the user decides to remove it. The arrow marker was thought to be useful in marking locations such as fixes or airports. Both the arrow and the pen were predicted to be used in the scatter charts and tables. The arrows point towards the top-left, as do most mouse pointers.

Abstraction from the file system

C-SLANT has a built-in option to send a slide show using the default E-mail system on a machine so the user is saved from having to explicitly make the slide show an attachment. This is desirable to reduce the number of steps involved in sending a slide show and also to avoid interfacing with the file system, which can be a problem for novice users. When a slide show is received most E-mail programs allow an attachment to be automatically opened by the appropriate program for that file type, so again the user can normally just double click on the slide received and have C-SLANT automatically open the slide show and thus avoid dealing with the file system. Of course, this abstraction from the file system works unless the user wishes to archive slide shows outside their mail system.

Slide Sequencing Window

There are a number of operations that need to be performed at the slideshow level as opposed to the slide level. Saving, loading, and sending are the most basic. Viewing and rearranging are two additional functions.

100



Figure 5.11 Slide sequencing window

This level of abstraction is not as well developed in C-SLANT as the slide level. Users can view a sequential list of thumbnails for each slide, summarize the slide, and move a slide using the up and down arrows, shown in Figure 5.11, but the user can not apply the direct manipulation technique of simply dragging thumbnails, as can be done in Powerpoint.

Architecture

C-SLANT does not contain any code for direct machine-to-machine communication; however, when C-SLANT messages are sent as E-mail attachments, it is operating in the clientserver environment of the E-mail system, as shown in Figure 5.12



Figure 5.12 C-SLANT asynchronous communications via E-mail

This is the scenario originally intended and C-SLANT is "mail enabled" in that it uses the Messaging Application Program Interface (MAPI) to interact with MAPI compliant E-mail systems, such as Microsoft Exchange, Microsoft Outlook Express, and Eudora. In such an environment the user can simply click on the E-mail button in C-SLANT to have the current slideshow automatically be an attachment to a new E-mail message opened and displayed using the underlying mail system. If the underlying mail system is not MAPI compliant the user must complete this process manually.

Implementation

Software Development Environment

Microsoft Visual Basic 6.0 was used as the implementation language, because it supports rapid prototyping of applications where the interface is more complex than the underlying algorithm through its visual Integrated Development Environment (IDE). The system was decomposed into modules, in order to achieve the benefits of *structured programming* (Kruse, 1994), but there were also some disadvantages in this environment:

- 1. Visual Basic 6.0 supports the object oriented paradigm, but not particularly well;
- 2. The compiled code will only run on Windows platforms;

3. It does not have the software engineering tools of more sophisticated environments. For instance, state transition diagrams for the recorder were drawn on paper rather than using a CASE (Computer Aided Software Engineering) tool.

Audio Format

The audio format used in C-SLANT is WAVE PCM (Pulse Code Modulated audio), 11,025 KHz sampling, 8 bit, mono format. These settings require 11,025 bytes for each second, and the compression of C-SLANT can reduce the image sizes significantly, but only reduces the audio size slightly.

Compression

C-SLANT uses the zlib compression-decompression library¹² to compress all files used to represent a single slide into one file (a .sld file) and then combines all slide files to produce one slide show file (a .sho file). This is hidden from the user unless he or she chooses to turn compression off or save or load slides, recordings or images individually. The compression feature was thought critical for asynchronous communication when a network connection is slow, but for fast connections or local machine access it is useful to have the option of turning off compression.

C-SLANT captures screen images using the BMP (BitMaP file) format, but also allows these to be saved as JPG (Joint Photographic Experts Group) files.¹³ Screen captures normally

¹² zlib is library compatible with the gzip format commonly used on Unix systems. It is written by Jean-Loup Gailly and Mark Adler, and is freely available. The home page for zlib is http://www.gzip.org/zlib/

¹³ JPG files are created using Intel's free JPEG library of routines. These are described at: http://developer.intel.com/software/products/perflib/ijl/index.htm

look better in this BMP format because BMP files are not compressed. JPG files have color information removed during compression, so that sharp edges tend to blur. However, JPG files capture 24 bits of color information, so are most commonly used for photographs and similar continuous tone bitmap images. With the .zlib compression algorithm used in C-SLANT the file size is reduced to be approximately equivalent to the JPG format, but the edges remain sharp when decompressed.

Summary

Category	Summary
Platform	Microsoft Windows
Communication Method	Asynchronous via application file transfer (e.g. E-mail attachments, FTP transfer, or non-streamed Web access).
Annotation Methods	Unimodal (via text, pen marks and arrows) and multimodal (via voice over images with deictic gesturing)
Message Structure	Linear sequence of annotated images (slides)
Compression	Zlib algorithm
Audio Format	Wave PCM
Graphics Format	JPG or BMP (can import GIF)

Table 5.3 Basic implementation summary of C-SLANT

Table 5.3 shows a summary of the basic implementation level characteristics of C-SLANT.

CHAPTER 6

EXPERIMENTAL PROCEDURE

This research has involved trying to develop a functional tool that could be usefully used in the target domain and context. However, developing this tool also provides a valuable opportunity to use it as a "tool for discovery" and to (1) test certain hypotheses about the effect of particular communication modes for collaborative post operations analysis in this domain; (2) learn more about the information gap that exists between these two organizations; and (3) evaluate the usability and usefulness of C-SLANT used in combination with POET, and as an application in general.

Subjects

36 AOC dispatchers took part in the study. Eight were female; twenty-eight were male. Nine did not provide their age, but of those who did the age range was 33 to 55 with a mean of 45 years old. The number of years working for Northwest Airlines was between 9 and 35 years with a mean of 18 years. 26% indicated the computer they used at work or home had a sound card. Experience with software applications either at work or home was as follows: word processing 81%; spreadsheet 61%; database 35%; presentation software 23%.

36 Traffic Managers took part in the study. Four were female; thirty-two were male. Three did not provide their age, but of those who did the age range was 31 to 57 with a mean of 43 years old. The number of years working for their respective centers was between 3 and 32 years with a mean of 17 years. 34% indicated the computer they used at work or home had a

105

sound card. Experience with software applications either at work or home was as follows: word processing 88%; spreadsheet 51%; database 48%; presentation software 37%.

None of the dispatchers or traffic managers had any previous experience with POET.



Experimental Design

Figure 6.1 Enroute Centers¹⁴

¹⁴ The 'K' prefix before the Center names represents the United States, but is not normally used, so that convention will be followed in this document.

POET searches were run on Northwest Airlines historic flight data to identify nine collections of flight instances (each collection involving direct flights in one direction between a city pair combination) where it was predicted there would be sufficient concern about the undesirable performance of those flights to cause a dispatcher to welcome the opportunity to discuss with a traffic manager at a relevant center what might be contributing towards those results and what might be viable actions to improve performance. The collections of flight instances each involved a different city pair combination and 8 different centers were identified as particularly relevant. The Minneapolis center was involved twice, but given the hub for Northwest Airlines is located in Minneapolis this did not seem inappropriate.

Figure 6.1 shows the location of the FAA Enroute Centers (except the center based in Honolulu). Those included in the study were: ZMP, ZOB, ZBW, ZNY, ZID, ZDC, ZTL, and ZFW. This involves no centers on the west coast because, for Northwest Airlines, the flights with the worst performance results, where POET appeared to reveal a systematic pattern, involved the other centers more. Nine slide show scenarios (see Appendix A.) involving POET screen captures for the nine respective flight collections were created and four dispatchers were selected at random for each of the nine scenarios and asked to create one message using their assigned slide show scenarios as the context for the message. For each scenario, two used C-SLANT with the textbox button hidden and two used it with the voice and pointing button hidden. For each of the nine scenarios, four traffic managers were selected at random from the relevant center and each used the same configuration of C-SLANT that the dispatcher used to create a response. This experimental design is summarized in Table 6.1.

107

	Voice+Pointing 1	Voice+Pointing 2	Text 1	Text 2
Scenario 1	1	2	3	4
Scenario 2	5	6	7	8
Scenario 3	9	10	11	12
Scenario 4	13	14	15	16
Scenario 5	17	18	19	20
Scenario 6	21	22	23	24
Scenario 7	25	26	27	28
Scenario 8	29	30	31	32
Scenario 9	33	34	35	36

Table 6.1 Experimental design (72 participants: 36 airline-FAA participant pairs each assigned to one of 9 scenarios and either the voice and pointing mode or the textbox mode)

In addition to selecting flights that involved poor performance results, flights were picked where the POET results showed some kind pattern in what was happening to those flights, e.g. most being rerouted from the filed route in a particular direction or a significant number being put into holding. Thus, because of the information provided by POET the discussion was predicted to not only involve communicating what the performance results was, but also at least in part why they were occurring. It was anticipated that the message would additionally involve discussing further causes not supported directly by information on the slides, and solutions to the problems identified, because of the knowledge in the minds of the dispatchers and traffic managers. For both the additional causes and the solutions it was anticipated the images would be used when they would help (e.g. spatial information could be used) in the communication.

It was hypothesized a priori that:

 The two modes of communication would interact differently with the collaborative problem solving and discussion. (E.g. the task time; the number of questions asked; the number of dispatcher ideas responded to by the traffic manager; and the number of ideas for improving performance produced by the pair.) 2. The two modes of communication would interact differently with the other annotation tools, namely the pen and the arrow marker.

The first hypothesis would be consistent with related research involving different tasks and participants. If evidence supporting this first hypothesis could be found it might help provide guidance for designers on how to support and integrate these communication modes.

The second hypothesis was based on the believe that the workload and communication differences between making deictic gestures and creating similar semantic references in the text mode would effect the way the pen and arrow is used.

Scenario	Number of slides	Search summary	Scatter chart	Filed route table or expanded filed route table	Flight route map
ZNY	6	X	X	x	X
ZOB	5		X	x	x
ZID	5	X		x	X
ZTL	7	X	X	x	X
ZFW	2	X		x	X
ZBW	6	X	X	x	X
ZDC	6	X	X	X	X
ZMP1	4	X		x	X
ZMP2	3	X		x	X

Scenarios used

Table 6.2 Scenarios summary

Appendix B shows which POET windows were used for each slide of the scenarios and Table 6.2 is a summary of that information. As this table shows, the same number of slides was not used for each scenario. Depending on the particular scenario a different number was deemed to be necessary to show a performance problem and provide supporting information that would seem likely to suggest some potential causes. In every scenario at least one flight route map was provided to show the actual route filed versus flown and to support discussion relative to a 2dimensional view of the relevant airspace. The one scenario that did not include the search summary was the ZOB scenario, but the filed route table and scatter chart on the first slide together showed the city pair, departure time, aircraft type, time of year, and the number of flights involved.

Detroit to Newark (DTW-EWR): the ZNY scenario

The DTW to EWR scenario shows flight information for flights with a departure time between 1800 and 2100. This involved 210 flight instances with an average over burn of 946.5lbs, but with 28 burning more than 2,000 lbs above planned. 194 flight instances were filed through Slate Run (an essentially direct route) and 14 of these were held with an average over burn of 2,490 lbs. The slides show this information and that the holding normally took place at the boundary between ZNY and ZOB, on the ZOB side, although holding for one instance closer to DTW is shown. Flights filed on this more direct route are shown to not have a direct approach into EWR (being routed first to the north or the south) and there are also two instances of rerouting to the south enroute. Vectoring is shown to sometimes occur in the ZNY airspace. It is also shown that when the remaining flights were filed on a more northerly route through ZBW they were not held and often flew a more direct route than filed. Seven slides are used in this slide show including three showing individual flight instances to more clearly show where the holding occurs, and the enroute vectoring.

Minneapolis-St. Paul to Newark (MSP-EWR): the ZOB scenario

The MSP to EWR scenario shows information for a flight with a departure time of 1955 between April and May, and a flight with a departure time of 1945 through June. These essentially represent the same flight, but a change in aircraft type where a 320 was replaced by a 757 and the departure time changed slightly. There were 51 and 19 instances respectively, and a scatter chart on the first slide showing planned versus actual air times shows a similar distribution for both flights, with an average difference of 13.1 and 14.9 minutes beyond planned respectively, but with 7 (10%) 40 minutes above the planned air time. The scenario shows that most instances were filed and flown on close to straight routes between the city pairs, but there were some flight instances filed and flown to the north, and two instances of flights flown on very large deviations to the south. Holding is shown to have occurred 12 times at SLT and to have occurred for the flights with the worst air times. The scenario also shows holding occurring for flights with both close to planned off times and off times significantly later (80, 72 and 29 minutes) than planned. All flight instances flown to the north are shown to involve no holding and air time delays below the overall average.

Detroit to Raleigh-Durham (DTW-RDU): the ZID scenario

The DTW-RDU scenario shows information for 35 instances of a flight that was scheduled to depart at 1050Z and was filed on an essentially straight route passing through central Ohio after departing DTW in a south west direction. The scenario shows the actual route flown on 34 of these occasions was much more to the south west, often on a route as far south as Cincinnati, Ohio or Lexington Kentucky, with two as far south as Tennessee. The 35th instance flew a route to the north of the filed route. In addition to these 35 instances, one instance is shown of a filed route departing DTW in a south east direction passing over Cleveland. The first slide in this scenario shows the overall impact on fuel burn with the nine worst cases burning between 1,065 and 5,265 lbs beyond planned, and a route map showing the filed versus actual routes. The remaining 4 slides show the same information with the four worst performing flight instances, in terms of fuel burn, selected in an expanded filed route table. These corresponded to those deviating from the filed route the most.

111

Detroit to Atlanta (DTW-ATL): the ZTL scenario

The DTW-ATL scenario shows information for five flights from Detroit to Atlanta. The first slide shows the planned versus actual average fuel burn figures for these flights sorted by average percentage difference. The next six slides focuses on the worst three performing flights, using two slides per flight starting with the worst first. The table on the first slide shows that the flight departing at 1105 was the worst for this performance metric, burning an average of 1,911.4 lbs (13.7%) more fuel than planed. (The fifth flight in the table had a departure time 10 minutes before this one, and burned 1,078.5 lbs (7.9%) more fuel than planned.) The second flight in the table had a departure time of 0035 and the third a time of 1630. The 1105 flight is shown on slides 2 and 3 to nearly always be filed out of DTW over ROD rather than APE, and to approach ATL from the northeast, but the routes flown are such that as many flight instances are rerouted to approach ATL from the northwest as stay on a northeast approach. It is further shown that both approaches cause the flight to be put into holding and that it occurs for 50% of the flights flown. Slides 4 and 5 show the 0035 flight is filed on the same route, but that nearly always that route is flown at that time of day. One instance of a severe reroute and holding is also shown for this flight. Slides 6 and 7 show flight information for the 1630 flight, where the same route is filed for every instance, but one (which is filed to depart DTW over APE) and the majority of flight instances appeared to be rerouted to approach from the northwest with holding again occurring for flight instances rerouted and staying on the original flight path.

Minneapolis-St. Paul to Dallas-Forth Worth (MSP-DFW): the ZFW scenario

The MSP to DFW scenario involves two slides. On the first slide two filed routes from MSP to DFW are shown approaching DFW from the northwest. One via Oklahoma City and the other via Tulsa. In both cases many flights actually flew a route arriving from the northeast (13/14 for MSP.J21.OVR.J21.ICT.J21.IRW.UKW3.DFW and 12/17 for MSP..MKC.J25.TUL.UKW3.FGW). The second slide shows three filed routes

(MSP..RST..ALO.J233.CNOTA..DFW; MSP..MKC.J25.TUL.BYP3.DFW; and

MSP..MKC.J25.TUL.BYP2.DFW) with 35 instances shown filed into DFW from the northeast BYP approach, and in most cases the flight instances stayed on the filed routes. All flights shown in the scenario have a departure time between 1400Z and 1500Z.

Detroit to Boston (DTW-BOS): the ZBW scenario

The DTW to BOS scenario shows flight information for 103 flight instances departing between 1800 and 1900Z. It is shown that the average fuel burn is 994.6 lbs beyond planned with 20 of these flight instances burning at least 2,000 lbs more than planned. 14 flights instances are shown to have been held, with this occurring most often close to the BOS airport, but one extreme case of holding is shown north of the airport and one halfway along the filed route. The flown routes are shown to normally be close to the filed route, but on one occasion a flight instance was flown on a route through Canada to approach BOS from the north.

Memphis to Newark (MEM-EWR): the ZDC scenario

The MEM to EWR scenario presents flight information for 30 flight instances departing at 1900 and 29 flight instance departing at 1915. The flights are shown to burn 1,521.0 and 1,601.9lbs more fuel than planned on average; to have air times 15.9 and 17.3 minutes more than planned on average; and to have off times 23.0 and 30.1 minutes later than planned on average, respectively. More detailed information is presented on the 1915 departure, and it is shown that there are filed routes used that take a northern route to approach New York through the Cleveland Center and there are routes that approach New York from the southeast passing through the Washington Center. The slides show that flights filed on a northerly route are sometimes rerouted to pass through the Washington Center, but when a flight instance is filed on a route through Washington it is not rerouted to fly on a route through the Cleveland Center. Flight instances flown on the northern route are shown to have deviated from it significantly

before returning to it on two occasions. Flight instances filed to pass through the Washington Center are shown to be held south of Washington where 4 routes used merge. The most northerly of these four routes, passing through the south of Western Virginia, is the one shown to be filed and flown the most.

Midway to Minneapolis-St. Paul (MDW-MSP): the first ZMP scenario

The MDW to MSP scenario involves 10 flight instances filed on the MDW..BAE..EAU4.MSP route and 16 on the MDW..BAE..EAU3.MSP route, with an arrival time between 2200Z and 2400Z in the month of July. The average planned air time was 62.7 and 62.6 minutes respectively, but the actual average air time is shown to have been 89.7 and 80.7 minutes respectively, with an average off-time 22.8 and 21.7 minutes later than planned, combining to create many significantly late on times for such a short flight. 8 flight instance where held, with 4 coming from each of the two filed routes. In addition to 3 POET screen captures showing this information, an Excel spreadsheet screen capture was included at the end of the slide show that uses POET data to show the worst average air-times for flights into MSP from any other city. This table shows the TWINZ arrival fix to produce the worst four air-time delays and for the two worst arrival hour bins to be 0100Z-0200Z and 2300-2400Z. The table also shows MDW-MSP to be third in this table.

Omaha to Minneapolis-St. Paul (OMA-MSP) : the second ZMP2 scenario

The OMA to MSP scenario shows flight information for 24 flights with arrival times between 2200Z to 2400Z in the month of July. All but two of these flight instances were flown on the filed route to approach MSP from the south. The two flight instances that did not approach MSP from the south, approached it from the west. The average planned airtime for this flight was 45.2 minutes, but 7 flights were held and averaged an additional 32.1 minutes. Those that were not held averaged 20.1 minutes additional airtime. The average off time for those held was
12 minutes later than planned (ranging from 7 to 19 minutes late) and the average off time for those not held was 20.9 minutes later than planned (ranging from 7 to 54 minutes late). The combined off time and air time delays produced many significantly delayed flight instances for such a short flight.

Procedure

Phase	Dispatcher at AOC	Traffic Manager at Center
1	Dispatcher signs consent form and is given problem statement	
2	Training with C-SLANT and introduction to POET screens	
3	The dispatcher creates a message	
4	The dispatcher completes the questionnaire	
5		Traffic manager signs consent form and is given problem statement
6		Training with C-SLANT and introduction to POET screens
7		The traffic manager creates a response
8		The traffic manager completes questionnaire

Table 6.3 Experimental procedure for each participant pair

There were eight phases to the experiment for each dispatcher – traffic manager pair, as shown in Table 6.3. The first four phases for each pair took place at the Northwest Airline Operations Center in Minneapolis. The next four phases for each pair took place at the respective center. During phase 1, the experimenter gave a problem statement to the dispatcher. Phase 2 introduced the dispatcher to C-SLANT's interface and POET screens. Phase 3 involved the dispatcher creating a message using C-SLANT. Phase 4 involved the dispatcher completing a questionnaire. Phase 5 involved the experimenter giving a problem statement to the traffic manager. Phase 6 introduced the traffic manager to C-SLANT's interface and POET screens. Phase 7 involved the traffic manager responding to the message created by the dispatcher he or she was paired with. Phase 8 involved the traffic manager completing a questionnaire.

1. The dispatchers' problem statement

In Phase 1 dispatchers were asked to sign a consent form and then given the following problem statement:

"This is part of a study of asynchronous communication technologies for supporting communication between AOC staff and traffic managers. You will be asked to work with a software application that allows screen captures to be taken from a computer to form a slide show, which may then be annotated to create a message. Such a message could be transmitted as an attachment in any email program.

You will be given an unannotated slide show of screen captures displaying historic data about a particular city pair and asked to annotate it in order to create a message to be sent to a traffic manager at a relevant center. You are asked to use any of the slides given to begin a discussion with the traffic manager of the problems you see with the city pair and possible solutions to the problems. This message will actually be given to a traffic manager and he/she will be asked to create a response".

2. Training with the C-SLANT interface and introduction to POET screens Dispatchers were given approximately 25 minutes of training with the C-SLANT interface using a practice slide show (see Appendix C.) that included POET screen captures to demonstrate the basic types of result that POET can produce and how it is presented on computer displays. Questions were allowed at any time during the training. Slide 1 contained a blank slide so that the dispatcher's attention could be focused on the C-SLANT tools window. Determining the number of slides in a slide show, the purpose of the left and right arrows, and the ability to move the tools window around on the screen above any slide was emphasized.

Slide 2 contained no POET data either, but was a screen capture from the Internet that included a standings table in the Big Ten Conference for American Football at the time the screen capture was taken. This slide included a portion of empty screen space on the right hand side that was used by the experimenter to demonstrate the pens and arrows in isolation, and the ability to change the color of either annotation marker using the color palette. After demonstrating the arrow and pen markers, and the color palette the dispatchers were asked to try them on the slide. If the dispatcher was to work with textboxes a green arrow was placed beneath the score for the Penn State - Minnesota game and the following text was typed into a textbox: "As you can see from the score marked with the green arrow, Minnesota were winning at the time this screen was saved. This was a surprise because Penn State was higher in the standings." Then the lines for Penn State and Minnesota in the standings table were highlighted with a yellow pen. A second textbox was produced and moving and resizing of the textboxes was demonstrated. The dispatchers were then asked to try to create a text box, type something in it and move it on the slide. If the dispatcher was to work with the voice and pointing annotation feature a green arrow was again placed beneath the score for the Penn State - Minnesota game and then the same statements were made as before, this time pointing with the mouse pointer at the Penn State - Minnesota game score and the position of Penn State and Minnesota in the standings table as the statements were made. The recording was played back, the other buttons of the interface to the recorder itself were described and then saving the recording demonstrated along with saving a subject ("Surprising Score") for the recording. A second recording was made underlining the lines for Penn State and Minnesota in the standings table with a yellow highlighter pen while recording to demonstrate that the pen marks are actually not dynamically recreated

during playback, but are static on the slide. Again the recording was saved (but this time no subject was given). The ability to move the player windows was also demonstrated. The dispatcher was then asked to create a recording, play it back and save it. The remaining slides all contained practice POET screen captures that were explained verbally. Slide 3 was used to explain that the text in the search window of a POET screen shows the particular search that was conducted – in this case, that the search was for flights from MSP to CMH between 04/01/98 and 07/01/98 at any time of day. The table of sorted flights by average actual – planned fuel burn was explained, as was the scatter chart for the highlighted flight in the table.

Slide 4 provides additional information about the particular flight that was highlighted in Slide 3. It was used to start describing the maps that POET can produce and their relationship to POET tables when both are displayed at once. In particular, the significance of the color and line thickness of routes shown in the map was explained. In Slide 4 the green thick line shows the filed route for the highlighted 57 flight instances and the block of thick black lines are the routes actually flown. The thin green line is the filed route for the single flight instance that filed a different route from the others shown and its actual flown route is shown with a thin black line. In general, thick lines correspond to those flights selected in the table and green lines represent filed routes whereas black lines represent those actually flown.

Slide 5 was used to demonstrate that if an individual flight is selected in a table its actual flown path appears blue in the map. Slide 6 was used to demonstrate the performance metrics that POET can display in addition to the fuel-burn data demonstrated on the previous slides. Slides 7 and 8 were used to demonstrate what information about race-track (called circular in POET) holding looks like in POET. Slide 7 demonstrates that a yellow diamond indicates at least one instance of holding. Slide 8 demonstrated holding

shown in a flight path might be difficult at times to see in the version of POET used in the scenario as it did not have a feature to zoom-in and see the flight pattern clearly.

3. The dispatcher creates a message

After the training dispatchers were given their assigned slide show to use as the basis for their message. They were told they could ask questions at anytime if they couldn't remember how to do something in C-SLANT or if they had trouble interpreting the data presented by POET. A time constraint was not directly placed on the dispatchers, although there was an indirect time constraint in that their supervisor told them that the task would take approximately one hour. After the dispatcher indicated the message was complete it was saved for them.

4. The dispatchers' questionnaire

After saving the message the dispatcher was given a questionnaire to get feedback on the perceived value and usability of C-SLANT and being able to use it in the type of context represented by the experiment.

5. The traffic managers' problem statement

In Phase 5 traffic managers were asked to sign a consent form and then given the following problem statement:

"This is part of a study of asynchronous communication technologies for supporting communication between AOC staff and traffic managers. You will be asked to work with a software application that allows screen captures to be taken from a computer to form a slide show, which may then be annotated to create a message. Such a message could be transmitted as an attachment in any email program. You have been paired with an AOC person (ATC Coordinator/Chief Dispatcher/Dispatcher) from Northwest Airlines who has already used the same software to create a message to you. This message was created after he/she was given an unannotated slide show of screen captures displaying historic data about a particular city pair and asked to annotate it in order to create a message to be sent to you. The AOC person was asked to use any of the slides given to begin a discussion with you of the problems he/she saw with the city pair and possible solutions to these problems. You are now asked to continue the scenario by responding to this message."

- 6. Training with the C-SLANT interface and introduction to POET screens The training given to the traffic managers was the same as given to the dispatchers, except for those given the textbox condition. The text in the textboxes created by the dispatchers was changed from the default text color of black to blue, before the traffic managers saw those messages. This was to allow the traffic manager to use the same text box to create a response by typing their text in black, between groups of sentences if desired – rather like people often do in an E-mail program, or a separate text box could be used for the response. Traffic managers were explicitly told they could respond using whichever method they preferred.
- 7. The traffic manager creates a response

After the training traffic managers were given the respective dispatcher's message to respond to. They were told they could ask questions at anytime if they couldn't remember how to do something in C-SLANT or if they had trouble interpreting the data presented by POET. A time constraint was not directly placed on the traffic managers, although there was an indirect time constraint in that their supervisor told them that the

task would take approximately one hour. After the traffic manager indicated the message was complete it was saved for them.

8. The traffic managers' questionnaire

After saving the message the traffic manager was given a questionnaire to get feedback on the perceived value and usability of C-SLANT and being able to use it in the type of context represented by the experiment. They were also asked if there was anything they thought the dispatcher could have done to improve the message they received.

Data collection

A special version of C-SLANT was created where pressing specific function keys hides the buttons for synchronized voice and pointing or creating textboxes. In addition, a function key was assigned to hiding the "Screen Capture" button so the participants could not switch C-SLANT into its screen capture mode. Participants created and responded to slide shows using a Gateway SOLO 9120 laptop computer connected to an external monitor and keyboard. This allowed the screen output to be duplicated on both the laptop display and monitor display so that the participant could work in front of the regular monitor in a typical desktop environment, but the output duplicated on the laptop could be video taped using a SONY Handycam TRV65 Hi8 camera. The modified slide shows themselves were saved to disk after each participant completed his or her message. The arrangement of this equipment for the ZDC scenario is shown in Figure 6.2.



Figure 6.2 Equipment setup

Data Analysis

Three types of analysis were performed: 1) An analysis of the outcome performance; 2) A questionnaire to get reactions to using the system for the task given; and 3) A qualitative analysis of the communications and the information exchanged.

CHAPTER 7

RESULTS AND DISCUSSION

Introduction

In this chapter results from the experiment are discussed, first focusing on discourse and problem solving characteristics and effectiveness for this task in this domain, followed by an analysis of lower-level interface issues. Stated another way, the focus first is on the semantic issues and then the syntactic ones. The questionnaire results are also analyzed. Following these analyses, the impact of promoting collaborative problem solving between these groups is discussed and then possible enhancements to POET and redesign ideas for C-SLANT are presented.

Discourse and Problem Solving Analysis

While it is a subjective assessment, the "tone" of the communications by both dispatchers and traffic managers seemed to be cooperative, constructive and positive in most cases. There were a few who did not seem to adopt the intended collective responsibility mode, and instead appeared to be assigning blame for the performance problems to the other group; however, this attitude is actually consistent with a "culture" toward each other that has occasionally been reported in the news media:

"'As long as airlines continue to overbook runways, especially during peak hours, air traffic delays will continue and passengers will wait,' said Randy Schwitz, executive vice

president of the National Air Traffic Controllers. The airlines say flight scheduling amounted to only 7.5 percent of delays." (CNN Report, October 15, 1999).

"One aviation expert said delays will drop only when the airlines and the Federal Aviation Administration accept responsibility for their share of the problem. "That means airlines have to schedule realistically," former FAA chief of staff Michael Goldfarb said. 'Air traffic control has to be modernized. Congress has to give the monies to the agency to truly do that. So everybody has something they have to give and right now we have a culture of blame."" (CNN Report, August 21, 2000)

An example of this attitude and the need for collaboration was demonstrated by the participants working with the ZFW scenario. One dispatcher working in the voice and pointing mode looking at a pattern of reroutes to Tulsa enroute to DFW made the comment, "If we had more prior knowledge of what you guys were doing we could have routed some of these flights straight south [to] Fort Dodge [to] Kansas City, and come in on a Fort Smith routing into Dallas -Fort Worth. That would have saved us a great deal of fuel and a lot of aggravation." However, in the traffic manager's response he pointed out that there would be a problem with heavy traffic from the North West through Fort Smith. The other voice and pointing traffic manager also mentioned the problem with filing through Fort Smith and provided more details on the problem. He added that there would be problems due to the fact that route would cross into Memphis Center airspace and they already sequence over Fort Smith and have miles in trail for traffic that ZFW must merge for a BYP arrival from Little Rock and another route (which the traffic manager doesn't name but traces with the pointer while recording his response). He added that if the Memphis Center is saturated the Kansas Center would reroute the flight over Tulsa anyway. He suggested contacting the Kansas Center each day to "get a heads up" on what ZFW is doing and whether or not arrivals over UKW or BYP are in effect and to then plan accordingly.

In the MDW to MSP, ZMP scenario, another example of blame occurred where the dispatcher for one of the voice and recording pairs commented on the first slide, "Midway to

Minneapolis. Difficult route. We plan it as conservatively as possible. Obviously, from these statistics we can see we are getting extended enroute times." Then on the second slide wrote, "In looking at the routes that were filed on each of these example flights, we can see the flight actually tracked on its filed routing. Still we experienced this kind of delay. Usually we try to file the preferred routings that ATC likes. If this isn't going to work, we need to figure out ways that we aren't going to have this kind of delays on inbounds to Minneapolis, because this is obviously going to cause problems. We will end up with diversions." In response the traffic manager first commented, "I agree with you this is a problem, but it's a Northwest problem. If you change your schedule by a few minutes you would miss the rush." After looking at the slide again he then commented, "Another thought. Most of the time the flights are taking this busy route over TWINZ. If we offloaded the flights and go up to a northern arrival over Branard or Edward Fields, or a south arrival, we could avoid the busy route at this arrival over TWINZ with fix balancing, because the airport is only busy on this side of the airport at that time of day." On the second slide the traffic manager referred to the airline's scheduling by commenting, "These routes at this time of day, and the kind of delays we are seeing here, are a part of marketing trying to get all the flights here at the same time in a hub and spoke system. If flow control could have more control of these flights, and offload at will, we could probably take care of these delays."

The fact that the majority of the participants however, did seem to assume collective responsibility during the experiment can therefore be viewed as positive.

Dispatcher task times

After each participant was introduced to POET and C-SLANT using the same training scenario the participant's particular scenario was loaded and the participant was asked to create a message. The time each participant took from this point until he or she completed the message was the task time recorded unless the participant asked for assistance interpreting the POET data

or performing a C-SLANT operation, or there was some other interruption from their problem solving and message creation time, in which case that time was deducted.

Problem	Text Mode	Voice and Pointing Mode
ZNY	1430, 3570	900, 980
ZOB	2010, 2410	2040, 1080
ZID	1040, 1290	320, 740
ZTL	1950, 2025	1480, 2400
ZFW	600, 510	1110, 1020
ZBW	2831, 3411	1320, 1140
ZDC	260, 1200	950, 510
ZMP-1	840, 830	565, 690
ZMP-2	2200, 1830	1680, 540
Average	1680	1081
Std. Dev.	965.71	544.27
Max.	3570	2400
Min.	260	320

• Dispatchers took longer to complete the task in the text mode

Table 7.1 Dispatcher task time (in seconds) summary



Figure 7.1 Dispatcher task time distribution

As Table 7.1 and Figure 7.1 show, dispatchers took longer to complete the task when working in the text mode compared to the voice and pointing mode. The average task times were 28 minutes (1680 seconds) and 18 minutes 1 second (1081 seconds) respectively, although there was variation between scenarios. 5 of the 6 longest task times were in the text mode, ranging from 36 minutes and 40 seconds to 59 minutes and 30 seconds. The longest task time in the voice and pointing mode was taken by a dispatcher working with the ZTL scenario who took 40 minutes. This scenario involves seven slides, which is more than any of the other scenarios, and the dispatcher made 8 recordings on those 7 slides for a total recording length of 9 minutes and 21 seconds (more than any other dispatcher). The dispatcher also played back several of the recordings and deleted some first attempts. For 5 of the 9 scenarios both text mode dispatchers took longer than either of the voice and pointing mode dispatchers for that scenario, but the ZFW scenario was unusual in that it was the only one were both text mode dispatchers took less time than either of the voice and pointing mode dispatchers. The ZFW scenario involves only two slides, which is less than any of the other scenarios, but it is also the only scenario where both text mode dispatchers created only one text box (which was placed on the first slide in both cases). The ZFW voice and pointing mode dispatchers both commented on each of the two slides, with one of them creating two recordings on each slide. The first text mode dispatcher made no additions and simply wrote on the slide:

it seems more often then not that our flights are geting rerouted from msp to dfw granted some may be for weather but i need to know if we need to file on a denerent route that which help atc and us for a better trafic flow and to cut down our fuel burns. Thanks

This dispatcher appeared to type more slowly than most text mode participants and also possibly didn't process the problem as thoroughly as the other participants given this scenario, because the slides actually showed that flights filed to approach from the west were being rerouted consistently, but not those from the east. The other traffic manager working with the ZFW scenario in the text mode wrote more:

I am concerned about flights inbound to DFW that are planned to arrive from the west. It seems that from ICT to destination routings are quite often being offloaded to the east resulting in inaccurate fuel burn data. Could I have a response as to why this is? Center transitions? Perhaps if this is happening as often as it should we should work out new routes or file the arrivals via the east fixes daily so we can have more accurate arrival times/burn.

However, both the voice and pointing mode participants discussed the routings filed and flown in more detail and actually proposed specific new routes which they drew. The first voice and pointing mode participant also mentioned a flight instance that appeared to be rerouted for weather and the second discussed the problems that the reroutes caused the airline in terms of fuel burn and planning.

The shortest task time of 4 minutes and 20 seconds was taken by a dispatcher working in

the text mode with the ZDC scenario, who placed one text box on the last slide. He simply wrote:

i have some concerns about the route mem ewr and most often if holdiing is done its done over the dca area would it be possible to move some of this traffic more north and and out of the dca areaalso the highlighted fits that took reroutes that used several thousands of lbs of fuel was this really necessary expect a reply as soon as practical..

The second shortest task time of 5 minutes and 20 seconds was taken by a dispatcher working in the voice and pointing mode with the ZID scenario who placed one recording of 1 minute and 6 seconds on the first slide. This participant did not create any pen or arrow marks and simply indicated that the largest deviations from the filed route were probably due to thunderstorm activity or heavy traffic. He then added that the heavy traffic flow of the north-east corridor could be a factor, but he said he thought that thunderstorm activity was most likely the main reason for the deviations.

Source	DF	Seq SS	Adj SS	Adj MS	F	P	
Treatment	1	3223222	3223222	3223222	12.00	0.003	
Scenario	8	11317380	11317380	1414673	5.27	0.002	
Treatment*Scenario	8	4736422	4736422	592053	2.20	0.078	
Error	18	4836475	4836475	268693			
Total	35	24113499					

Table 7.2 ANOVA for dispatcher task times

Table 7.2 shows that using an Analysis of Variance (ANOVA) statistical test on the data in Figure 7.1, the hypothesis that there is no difference between the dispatcher task times in the two respective modes (treatments) can be rejected (p-value < 0.05).



Figure 7.2 Normal probability plot of the residuals for dispatcher task time

As a condition for using the ANOVA test, the residuals should be normally distributed, and therefore the normal probability plot, shown in Figure 7.2, should be close to a straight line. The plot is approximately a straight line, but as shown in Table 7.1 and Figure 7.1 the task times for the ZNY participants differed by 35 minutes and 40 seconds, which produces the two significant differences from the observed values and the fit (which is simply the average of the two observed values). The average time for all dispatchers working in the text mode was 28 minutes. In the ZNY scenario one participant took 23 minutes 50 seconds and the other took 59 minutes and 30 seconds, which was the longest time for all participants working in the text mode. This latter dispatcher working in the text mode created pen or arrow marks on all but the first slide and also wrote more than any other dispatcher, producing 1,836 characters. In general, the number of characters has a linear relationship with the task time for dispatchers (with Pearson's correlation = 0.82; and sufficient evidence at $\alpha < 0.01$ that the correlation is not zero). The writing

style of this dispatcher also involved complete detailed sentences, such as the following on slide

6:

The worst case of en route holding is depicted here. The flight was held very shortly after departure from DTW. This usually happens when ZNY stops traffic from ZOB at the center boundary and ZOB has to begin holding traffic further from EWR. In cases like this, would it be possible to offer a reroute to one of the northern alternate routes? This soon after departure from DTW, an alternate route could be easily analyzed and there would be enough fuel to accept such a route in most cases. The added flying on the alternate route would probably consume less fuel than the 2900 lbs this flight burned in holding and vectoring.



Figure 7.3 Residuals versus the fitted values for dispatcher task times

Figure 7.3 shows that the largest pair of residual values occurred for the second largest average task time for a particular scenario, but it shows there is not a strong relationship between the average task time and the residuals. Although there does not appear to be a compelling reason to remove the ZNY scenario as an outlier, Table 7.3 shows the result of performing an ANOVA with that scenario removed for completeness. The effect is that (1) the hypothesis that there is no difference between the dispatcher task times in the two respective modes (treatments) can still be rejected (p-value < 0.05); (2) the hypothesis that there is no difference between the

dispatchers task times across the scenarios can still be rejected (p-value < 0.01); the hypothesis that there is an interaction between the mode (treatment) and the scenario can no longer be rejected (p-value < 0.05).

Source	DF	Seq SS	Adj SS	Adj MS	F	P	
Treatment	1	1829785	1829785	1829785	11.51	0.004	
Scenario	7	10799048	10799048	1542721	9.70	0.000	
Treatment*Scenario	7	3696259	3696259	528037	3.32	0.022	
Error	16	2543475	2543475	158967			
Total	31	18868567					

Table 7.3 ANOVA for dispatcher task times (with ZNY data removed)

Figure 7.4 shows the main effects of the mode (treatment) and the scenario. The average task time for the text mode participants was 28 minutes 0 seconds and the average task time for the voice and pointing mode participants was 18 minutes 1 second, which is difference of 9 minutes 59 seconds.

Figure 7.4 shows the average time for each scenario, whereas Figure 7.5 separates the average time for each scenario using a particular mode, and thus suggests interactions between the mode and the scenario where there are large differences. (It is important to note that the average for each unique mode and scenario combination is the average of just two participants however, so an outlier can have a significant impact on that average.) The main effects of the scenarios suggest there were two levels of complexity for the problem solving and communications task, assuming complexity is correlated positively with the task time. The ZDC, ZFW, ZID, and ZMP-1 scenarios produced the four shortest average task times for both modes, and ranged from 12 minutes 10 seconds to 14 minutes 7.5 seconds. The remaining scenarios' averages' ranged from 26 minutes 2.5 seconds to 36 minutes 15.5 seconds. The number of slides in each scenario varied, and the four scenarios with the lowest average task times had 4.25 slides on average, whereas the remaining five had 5.4 slides on average, but there is evidence that other factors influenced the task time. Pearson's correlation between the number of slides and the

average voice and pointing mode task time = 0.251; with evidence at α = 0.315 that the correlation is not zero. Pearson's correlation between the number of slides and the average text mode task time for each scenario = 0.426; with evidence at α = 0.078 that the correlation is not zero. In addition, it can't be concluded that the task times within the same scenario for the two modes are correlated. Pearson's correlation between the voice and pointing and the text mode task time for each scenario = 0.462; with evidence at α = 0.053 that the correlation is not zero.



Figure 7.4 Main effects plot for dispatcher task times



Figure 7.5 Interaction plot for dispatcher task times

The ZID scenario involved five slides, however after the first slide the remaining four only show what the mapping is between specific fuel burns and instances in the route map for the worst of these instances. The overall complexity of the slide show is therefore not that much more than the first slide. The ZFW scenario was just two slides with flown versus filed routes for those that were filed to the northwest of DFW on one slide and to the northeast of DFW on the other. The ZMP-1 scenario involved four slides; however, the information on the maps was essentially redundant after the first slide. The ZDC scenario involves six slides and with a scatter chart, flight instance information, holding, and instances most likely influenced by weather it is surprising this scenario did not produce longer dispatcher task times.

Dispatcher situation assessments

• Dispatchers agreed that POET data demonstrated performance problems

In this experiment dispatchers working with the same scenario were given a common set of slides to work with as a control, although if a dispatcher really had access to C-SLANT he or she would decide when a performance problem existed and create slides that would seem to help the communication. While not completely authentic, the experiment would clearly be more valuable if the dispatchers agreed there was a problem represented by the slides that were given to them. All but one dispatcher used language to express "concern" about one or more performance issue shown on the slides for their respective scenario. For instance, "some of these fuel burns are excessive"; "this flight is frequently on a route other than the filed route"; "a large percentage of these flights are getting held at this location"; or "the airtime for these flights suggests miles in trail are often applied", followed by one or more comments regarding the negative impact to the airlines and/or a question, suggestion, or idea regarding alternatives to avoid the problem. The one dispatcher who did not express concern merely explained how to interpret the POET screen captures without using such language to indicate he considered that a problem had actually been represented.

• Dispatchers normally did not introduce the flights with some basic classifying data defining the situation

Category	Dispatcher Text Mode	Dispatcher Voice & Pointing Mode
City pair	5/18	11/18
Time of day	1/18	4/18
Time of year	0/18	5/18
Number of flights involved	1/18	4/18

Table 7.4 Number of dispatchers mentioning certain flight classifying data

As Table 7.4 shows, when dispatchers started making comments about the scenario on the first slide they often did not mention certain basic flight classification data. Although this information was available by searching through the POET data on the first slide in each scenario, the context for the discussion could have been made clearer by drawing attention to this information with comments. In addition only one dispatcher used pen or arrow marks to annotate this information, (a text mode participant working with the ZMP2 scenario who used a yellow pen to highlight the city pair and date range), and none of the dispatchers working in the voice and pointing mode pointed to this data while recording their comments.



Figure 7.6 Loss of time of day information

Figure 7.6 shows the first slide communicated by a dispatcher working with the ZID scenario in the text mode. This was the only slide the dispatcher annotated and he wrote:

its obvious we have a considerable amount o fits which deviated from the fit planned routes. with out having access to old weather data and time of day, its difficult for me to conclude if the situation was completel weather related or traffic in general. I would expect the highlighted fit route in red was because of wx. please claify that one for me, thanks

Thus, the dispatcher clearly considered the time of day important, but didn't notice that information in the POET screen capture. Further, by creating the text box and not moving it from its default position of the top left hand corner he made it more difficult for the traffic manager to notice that he was wrong in stating that there was no information regarding the time of day in the slide show. In fact, the traffic manager did move the dispatcher's text box, but did not make a comment about the time of day, so it can't be determined if the traffic manager noticed it or not. The traffic manager left his text box over the time of day information, so in a further message iteration it might still have not been noticed by the dispatcher. In the ZDC scenario, one of the traffic managers looking at the information for a flight instance on slide four made the comment that the holding he saw might be connected to the time of day, but he did not see that information on the slide. In fact the time of day information for all the flight instances in that scenario were shown on slide one, but the dispatcher didn't mention it and the traffic manager either didn't notice it or remember it by the time he reached slide four. These examples demonstrate the need for this important information to be made more salient in POET and also the value of the dispatcher mentioning it in the initial message.

The distribution of dispatchers mentioning the categories shown in Table 7.4 across each scenario is shown in Appendix D. Using a Chi-Square test to measure the relationship between the mode and whether or not the city pair was mentioned, the p-value is 0.044, which indicates that there is evidence that mode and whether or not the city pair was mentioned are related at the $\alpha < 0.05$ level. The contingency tables for the other categories shown in Table 7.4 produces two

cells with expected values of less than 5 in each case, but applying the randomization test (Box, Hunter & Hunter, 1978) produces the time of day p-value as 5/32 = 0.16; the time of year p-value as 1/32 = 0.03; and the number of flights p-value as 5/32 = 0.16. The time of year p-value is the only one of these that would normally be regarded as statistically significant. A difference between the two modes might be because the cost/benefit ratio is different, given the relative ease with which the information can be stated verbally versus typed. It is also perhaps a factor that the message creator in the voice and pointing mode feels more control over the message recipient's attention and the order and rate at which information is communicated, and thus feels more responsibility to introduce the scenario before discussing the details. In some ways reducing the recipient's workload is also more polite, and that has reported to be a difference between voice and text based communications (Neuwirth et al., 1994). A difference between the modes with regard to mentioning this type of information is important, but even the voice and pointing mode did not create an environment where the dispatcher chose to mention more than the city pair more often than not, possibly because it was assumed to be something the traffic manager would note for him or herself. In other words, to use Clark and Wilkes-Gibbs (1986) distinction discussed in Chapter 2, in both modes the dispatcher appeared to consider the situation one of mutual responsibility rather than distant responsibility to note this basic information on the slides.

• Dispatchers problem solving focus varied

To collaboratively identify the performance problems, the actual or probable causes and the alternative actions to improve performance, the following was necessary by the dispatcher:

- 1. The dispatcher needed to interpret the information on the slides;
- 2. The dispatcher needed to focus the traffic manager's attention on information the dispatcher considered important to completing the task, including the performance

problems, and the actual or probable causes of the poor performance, based on information on the slides or in the dispatcher's head;

3. To progress further the dispatcher could ask for more information to determine what is happening to the flights; ask why the ATC are taking the actions they are taking; ask how the problem could be avoided; or offer suggestions and ideas for improving performance.

Dispatchers varied in how much they focused on each of these subtasks. This will be shown in the following sections that identify the types of questions asked and the number of dispatchers who made suggestions to improve NAS performance. These differences could have occurred because they had different interpretations for how far in the process a first communication should go, even though they were all given the same set of instructions before creating their message.

Dispatcher direct questions

Question: "A sentence in an interrogative form, addressed to someone in order to get information in reply."

(Webster's Encyclopedic Unabridged Dictionary of the English Language)

In collaborative problem solving tasks, questions can serve to explicitly request information the person asking doesn't have, but they may also be used to ask if a proposed solution would be considered a solution by the other participant. Thus, some questions are simply requesting acceptance or rejection, while others request different information. However, as Clark and Schaefer (1989) note, there are sentences that are not questions that also are normally followed by acceptance or rejection during interaction. For instance, in this context, there were sentences of the form, "I wonder if we were to ….." and "Perhaps if we …..", when considering changes that might improve performance. During synchronous communications such expressions are often followed by a pause, to see if the comment is accepted or rejected, but it is possible the person is just thinking aloud (or in text) and actually is not proposing the other person respond. Thus, in this section a strict interpretation of a question is used, even though this means some sentences that were used with the intention of receiving a response will not be counted. Specifically, a question in the text mode had to be terminated with a '?' character, or require one to be grammatically correct, and in the voice mode a sentence had to require a '?' to be grammatically correct, when written down.

Problem	Text Mode		Voice and Po Mode	ointing
ZNY	3C's	1C	2A's, 1B, 3C's	-
ZOB	1B, 2C's	1A, 1B, 3C's	-	-
ZID	1A	2B's, 1C	-	1A, 1B
ZTL	3B's, 1C, 1D	-	-	1A, 1C
ZFW	-	1B	-	1A
ZBW	6A's, 1C, 2D's	-	-	-
ZDC	1A, 1C	2B's, 1C, 1D	-	-
ZMP-1	1A, 1C	2B's 1C	-	-
ZMP-2	-	-	-	-
Total:	42: 10A's, 12B's, 16C's, 4D's		11: 5A's, 2B'	s, 4C's

Table 7.5 Number of direct questions asked by each of the 36 dispatchers

The majority of questions asked fell into one of three categories:

Question Type A: "Why is X happening to our flight(s)?"

Sometimes X was general, such as "being held so often", or "being rerouted so often"; and sometimes it was more specific, such as "being held here so often", "being rerouted to the north through ZBW".

• Question Type B: "Was the problem caused by Y?"

Again, sometimes Y was general, such as "weather" or "traffic"; and sometimes it was more specific, such as "thunderstorms", "traffic from the east", or "traffic departing from Cincinnati".

Question Type C: "Would doing Z help?"

Again, sometimes Z was general, such as "taking a different route"; and sometimes it was more specific, such as "taking this route".

Of the 17 dispatchers who asked direct questions, 13 included at least one of type C - 11 in the text mode and 2 in the voice and pointing mode. As Table 7.5 shows, of the 53 direct questions asked, 42 were asked by dispatchers in the text mode (10 A's; 12 B's, and 16 C's), and 11 in the voice and pointing mode (5 A's; 2 B, and 4 C's). Overall, 15 were of type A, 14 were of type B, and 20 were of type C. The other four questions asked in the text mode (shown as type 'D' in the table) were, "Was the reroute predictable before the flight departed?"; "What is the reason for the gap in the data?"; "Is there any airspace available around the DCA area where the holding occurs?" and "Are we checking if these flights are over water equipped?" All the questions asked are shown in Appendix E.

Dispatcher ideas for improving performance

As was mentioned in the last section, there were 20 questions that specifically asked if doing X would improve performance (16 in the text mode by 11 dispatchers and 4 in the voice and pointing mode by 2 dispatchers); however, there were also suggestions and ideas raised that were not in the form of a direct question. Grouping these together produces the numbers shown in Table 7.6.

Problem	Text Mode	Voice and Pointing Mode
ZNY	4, 2	3, 1
ZOB	2, 3	3, 1
ZID	0, 2	0, 1
ZTL	2, 1	2,7
ZFW	1, 1	2, 2
ZBW	1, 0	1, 1
ZDC	1, 2	0, 2
ZMP-1	2, 2	1, 1
ZMP-2	1,0	1,0
Total	27	29

Table 7.6 Number of dispatcher ideas for performance improvement

Table 7.6 shows 15 dispatchers in the text mode and 15 dispatchers in the voice and pointing mode made suggestions for improving performance. The total number of suggestions was 27 in the text mode and 29 in the voice and pointing mode. Thus, dispatchers generated a similar number of ideas, but less often in the form of a direct question when working in the voice and pointing mode.

Category	Dispatchers	
	T	VP
New preflight route	11	9
Enroute reroute	2	3
Change schedule	1	1
Ground delay/stop	3	4
Same route but miles in trail	0	1
Change Center staffs' workload / traffic volume	2	0
Share information for planning (fuel burn, filed route, etc.)	3	3
Hold at a different location	1	1
Change altitude	2	2
Fix balancing	0	1
Other	0	1
Totals:	25	26

Table 7.7 Classification of dispatcher ideas regarding performance improvement

Appendix F lists each performance improving idea that was produced by each dispatcher.

Table 7.7 is a classification of these per dispatcher. Thus, for instance, 11 dispatchers made at least one comment about a new preflight ("preferred" or "filed") route working in the text mode, and 9 working in the voice and pointing mode. Overall, it reveals that the routes were seen as an issue more often than other factors influencing performance.

Problem	Text Mode	Voice and Pointing Mode
ZNY	2, 2	1,0
ZOB	0, 1	1, 1
ZID	0,0	0, 1
ZTL	0, 1	0, 2
ZFW	1, 1	2, 2
ZBW	0,0	1, 0
ZDC	1,1	0, 1
ZMP-1	1, 1	0, 1
ZMP-2	1,0	0, 0
Total	13	13

Table 7.8 Count of dispatchers' proposed routes

Appendix G lists each of the dispatchers' proposed routes, whether pen or arrow marks were used, and what deictic gesturing occurred. Table 7.8 shows the tabulated number of routes proposed to improve performance, which is the same in each of the two modes. The distribution was also similar with all dispatchers proposing between 0 and 2 routes.

Text Mode	2/13
Voice & Pointing Mode	4/13

Table 7.9 Count of dispatchers using pen or arrow marks with a proposed route

Table 7.9 shows the number of proposed routes that were annotated with arrow marks or the pen marker. In the voice and pointing mode the 4 instances where pen or arrow marks were used all involved doing so while speaking at the same time. Of the nine other instances, 5 involved pointing to or tracing the route referenced. The remaining four did not involve pointing, included two instances for the ZOB scenario where the comments clearly identified the occasionally filed routes to the north of the normally filed routes. The comment, "A filed route that can be flown without being rerouted consistently" by a dispatcher in the ZID scenario appears to reference a route the dispatcher couldn't point to on the map, because he doesn't have an instantiated route in mind. The fourth comment which didn't occur with simultaneous pointing to the referenced route occurred while the dispatcher placed a yellow arrow marker beneath another route he had just previously commented about.

Dispatcher	Voice and pointing mode proposed route description
ZNY V&P1	A route to the north or south of this area where holding occurs
ZOB V&P 1	This northerly route because it appears to run into less traffic
ZOB V&P 2	A northern route because they don't appear to have holding
ZID V&P 2	A filed route that can be flown without being rerouted consistently
ZTL V&P 2	A different preferred route on a lower altitude out of DTW, coordinated with
	Cleveland and Chicago Centers
ZTL V&P 2	The route most often actually flown
ZFW V&P1	Ideally a route through Kansas City to Fort Smith
ZFW V&P 1	From the routes currently flown, Kansas City to Tulsa.
ZFW V&P 2	A more direct route than the dogleg from Tulsa to DFW
ZFW V&P 2	This route we are actually flying after being deviated from Kansas City
ZBW V&P 1	A different preferred route to the north or south to avoid holding over
	Gardner
ZDC V&P 2	A north west approach on a route currently used with no shown holding,
	rather than one from the south west
ZMP-1 V&P 2	A route through the oclare3 arrival to reduce the enroute and holding time
	associated with oclare4

Table 7.10 Dispatcher voice and pointing mode proposed route descriptions

Table 7.10 lists the route descriptions in the voice and pointing mode. Two of these benefited from simultaneous pointing to not only help the recipient focus attention to the correct route quickly and unambiguously, but also to add information to the comments made. After noting that most of the traffic appeared to be rerouted to the west enroute to Atlanta, a voice and pointing mode dispatcher suggested that perhaps a new preferred route could be coordinated with the Cleveland and Chicago Centers. In doing so she suggested considering lower altitude restrictions out of Detroit to avoid Chicago's overhead traffic. While making this recording the dispatcher traced a partial route out of Detroit to the east when mentioning the possibility of developing a new preferred route, even though there was nothing in what the dispatcher actually said that suggested a route to the east. Thus pointing made it clear that she had a route in mind when making that comment and more specifically that it was a route to the east. A dispatcher working with the ZBW scenario commented that perhaps the flight could be moved a little more to the north or south to avoid holding over Gardner. While making this comment he traced two routes, one to the north and one to the south, both deviating from the filed route relatively close to BOS. Thus, he showed more precisely what routes he was thinking off, however, when tracing those routes he moved the pointer much faster than during his other pointer, which appeared to gesture that these were approximate routes.

Traffic manager task times

The traffic manager's task time involved time spent interpreting the POET slides, interpreting the dispatcher's comments, deciding appropriate responses and communicating those responses. Thus, the task time was recorded as the difference between the time the traffic manager started viewing his or her scenario and the time they finished the message, less any time taken by the traffic manager for other activities such as asking for and receiving assistance in interpreting a POET slide or performing an operation in C-SLANT.

Problem	Text Mode	Voice and Pointing Mode
ZNY	3830, 2370	1170, 840
ZOB	1210, 1600	1140, 1230
ZID	780, 720	360, 240
ZTL	2700, 2580	1510, 1360
ZFW	810, 990	930, 1050
ZBW	2235, 1670	2010, 2415
ZDC	1410, 1020	1070, 1100
ZMP-1	1110, 800	540, 280
ZMP-2	500, 1070	1170, 660,
Average	1523	1060
Std. Dev.	886.99	561.45
Max.	3830	2415
Min.	500	240

• Traffic managers took longer to complete the task in the text mode





Figure 7.7 Traffic manager task time distribution

As Table 7.1 and Figure 7.7 show, traffic managers took longer to complete the task when working in the text mode compared to the voice and pointing mode. The average task times were 25 minutes 23 seconds (1523 seconds) and 17 minutes 40 seconds (1060 seconds) respectively, although there was variation between scenarios. 5 of the 6 longest task times were in the text mode, ranging from 37 minutes and 15 seconds to 63 minutes and 50 seconds. The longest task time in the voice and pointing mode was taken by a traffic manager working with the ZBW scenario who took 40 minutes and 15 seconds. This traffic manager was working with a scenario involving 6 slides and a dispatcher's comment on each slide. The traffic manager played each recording one time (for a total of 5 minutes and 46 seconds) and created a recording on slides 2 through 6 in response (for a total of 25 minutes and 24 seconds). He never played back any of his own recordings. For 4 of the 9 scenarios both text mode traffic managers took longer than either of the voice and pointing mode traffic managers for that scenario. There were no scenarios where both traffic managers working in the text mode took less time than both traffic managers working in the voice and pointing mode. Five of the six shortest task times were in the voice and pointing mode with the shortest time being 4 minutes by a traffic manager working with the ZNY scenario. This scenario involved 2 slides, with a recording of 2 minutes and 23 seconds on one slide by the dispatchers and a response of 1 minute and 20 seconds by the traffic manager.

Problem	Text Mode	Voice and Pointing Mode		
ZNY	5260, 5940	2070, 1820		
ZOB	3220, 4010	3180, 2310		
ZID	1820, 2010	680, 980		
ZTL	4650, 4605	2990, 3760		
ZFW	1410, 1500	2040, 2070		
ZBW	5066, 5081	3330, 3555		
ZDC	1670, 2220	2020, 1610		
ZMP-1	1950, 1630	1105, 970		
ZMP-2	2700, 2900	2850, 1200		
Average	3202	2141		
Std. Dev.	1545.16	957.88		
Max.	5940	3740		
Min.	1410	680		

Table 7.12 Dispatcher-Traffic Manager pair task time (in seconds) summary



Figure 7.8 Dispatcher + traffic manager task times

Table 7.12 and Figure 7.8 show total task times for each dispatcher-traffic manager pair. The average task times were 53 minutes and 22 seconds (3202 seconds) and 35 minutes and 41 seconds (2141 seconds) respectively. The highest 7 task times were in the text mode ranging from 66 minutes and 50 seconds (4010 seconds) to 99 minutes (5940 seconds). The longest task time in the voice and pointing mode was 62 minutes and 40 seconds by a pair working with the ZTL scenario. In this scenario the dispatcher created 8 recordings on 7 slides and produced 9 minutes and 21 seconds of audio. She also played back her own recordings several times and deleted some first attempts, taking 40 minutes to complete her part of the task. The traffic manager created 6 recordings on the slides and produced 7 minutes and 30 seconds of audio. The traffic manager only played each recording one time, but also spent time looking for the arrival times because the dispatcher did not mention the departure or arrival times and a player window was left covering the departure time constraint on the POET search conducted. The traffic manager's task time was 20 minutes and 40 seconds. For 6 of the 9 scenarios both text mode pairs took longer than either of the voice and pointing pairs for that scenario, but the ZFW scenario was the one scenario where both text mode pairs took less time than either of the voice and pointing pairs, as it was for dispatcher task times. The shortest five task times were for pairs working in the voice and pointing mode.

Source	DF	Seq SS	Adj SS	Adj MS	F	P	
Treatment	1	10135733	10135733	10135733	60.41	0.000	
Scenario	8	42235407	42235407	5279426	31.46	0.000	
Treatment*Scenario	8	10930115	10930115	1366264	8.14	0.000	
Error	18	3020250	3020250	167792			
Total	35	66321505					

Table 7.13 ANOVA for dispatcher + traffic manager task times

Table 7.13 shows that using an Analysis of Variance (ANOVA) statistical test on the data in Figure 7.8 the hypothesis that there is no difference between the dispatcher + traffic manager task times in the two respective modes (treatments) can be rejected (p-value < 0.05). The second text mode traffic manager's response in the ZMP-1 scenario was unusual in that she responded by stating the delay times were hard to believe as valid and then later that they must have been caused by weather. If this person's data is considered an outlier and removed however, it has no affect on the p values in the ANOVA table shown in Table 7.13 and the average task time for the text mode changes from 53 minutes and 22 seconds to 54 minutes and 55 seconds (3%).



Figure 7.9 Normal probability plot of the residuals for dispatcher + traffic manager task times The plot is approximately a straight line, but as shown in

Table 7.12 and Figure 7.8 the task times for the ZMP-2 participants differed by 27 minutes and 30 seconds, which produces the two significant differences from the observed values and the fit. Figure 7.10 shows the largest pair of residuals, but also that there is not a relationship between the average task time and the residuals. Table 7.14 shows the result of performing an ANOVA with that scenario removed.


Figure 7.10 Residuals versus the fitted values for dispatcher + traffic manager task times

Source	DF	Seq SS	Adj SS	Adj MS	F	P	
Treatment	1	9627272	9627272	9627272	93.98	0.000	
Scenario	7	41933024	41933024	5990432	58.48	0.000	
Treatment*Scenario	7	10837951	10837951	1548279	15.11	0.000	
Error	16	1639000	1639000	102437			
Total	31	64037247					

Table 7.14 ANOVA for dispatcher-traffic manager task times (with ZMP-2 data removed)

The effect is that (1) the hypothesis that there is no difference between the dispatcher + traffic manager task times in the two respective modes (treatments) can be rejected (p-value < 0.01); the hypothesis that there is no difference between the dispatcher+traffic manager task times across scenarios can be rejected (p < 0.01); the hypothesis that there is not an interaction between the mode (treatment) and the scenario can not be rejected (p-value < 0.01).



Figure 7.11 Main effects plot for dispatcher + traffic manager task times



Figure 7.12 Interaction plot for dispatcher + traffic manager task times

Traffic manager responses to the dispatchers' performance questions and ideas

Given that the voice and pointing mode dispatchers produced a similar number of ideas for improving performance, but did not do so in the form of direct questions as often as those working in the text mode, an important question is whether or not the traffic manager's responded differently to ideas in the form of direct questions versus other forms. For instance, if they responded to significantly more ideas in the form of a direct question that would suggest an advantage of text based communications. Determining whether or not an idea is responded to is however more difficult in this asynchronous, slide show based communications environment than in a synchronous voice based environment (e.g. via telephone) or text based environment (e.g. via a chat program). This is because the dispatchers' comments are not as interleaved in time with the traffic managers comments as much as they would be in most conversations. This is not only a problem because the response may appear in a separate text box or recording, and on a later slide, but it may also occur on an earlier slide. Either because the traffic manager chose to do that, for instance, when that slide provides relevant information, or possibly because the traffic manager was actually providing a response to an idea he or she had not read or heard yet. (The mapping of dispatcher comments to traffic manager comments was less complicated when text mode traffic managers used the same text box as the dispatcher, and particular when the text was interleaved.)

In addition, in a dialog a person may respond to one idea by providing an alternative idea without explicitly referring to the first person's idea. For example, "Perhaps doing X would improve Y". The response, "Doing Z is the best way to improve Y" could be presented in response to the first statement, but it does not reference the idea of doing X. Because it does reference the dispatcher proposed action and only the goal, it may in some situations be less valuable. For instance, in achieving common ground or exchanging knowledge.

	Direct questions	Other form
Text mode	16/16	11/11
Voice & pointing	4/4	23/25

Table 7.15 Count of the performance improving ideas responded to by the traffic managers

Appendix F lists each idea presented by the dispatchers and each is mapped to what appeared to be the corresponding response primarily based on the semantics of the comments made. This may mean some traffic managers' comments that were intended to be responses to particular dispatcher ideas were missed, but most mappings appear to be unambiguous. Two ideas in the form of a non-direct question were not addressed directly. One voice and pointing mode dispatcher working on the ZTL scenario made the comment, "A delay program could have helped to improve spacing", but the traffic manager made no comment that appeared directly connected to this idea. This was one of seven ideas mentioned by this dispatcher, which was the most in either mode. However, the specific suggestion not responded to (on slide 6) was the only suggestion in a recording of 56 seconds. One voice and pointing mode dispatcher working on the ZOB scenario made the comment, "Another solution might be to fly at different altitudes". This was one of three ideas for this dispatcher, which was the second highest number of ideas by a dispatcher working in the voice and pointing mode. The actual recording containing the suggestion (on slide 4) was one containing two suggestions and was 1 minute 17 seconds long. The traffic manager made no reference to a different altitude in the suggestions he considered. All direct questions related to performance improvement were answered by the traffic managers. Thus, as shown in Table 7.15, in the text mode all 16 direct questions and 11 ideas in a different form were responded to, and in the voice and pointing mode the 4 direct questions and 23 of the 25 ideas in a different form were responded to, so there is no evidence of a relationship between the form of an idea and whether or not the idea received a response by the traffic manager. While having responses for each idea presented by a dispatcher is a positive attribute, as will be

discussed in a later section of this chapter, there were instances of redundancy in the traffic managers responses to the dispatchers' messages.

Although it was not a question related to improving performance, it is worth noting the question, "Are we checking that these flights are over water equipped?" was not answered. It was raised by a dispatcher working in the text mode and given the ZBW problem, who asked 8 other direct questions (more than any other dispatcher). It is possible the question wasn't answered, because the traffic manager didn't notice, or remember the question; that the traffic manager didn't know the answer and chose not to state that; or that the question was intentionally ignored because it is clearly a delicate question concerning safety.

Category	Category Dispatchers		Traffic Managers	
	Т	VP	T	_VP
A. New preflight route	11	9	9	10
B. Enroute reroute	2	3	2	0
C. Change schedule	1	1	7	3
D. Ground delay/stop	3	4	1	1
E. Same route but miles in trail	0	1	0	0
F. Change Center staffs' workload / traffic volume	2	0	0	0
G. Share information for planning (fuel burn, filed route, etc.)	3	3	0	1
H. Hold at a different location	1	1	0	1
I. Change altitude	2	2	0	0
J. Fix balancing	0	1	0	0
K. Other	0	1	0	0
Totals:	25	26	19	16

Table 7.16 Classification of dispatcher and traffic manager ideas regarding performance improvement

Table 7.16 shows the number of traffic managers who accepted the category of their

dispatcher's idea as a possible performance improving approach or added that category as an

approach through their comments.

Scenario	Text Mode		Voice and P	ointing Mode
ZNY	aA, rB, rG, rF, nB	aA, aB	rA, rB, rD	rH
ZOB	rD, rH, n(ref)H	a2A, rI, nA	a2A, mI, nC	aA
ZID	(Leave as is)	ref(G), (ref)I	n(ref)A	rA(Leave as is)
ZTL	rD, rG, nA, nC	rA	rG, rK, nA	rG, aA, rJ, aA, aC, mD
ZFW	aA	aA, nC	rA, aA	rB, aA, n(ref)G
ZBW	rD	nC	rA(Leave as is)	rD
ZDC	aA, nC	aA, rA, n(ref)A	nA	rE, rA, n2A
ZMP-1	aC, rA, nA	rA, rF	aD, n2A, nC	rB
ZMP-2	aA, nC	nC, nD	rG	(none)
Totals:	a = 11 ; r = 14	; n = 11 ; m = 0	a = 9; r = 18;	n = 10; m = 2

Table 7.17 Classification of communications outcome with respect to ideas for improving performance

(a: accepted; r: rejected; n: new idea; m: no response. (ref): referred to another center or military airspace manager. Classification of ideas A-K shown in Figure 7.16)

Table 7.17 summarizes the outcome for each dispatcher-traffic manager pair according to how each traffic manager reacted to the ideas presented by the paired dispatcher, and also what ideas he or she added, if any. Thus, for instance the second ZFW voice and pointing mode traffic manager is coded as rB, aA, nG to present the facts that the traffic manager rejected the dispatchers idea of an in-flight change of route; accepted the dispatchers idea of a preflight change to the filed route; and proposed the idea of providing new information to the airline via another Center to help planning. The table shows that, in the text mode, there were 11 dispatcher ideas that the traffic manager accepted, 14 that were rejected, 11 that were proposed by the traffic manager and all were responded to. The table also shows that, in the voice and pointing mode there were 9 dispatcher ideas that the traffic manager accepted, 18 that were rejected, 10 that were proposed by the traffic manager; and 2 not responded to. Thus the distributions are similar.

Other information exchanged

Most of the analysis thus far has focused on the number of performance improving ideas generated and the time taken to generate these ideas. In the section discussing dispatchers mentioning certain flight classifying information however, it was shown that more dispatchers commented on the city pair (p = 0.044) and the time of year (p = 0.03) in the voice and pointing mode than the text mode. This is significant because it may help the traffic manager achieve greater situation awareness for the problem solving task and suggests the dispatcher took greater distant responsibility (Wilkes-Gibbs, 1986). However, there was not sufficient evidence to conclude that more dispatchers commented on the time of day (p = 0.16) or the number of flights involved (p = 0.16).

Scenario	Text	Text mode		ointing mode
Scenario	Pair 1	Pair 2	Pair 1	Pair 2
ZNY				
ZOB		-		X
ZTL				X
ZFW	-		X	
ZBW		X		X
ZDC		X		X
ZMP-1				X
ZMP-2	X		X	
Totals:	3/	16	7	/16

Table 7.18 Pairs commenting on the effect of the arrival airport's runway on performance

Other information highlighted and added in the process of generating performance improving ideas will be discussed in the section describing the annotation of flight route maps, where the map itself can be used as a reference for that information, such as traffic on other routes, departure sectors, and military airspace regions; but some useful information can also be exchanged that is not tied so directly to information on these maps. Table 7.18 shows which pairs made a comment regarding how the runway in use at the arrival airport could have an effect on performance. The mode is not statistically significant (Chi-squared, p= 0.127) in its effect, but the table shows that most pairs in both modes did not mention the effect of the arrival runway without information regarding the runway in use. Data from the ZID scenario was not included because that is the one scenario involving a city pair where the arrival airport is not a major airport and thus it is not surprising no participant mentioned the arrival runway. It is also perhaps worth mentioning that it was actually the traffic manager who raised the issue of the arrival runway in all cases, but the ZTL voice and pointing mode.

Analysis of Annotation Modes

Speech and pointing

Almost all voice ad pointing mode participants made deictic references while speaking

	Dispatchers	Traffic managers
Participants making deictic gestures	18/18	17/18

Table 7.19 Count of voice and pointing mode participants making deictic gestures

As shown in Table 7.19, 18/18 (100%) of the dispatchers in the voice and pointing mode made deictic gestures during the creation of their messages. 17/18 (94%) of the traffic managers responding in this mode also made deictic gestures. The one traffic manager who did not make any deictic gestures (a traffic manager given the OMA to MSP problem) made only two relatively short recordings (42 seconds and 33 seconds).

• Participants did not normally point continuously

While the majority of participants using this mode made deictic references, the mouse pointer did not remain in motion continuously, as in general, most did not move the mouse unless there was something to point to related to their comments. Thus, for instance, when the traffic manager working with the ZFW scenario recommended contacting the Kansas Center each day to "get a heads up" on what ZFW is doing and whether or not arrivals over UKW or BYP are in effect, he did not move the pointer until mentioning UKW and BYP, when he pointed to both those locations.

• Objects and locations pointed to

The deictic gesturing that took place can be divided into two kinds: (1) deictic gesturing to objects and locations visible on the slide; and (2) deictic gesturing to objects drawn or locations marked with visible annotations. As will be discussed later in this chapter, most participants created pen and arrow marks and in the voice and pointing mode most did so while speaking.

• Information added by deictic gesturing

1. Locations

In this context the location of objects (such as named or unnamed holding locations, fixes, particular routes, arrival or departure approaches, Centers, "being metering" locations, merge traffic points, sequencing segments of a route, miles-in-trail segments, and route/jetway selection decision points) and events (such as vectoring, holding, deviations from the filed route onto a different route/jetway) can be made explicit in relation to an underlying map with synchronized speech. Information contained in tables and graphs could similarly be identified, and for these gesturing can clarify what is meant by phrases such

as "high fuel burn" or "unacceptable delay". Depending upon the knowledge/beliefs of the recipient this could be new information or not, but regardless, it reveals the speakers knowledge/beliefs more precisely. Thus, for instance, when the dispatcher working in the voice and pointing mode with the ZTL scenario suggested considering a new preferred route leaving Detroit at a lower altitude to avoid Chicago's overhead traffic, and traced a partial route to the east while doing so with no comments about the path for the route at all, she revealed that she had at least a partial route in mind.

2. Reducing ambiguity

Specifying the location of objects and events shown on a slide can reduce or eliminate ambiguity when there are multiple interpretations for the comments made. Thus, the comment "some of these flights have unacceptable delays that create significant problems for our operation and we would rather see a ground delay in those situations" would be ambiguous on a slide describing multiple flights, each with delays at different stages of the flight.

Thus, for instance, the dispatcher working with the ZDC scenario in the text mode, who wrote "would it be possible to move some of this traffic north and out of the dca area?", made an ambiguous comment if he was actually thinking of a a route or routes that could be described more specifically. The same comment tied to a drawn route or traced route in the voice and pointing mode would not be ambiguous. Similarly, the Traffic manager who referred to Volunteer as "about half-way between the Macey and Rome arrival" made his description more specific when he simultaneously pointed to a location half-away horizontally, but to the north of Macey and Rome.

3. Revealing incomplete or erroneous knowledge



Figure 7.13 Example of a verbal error during pointing

The comments and pointing were synchronized for most participants most of the time, and the semantics of the pointing were also consistent with the semantics of those comments. Sometimes however, there were conflicting semantics. For instance, a dispatcher working in the voice and pointing mode with the ZOB scenario started a recording on the slide shown in Figure 7.13. by stating that the scatter chart showed flight times for instances between April 10th and the end of May for the 757 flight from MSP to EWR. In fact, this was not correct because the data she pointed to was for flights from April 1st and further, it was for the A320

flight. This error was likely caused by the top-to-bottom listing of the flights not matching what would intuitively be a left-to - right ordering of the flight data in the scatter chart, even though the key makes the mapping explicit. Because she pointed to the locations she was looking at when she was misreading the information on the slide it is easier to see the inconsistency than if just listening to the audio and possibly not being able to match each of the comments made to the data because of the additional workload involved in determining where to look. Similarly, a voice and pointing mode dispatcher working with the ZMP2 scenario commented that the airborne delays for the flight instances between OMA and MSP ranged from 20 to 32 minutes and that this seemed to be a little "extensive". In fact, the range was actually from 16 minutes, but the dispatcher read the 20 from the line below, which was the average airborne delay for a different group of flight instances. (Further errors of this kind when working with POET tables could be reduced by alternating the background color of each row in the tables that were all white.) In this case the dispatcher did not actually point to the '16' and '32' while speaking, but did place an arrow under each number, which made it easier to locate when the corresponding numbers were spoken.

4. Overriding verbal slips

Very similar to the mistakes just discussed were slips that were made. In these cases, the person speaking was not stating information that was incorrect because they believed that to be true, but because of errors made during the communication of those beliefs. Thus, for instance a traffic manager working with the ZTL scenario stated that the Volunteer transition is the shortest route. He also commented that "only when there is a major push will you be sent over Nashville on the Choochoo transition". As he said this he pointed to the Rome

transition, but later in the recording when he referred to the same location he paused briefly and this time called it the Rome transition.

A traffic manager working with the ZDC scenario made the comment, "Generally, if you come up the east side the chances of coming in with a delay are less than if you come in off the west side", while tracing a path through ZDC to EWR and a path through ZOB respectively. In this case, he seems to have said 'west' instead of 'south' because 'the west' shares the attribute that it contrasts 'the east'.

A voice and pointing mode dispatcher working with the ZFW scenario gave the subject "dfw east arrival deviations" to a recording he made when he discussed deviations for flights filed to the west. This comment could mislead the reader, because flights filed to arrive from the east are shown in the scenario.

- Negative effects from pointing
 - 1. Pointing was captured when moving to and from recorder buttons Because the pointer's positions were captured between pressing the start and stop buttons, movement to and from these buttons was captured in addition to any intended deictic gesturing. The problem was made worse by the fact the window containing these buttons was not visible during playback, so the viewer's attention is drawn to motion to and from locations that normally have no apparent significance.
 - 2. "Doodle pointing"

While the majority of participants did not move the pointer unless they were making a deictic reference or making a menu/button selection, some participants

did tend to keep the pointer moving even if there was nothing to point at in the image that corresponded to what was being said. For instance, one traffic manager from the New York Center pointed to some text data on the last slide of his scenario, then while speaking for the remainder of the recording tended to move the mouse pointer slowly back and fourth while "drifting" in an area on the slide where there was no data.

3. Unsynchronized pointing

Although no attempt was made to precisely measure the synchronization of the speech and pointing that occurred, there were two participants where a lack of synchronization was very noticeable when following the pointer. In one case this occurred because the speaker actually pointed at objects well before he talked about them (a traffic manager working with the ZOB scenario) and in the other case the speaker was using the arrow marker to annotate a location and didn't adjust the speed of his speech to avoid stating the location followed by other words before placing the arrow marker. The former case is consistent with the finding of Oviatt (1997) that pointing generally precedes speech, although in this case the synchronization was noticeably worse than the others in the amount of the difference.

4. Slowed speech

Again, while no attempt was made to precisely measure the speed of speech and annotation with arrow markers or the pen, the speed of both varied across participants as for instance, some participants did more problem solving before speaking, gave more attention to the problem solving itself, found the problem solving more difficult, and found the task of using the pen and arrow marker to

take more effort. In the majority of cases when the participants speech was about to get ahead of the pointing or using the pen and arrow markers the participant slowed down or paused the speech to achieve synchronization.

- 5. Pointing to complex imaginary objects not related to the image In another instance a traffic manager from the Cleveland Center discussed information on a scatter chart then appeared to have imagined a map in the "white-space" area of the slide as he pointed to imaginary locations and paths that corresponded to his verbal comments. However, he made no visible markings while he did this. It is worth noting that this was the only instance were the participant pointed to an imaginary complex object, but there were other instances of pointing to small numbers of unmarked objects or locations. This would be a negative effect if the listener was able to process the communication at a faster rate.
- The playback slider was very rarely used

Although the effect of moving the slider on both the audio and pointer's position were demonstrated, participant rarely used it. There could be a number of reasons for this, including:

- 1. The demonstration during the training was too brief;
- It is a novel concept, in an otherwise intuitive environment¹⁵, so it didn't occur to participants to use it;

¹⁵ The claim is made that it is intuitive based on the debriefing questionnaire.

- It wasn't obvious when it might be useful, because the participants had to remember if pointing occurred in distinct regions / stages, so that it would therefore be useful to speed up the search process;
- 4. There was not constant dynamic pointing on a slide;
- 5. The recordings were clear enough there was no need to playback the recording.
- "Subjects" were rarely entered for recordings

14 of the 18 dispatchers using the voice and pointing mode did not create any subjects for their recordings. 1 participant from the Washington center created a subject on only the first of five recordings ("intro"). 3 dispatchers created subjects for all of their recordings (although one of them just used, "dtw-bos", "dtw-bos2", and "dtw-bos3"). Most of the dispatcher subjects were *issue related*, such as "dfw east arrival deviations", and "mem-ewr southern routing", or *message structuring* such as "introduction to memewr flight data".

11 of the 18 traffic managers using the voice and pointing mode did not create any subjects for their recordings. 1 participant from the New York center created a subject on only the first of three recordings. 6 traffic managers created subjects for all of their recordings. Again some subjects were not very meaningful, such as "slide one", "slide 2", but others were *issue related*, such as "route over tul" or *message structuring* such as "suggested remedy".

Subjects were intended to be static visual descriptions of dynamic speech and pointing, so without this, a potential summary and external memory aid is lost to what cannot be viewed "at a glance".

• Dispatchers and Traffic Managers normally created no more than one recording on a single slide

14 dispatchers and 15 traffic managers never created more than one recording on a slide.
4 dispatchers created more than one recording on at least one slide. (3 creating 2 recordings on a slide twice in their slideshows and 1 creating 2 recordings on one slide).
3 traffic managers created more than one recording on a slide (1 created 3 recordings on one slide) one slide and 2 created 2 recordings on one slide).

• Some participants created very long recordings



Figure 7.14 Dispatcher recording lengths

Figure 7.14 shows how many recordings each dispatcher made and how long each recording was. This figure shows that the most recordings made in a slide show was nine and

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

this was made by one of the dispatchers given the Atlanta Center scenario. The second largest number of recordings was 8 and the other dispatcher given the Atlanta Center scenario created these. The smallest number of recordings was 1 and it was both dispatchers given the Indianapolis Center, and one dispatcher given the first Minneapolis Center scenario who created one recording. The longest recordings (5 mins : 33 secs; and 2 mins 49 secs) by a dispatcher were made by one of the dispatchers given the Fort Worth Center scenario. The shortest recordings by a dispatcher were 13 seconds. These were made by one dispatcher given the New York Center scenario (who made four relatively short recordings: 19 secs; 13 secs; 23 secs; 28 secs) and one of the dispatchers given the first Minneapolis Center scenario (who made three relatively short recordings: 13 secs; 30 secs; 34 secs).

The average recording length for all the recordings was 61 secs (with a standard deviation of 45 secs). If the data for the one dispatcher who created the two longest recordings is considered to represent an outlier and removed this average changes to 56 secs (with a standard deviation of 29 secs).

Given the ZTL scenario caused the two largest number of recordings and the ZID scenario the smallest number, this number of recordings would appear to be a function of the scenario in these cases (even considering the fact the ZID scenario involved 5 slides and the ZTL scenario 7). This demonstrates that even though some participants appeared to feel obliged to make at least one recording on each slide, even if it meant duplicating information, there were times when the scenario supported referencing information on multiple slides collectively, and some participants did that.



Figure 7.15 Dispatcher slide show recording lengths

Figure 7.15 shows the same data as in Figure 7.14, but shows the cumulative effect of each recording made in the slide shows on the total recording length for each slide show. The longest slide show, in terms of total audio recordings length, was that produced by one of the dispatchers given the Atlanta Center scenario, who made 8 recordings on 7 slides and produced 9 mins 21 secs of audio. The second longest slide show was 8 mins 22 secs, but was just 2 recordings on the 2 slides in Fort Worth Center scenario. The shortest total length was 38 secs, which was the only recording made on the first slide of the 4 slide Minneapolis Center scenario (MDW to MSP). The second shortest total length was 66 secs, which was created by one of the dispatchers in the Indianapolis Center scenario, who also only made one recording on the first slide (of the 5 slides in the scenario).

The fact the two longest slide shows, in terms of audio, were slide shows of 7 and 2 slides, demonstrates that the total recording length is not just a function of the number of slides.



Figure 7.16 Dispatcher and traffic manager slide show recording lengths

Figure 7.16 shows the recording length for each dispatcher-traffic manager pair. The average length of the traffic managers' individual recordings was 1 min 57 secs, with a standard deviation of 85 secs. The longest individual recording was 7 mins 29 secs, and was made by a participant given the ZBW scenario. This participant made 4 other recordings, all above the average recording length. If his numbers are removed the average traffic manager's recording is reduced to 1 min 40 secs with a standard deviation of 59 secs. The shortest recording was the last one made by a traffic manager given the ZDC scenario, but the same participant made another recording of 2 mins 57 secs.

The average total amount of audio recorded by the traffic manager in a slide show was 6 mins41 secs with a standard deviation of 5 mins 56 secs. The longest was 25 mins 24 secs and the shortest was 1 min 15 secs.

The average amount of audio generated by the dispatcher - traffic manager pairs was 10 mins 48 secs. The shortest was 2 mins 24 secs and the longest was 31 mins 10 secs.

• Recordings were not normally played back after being made

Participants rarely played back their own recordings after making them. This sometimes meant a problem with the recording itself was not noticed and sometimes meant a problem with the recording in relation to other recordings was missed.

One of the voice and pointing mode dispatchers given the ZDC scenario had two extremely long pauses during his recordings: 17 seconds during one recording, after speaking for 28 seconds; and then 14 seconds during another recording, after speaking for 48 seconds. If he had played back his recordings he might have realized how long the pauses were and adjusted his technique to use the buttons to start and stop recording.

	Dispatcher	Traffic manager
ZNY-1	1, 2, 3, 4	2, 3, 4
ZNY-2	1, 3, 4, 6	6
ZOB-1	2, 4, 5, 3	1, 2, 3, 4, 5
ZOB-2	1a, 2, 4a, 4b, 1b	1,5
ZID-1	1	2
ZID-2	1	1
ZFW-1	1a, 1b, 2a, 2b	2a, 2b, 2c
ZFW-2	2,1	1, 2
ZTL-1	2, 3, 5a, 6, 7a, 1, 4, 5b, 7b	2, 3, 4, 5, 6, 7
ZTL-2	1a, 2, 1b, 7, 3, 4, 5, 6	2, 3, 4, 5, 6, 7
ZBW-1	1, 2, 6	1, 2, 3, 4, 5, 6a, 6b
ZBW-2	1, 2, 3, 4, 5, 6	2, 3, 4, 5, 6
ZDC-1	1, 2, 3, 4, 5, 6	3, 4, 6, 5
ZDC-2	1, 3, 4, 5, 6	2, 3, 4, 5, 6
ZMP-1-1	1, 2, 4	1a, 1b, 2, 3, 4
ZMP-1-2	1	1
ZMP-2-1	2, 1, 3	1, 2, 3
ZMP-2-2	1,2	2,3

Table 7.20 Order of recordings relative to the slides order

Table 7.20 shows that recordings were not always created in the same order as the slides. Five dispatchers and one traffic manager created a new recording on a slide earlier in the slide show than another slide that contained a recording already made. An example of a memory and/or mapping of comments to particular slides problem occurred with a dispatcher working with the ZTL scenario, who made a comment on slide 5 then made comments on slides 6, 7, 1, and 4, before returning to slide 5 to redundantly state (relative to the previous recording on the same slide) that at this time of day the traffic is handled very efficiently, staying on the direct filed route, so if the airline can be told what is being done differently at this time of day they should be able to get the same results at other times of day.

• Traffic managers normally played the dispatcher's recording one time

Traffic managers normally only played each of the dispatcher's recordings one time, which might have been because they felt they could remember the content of those recordings, and or because they did not want to take the time to listen to the recordings again. As discussed earlier, in this study all the dispatchers' questions and ideas for improving performance were responded to except two, which occurred in the voice and pointing mode. While this is not statistically significantly different from the text mode, there is evidence that the traffic managers would have benefited from a visual memory add of the most important issues and questions raised in the response of one traffic manager given the ZNY scenario who started a recording by saying, "Okay. You asked several questions and I'm trying to remember them all, but the first question that hit me was the fact you held and the actual location that you held."

172

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

Text comments

• Normally no more than one text box was put on a slide by a participant

15 of the 18 dispatchers working in the text mode created no more than one text box on any individual slide. 2 put 2 on 1 slide and 1 put 2 on 2 slides. 13 of the 18 traffic managers created no more than one text box on any individual slide. 4 put 2 on one slide and 1 put 2 on 2 slides and 3 on 1 slide.

• Most text boxes were moved from their default position

When a text box is selected in C-SLANT it appears in the top left corner of the screen. 11 of the 18 dispatchers moved all the text boxes they created, 1 moved some, but not all, and 6 never moved their text boxes. 13 of the 18 traffic managers moved all their text boxes, 2 moved some, but not all, and 3 never moved their text boxes.

• Most traffic managers used a separate text box to create their responses

6 of the 18 traffic managers always responded in the same text box as the dispatcher, 2 sometimes used the same text box, but not always, and 10 never used the same text box. Only one of the traffic managers who used the same text boxes intermixed responses in those text boxes. (This might be partly because this dispatcher's style was to make a numbered list of points that could easily be distinguished and responded to separately.) Not putting the responses in the same text box makes the order in which the text should be read non-obvious to any potential third user of the slide show, or the dispatcher if the traffic manager really had sent his or her response back to the dispatcher. • Text boxes were often enlarged

When more text is entered into a C-SLANT text box than can be displayed, it automatically produces a scroll bar; however, many participants chose to enlarge the text box making more of the text visible without scrolling.

Sometimes text boxes were left covering significant data on the screen



Figure 7.17 Text boxes left covering significant data

As demonstrated in Figure 7.17, which is the first slide from one of the text mode pairs given the ZNY scenario, enlarging the text boxes and leaving them in a position that the

recipient notices quickly while scanning the image can leave important data covered and potentially not noticed during subsequent discussion.

Pen marks

The use of pen marks is first discussed here without reference to the particular contents of the slides. How window and data types were annotated with the pen and arrow markers are both discussed in a later section of this chapter.

• Pen marks were used more by dispatchers than traffic managers

	Dispatchers	Traffic Managers	Dispatchers and Traffic Managers
Text mode	15/18	3/18	2/18
Voice and pointing mode	11/18	6/18	4/18

Table 7.21 Participants that used the pen

As shown in Table 7.21, 15 of the 18 dispatchers working in the text mode used the pen at least one time in their slide shows. 3 of the traffic managers in this mode also used the pen, and 2 of those 3 were working with slide shows where the dispatcher had made pen marks, although only one (a traffic manager working with the ZDC scenario) made any pen annotations on an individual slide that had dispatcher pen marks and he used a different color and wrote, "... circled in red". 11 of the 18 dispatchers working in the voice and pointing mode used the pen at least one time in their slide shows. 6 of the traffic managers in this mode also used the pen, and 4 of those 6 were working with slide shows where the dispatcher had made pen marks, with 3 of those 4 using the pen on the same slide at least one time. These were three traffic managers working with the first ZMP scenario, the ZID scenario and the ZFW scenario and in each case the pen color selected was different from the dispatchers.

While it was not an issue in this experiment, when the pen was used it would be ambiguous, to a potential third person, who had made those markings unless they were made while pointing and speaking, or they were referred to directly or indirectly in the speech or text.

• Dispatchers and traffic managers selected pen colors that contrasted with the background

Yellow and red were the colors used by the most number of dispatchers. This is perhaps because these are common highlighter colors and because they contrast the most with the colors used in POET. Yellow was used the most in the text mode (10 used yellow; 7 used red; 3 used blue; 3 used black; and 2 used green) working in the text mode. More than one pen color in a slide show was used by 7 dispatchers, but only 3 used more than one pen color on any individual slide. In the voice and pointing mode red was used the most (5 used yellow; 7 used red; 1 used blue; 1 used black; and 1 used green). More than one pen color in a slide show was used by 4 dispatchers, but only 1 used more than one pen color on any individual slide.

Yellow and red were also the colors used by the most traffic managers. In the text mode, where 4 used yellow, 4 used red, 0 used black, 4 used blue, 1 used green and 1 used white, 4 used more than one color in a slide show, with 3 doing so on at least one slide. In the voice and pointing mode, where 2 used yellow, 3 used red, 0 used black, 0 used blue, and 0 used green, 1 used more than one color in a slide show, but did not do so on more than one slide. In the text mode, 1 traffic manager used the pen on a slide with pen markings by the dispatcher, but chose to use a different color. In the voice and pointing mode, 4 traffic managers used the pen on at least one slide with dispatcher pen markings, and in all cases except one the traffic manager selected a color not used by the dispatcher.

Arrows

	Dispatchers	Traffic Managers	Dispatchers and Traffic Managers
Text mode	12/18	3/18	2/18
Voice and pointing mode	11/18	7/18	5/18

_ .

Table 7.22 Participants that used the arrow marker

As shown in Table 7.22, 12 of the 18 dispatchers working in the text mode placed arrow markers at least one time in their slide shows. 3 of the traffic managers in this mode also used arrow markers, and 2 of those 3 were working with slide shows where the dispatcher had made pen marks, although only one (a traffic manager working with the ZNY scenario) made any arrow marker annotations on an individual slide that had dispatcher arrow markers and he used a different color and used direct references to them. 11 of the 18 dispatchers working with slide shows. 7 of the traffic managers in this mode also used arrow markers at least one time in their slide shows. 7 of the traffic managers in this mode also used arrow markers, and 5 of those 7 were working with slide shows where the dispatcher used arrow markers, with 2 of those 5 using arrow markers on the same slide at least one time. These were traffic managers working with the second ZMP scenario and the ZTL scenario. A different color was used by the ZTL traffic manager, but not the ZMP traffic manager. The ZMP traffic manager put the same color arrows on a scatter chart as the dispatcher, but he put them just below the horizontal axis at the same horizontal position as the dots he wanted to annotate to group them separately from the arrows of the dispatcher.

• Participants wanted to be able to place arrow markers pointing from a variety of directions

On several occasions participants using the arrow markers wanted to have an arrow marker pointing in a direction other than in an up direction. This was perhaps to be able to place the arrow in a location where it would not cover data or cover the least important data.

Annotation mode interactions

• More visible annotations were made by dispatchers

	Dispatchers	Traffic Managers	Dispatchers and Traffic Managers
Text mode	17/18	5/18	5/18
Voice and pointing mode	15/18	9/18	8/18

Table 7.23 Participants that used arrows or pen marks

Table 7.23 shows that more visible annotations (annotations with the pen or arrow marker) were made by dispatchers than traffic managers in each of the two modes, but that more dispatchers made annotations in the text mode (17 versus 15) and more traffic managers made annotations in the voice and pointing mode (9 versus 5). It also shows that 5 pairs in the text mode and 8 pairs in the voice and pointing mode made annotations with the pen or arrow markers. 6 dispatcher-traffic manager pairs made annotations while simultaneously speaking.

• Most voice and pointing mode participants made visible annotations while speaking

	Dispatchers	Traffic Managers	Dispatchers and Traffic Managers
Voice and pointing mode	9/15	9/9	6/9

Table 7.24 Participants that made arrow or pen mark annotations while speaking

As Table 7.23 and Table 7.24 show, 15 of the 18 voice and pointing mode dispatchers made arrow or pen mark annotations and 9 of the 15 made annotations while simultaneously speaking. 9 of the 18 voice and pointing mode traffic managers made arrow or pen marks annotations and all 9 made annotations while simultaneously speaking. This is consistent with the results of Oviatt (1999, mentioned in Chapter 3, who found that 95% to 100% of users preferred to interact multimodally when they were free to use either speech or pen input in a spatial domain, but that users typically intermix unimodal and multimodal expressions.

Sconario	Dispa	atcher	Traffic manager	
Scenario	Text	V & P	Text	V & P
ZNY	5, 21	0, 6	0, 10	0,0
ZOB	5, 12	0, 5	0, 0	0, 19
ZID	3, 5	0, 11	0,0	3, 1
ZTL	2, 14	30, 12	0, 1	4, 17
ZFW	0, 1	17, 1	0,0	0, 10
ZBW	9, 33	3, 2	0,0	0, 0
ZDC	2,3	7,7	2,0	0, 31
ZMP1	6, 17	15, 1	3,0	10, 0
ZMP2	9, 14	7, 14	0,1	8,0
Average	8.94	7.67	0.94	5.72

Table 7.25 Number of pen or arrow annotations per slide show

Table 7.25 shows the distribution and average number of annotations made per slide show. The average number made by traffic managers in the voice and pointing mode was higher than the average in the text mode, although this was in part because a few traffic managers made a large number of annotations.

The fact that a higher percentage of traffic managers (9/9 = 100%) made annotations than dispatchers (9/15 = 60%) while working in the voice and pointing mode, and that more annotations were made by traffic managers in the voice and pointing mode (average = 5.72) than by traffic managers in the text mode (average = 0.94) is possibly influenced by not remembering their pen and arrow marks would not appear synchronized with the speech and pointing.

• Text mode participants did not always use direct references to their annotations

Arrow markers or pen marks can be used to draw attention to something on a POET screen or to add information to the screen capture, but then a reference to that annotation needs to be made in a text box when working in the text mode in order to make the purpose for that annotation explicit. 8 of the 17 dispatchers, and 2 of the 5 traffic managers, working in the text mode who made annotations created some with some direct references to them such as, "the red arrows show...", "the proposed route drawn in green...", or "the fuel burns highlighted in the table....". When direct references were not made more cognitive work is required to connect the relevant text to the annotations and sometimes ambiguity resulted. For instance, one of the dispatchers working with the ZNY scenario in the text mode asked the question, "Would it be possible to move some of this traffic north and out of the dca area?" The traffic manager voluntarily thinking aloud after reading this comment expressed confusion as to how far north the dispatcher meant. In this case the traffic manager decided the dispatcher could have been referring to routes through ZOB, or routes through ZDC (much closer to the DCA area), and covered both interpretations in his response. In a different situation the traffic manager might have assumed the wrong interpretation and not actually answered the question that was intended.

Annotation of POET windows

Annotation of Search Summaries

A search summary was included on the first slide of 8 of the 9 scenarios, because it shows the city pair involved in the scenario and also provides temporal context information. (The one scenario that did not include a search summary was the ZOB scenario, but the filed route table and scatter chart on the first slide together showed the city pair, departure time, aircraft type, time of year, and the number of flights involved.) Only one of the 36 dispatchers made visible annotations (with the pen or arrows) to draw attention to the city pair or the temporal context information, although some mentioned this information in their initial comments, as discussed earlier in this chapter. The one participant that did was a dispatcher working in the text mode with the ZMP2 scenario, who used a yellow pen to highlight the city pair and range involved. Given that other windows could be left covering this information, it would seem that in many cases the city pair involved would not be immediately obvious to a recipient of a message, and the temporal context may be impossible to determine with certainty.

Annotation of Scatter Charts

Scatter charts were used in 5 of the 9 scenarios. In each of these 5 scenarios a scatter chart was part of the first slide, to provide an overall summary on a performance metric (and in the ZTL scenario a scatter chart was also used on two other slides to introduce two different groups of flights for the same city pair).

In the ZNY scenario a dispatcher working in the text mode drew a horizontal line to indicate that flights above that level were considered to have serious performance problems. A traffic manager working with the same scenario (and a scatter chart not annotated by the dispatcher he worked with) in the voice and pointing mode, placed seven arrow markers below dots representing flight instances that were considered a problem. The traffic manager seemed to be suggesting again that it was a group above a certain level that were most worthy of investigation, but actually didn't mark all of them above that level. This is possibly due to cognitive work involved in placing markers and speaking at the same time, and trying to synchronize speech with the gesturing. One dispatcher working with the ZOB scenario drew a box with the pen to indicate it was flights within a certain range that should be the focus of attention. In the ZTL scenario a dispatcher working in the voice and pointing mode drew a

continuous line through a collection of dots representing flight instances he was describing, and then marked three with arrows that were particularly bad, while speaking at the same time. The traffic manager referring to these flight instances placed arrows in the same color at the same horizontal position, but just off the graph to refer to the flight instances. The other voice and pointing mode dispatcher for the ZTL scenario drew short lines in yellow through each of the individual flight instances that he referred to while speaking. In the ZBW scenario a text mode dispatcher used a horizontal line to indicate a temporal gap in the data. The other text mode dispatcher used a yellow arrow to refer to a cluster of flights, a green arrow to refer to the worst performing flight instance, and multiple blue arrows at the same horizontal level to refer to a group of flights performing at that level. This dispatcher then made direct references to the first two arrows to help make the text less ambiguous and quickly interpretable. No participants made a visible annotation on the scatter chart of the ZDC scenario.

In general, annotations made on the scatter chart showed participants found both the arrow and pen useful when making visible annotations. They also used color or boundary forming lines to group and separate flight instances into categories. Line and rectangle drawing annotations tools in C-SLANT might improve the efficiency and appearance of some of the annotations participants were trying to produce.

Annotation of Tables

Every scenario had at least one expanded filed route table. 5 involved at least one flight table and 6 involved an expanded filed route table with an instance highlighted.

7 dispatchers in the text mode used a pen to make a visible mark in a table; 4 used an arrow; and 3 used both. 4 dispatchers in the voice and pointing mode used a pen to make a visible mark in a table; 3 used an arrow; and 1 used both. 1 traffic manager in the text mode used a pen to make a visible mark, but none used any arrow marks. 1 traffic manager in the voice and

pointing mode used a pen to make a visible mark in a table and 1 used the arrow marker, but they were not the same traffic manager.

The pen was most often used as a highlighter in the tables. The pen was also sometimes used to circle a block of numbers. Arrows were used to mark individual figures, including doing so when a line was highlighted in blue in POET making it difficult to annotate clearly with a highlighter.

Annotation of Maps

Every scenario includes at least one flight route map to allow discussion of routes, traffic flows, and the location of other related factors such as holding fixes. 13 dispatchers in the text mode made visible annotations on a flight route map. 12 dispatchers in the voice and pointing mode made visible annotations. 4 traffic managers in the text mode and 8 in the voice and pointing mode made visible annotations on a flight route map.

Semantics for object	Annotation for syntactic representation	Example Scenario
Route not shown in	Drawn with the pen or a sequence of	ZTL Traffic Manager
screen capture	arrows	V&P pair #2, Slide 2
Route shown in	Highlighted with the pen or marked with	ZBW Dispatcher Text
screen capture	one or more arrows	pair #1, Slide 2
Proposed route	Annotated as a route shown or not shown	ZTL Traffic Manager
-	in the screen capture, dependant upon	V&P pair #2, Slide 2
	whether or not the route already exists in	-
	the image	
FAA 2nd, 3rd etc.	Same as proposed route	ZDC Traffic Manager
choice route		V&P pair #2, Slide 2
Enroute Fixes	Marked with an arrow	ZOB Traffic Manager
		V&P pair #2, Slide 5
Traffic merging	Circled with pen	ZDC Traffic Manager
point/region		Text pair #1, Slide 6
Traffic flow "switch"	Marked with a line using the pen	ZFW Traffic Manager
point for split		V&P pair #2, Slide 2
Traffic flow decision	Marked with an arrow	ZTL Traffic Manager
point for split		V&P pair #2, Slide 7

(Continued)

Table 7.26 Catalog of map annotations

183

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

Table 7.26 (Continued)

Sequenced route	Line with pen to indicate start of segment	ZDC Traffic Manager
segment		V&P pair #2, Slide 2
Holding (1 st choice,	Circled with the pen, or marked with one	ZBW Dispatcher Text
2 nd choice, etc)	or more arrows	pair #1, Slide 3
Departure sector	Mark with an arrow	ZFW Traffic Manager
		V&P pair #2, Slide 2
Turn for runway	Path drawn with pen	ZMP2 Traffic Manager
direction		V&P pair #1, Slide 2
Over-water holding	Circled with the pen, or marked with one	ZBW Dispatcher Text
	or more arrows	pair #1, Slide 4
Vectoring	Marked with an arrow	ZNY Dispatcher Text
		pair #2, Slide 3
Center boundary	Marked with the pen or multiple arrows	ZTL Traffic Manager
		V&P pair #2, Slide 7
Military airspace	Filled polygon drawn with pen	ZID Traffic Manager
		V&P pair #1, Slide 2
Altitude cap regions	Circled with the pen	ZID Traffic Manager
	_	V&P pair #1, Slide 2

Table 7.26 shows the types of annotation that were made on the flight route maps with the arrow marker and pen. (In the ZBW scenario the international traffic flow into the BOS area and the "begin metering point" were referenced, but no pen or arrow marks were used to annotate them.) Because the annotation tools in C-SLANT were very simple and generic the result was that the annotations made focus attention, but are not immediately meaningful beyond that. Thus the comments must be used to interpret their significance. The annotations and comments made in this study could be used to determine additional information to be added to the POET maps directly and or be used to design a richer set of annotation tools in C-SLANT. Such information could also be useful for pre-flight analysis. To protect the integrity of C-SLANT's ability to be used for multiple domains and tasks, this might be implemented via templates. The design challenge would be to add a richer set of annotation tools while supporting novice users.

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

.

Participant	Path	Pointed to	Pen/Arrow marks
ZNY text mode traffic manager #1	PSB traffic merging with MIP traffic in an arrival sector	N/A	No
ZNY text mode traffic manager #2	Sector #75 has LGA arrivals via J46	N/A	No
ZNY voice & pointing mode traffic manager #1	Traffic flow from the west into the DTW area	Yes	No
ZOB voice & pointing mode dispatcher #1	Transient traffic crossing the filed route	Yes	No
ZOB voice & pointing mode traffic manager #1	Indianapolis Center holds Continental and other flights for us [said while tracing a traffic flow connecting to the filed route	Yes	No
ZOB voice & pointing mode traffic manager #1	The route used by arrivals from Europe	Yes	No
ZOB voice & pointing mode traffic manager #1	Traffic from the south which would arrive via [29	Yes	No
ZOB voice & pointing mode traffic manager #1	A route over Morgantown	Yes	No
ZOB voice & pointing mode traffic manager #2	Traffic from the south through the Washington Center	Yes	No
ZBW voice & pointing mode traffic manager #1	Departing route out of BOS to the north west	Yes	No
ZBW voice & pointing mode traffic manager #1	Departing route out of BOS to the north east	Yes	No
ZBW voice & pointing mode traffic manager #2	North Atlantic traffic	Yes	No
ZBW voice & pointing mode traffic manager #2	Departures out of BOS to the south west	Yes	No
ZFW voice & pointing mode traffic manager #1	Heavy traffic from the north east through Fort Smith	No (but there was little room to do so)	No
ZFW voice & pointing mode traffic manager #1	Heavy filing to the north west corner of DFW from the south west	Yes	No
ZFW voice & pointing mode traffic manager #2	Traffic that ZFW must merge for a BYP arrival from Little Rock and another route	Yes	Yellow pen mark
ZTL voice & pointing mode traffic manager #1	Other traffic from the west merge on the Rome arrival	Yes	No

(Continued)

Table 7.27 Comments about flight paths for other traffic

Table 7.27 (Continued)

ZTL voice & pointing mode traffic manager #1	Other traffic from the east merge on the Macey arrival	Yes	No
ZID text mode traffic manager #2	The CVG inbound push via the APE VORTAC	N/A	No
ZID voice & pointing mode dispatcher #1	The heavy traffic flow of the north east corridor	Yes	No
ZMP-1 text mode traffic manager #2	Blending the BAE and GRB streams together	N/A	No
ZMP-1 voice & pointing mode traffic manager #1	Other Northwest flights arriving at this time from the east coast	No (but he was talking over a slide without a map)	No
ZMP-1 voice & pointing mode traffic manager #2	South east departures out of MSP	Yes	No

Table 7.26 is a catalog of visible annotations that were made on the flight route maps,

however Table 7.27 shows that when comments were made about other flight paths that were not the filed route or part of any other route being discussed as an alternative between the two cities, it was not normally drawn (i.e. 0 out of 4 for the text mode and 1 out of 11 for the voice and pointing mode).

Scenario	Text mode pair		Voice and pointing mode pair	
ZNY	0+1	0+1	0+1	0+0
ZOB	0+0	0+0	1+4	0+1
ZBW	0+0	0+0	0+2	0+2
ZDC	0+0	0+0	0+0	0+0
ZFW	0+0	0+0	0+2	0+1
ZTL	0+0	0+0	0+2	0+1
ZID	0+0	0+1	1+0	0+0
ZMP-1	0+0	0+1	0+1	0+1
ZMP-2	0+0	0+0	0+0	0+0
Totals:	4/18 p	airs > 0	11/	/18 pairs > 0

Table 7.28 Count of dispatcher + traffic manager pairs mentioning flight paths for other traffic
There is also a difference between the two modes as shown in Table 7.28, with 4/18 pairs in the text mode and 11/18 pairs in the voice and pointing mode mentioning such a path. This is a statistically significant difference at p < 0.05 (Chi squared, p = 0.018).

This is consistent with the fact discussed earlier that certain basic flight classifying information was described by dispatchers more often in the voice and pointing mode. The fact that participants normally pointed when mentioning these paths in the voice and pointing mode again makes the semantics more meaningful than just the same verbal comments alone.

Within Slide Window Management and Navigation

• Players and text boxes left covering data

Text boxes were normally moved from their default position at the top left corner of the screen, but they were very often still left covering data even when there were "empty" positions on the screen available. For instance, 66.6% of the dispatchers covered data on their first slide although there was always room to avoid this. The example of the text mode dispatcher working with the ZID scenario was mentioned earlier in this chapter as one where this caused a problem because he reported he didn't know the departure time of the flight, but in fact it was covered by the text box he was using to make that comment.

• Some participants tried to interact with pictures of controls that were part of the screen capture

As Figure 7.18 demonstrates there were times when participants mistook a control that was part of a POET image used in a screen capture for an active control. In this example the user tried to move the contents of the image map to the left while using the arrow marker, but the result was that an arrow marker was placed on the picture of a scroll bar.

THE AND HIS MORE	
> 9(b) 1	
Fie Ed Vews Go Addins Window Hep	
Store of Fragmonry Fright 1 Filed Warder A	rarch2. wh: Summary: Highty, 10
Location South2 str. Standay, Fights1: DDV/A151105/2: Fiel Routes Ser. 37151	
07/01/1 998: Depet Line: 0000.16/2400:1	TAATHE Post
	Actual Baute
Explore Data Minute	
Fiel Route Contraction of the American State of the	
24 DTW: CAVVS::VWV: APE::HVGO 1 Air FuelBurn Uncor. (bs) 15,900.0 20.020.0 4,120.0 25.92	
A Flading	
Holding 0 6 Air FuelBurn Uncorr. (bs) 13,466.7 16,153.3 2,666.7 20.04	
Air Fueldum Uncorr. [bs] 12,500,0 25,70 25	
Air FuelBurn Uncorr. (bs) 14,100.0 17,120.0 3,020.0 21.42	
Air FuelBurn Uncort. (bs) 13,500.0 15,560.0 1,980.0 14,52 Air FuelBurn Uncort. (bs) 13,500.0 12,550.0 1,250.0 12,550.0 1	
7 Air Fuellium Uncorr. (bs) 14.042.9 14.974.3 931.4 6.62	
Play Frity Stop Streeming 20 mms 0 secs	
From the second s	
C THE REAL TOWNED OF A DEPENDENCE SUBJECT STATE	
	10
	112

Figure 7.18 Example of confusion between controls in the image and C-SLANT

• When there were multiple recordings on a slide, traffic managers sometimes were not sure which

one to look at first

Departure Anival	Departure	raft Numbe		2.0.0 F		্ৰ না	Dillerence	HAR T
e Airport Airport		ie instan O	ces Performa	nce Metrics	Planned A	ctual (A	ctual - Planned	
	_							
shi jahi summary Eligi	a 1 Isiplayi							
cater Chart			16.23					
CYAN STAT	2		Similarenv2 dia		101200	E.S.	52.34	-
							CTARE C	1
12, 70.0						<u>_</u>	NWA MSP EV	VR 1945 757
		Golo Cap				1.1	NIVA ISP EV	WC 1955 320
60.0	-1	ex course in			1. A. C. A.			$\epsilon < 3$
£ 500				September 19				53.5° .
5 45.0	•	HE COMPANY		2542.244	2. A 1. A	÷ 2,59		7.5 X X
5:40.0							974 YZ	7-2-3-3
£ 350								
\$ 300		•				57 S. A.		A. T. 1. F. 1.
200		•	e Bassier,				J. Same	
E 150								
10.0					0.000 10.20			
50 0				Sizebal Tan 30		-00		
Sol			•	Tone Alternation	Lengt	- 0 mins 40	30C1	
10.0			and the second second			ADD DALLA		
04/10/1998	04/25/1998	05/10/1998	25/1998	Subject			H	1944 - Y

Figure 7.19 Poor arrangement of player windows

As Figure 7.19 demonstrates, the ordering of multiple recordings on a slide became difficult to determine and work with when player windows were not arranged in a logical non-overlapping order (such as left- to-right or top-to-bottom). In this case the dispatcher created recordings at 11:21, 11:23 and 11:31, but playing them back in the correct order requires revealing the hidden 11:21 window, then selecting each of the other two player windows in order to reveal the play button. The original design idea was to allow recordings to be placed close to that which they are related on a slide, but in this study there appeared to be little benefit of that kind.

Between Slide Navigation

The slide show structure allowed single or multiple comments to be associated with a slide, but comments were often related to the images on other slides too, and comments were also related to each other on the same or other slides. Comments could also be placed on slides in any order, and this would not necessarily be the order the recipient read/listened to them. Therefore slides were not as independent as necessary for the simple structuring of comments that was used, as demonstrated with the following examples.

• Creating comments out of order on the slides could confuse the recipient

A voice and pointing mode dispatcher working with the ATL scenario created recordings on slides 2, 3, 5, 6, 7, 1, 4, 5, and 7, in that order. On slide 1 the dispatcher commented that the graph indicated there were instances when the flight stayed on the filed route and there were other instances that had extreme reroutes causing very high over-burns. He then stated, "Again, information would go a long way in telling us when to use the filed route and when to use the overflow route". The "again" comment was made because the dispatcher commented first on slide 2 that "we should be advised of changes to the route so we can plan for fuel-burns"; on slide 3, "providing information to us in a timely fashion would allow us to save the gas", and on slide 7 that if the airline was told the filed route couldn't be flown before departure or before crossing Dayton the reroute and holding could hopefully be avoided.

A traffic manager working in the voice and pointing mode with the ZFW scenario showed he was aware of the potential ordering problem when after making a recording on the second slide he started his next recording with the comment, "I went back to slide one for a moment. I didn't want to put a recording there because it would be the latter of the two recordings."

• Participants sometimes referenced information on other slides

One of the traffic managers working in the voice and pointing mode with the ATL scenario started a comment on slide 6, "Here are the two instances of holding" because there was information about a flight that had two instances of holding on the previous slide. In fact, he actually was looking at information for a different flight. This is a simple example of a participant referring to information on another slide and one where he assumed the recipient will work through the slides in order.

A traffic manager working in the voice and pointing mode with the ZOB scenario started his first recording by commenting, "I've previewed the next four slides and in response I'll leave it on this slide for starters". This shows the traffic manager wanted the recipient to understand the traffic manager's comments were made in the context of more than the current slide, which can only be achieved in C-SLANT with a comment such as this one.

• Traffic managers sometimes started responding after only viewing or listening to the recording(s) on the first slide

One text mode dispatcher given the New York scenario used the scatter chart data on the first slide to point out that flights from Detroit to Newark were burning 11.6% more fuel than planned and that there were some burning much more than this because of holding. In subsequent slides he used the maps and further numeric data to indicate where the holding was taking place and that flight instances filing a northerly route through the Boston Center did not seem to have the same problem. The traffic manager responded after looking at the first slide only as follows:

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

As we all know, EWR is a very busy airport during the 1930 to 2300 period - times where these flights are arriving. There is always a fine line whether or not a ground delay program should be implemented. For this day, the weather indicated a ground delay program was not necessary and we (ZNY) would go with miles-in-trail. Unfortunately, the flights from many places got clumped more so than usual. A possibility would be to file via the SHAFF arrival routing and inbound through ZBW - rather than via ZNY. Another possibility would be to rearrange the schedules to come either slightly earlier or slightly later (i.e. after the big arrival period).

I take a look at the remaining slides to see what else you've offered. I'm curious how that 11.6% figure compares with flights to LGA or to other destinations. Also, is that 11.6% figure high in comparison to more fuel burned during other periods of the day.

After later viewing the other slides and seeing that the dispatcher then asked about filing the northerly route, the traffic manager could have modified this initial text (more easily than a person in the voice and pointing mode could do the equivalent), but he actually did not, and redundantly recommended the northerly route again.

Another example of the traffic manager responding to the comments of the dispatcher one slide at a time occurred with the ZBW scenario in the voice and pointing mode. In the dispatcher's recording he stated that he was sending a number of slides and then highlighted the data he saw as significant on the scatter chart of fuel burn data. In response, the traffic manager commented that it was hard to say why the extreme over-burn and under-burns occurred with just fuel burn data. He then commented that the jet stream may have been in favor for some instances and weather at Logan caused a lot of holding for others, but it was difficult to know without more information. He then went on to respond to each of the dispatcher's comments in order, but this was an example where he need not have guessed what the reasons were if he had looked at the information on the remaining slides.

• Treating each slide separately creates redundancy in the comments for related slides

As has been discussed already, some traffic managers treated each slide separately, in that they responded to the dispatchers' comments on each one before looking at the next slide. Some participants also repeated their previous comments if they again seemed relevant to the current slide, even though their information was available on the previous slides. This occurred for instance with the ZBW scenario where one of the voice and pointing mode traffic managers made a comment about Gardner as the primary holding location on each slide. When he reached the last slide he played the dispatcher's comments which started with the observation that most of the holding appeared to be taking place at Gardner. The traffic manager then stated that, "Yes, the area you have circled in red is the Gardner VOR, which is where we do most of the holding for flights filed on the Albany route", followed by the comment, "we are beating a dead horse here!" He then went on to respond to the rest of the dispatcher's comments and the information on this slide and took 40 minutes and 15 seconds to complete the task which was the longest for the voice and pointing mode traffic managers.

• Some participants wanted to record across multiple slides

A few participants tried to change slides while recording because they wanted to "tie" together something on one slide with something on another, using speech and pointing. C-SLANT wouldn't allow this and stopped the recording, upon clicking the next slide button, with a warning message. While not supporting this option may at times stop users from creating very long recordings, not being able to do this sometimes created negative effects in this study. For instance, a traffic manager working with the ZDC scenario created no comments on slide 1, then commented on slide 2 that he would explain the reason for the additional fuel burn shown on slide 1 on this slide and then proceeded to use the flight route map to explain how particular routes flown could be tied to the high fuel burns.

Analysis of Implementation Issues

An important implementation level issue is how large the files were that C-SLANT used to represent the messages created by the participants. Given the storage capacity of even modest computers today, the storage requirements are actually less of an issue than the transmission time between these target organizations. Because it can not be assumed that Local Area Network (LAN) transmissions speeds of up to 10 Mbps or 100 Mbps will always be available, the analysis below assumes only a 56 kbps (equivalent to 7 kilo bytes per second) connection, which is a typical modem speed.

File size and transmission time

Slideshow Description	Minimum		Average		Maximum		Standard Deviation
ondesnow Description	КВ	Min : Secs	KB	Min : Secs	КВ	Min : Secs	КВ
A. Images Only	58	0:8	128.0	0:18	180	0:26	36.8
B. Annotated Dispatcher's Message (Text Mode)	59	0:8	133.3	0 : 19	185	0 : 26	37.0
C. (B – A)	1	< 0:1	5.3	0:01	18	0:03	4.1
D. Annotated Dispatcher's Message (Voice & Pointing Mode)	414	0 : 59	1913.1	4:33	4275	10 : 11	1168.8
E. (D - A)	286	0:41	1785.1	4:15	4095	9:45	1132.1
F. Annotated Traffic Manager's Message (Text Mode)	59	0 : 08	135.3	0 : 19	187	0 : 27	37.7
G. (F – B)	1	< 0 : 1	1.9	<0:1	11	0:02	2.6
H. Annotated Traffic Manager's Message (Voice & Pointing Mode)	1151	2:44	4837.6	11 : 31	13254	31 : 33	3216.6
I. (H - D)	488	1:10	2924.5	6 : 58	10415	24:48	2517.0

Table 7.29 Slideshow storage and transmission time requirements (assuming a 56 kbps connection)

Table 7.29 summarizes how large the slideshows created were and how long it would take to send those messages with a modem connection. The storage requirements for the pen marks and arrows are not separated in the table, but they would be relatively small as they are represented by a set of x, y coordinate positions in a file. Row A shows that the average slideshow size before any annotations or comments were made was 128 KB. Row C shows the average storage needed for text mode comments and annotations made by dispatchers was 5.3 KB, and row E shows the average storage needed for voice and pointing mode comments and annotations made by dispatchers was 1785.1 KB. The respective average transmission times for the slideshows produced would be 19 seconds versus 4 minutes and 33 seconds as shown in rows F and D. The average additional storage requirement for the traffic manager's response itself in the text mode is shown in row G to be 1.9 KB. The traffic managers working in the voice and pointing mode created an additional 2924.5 KB on average, although the two ZBW traffic managers added the most to their slideshows and these were considerably more than the average (10415 KB and 5869KB). If these two outliers are removed the average additional storage is reduced to 2272.3 KB. The average size of the slideshow produced was 4837.6 KB, as shown on row H, which would take 11 minutes and 31 seconds to send with a modem. The largest slideshow was 13254 KB, which would take 31 minutes and 33 seconds to send with a modem. This extreme case on a LAN would take less than 20 seconds to send, if only half the maximum bandwidth of a 10 Mbps network is available; however, for a modem connection that might require a person to initiate a connection and disconnection, or might timeout, a transmission time of more than a few minutes may not be acceptable.

Slide Description	Ave	rage	Standard Deviation
onde Description	КВ	Min : Secs	КВ
A. Image Only	27.2	0:04	3.0
B. Annotated Dispatcher's Slide (Text Mode)	28.4	0:04	3.1
C. (B - A)	1.2	<0:01	
D. Annotated Dispatcher's Message (Voice & Pointing Mode)	444.4	1:03	407.8
E. (D - A)	417.2	1:00	
F. Annotated Traffic Manager's Message (Text Mode)	28.7	0:04	2.9
G. (F - B)	0.3	<0:01	_
H. Annotated Traffic Manager's Message (Voice & Pointing Mode)	1065.6	2 : 32	783.3
I. (H - D)	621.2	1 :29	

Table 7.30 Slide storage and transmission time requirements (assuming a 56 kbps connection)

Table 7.30 shows per slide average storage and transmission time requirements, and provides a rule of thumb estimate of 1Mb and 2 ½ minutes per slide in the voice and pointing mode for each slide with both dispatchers and traffic manager comments.

The two tables show:

- The voice and pointing occupied most of the storage required for voice and pointing mode slide ((417.2/621.2)/1065) = 97.5%);
- The text mode slideshows were approximately 2.8% (135.3/4837.6) the size of the voice and pointing slideshows.

Improvement in the transmission time can be made by (1) improving the bandwidth of the connection; (2) changing the implementation of messages; (3) changing the interface of C-SLANT to encourage more succinct messages; or (4) train users to create more succinct messages.

Questionnaire Results

After saving their messages the dispatchers and traffic managers were given a questionnaire to get feedback on the perceived value and usability of C-SLANT and being able to use it in the type of context represented by the experiment. Traffic managers were also asked if there was anything they thought the dispatcher could have done to improve the message they received. Both groups of participants were asked questions to determine experience with relevant hardware and software and work experience.

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

Participants Profile

	Dispatchers	Traffic Managers
Gender	8 Female, 28 Male	4 Female, 32 Male
Age Range, Average Age	33-55, 45	31-57.43
Experience with Organization (range, mean)	9-35, 18 years	3-32, 17 years
Computer w/sound	26%	34%
Word-processing Experience	81%	88%
Spreadsheet	61%	51%
Database	35%	48%
Presentation Software	23%	37%
Computer at Home	85%	91%

Table 7.31 Participant background and experience summary

As Table 7.31 shows both groups of participants involved more males than females, but that is representative of the respective populations. The age and experience levels between the two groups were also similar. More of the traffic managers had computers with sound cards, and generally had experience with more popular generic computer applications, although more dispatchers had experience with spreadsheet programs. These questions were asked in case background differences between dispatchers and traffic managers, might help explain observed differences and perceived usability and value of the system by participants.

Question: On the following scale, please indicate how comfortable you are using voice mail?						
Results	Very uncomfortable	Uncomfortable	Neutral	Comfortable	Very comfortable	
Dispatchers	4	3	4	14	8	
Traffic Managers	3	2	3	13	14	

Table 7.32 Participant comfort levels for voice mail

Both groups of participants communicate frequently using voice-based communications as part of their work and are used to having their communications recorded. It was therefore

expected that they would generally be comfortable or very comfortable using voice mail, which would suggest that the voice-recording feature of C-SLANT would not be an obstacle. Table 7.32 shows that this prediction was correct, although 21% of dispatchers and 14% of traffic managers were very uncomfortable or uncomfortable.

Question: How ofte	n do you currently use E-r	nail?		
Dispatchers		Traffic Managers		
19 (61%)	Daily	23 (66%)	Daily	
1	3-5 times a week	1	Regularly	
2	2 times a week	1	Often	
1	Weekly	3	Weekly	
1	Monthly	1	Monthly	
2	Rarely	2	Sometimes	
1	Yearly	3	Rarely	
4	Never	1	Never	

Participants use of E-mail

Table 7.33 Frequency of E-mail use

Table 7.33 shows that approximately two-thirds of both groups of participants responded that they use E-mail on a daily basis. This suggests that participants were equally comfortable with the concept of E-mail for communications. While this is a majority, Table 7.33 also shows that a significant minority does not use E-mail very often, if at all, so some training would be involved in introducing this form of asynchronous communications for frequent use. For instance, 13% of the dispatchers responded that they never use E-mail and 11% of traffic managers responded they rarely or never use E-mail.

Question: [If you use E-mail] How often do you send attachments?					
Disp	atchers	Traffic Managers			
14 (45%)	Never	7 (20%) Neve			
4	Rarely	4	Rarely		
1	Infrequently	1	Occasionally		
1	Not Often	2	Fairly Regularly		
2	Monthly	1	Regularly		
2	Weekly	5	Weekly		
1	10%	1	20%		
4	Sometimes	3	Sometimes		
1	Often	5	Often		
1	50%	1	50%		
		5	Daily		

Table 7.34 Frequency of sending attachments

Table 7.35 shows that 18 (58%) of dispatchers said they never or rarely send attachments,

11 (31%) of traffic managers said they never or rarely send attachments.

Question: [If you us send?	e E-mail and send attachn	nents] What kind o	f attachments do you
Dis	patchers	Traffi	c Managers
11/13	Images	10/24	Images/Pictures
7/13	Text	9/24	Word
1/13	Spreadsheet	8/24	Text
1/13	Audio	3/24	Excel
		2/24	Variety
		1/24	Data

Table 7.35 Type of attachments sent

Table 7.35 shows that 85% of the dispatchers sending attachments indicated they sent graphics files and 42% of the traffic managers indicated they sent graphics files. 54% of dispatchers indicated they sent text attachments and 79% of traffic managers indicated they sent text or Word attachments.

Question: [If you us attachment?	e E-mail and send attachn	nents] How often do	you include an audio
Disp	patchers	Traffic	Managers
11/13	Never	15/24	Never
1/13	Infrequently	5/24	Rarely
1/13	Sometimes	3/24	Sometimes
		1/24	Often

Table 7.36 Frequency of audio attachments

Table 7.36 shows that 85% of the dispatchers sending attachments never send audio attachments and 63% of traffic managers never send audio attachments. Overall, this suggests that both groups are not experienced using computers to send audio messages, but because they are generally comfortable using voice-mail, training to do so would likely be successful.

Perceived usability and value of C-SLANT

Question: On the follow	ing scale, please indicate how	easy/difficult you four	nd this software
to use?			

Results	Very easy	Easy	Neutral	Difficult	Very difficult
Text mode dispatchers	8	7	2	0	0
Voice & pointing dispatchers	2	11	4	0	0
Text mode traffic managers	8	8	1	0	0
Voice & pointing traffic managers	12	6	0	0	0

Table 7.37 Perceived usability of C-SLANT

Question: On the following scale, please indicate how useful you think this kind of software might be in your job?

Results	Not useful at all	Not very useful	Neutral	Fairly useful	Very useful
Text mode dispatchers	0	0	2		6
Voice & pointing dispatchers	0	3	3	9	2
Text mode traffic managers	0	1	4	8	4
Voice & pointing traffic managers	0	0	2	4	12

Table 7.38 Perceived utility of C-SLANT

200

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

Table 7.37 and Table 7.38 show the results from usability and utility rating questions, respectively. No dispatchers rated their version of the system as "Difficult" or "Very difficult" to use, and 82% overall gave a rating of "Easy" or "Very easy". Overall, 76% of the dispatchers rated their version of the system as "Fairly useful" or "Very useful". The "average" text mode dispatcher's usability and utility rating was higher than that of the voice and pointing mode dispatcher's.

No traffic managers rated their version of the system as difficult or very difficult to use, and 97% overall gave a rating of "Easy" or "Very easy". Overall, 80% of the traffic managers rated their version of the system as "Fairly useful" or "Very useful". The "average" text mode traffic manager's usability and utility rating was lower than that of the voice and pointing traffic manager's, which is in contrast with the relationship between dispatchers.

A preference by traffic managers for the voice and pointing mode over the text mode, but the reverse for dispatchers might be explained as follows:

- The task of responding to a message may have been easier than creating the original message, because traffic managers were normally discussing problems in the airspace they were most familiar with, whereas dispatchers were assigned scenarios at random. Responding to an initial set of observations and questions rather than creating them, could also be argued to be a simpler task in itself.
- If it is accepted that the task was easier and more "natural" for the traffic manager, it could be argued they would prefer the communication mode that supported the more rapid production of comments and would not need the greater flexibility in editing previously created comments that is provided by text based communications as they would be less likely to change their minds on what needed to be communicated.
- The tasks were also different because the traffic manager not only created a message, but also had to interpret the dispatchers message. Even though the dispatchers tended to rate the text mode higher for usability and utility the traffic managers may have

preferred to receive a message in the voice and pointing mode than the text mode and that could have influenced their ratings.

Question: If you had access to this software regularly how do you think you would use it?

Traffic Managers: All but two traffic managers gave an answer to this question. The categories for their answers are shown below with example comments. <u>To analyze performance</u>

- Analyze specific traffic flows; attempt to identify specific causes of delays.
- To find out where the traffic bottlenecks are.
- Evaluate present routes/alt usage compared to sector loading issues i.e. does excessive vectoring/alt. Changes in lieu of new routing prove to be cost effective and can the new routing be efficient.

To improve communications in general

- I would probably not initiate a request/question, but would be grateful for the opportunity to respond to a question, especially where our customers felt that they were being delayed.
- Exchange of info/ideas with other air traffic facilities.
- I think this would be a very helpful tool during severe weather season to explain to the users how/why we chose rte methods.

To communicate about a specific recent event

- To help answer specific questions, immediately (the next day) after the occurrence involved.
- I think this will be very useful in sending several types of messages (audio, maps, routing, procedures) in a timely manner and being able to receive feedback easily.

For training

• Reproduction of offload scenarios for flow effectiveness - as an additional training aid

For personal or shared annotation of events

• Just by leaving notes to fellow employees or in documenting situations that might be reviewed at a later date.

Dispatchers: All but three dispatchers gave an answer to this question. The categories for their answers are shown below with example comments.

To analyze performance and strategic planning

- I would use it to plan NWA's contingency fuel above and beyond normal reserve and alt (if needed).
- As a dispatcher not much. For planning purposes based on past performance, it would be a great tool for recognizing trends.
- Use it to help identify problems as well as show what is working in the aviation system. <u>To improve communications</u>
- Allows for very quick dissemination of information to proper persons/departments.
- Capture live data insert dialog and prompts share with chief dispatcher / ATC coordinators.
- Saving / capturing info for reports following flight progress more closely.

For training and briefings

- As a question / response system as well as in-house historical info such as playing back the previous day to improve how we do things.
- Document operational situations for use to best justify equipment training and to use scenarios to train other dispatchers.
- Communicating with ATC facilities and eventually communicating with flight crews during a pre-flight briefing.

Wouldn't use it, because not real-time

(This suggests some participants didn't separate the functionality of POET from C-

SLANT, because C-SLANT could be used to capture information from real-time displays such as

ASDs - or possibly that some didn't consider the task of finding patterns of performance in previous flights their responsibility. A number of dispatchers said they thought the software would be more valuable to the ATC Coordinator or others who normally look back at performance data. Many of them used the term "Monday morning quarterbacking" when discussing looking at historic data of individual flight instances. A number of traffic managers also indicated that the Traffic Management Officer (TMO) or person charged with looking back at performance data would find the application more useful.)

- It loses the actual dynamics of the "real-time" situation so I don't think I would use it.
- Depends if it was new or after the fact. Now would be great.

Question: Was there anything you particularly liked about this software?

Traffic Managers: 16 traffic managers using the voice and pointing mode and 12 traffic managers using the text mode answered this question. The categories for their answers are shown below with example comments.

Simplicity

7 of the traffic managers using the voice and pointing mode, and 7 using the text mode, chose to mention that they found the software easy to use. This does not include those that only mentioned a specific feature.

- User interface very simple! Any idiot could use this tool.
- Very easy to use.
- It worked well and was easy to understand how to use.

Voice and pointing

- Voice and pinpoint this helps explain situations.
- Yes, the ability to utilize the voice feature associated with pointing and specific issues.
- Voice recording pen. I like to be able to point and talk at the same time.

Other annotation tools

- I liked the ease of using the red arrow. The pointer.
- Highlighting.

Slideshow structure and images

- The ability to incorporate various slides you usually have on your desktop PC.
- The visuals allows for a better mental picture.

Electronic communications

• Yes – voice over to ask questions. Capability to send electronically.

Personal communications

• I liked it because it makes the presentation more personable.

Dispatchers: 15 dispatchers using the voice and pointing mode and 12 traffic managers using the text mode answered this question. The categories for their answers are shown below with example comments.

Simplicity

4 of the dispatchers using the voice and pointing mode, and 4 using the text mode, chose to mention that they found the software easy to use. This does not include those that only mentioned a specific feature.

- Very easy to use.
- Easy to use; simple.

Voice and pointing

- Ability to be able to refer and highlight.
- Voice recording and highlighting seem useful.
- The sound and drawing capabilities.

Graphical user interface

7 dispatchers specifically referred to the graphical user interface. This is perhaps because dispatcher's visual communications with flight crews and other airline personnel is text based.

- Clarity ability to display lots of info in windows form.
- Graphical data is always useful.
- Good graphics, fast, easy to read.

Other annotation tools

- Could attach post-it and highlight or arrow.
- Custom options highlighter etc.
 - Electronic communications
- Like identifying the numbers than typing commands that go directly to someone.

Participant suggested enhancements

Question: Do you have any suggestions for improving this software?

Traffic Managers: 17 traffic managers responded to this question with ideas. The

categories for the traffic managers and example comments were.

Interface suggestions

- Add timer clock to view.
- The ability to draw line segments.
- Equip the voice feature to type the spoken word into the post-its.
- Perhaps adding a "pointer" tool box to access after you've used pen or other tools.
- More colors. Arrows pointing in different directions. Eraser head.
- Indicate the origin of recorded message. The only way to differentiate between my or NWA recording is the date.
- It would be useful to be able to click on airways or fixes to highlight them.

206

- Maybe an "undo" function to cancel (erase) just the last mistake. <u>Information not in POET</u>
- It would be beneficial to also include distance traveled i.e. filed route versus flown route.
- Need more data (wx, rwy configuration, details of the daily operation)
 <u>Information in POET, but not included in the scenarios</u>

(In reality, some of this information would likely be asked for and provided in

subsequent communications as exchanges occur back and forth.)

- Would like to see the routes filed and flown via flight plan into ex: dtw..fdy..rod..flm..bkw..sbk3..rdw.
- ETA Times area.

Dispatchers: 15 dispatchers responded to this question. The categories for the dispatchers and example comments were.

Interface

- To be able to review my verbal message (in writing) prior to sending it. I would also like to be able to go back and forth between slides, and to have all the slides on one page.
- Make buttons easier to follow for listing flights/filters.
- Access to all screens at once.
- Have an "undo" function for the arrow / pointers.

Not in POET

- Great tool for real-time briefing if it had a weather overlay. Could brief pilots at remote stations or their homes.
- Need wx overlay / turb info / more of traffic runway closures / dest wx winds / enroute wx the influencing factors / variables. The audio was interesting, but I was not comfortable with not having an immediate response: "was I clear in my statement?" Could I be misinterpreted? That could be a problem with communication.

Traffic Managers' critique of dispatcher messages

Question: [Traffic Managers] What advice would you give an AOC person sending you a message using this software?

a. With regard to using the interface itself to convey a message.

b. With regard to the content of the message in the aviation context.The categories for the traffic managers' responses and example responses were:

Be specific

9 out of 29 answers to this question indicated the dispatcher should be specific. This might partly be because normally reviews would be at the individual flight level or limited to a particular time period on one day, relatively shortly after the event, when factors such as weather may by more significant than for a collection of flight instances over a longer period of time.

- Don't try to involve too many situations at once. One issue at a time.
- Timely: try to send data when event is still fresh. Identify specific concern.
- Point out specific problems and solutions you are seeking. Also include your preferences and any advice possible.

Provide sufficient detail

- Make sure all details are either drawn or verbally expressed.
- Be as thorough and complete as possible and tell me your name (1st name is ok) so that I can talk to you.
- Be very specific with your question. Supply all necessary data, i.e. weather along route of flight, any statistics.

Be concise

- Keep it short.
- Be concise what are your goals what is your request / final outcome?

208

• Use concise messages when possible.

Ask questions

- Asking for help on different circumstances. Ex. If we file this route would it help? What about time frame?
- Ask definitive questions. How/why/what can we do?
- Ask specific questions, and identify concerns. I cannot relate to 3,000 pound extra fuel burn. What impact to operation are we addressing?

Explain what is on the slides

- Explain the purpose of the included slide the receiver needs to know the point of the exchange.
- Explain your concerns or points on all the data or slides that are presented.

Other interface issues

- Preferable to the "post-its", use the arrows and drawing tools. "A picture is worth a thousand words".
- Speak clearly + use pointer to annotate what they are discussing.

Structured Interview with the ZNY TMO

The results of this experiment provide evidence for how novice users of C-SLANT and POET might interact, because none had previously used either application. The TMO from the New York ARTCC used C-SLANT for several months as a beta tester, and also gained experience with POET during that time. The TMO's position is particularly interesting because that person communicates frequently with the airlines and other FAA organizations and is often asked to review the performance of the New York ARTCC after a significant event. The TMO's responses to questions during a structured interview via E-mail to determine the nature of the questions asked by these other organizations, and his prediction for the impact of using C-SLANT, is

presented below.

1. What types of question do you get, asking you to explain the decision making that took

place to strategically manage traffic through ZNY?

E.g. Why did you need 30 MIT from Boston Center to (ATL) during a restriction period in which Washington Center placed 15 MIT to ATL on New York Center? Perhaps there were only 3 or 4 airplanes coming from Boston Center so it seems like the 30 MIT is an overreaction.

E.g. Why didn't you get traffic on to J-XX sooner. We showed the weather clearing that route at XXXX. It could take hours to run replays of the sector and search for the Terminal/TRACON and Center picture at the time to determine if the question is even valid. Sometimes it is relatively easy, other times it can get very complex. Then how do you explain it? Verbally over the phone with no visual references? In an E-mail trying to describe where the weather was, when and what other traffic was in the sectors? Normally we are in a defensive posture, defending the actions of our facility.

2. What percentage of these types of question do you get asked as opposed to others

working in the facility?

The TMO is the focal point for the operation. Generally speaking, unless it's a purely Quality Assurance issue, i.e. systems error or pilot complaint, my office receives the questions. The TMO, or representative is the focal point for the Airlines and other system users, as well as adjacent facilities and the ATCSCC.

3. Who asks these questions? (i.e. what organizations are involved and what positions do

those people hold?)

Other TMOs, ATT4, who is the ATCSCC individual responsible for coordinating and communicating areas of concern to the field facilities, ATCSCC National Operations Manager (NOM) for Severe weather or NOM for the EAST side in the ATCSCC, User groups such as the Teterboro Users Group Rep., NBAA, Airline ATC reps., Tower Managers at the NY airports, those types.

4. How many of these questions would you be asked in a day on average?

Things have changed since 9/11. During the summer, after a SWAP event there could be several questions the next day. Lately we have been so engaged keeping our head's above water, and working with the military blocks that these things have dwindled. Let's say it averages one per day.

5. How often are you asked the same question?

It can be streaky. We tend to get wrapped around an issue for a week or two, then move on as we react to the latest crisis. Questions about the need for, and effectiveness of MIT restrictions are continuous. That is why I want to get better at POET as a training tool for the ZNY TMCs to see the results of their actions, as well as demonstrate what the issues are to other people.

- 6. How often are the questions asked by telephone versus email versus any other form? *Issues are generally raised on a Telcon or by telephone.*
- 7. How often can you answer the question using just the information you have in your head

about the events that took place?

Rarely. Unless it was brought up during the event and I have explanations from the people who worked the event in the TMU.

8. When you have to investigate further, what is typically involved?

A SATORI replay review. Contacting the STMC and/or TMCs involved. Reviewing reroutes that were used. Sometimes talking to Controllers and Supervisors. It could also involve a call to a customer or adjacent facility to see what their perspective was at the time of the event.

9. How long do these telephone conversations usually last?

Too long! We've spent 8 man hours, over the course of days to respond.

10. Can you think of ways that using C-SLANT might improve the efficiency of answering these particular types of questions?

Yes, especially if I could get good convenient slides of weather overlaid on the airways. You could do a review every day, with a little practice I don't believe it would be a great deal of work. The MIT review relies more on POET. POET screen capture with voiceover would be a good way to go. The analysis could then also serve as an archived training tool in the off season. "OK let's look at this scenario from last summer. Here's what you can expect when the weather is here, etc, etc."

11. Can you think of any disadvantages of using C-SLANT to replace or supplement these

phone calls?

A process must be developed to familiarize users with the tool and all need the software.

12. Any other comments?

I think that it is very powerful when weather is involved, because the person can see the picture and the expert can interpret it via voice to explain the impact. I experimented doing a voiceover using the CCFP. That's another area to explore. Then you archive it via the subject on the cover page¹⁶. You might have a training class two years after the event, but the weather is consistent in that it's always there (brilliant)! Another item that just recently came up in the meetings I attended was feedback to the Severe weather people in the ATCSCC. They don't get any of this review and analysis. Think about it. They work the shift, go home and most often never hear or see what other stakeholders have to say about their operation. They are not available for the telcon the next day and are usually completely out of the review loop. That's significant, but no one has figured out an effective, timely, workable way to create the feedback, and deliver it. We could be using CSLANT right now to send voiceovers to explain to the military and users the impact of military blocks, because we have a picture of the block on an airway chart. The picture itself is somewhat meaningless without an explanation.

The TMO's comments provide important information for potential further research using C-SLANT or similar applications in this domain:

- The study here involved analysis of performance for particular city pairs based on historic data normally involving several months and one airline. In the comments made by the TMO established and ongoing discussions were described that focus on the events for some portion of one day.
- 2. Miles in Trail (MIT), sector volumes, and weather are important factors for traffic going through ZNY and they need to be represented clearly when performing analysis.
- 3. The TMO and those he listed as individuals he interacts with, when reviewing decision making to strategically manage traffic through ZNY, could form an appropriate group of users for C-SLANT. The group is already established, their task is clearly defined, and they already comment at a distance. One of the reasons cited in Chapter 3 for the failure of FREESTYLE was that a critical mass of appropriate users was not given the system to use, so knowing the existence of a group such is this is very valuable. They currently communicate synchronously, but there appear to be problems with that as the primary form for communications, because it requires significant preparation and the information

¹⁶ The idea of a modification to C-SLANT where a "cover page" would overview the slide show and provide data that could be used for archiving the slide show was mentioned to the TMO before he responded to these questions. This is discussed in more detail later in this chapter.

is often graphical so it is difficult to visualize this with verbal descriptions. The TMO's idea of developing daily reviews using C-SLANT, POET images and images from other applications would be one approach. These could be made available on the web for responses by, for instance, telephone, C-SLANT, or an application like Microsoft Netmeeting, if a synchronous communication is needed between participants who need to see the images at the same time. In this case the TMO discussed reviewing operations, but a similar approach could be taken when projecting forward while, for instance, looking at weather maps and discussing what will be done or is proposed to deal with that.

4. C-SLANT could be used in combination with the CCFP to discuss weather in the system when projecting forward or looking back to discuss the quality of the forecast made and its impact.

Discussion of Results

Comparison between the text and voice and pointing modes

Although C-SLANT will support the creation of messages involving text and/or voicebased communications with deictic referencing, one or the other of these features was disabled in this experiment in order to understand better the effect of each mode in this context where the efficiency of working with data represented by screen captures is important. To estimate how often the modes would be used when users are given a choice, eight dispatchers, not involved in the main experiment described here, were given slide shows to annotate using C-SLANT with no modes disabled. Four chose to use voice and pointing alone, three chose to use text alone, and one used both modes. This suggests there is value in supporting both modes to accommodate user preferences, as well as flexibility for task and environment variations.

The text based mode is also a more traditional form of computer supported asynchronous communications, so now that the technology exists to support voice and pointing based communications it is valuable to understand the impact of using it.

Aviation context issues

This study was not only an investigation of the differences between a text based and a voice and pointing based communications mode in this context, but also an investigation of the interactions that would occur when these two groups were given POET data and allowed to discuss the actual performance problems that it represented. It thus provided an opportunity to learn more about the effectiveness of POET as well the information these two groups could exchange when working collaboratively.

Enhancements suggested for POET

Weather and traffic density information were the two most commonly requested additional sources of information. The runway configuration in use was also often mentioned because it can impact the arrival rate, the taxi-in and out time, and the direction of approach. In general, this type of information can help explain not only what happened to particular flights, but also why it happened. Figure 4.16 in Chapter 4 showed many of the factors that influence the performance results stored in POET. The more of them that the system has information on, the more of the "story" for a particular flight instance can be told; the more patterns can be revealed for multiple flight instances; and the more criteria are available to conduct searches. Thus, all of the additional factors shown in Table 7.39 could have been useful when using POET 1.0.

Data	Example Search Criteria
CCFP weather information	coverage > 50% and sector = ABC
Sector traffic density	density > 15 and sector = ABC or DEF
Planned and actual distance traveled	actual - planned > 500
Playbook play	play = ZNY2
Ground stop	groundstop = LGA
Runway	arrival = EWR and runway = 2
Holding fix	holdingfix = ABC

Table 7.39 Potential flight or airspace attributes for POET 1.0

A basic problem solving strategy that can be used with POET when looking to identify poor performance and potential improvements is to conduct comparisons between what are or might be alternative options. For instance, alternate routes between the same city pair, alternate times of day, and flights held versus those not held. When information can be displayed about two alternatives on the same screen, those options can be compared without having to change screens and remember the attributes of one of the options. POET supports displaying many alternatives of this kind on one screen, but it does not support the display of multiple instantiations of the flight route map window to allow juxtaposition where, for instance, the filed versus flown routes can be unambiguously be seen for multiple filed routes without interacting with the window. This would appear to be a useful alternative method for comparison, and particularly useful if trying to produce a snapshot of information on POET screens to become the focus of discussion, where in general, a smaller number of screen captures will be easier to work with. In creating the ZFW and ZMP scenarios, there was some "tweaking" of the screen captures to actually simulate that ability and reduce the number of screen captures needed.

The comments and annotations made in this study show certain information (such as shown in Table 7.26 is used or discussed in explaining the events that have taken place when analyzing or predicting performance. Including information the traffic managers used to explain the system to dispatchers in POET (such as altitude cap regions [e.g. the ZID and ZTL scenarios], the prioritization for traffic on certain routes [e.g. the ZDC scenario] and decision points and reasoning for determining whether to reroute enroute or not [e.g. the ZFW and ZTL scenarios]) should make the system more useful.

The study also revealed certain minor interface problems with POET as participants misunderstood some information on the displays. As discussed earlier, the scatter chart was misread because of problems reading the scale for the date range, the mapping of the flights table to the scatter chart data caused confusion because the table ordering of top-to-bottom didn't map to left-to-right for multiple flights, and a table was misread when the user incorrectly matched information on one line to information on another.

Further Enhancements to C-SLANT

C-SLANT 0.3.7 was used by all the participants in the experiment described in this chapter; however, it has continued to be developed further, with several significant changes implemented as a direct result of the observations made during this study. In this section the results described earlier are discussed and the redesigns implemented or proposed for C-SLANT in response are described.

Improving discourse

Problem	Redesign Idea
Annotations and text box comments did not have their author's name recorded	 Assign an author, date and text value to text box windows Support session layering and key creation for annotation markings
The ordering of discourse contributions is not recorded for text and requires too much cognitive work for voice and pointing	Maintain both text and voice & pointing comments within a table of a separate window. Support text labeling as another form of annotation

Table 7.40 Resign ideas to improve discourse structure

• Improving the distinction between annotators

For the experiment, participants were not asked to enter their name as is an option when starting C-SLANT. If this were done the audio recordings would have had the name of the person shown in any player windows created, as well as the date and time of creation (as was shown). However, text boxes, pen marks and arrows carry no explicit indication of who created them. When analyzing the slide shows created it was sometimes necessary, to go back to saved copies of the dispatcher's message before the traffic manager responded to separate the authors (although the color of the text can be changed and for this experiment the default color for the traffic managers' text was changed to be different from the dispatchers'). Because a traffic manager's response might easily be sent to a third person in reality, for instance a traffic manager at a different center, this is a potential problem. One approach to solving this problem would be to put all annotations in a temporal context and reveal them as layered sessions. Another approach would be to include an annotation key that assigns color (or some other distinguishing characteristic) to specific annotators.

The author, date and time could also be associated with each text box, as is done with Email messages. Then a separate window could be used to manage those comments in a table as is also done in E-mail programs. The same approach can be taken with voice and pointing recordings. As was shown in the analysis for this experiment, the sequencing of those recordings can be difficult to determine when multiple recordings are made on a slide and player windows overlap.

• Threading recordings to aid context and navigation

Given that participants didn't always make comments on every slide (and with further iterations and possibly more people involved in the discussion that might be expected to continue), having a single window for all text based comments and voice & pointing based comments should help navigation through those comments and context. In fact the current version of C-SLANT supports this management of voice & pointing recordings, as shown in Figure 7.20, but not text based communications.

÷., н.,		X
Play F	Stops	Hewind Respond Delete 0 mms 0 secs
Auto	natically	play next recording Length: 0 mins 45 secs
Author	- Slid	EL Date Subject
Roger	1	06/18/00 18:25 High fuelburns
Phil	-1	06/18/00 18:37 Note military airspace
Roger	2	06/18/00 18:26 Inpacted on times
Roger	3	06/18/00 18:27 Other traffic flows
Phil	-3	06/18/00 18:40 Center interactions
		and the provide the factor of the second

Figure 7.20 C-SLANT's recordings window

Multiple recordings are allowed on a slide to provide a mechanism to make distinguishable topics and sessions tied to the same slide disjoint, and responses to specific recordings are visibly noted as such to help capture the context of the response. The user has the option to sort a table of recordings by annotator, slide, date and time of recording, or topic. The user may also select an "automatically play next recording" option in this table to be presented with an interruptible "show" of slides and recordings, using the current ordering in the table. The combination of these last two features allows the user to, for instance, listen to all the recordings by a particular person across the slides; listen to the recordings by temporal ordering; or listen to the recordings by slide ordering.

Given that some participants wanted to refer to multiple slides in a single recording, while describing related information, it would appear to be an enhancement to C-SLANT to support this feature. The tradeoff is that long individual recordings could be created when the recording could actually have been divided into separate recordings, because the comments were significantly disjoint. Separate recordings when appropriate would be simpler for the recipient to manage. Training and automatic feedback when recordings are becoming long would appear to be important when supporting this feature. Richer frames of reference for the recordings could also help with their management as discussed in the section "improving voice and pointing communications".

Allowing a responder to interleave responses within a single recording could also be supported. This would be analogous to interleaving responses in a text-based E-mail message. How well that would really work requires further research, but one potential benefit could be more efficient matching of users' individual comments.

Improving problem solving in this domain

Problem	Redesign Idea
Flight groupings, performance measures, questions and suggestions, and answers are significant types of contribution to discourse that should be more salient and distinguishable. Participants considered the communications useful, and almost all the questions asked and issues raised received a response, but the communications efficiency was sometimes poor.	Add configuration options to C-SLANT to support message structuring. (Like FREESTYLE, C-SLANT was not initially designed for structured workflow, but templates could be used to selectively provide structure to achieve the benefits of structured collaboration described by Kraemer and King, and discussed in Chapter 2.)
Visible annotation marks on the maps were used to represent a small set of semantic types.	 Implement an (advanced) user-defined set of symbols to supplement the arrow markers Implement an (advanced) user defined key to allow semantics to be associated with pen marking types

Table 7.41 Redesign ideas to improve problem solving

• Adding structure to C-SLANT messages to improve efficiency

The traffic managers' responses to the questionnaire question, "what advice would you give an AOC person sending you a message using this software?" were summarized earlier and are noticeably similar to Grice's Maxims of Relevance, shown in Table 7.42. Providing true, unambiguous information and comments put in context is a significant goal when designing asynchronous interactions about POET data using C-SLANT. The results from this experiment provide evidence that multimodal communications involving voice based communications with deictic gesturing can be used to solve problems of this kind and participants responded favorably to doing so, but there was also evidence of some inefficiency in the voice-based communications and the use of a slide show structure.

Maxim	Application to this context
Quantity: provide as much information as needed in a context, but not more information	 POET uses filters to provide focused results, showing influencing factors and alternative options Recipients can provide more information or request it Assigning blame or complaining without details in the discourse provides no information
Quality: speak true information	POET data reduces speculation
Relation: make your contribution relevant to the context in which you are speaking	 Recordings associated with particular slides suggest that slide is the context Recordings related to each other should be linked to aid context
Manner: speak as clearly as possible, avoid ambiguity, say things as simply as possible	 Maps, graphs and tables aid perception Speech and pointing provide converging evidence of meaning and reduce complexity Speech is not visible and slower to interpret or scan than text Asking direct questions simplifies communications and reduces ambiguity for what is requested Structuring messages may make them simpler to manage

Table 7.42 Summary of C-SLANT based on Grice's Maxims of Relevance (1975)

220

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

One approach to reducing those inefficiencies would be to provide task and domain specific structure to the design, to make it more focused, organized and simple to navigate. However, in adding such structure, C-SLANT could lose value for other types of communication (e.g. training) between these groups or for other domains. Adding structure through configuration options (as can be done by system administrators with Microsoft Netmeeting) rather than hardwiring it would however, allow a generic system to be available, while supporting optimization for particular tasks and domains. The concept is also somewhat similar to the templates available for particular types of presentation in Microsoft PowerPoint.

One approach to supporting structure in C-SLANT would be to allow, through configuration options, a "cover page" to be designed for slide shows. For this task and domain it might consist of fields for each of the following: flight grouping information; poor performance indicators; questions and suggestions, and answers, author and contact information. Its purpose would be to serve as an overview for the content of the slide show, with key information summarized succinctly with text. A "table of contents" might also be useful to give summarizing title to each slide. The slide show itself would be an "expansion" of what is summarized, but it would exist within a framework that is established quickly and initially when working with the C-SLANT message. Participants tended to put summary information on the first or last slide, but one text mode traffic manager given the ZNY scenario explicitly asked, "If I want to make a general comment to provide an overview, where can I do that?".

An alternative approach might be to modify the list of recordings window to try to achieve much of the same effect. The flight grouping information could be displayed in a fixed location, and "comment types" such as performance indicators, questions, suggestions, and answers, could be associated with individual recordings (or text boxes assuming they are combined with the list of recordings). When starting a new comment the type would be first selected.

221

• Domain specific symbols and annotations

The semantics of the arrow marker is a static deictic gesture, but combined with some verbal or text based description of what is being referenced at that point the intention is more specific. As was noted in the analysis, these were often placed while speaking and talking, but when used in the text mode the language was more ambiguous, and the recipient had more cognitive work to connect the referent to references (that were often indirect). Supporting label annotations in addition to text boxes would provide another method for avoiding such ambiguity, but specific symbols would also reduce ambiguity. Thus, symbols for holding fixes, airports, corner-post, and metering initiation points, placed in a "symbol markers" configuration file could all be useful.

The pen marker was often used specifically to propose a new route, but so too was a collection of arrows, possibly because the pen mark proposed routes were too difficult to distinguish from the POET produced filed and actual routes. The ability to have a configurable set of pen colors and line styles (e.g. dotted or dashed), and assign these to a key, would reduce ambiguity in the message communicated.

Improving annotations

Wojahn et al., (1998) present a set of requirements for annotation interfaces that were compiled with text-based annotation systems in mind, but are also relevant here for the voice and pointing mode, and can serve to provide further guidance considering the actions participants took and the redesign of C-SLANT. Some of the issues described in Table 7.43 have already been discussed and the others will be in their respective sections.

222
Requirement	Comment
There is a minimum of motion required to make an annotation	 Connecting a reference to the referent in a message requires little effort once recording starts in the voice and pointing mode
	 Text requires typing which is slower than speech, unless modifying previously created comments Text requires additional words to unambiguously reference an annotation Pen marks don't involve using a stylus, but a mouse pointer
	 Initiating annotation modes requires mouse movement and visual button clicking – middle and right buttons could be assigned additional functions
The primary text is easily distinguishable from the annotation text	 Speech and pointing contrast with visual images Pen marks contrast with most objects composed of straight lines and with text, but not routes in a map Arrow markers contrast with most images Text boxes contrast better with images without captured windows
The annotations are visible "at a glance" while reading the primary text	 This is not achieved with the voice and pointing mode, because it is not possible to "glance" at it and see a sufficiently rich representation of the annotation Scrolling text is not completely visible Pen marks and arrow markers are visible
The relationship between the primary text and the annotations is easy to see	 Deictic referencing ties verbal comments with points and objects Text boxes can't always easily be placed near to that which they refer - labels could be The value of the textbox reference to a referent depends on how descriptive the semantics are to the reader Pen marks and arrow markers are placed at the point or area to which they refer
Different annotators are readily distinguishable	 User login names are associated with recordings through labeling There are no distinctions made between different users' text boxes, pen markings and arrow marks

Table 7.43 Summary C-SLANT based on guidelines by Wojahn, Neuwirth and Bullock (1998)

Improving voice & pointing communications

Problem	Redesign Idea
 The recipient's attention is sometimes misled in the image due to: 1. Unintended pointing during movement to interface options; 2. A visible pointer when pointing is not desired; 3. A visible image when the image is not being referenced. 	 Move the C-SLANT record buttons to a fixed location. Allow the deictic gesturing pointer to be hidden and reappear at the click of a mouse button or use of a function key Turn on and off recording with a mouse button click or use of a function key Allow the background image to be hidden,
	dimmed or masked in some other way
Subjects were often not given to recordings	Force something to be entered or require a "cancel" command
The recorder and player windows don't capture the speech and pointing dynamics in a static representation	 Incorporate SoundBrowser's design for representing audio dynamics Use speech recognition to generate text and tie setting the text cursor's position to the audio and pointing positions' Include an option to display a trace of the pointing path Mark temporal segments in the slider where pointing occurred
More polished presentations require additional speech and pointing preparation	Provide better support for pausing and resuming recording
The user is not discouraged from creating very long recordings	Implement a configurable maximum recording length and show a countdown display, or accumulating recording length and storage requirements.

Table 7.44 Redesign ideas for improving voice and pointing based communications in C-SLANT

• Avoiding distractive pointing and supporting dialog composition

One option in the current version of C-SLANT that would seem specifically useful for an audio and pointing recorder is the ability to start and stop recording by use of a mouse button rather than a visible button. The problem with the latter approach is that the viewer's attention is mislead watching the pointer go to and from the start and stop recording buttons when the recording is played back. Further, because those buttons are positioned relative to C-SLANT's main control window, and that can be moved, pointing when played back can appear to move towards or from a particular point for no particular reason. (For this reason the recorder controls

should be given a dedicated screen position, if the interface involves these buttons.) Making it simple and accurate to start and stop recording also might help the user to improve the narrative as there is the potential to focus on a point to be made, then stop and think about the next point carefully, and then focus on the narrative again. In the research of Bly (1988) and (Tang 1989; Tang 1991; Tang & Leifer 1988) summarized in chapter 3, they both emphasized the importance of smooth transitions between annotation methods as they observed participants do so rapidly and smoothly in team paper and pen design situations. Transitions between speech and pointing stages/events also need to be "smooth". The use of a mouse button to start and stop recording would produce better results for the recipient, but there is not a visible memory aid for this interface feature, suggesting that both methods for controlling recording should be available to support the novice and more experienced user. It would be difficult to achieve the simplicity of GroupSketch, described in Chapter 3, because C-SLANT has a lot more functionality, but making document ed more advanced features available, but not immediately obvious (to protect the novice from being overloaded) is the design approach suggested here.

• Improving the frames of reference for recorded audio and deictic gestures

In a study of the interface to SoundBrowser (Degan et. al., 1992), shown in Figure 7.21, it was reported that users found the following useful:

- The ability to place coded 'markers' in the recording to make it easier to retrieve particular audio segments.
- 2. A visual representation of amplitude to make it easier to retrieve particular audio segments.
- The ability to zoom in on a portion of the recording when the amplitude representation was too dense, as long as a global reference to the entire recording was mentioned.

225

4. The ability to play the recording back at up to 200% the original recording speed using a pitch maintaining algorithm.



Figure 7.21 The SoundBrowser window (Degan, et. al., 1992)

It would appear that C-SLANT might benefit from applying these findings, as the recorder and player windows have no visual information other than possibly a completed subject field and the current position and length of the recording. Further, another horizontal axis might be used to indicate which periods of time there was motion of the pointer. This type of display could be particularly useful when a sender has paused for long periods while thinking and does not speak or point, but keeps recording – as was the case with the traffic manager given the ZDC scenario and discussed in the analysis. The visual display would reveal such pauses and therefore potentially cause the creator to remove them or allow the recipient to stop over them. The ability to place coded markers could, for instance, allow the listener to note where there was a question or issue to respond to and then continue listening, having made a visible note of something to be returned to. This might have been useful for the ZNY scenario participant who appeared to be having problems remembering all the questions asked.

A speech recognition module could be added to C-SLANT to generate a text version of the audio, and the user could then be given the option of setting the current position, or range, in the audio by doing so to the corresponding text. If this were done it might be better to only have the text appear after the recording is completed or it could be too distracting. Another visual reference to assist in the process of scanning saved deictic gestures would be to have the option to display a static trace of the path taken and then perhaps allow the user to click on a particular point to set the current position in the recording. This would be somewhat analogous to scanning text, which can be done so much more efficiently than scanning audio (Streeter, 1988).

• Require or encourage a text subject for the speech and pointing recordings

Subjects were not given to any recordings by 14 of the 18 dispatchers working in the voice and pointing mode. In the version of C-SLANT used in this study the player window appeared after a recording was made with a blank subject and participants needed to notice it was blank and complete it. In the current version of C-SLANT the user is prompted for a subject after the recording is saved and must select "Cancel" to have a blank subject.

• Discouraging long recordings

A message consisting of a small number of recordings may require less recordings management than a message with a lot of short recordings, but long recordings can present significant cognitive demands because they don't have the chunking benefits of multiple small recordings. Providing better frames of reference on the recording may help, as discussed earlier, but if they are long because the speaker had not paid sufficient attention to succinctness, they can present storage and transmission time problems. To encourage shorter recordings the current length and storage could be displayed and updated during recording or a countdown could be displayed. The most aggressive approach would be to impose a strict recording length and simply stop recording when it is reached.

227

Improving text based communications

Problem	Redesign Idea
 Text boxes were left covering data in the 	Separate "message boxes" from "labels"
image	 Put links to displayable text messages in
• The order text boxes should be read was	the recordings window, as an alternative
not always obvious	to a voice and pointing message

Table 7.45 Redesign ideas for improving text based communications in C-SLANT

• Text boxes were sometimes obstructive and difficult to navigate

CAANT [Editor]		X
,s (0) 1	6	X
File Eak Yew Go Addins Window Help		- 1
	Side 2 J S (b) shol	
Contraction Comments States and S	of lest Lold View Uptions Help	
Search: Addine and FAA Data Fights that were hald appear most often in		. 1
have been held at the boundary between		
ZOB and ZNY. This would indicate to me a		
Explore Data Mining		
We've been encountering some holding at the ZC	DB/ZNY boundary generally once the traffic becomes	
Fiel Route	nd when ZNY cannot accept the fights (often because of	ļ
+ DTW: PISTN.DXII020.ADRIE.V other traffic in the first ZNY sector (ZNY#75), it is	not uncommon to have 208 hold at the boundary. This is not 🚳	
+ DTW-TYCOB YOGOSE DOLENK I desirable. It is important to note that ZNY W/S alw	ways works LISA anreads via J146 along with some crossing 51	
only apport involved.		
Ideally, some of those fights being delayed at the	e boundary would have been belter off if they had been	
ZUB reroute (last minute if necessary) via SHAF	FF. This normally eliminates the need for last minute holding at $-\frac{1}{4}$	
the boundary. We are all striving to eliminate last	t minute holding, but often it is unavoidable. See the red line 🛛 🗾	
05/04/1998 Air FuelBurn Lincon. (bs) 8,100.0	0 11,000.0 2,900.0 35.8%	
05/02/1998 Air FuelBurn Uncorr. (bs) 7,700.0	0 10,390.0 2,690.0 34.9%	
Air FuelBurn Uncorr. [bs] 11,200.0		
G4/13/1998 Air FuelBurn Uncorr. (bs) 7,400.0		
	W. UCA	
		. 1
	A DECK	

Figure 7.22 Example of the demonstrated need for "labels"

All the text boxes created in this experiment were message comments (with sentences) rather than "labels". As discussed earlier, supporting a separate text label annotation type would be an enhancement because it efficiently enables information to be added in relation to a specific location in the image, and Figure 7.22 demonstrates a situation where a traffic manager would have benefited from using one in the ZNY scenario. In combination with a line annotation, the label can also be linked to a specific point or object with some flexibility when trying to avoid covering data. As also discussed earlier, traffic managers tended to open new text boxes when responding rather than reusing the dispatchers' text boxes they were responding to. Even though the text of the authors was assigned a different color, multiple text boxes were sometimes created by one person (and if there were further message iterations this would continue) making the order in which they should be read difficult to quickly determine. These problems could be eliminated by organizing this type of text comment as part of a recordings window, and thus actually support mixed mode communications with a consistent interface. This flexibility might be useful to allow selection based on the characteristics of what is being communicated and the current environment. If the comments involve details such as a list of fixes representation of a proposed flight path where the sender wants to be sure they are communicated clearly, because they will be entered into another system, the text format is better. Alternatively if the sender wants to discuss what will happen at several locations along the proposed path, in terms of other information on the map, it would be more efficient to do so while pointing and talking. In other situations a person may not have a sound card, or be in a situation where they can make verbal comments because of the environment they work in (although that seems unlikely in this domain). It may also be a matter of personal preference as to which mode the user prefers.

Improving visible annotation markings

As discussed earlier both pen markings and arrow marks were anonymous and this might cause owner identification problems when messages are exchanged several times. Session

layering would be one approach. Assigning sets of markers to different authors would be

another.

Problem	Redesign Idea
Pen Marks:	
Pen marks for proposed routes didn't contrast sufficiently with POET filed and actual routes	Provide a wider range of colors and line forms
Wobbly lines were drawn where straight lines would be appropriate	Support a line annotation tool
A rough box was drawn with the pen	Support a box annotation tool
Long arrows were drawn with the pen	Support a flexible length arrow annotation tool
Arrow Markers:	
Arrows could only point in one direction	Provide at least four directions

Table 7.46 Redesign ideas for improving visible annotations

• Pen markings for drawing routes

As mentioned previously, participants sometimes used a sequence of arrow markers instead of a pen for proposed routes and it was suggested this might be because they used the annotation method that contrasted most with the map and would be the least ambiguous for the recipient. A pen tool that produced dotted or dashed lines (or some combination thereof) might not match the real-world metaphor of a pen as well as solid-line pens, but would allow for more contrasting pen marks. Alternatively, a tool allowing a series of connected straight lines (with the dotted or dashed pattern) between points could be implemented and this would be consistent with lines and boundaries used on many maps and potentially be a more accurate representation of proposed routes.

• Other uses of the pen

The pen was sometimes used to produce approximations on horizontal lines and rectangles in both the scatter charts and tables, and on two occasions it was used to draw a longer

arrow than a arrow marker. This suggests annotation tools specifically for those types of

annotations would be used if implemented.

CAANT (Faira)	
POE T	
The Edit View Co. Addim: Writem Edit 11	
S Flight Instance(15159099) Shoute Force	
	12.22
Addine: NVA	
Department DTW A	5.5
Research Ind 21 200	
Follomance Metrice: Routing Information	<u> </u>
Participation Matters Plant Actual Actual Actual Actual Actual Actual	
Ar FueBurn Uncorr.(bs) 11.900.0 19.680.0 7,780.0 55.42 2	
Air FuelBurn Corr.(bs) 11.900.0 19.690.0 7.790.0 65.4%	
Ail Time Continuinsi 70.0 126.0 56.0 80.02	
Taxi Infimina 7.0 17.0 10.0 142.92	
Las Uutimmis 22.0 33.0 11.0 90.02 20 43.0	
0// Time(2) 2122 2312 1100 Subjects	
On Time(Z) 2232 0118 166.0	
	T
	5502 Till
A Test loans and an	
The data be off	

• Arrow markers in multiple directions

Figure 7.23 Example of the need for more sophisticated arrow annotation tools

Arrow markers were frequently used, but a number of participants wanted to place them in a direction other than the one direction supported. Providing either 4 or 8 directions would enable arrow markers to be more easily placed in positions to avoid overlapping data or other arrow markers. Figure 7.23 demonstrates a situation where a dispatcher wanted to place an arrow pointing to the left. Providing a flexible length arrow marker, similar to those used in the other programs with annotation features, would also be useful.

Improving single slide operations

There are issues regarding the design of C-SLANT related to events that occur at the individual slide level and others that exist when there is more than one slide. At the individual slide level the annotation options have already been discussed, but there are also design issues related to the management of the underlying image itself (some of which were not captured in the experiment because participants were given a slide show already created).

Problem	Redesign Idea
Users sometimes thought the screen captured	Use a mask over the background image or use
image was an active window	distinctive skins for C-SLANT's windows
C-SLANT does not handle images of a different size than the screen very well	Support zooming, and automated scrolling, with multiple views and support these actions as part of a recording
C-SLANT doesn't allow multiple images to be pasted on a single slide	Include a simple Paintbrush-like interface to combine images

Table 7.47 Redesign ideas for single slide operations

• Contrasting more with the background

To help the user distinguish between the real controls of C-SLANT and those that might just be part of the image from a screen capture, the latter could be made slightly less "real" by masking it to, for instance, make its color lighter. Alternatively, the C-SLANT controls could be made more distinct by giving them a "skin" different from standard Windows controls.

• Supporting small and large images

The version of C-SLANT used in this experiment did not support any zoom in or out operations. The current version supports this to a limited extent, with pull down menu options for percentage zoom levels, but not by zooming at an x-y position, selected with the mouse,

which would be preferable. When working with large image maps, in particular, it would be easier with a zoom feature to provide a micro and macro view. Voice and pointing annotation while zooming in and out is analogous to annotating over video, which would present visibility challenges in that video stream, and not just the audio as is currently the case. Similar challenges would exist if attempting to support annotation while scrolling.

• Enhancing background image creation

When screen captures are taken the resultant image is the size of the current resolution (e.g. 800x600 or 1024x768), if a full screen capture is taken, or the size of a window, if only an individual window's image is captured. The ability to paste more than one image on a slide would potentially allow more information to be viewed at one time and thus reduce navigation and memory requirements. An interface similar to Microsoft Paintbrush's for this feature would be simple and consistent with that program that is a standard application in the window environment.

Improving navigation across slides

Problem	Redesign Idea
Recipients of a message started responding before viewing/listening to all of it	Provide a better overview of the slideshow structure with thumbnails and the recordings window. Warn the user if necessary
Recording was not allowed over multiple slides	Allow recordings over multiple slides, but show them as two linked entries in the recordings window

Table 7.48 Redesign ideas for slide navigation

• Providing a better overview of the slideshow and warning if "interrupting"

Responding to an original message (spread over multiple slides) after the first slide suggests the person either forgot there were more slides, or didn't realize that his or her response may no longer be as appropriate after viewing or playing the recordings of the other slides. Putting the current slide more clearly in perspective of the slide show should help with this problem. This could be achieved by using thumbnails for each slide, as is done in programs like Microsoft PowerPoint. The recordings window also makes it much clearer that there are additional slides and if the user has the automatically play new recording option set, he or she can hear all the recordings at one time. If the user still creates a response without viewing or listening to all the originator of the message sent, a warning message could also be issued, but that should be a configuration option as some users may not consider it helpful. The reason being that there is a tradeoff – the more the user listens to before responding the less likely there will be something later causing the response to be less appropriate, but there is also more to remember from what is being listened to in one session. Allowing memory aiding markers to be placed on the players at significant points and adding more structure to the messages themselves are two approaches already discussed that might help address this problem.

Improving message management

Problem	Redesign Idea
There is no C-SLANT specific message management system	Add features appropriate for slide show manipulation and message management
There is no method for searching slideshows	Support search of text based message components such as text boxes and message attribute fields

Table 7.49 Resign ideas for message management

• A C-SLANT message management system

The C-SLANT program opens and saves individual slide shows, but there are no options to manage multiple slide shows at one time. As individual files, the slide shows can be manipulated like any other files in the Microsoft Windows operating system and can be sent as an attachment to an E-mail message or put on a file server for transmission, but additional functionality could be implemented considering the specific properties of these files. For instance, with multiple slide show views, slides could be copied from one slide show into another. If the management system included mail or file transfer protocols, message exchanges for a particular slide show could be tracked specifically. In a client server, rather than a peer-to-peer communications architecture, a slide show could be a "topic" in a pseudo newsgroup environment, where multiple participants could contribute to the discussion. In such an environment the recordings window could be managed by the server to coordinate the exchange of comments.

• Searching slide shows

As part of a C-SLANT slide show management system, searching through multiple slide shows could be supported. This could involve searching text box content, but if the structured messaging fields were part of a slide show, as discussed earlier, those fields could also specifically be searched. Thus, messages discussing particular city pairs, performance concerns, or performance improving strategies could all be searched for.

235

Improving storage and processing time requirements

Problem	Redesign Idea
The system was sometimes slow	Provide more compression options
Transmission time can be long	Stream the audio or only transmit changes to a slide show
The system is platform dependent	Use Java and/or a Web based implementation

Table 7.50 Redesign ideas for low-level computer efficiency

• Improving system response time

C-SLANT files are compressed to reduce the size of the slideshow that is sent and to reduce the amount of disk space needed. However, the compression and decompression processes take time, and on slower machines this can be noticeable. To provide the most flexibility the current version of C-SLANT allows several configuration options. Slides can (1) be compressed and decompressed automatically as the user moves between slides; (2) be compressed only at the slideshow level; or (3) not be compressed at all. As machines get faster and storage larger, compression for transmission becomes the most important speed related issue.

• Improving transmission time

C-SLANT files transmitted by E-mail are subject to two potentially significant delays: (1) the delay due to the queuing of the E-mail system itself, and (2) the delay due to the bandwidth of the communication's connection relative to the size of the C-SLANT message. The first problem could be addressed by using a more direct connection. The second problem could be reduced by only sending changes to a message or by streaming message components. Solutions to these problems are discussed in more detail below:

1. Immediate file transfers

A peer-to-peer environment could be created, if for instance two parties both ran FTP (File Transfer Protocol) client and server software. The advantage in doing that would be the message could be sent more directly between the machines and therefore normally more quickly, but the disadvantage would be the lack of the inherent queuing capability of E-mail systems, (which means both parties don't need to have a connection established before the communication can begin). If a peer-to-peer environment were highly desirable and queuing were not required, FTP client and server functionality could also be built into C-SLANT, which should make the process less cumbersome for the user than using separate FTP applications.

Another way to speed up the file transfer would be to use a messenger program like Microsoft's MSN Messenger or Yahoo Messenger to transfer the file through that system. This approach might also facilitate switching between asynchronous and synchronous communication modes when both parties agree.

2. A better E-mail based system

Assuming the E-mail systems used provide a sufficiently responsive service, but the time to send a C-SLANT message is more a problem because of the file size itself, C-SLANT could be modified to not send message components to a recipient who should already have them. For instance, if person A creates a message and sends it to person B, and the message components, such as a voice and pointing segment or an image are given unique machine: component identification tags, the system should realize there is no need for B to send A's original content back to A.

237

3. A Web based version of C-SLANT

A Web based version of C-SLANT would have a number of advantages over the current version:

- Messages could be managed on the web server and potentially be made accessible to anyone on the Internet. This might be particularly appropriate when messages could usefully be shared by groups of people with a common interest. Thus, for instance, there could be a Web page for each of the FAA Enroute Centers;
 - The audio portion of the recorded voice and pointing could be streamed, reducing the transfer time;
 - If the underlying format for a slide was changed to SMIL (Synchronized Multimedia Integration Language¹⁷), it would be playable by any SMIL player, although for creating new slides and for providing the full set of C-SLANT functions an SMIL based version of C-SLANT would be necessary.



Figure 7.24 C-SLANT asynchronous communications via the Web

¹⁷ Defined at: http://www.w3.org/TR/REC-smil/

The basic architecture for a Web based implementation of C-SLANT is shown in Figure 7.24. In this particular scenario a separate server is dedicated to providing Real Time Streaming Protocol (RTSP) service, for efficiently streaming audio, and an HTTP (Hypertext Transfer Protocol) service is provided separately for all other objects on a Web page.

• Developing a platform independent system

C-SLANT will only run on Microsoft Window's machines, which excludes potential users who have, for instance, Unix or Macintosh computers. A Java based version of C-SLANT would not have this problem, because its code would be compiled to produce Java Virtual Machine code, which would then be translated to the machine language of a particular computer by a platform specific translator. This is shown in Figure 7.25.



Figure 7.25 Java based applications (adapted from Slack, 2000)

CHAPTER 8

SUMMARY AND CONCLUSIONS

Introduction

In this chapter the most significant research findings are summarized and discussed. The context of the research is also noted and used in combination with the results to consider the application of the findings presented.

Summary of results

The most significant results are organized into two sections: (1) the problem solving summary focuses on the most tangible results for the overall collaborative task of identifying applicable strategies to improve performance; and (2) the context referencing summary focuses on results related to the fundamental design goal of supporting asynchronous communications rich in context, based on the theory that doing so would lead to more effective communications.

Problem solving summary

• The communication mode failed to demonstrate a significant difference in the number of ideas for improving performance generated by dispatchers

It was decided a priori that the number of dispatcher questions asked would be a useful measure in the analysis of the collaborative problem solving because, in general, it was assumed more questions would lead to more feedback for the issues raised by dispatchers. In

analyzing the comments made by dispatchers it was determined there were 42 questions asked by participants working in the text mode and 11 in the voice and pointing mode. Further analysis revealed that the majority of these questions could be classified as semantically equivalent to one of three question types: *Why is X happening to our flight(s)? Was the problem caused by Y?* and *Would doing Z help?* The 42 questions asked by dispatchers in the text mode were divided into each category and the count was 10, 12 and 16 respectively, with 4 questions that could not be classified as one of these three types. Similarly, the voice and pointing mode produced 5, 2, and 4 questions of each type respectively. Particular attention was given to the third type of question and while the number of *Would doing Z help?* questions was greater in the text mode, when the number of performance improving ideas was counted for dispatchers in the form of a question and not in the form of a question (typically stated as a hypothesis such as "A different preferred route in bad weather might help"), 27 were found in the text mode and 29 in the voice and pointing mode as was shown in Table 7.6. (A list of these ideas are shown in Appendix F.). Thus from a pragmatic perspective there was no significant difference detected if they were responded to similarly. Further analysis revealed that normally that was the case.

Problem	Text Mode	Voice and Pointing Mode
ZNY	4, 2	3, 1
ZOB	2, 3	3,1
ZID	0, 2	0, 1
ZTL	2, 1	2,7
ZFW	1, 1	2, 2
ZBW	1, 0	1, 1
ZDC	1, 2	0, 2
ZMP-1	2, 2	1, 1
ZMP-2	1,0	1,0
Total	27	29

Table 7.6 Number of dispatcher ideas for performance improvement

	Direct questions	Other form
Text mode	16/16	11/11
Voice & pointing	4/4	23/25

• Almost all the dispatchers' ideas were responded to by traffic managers in both modes

Table 7.15 Count of the performance improving ideas responded to by the traffic manager

As Table 7.15 showed, all the dispatchers' ideas presented in the form of a direct question were responded to by the corresponding traffic managers. Almost all the ideas not presented in the form of a question were also responded to: 11/11 for the text mode dispatchers and 21/23 for the voice and pointing mode dispatchers. The two ideas that were not responded to were "A delay program could have helped to improve spacing" (in the ZTL scenario) and "Another solution might be to fly at different altitudes" (in the ZOB scenario). The difference between modes for the number of ideas in a non-question form is not statistically significant, but the fact that 2 out of 15 dispatchers who raised issues regarding performance improvement did not receive a response to all of them may be enough evidence to warrant attention to this potential problem when considering environments were greater memory or attention demands are involved for the recipient of a voice and pointing based message.

•	The mode did	not impact l	the performance	improving	strategies p	presented
---	--------------	--------------	-----------------	-----------	--------------	-----------

Category	Dispatchers		Traf Mana	Traffic Managers	
	Т	VP	Т	VP	
A. New preflight route	11	9	9	10	
B. Enroute reroute	2	3	2	0	
C. Change schedule	1	1	7	3	
D. Ground delay/stop	3	4	1	1	
E. Same route but miles in trail	0	1	0	0	
F. Change Center staffs' workload / traffic volume	2	0	0	0	
G. Share information for planning (fuel burn, filed route, etc.)	3	3	0	1	
H. Hold at a different location	1	1	0	1	
I. Change altitude	2	2	0	0	
J. Fix balancing	0	1	0	0	
K. Other	0	1	0	0	
Totals:	25	26	19	16	

Table 7.16 Classification of dispatcher and traffic manager ideas regarding performance

improvement

As Table 7.16 showed, the type of idea the dispatchers had was also found to not be dependent on the mode, and in both cases changing the route was produced as an idea more often than other ideas such as changing the schedule, or the altitude for some segment of the route.

Scenario	Text	Mode	Voice and P	ointing Mode	
ZNY	aA, rB, rG, rF, nB	aA, aB	rA, rB, rD	rH	
ZOB	rD, rH, n(ref)H	a2A, rI, nA	a2A, mI, nC	aA	
ZID	(Leave as is)	ref(G), (ref)I	n(ref)A	rA(Leave as is)	
ZTL	rD, rG, nA, nC	rA	rG, rK, nA	rG, aA, rJ, aA, aC, mD	
ZFW	aA	aA, nC	rA, aA	rB, aA, n(ref)G	
ZBW	rD	nC	rA(Leave as is)	rD	
ZDC	aA, nC	aA, rA, n(ref)A	nA	rE, rA, n2A	
ZMP-1	aC, rA, nA	rA, rF	aD, n2A, nC	rB	
ZMP-2	aA, nC	nC, nD	rG	(none)	
Totals:	a = 11 ; r = 14	; $n = 11$; $m = 0$	a = 9; r = 18; n = 10; m = 2		

Table 7.17 Classification of communications outcome with respect to ideas for improving performance

(a: accepted; r: rejected; n: new idea; m: no response. (ref): referred to another center or military airspace manager. Classification of ideas A-K shown in Figure 7.16)

Table 7.17 summarized the traffic managers responses relative to the type of strategies the dispatchers proposed. The table reveals no detectable differences between the distribution across the ideas rejected, accepted, added and missed by the traffic managers (i.e. 11, 14, 11, and 0 versus 9, 18, 10, and 2, respectively). • Dispatchers took longer to create a message in the text mode than the voice and pointing mode



Figure 8.1 Main effect of the mode on dispatcher time (adapted from Figure 7.4)

Figure 8.1 shows the main effect of the mode for dispatchers task time. The average task times were 18 minutes 1 second (1081 seconds) for the voice and pointing mode and 28 minutes (1680 seconds) for the text mode. This is a difference of 9 minutes 59 seconds or 55%. Because it was unusual for dispatchers to playback their own recordings or rerecord them much of this time difference can be accounted for by the difference between typing and modifying comments versus making a voice and pointing recording.

• The combined task time was longer for participants working in the text mode than the voice and pointing mode



Figure 8.2 Main effect of the mode on dispatcher+traffic manager time (adapted from Figure 7.11)

Even though voice and pointing mode traffic managers had to listen to the dispatchers comments, rather than read them, and listening is generally performed slower than reading, Figure 8.2 shows the main effect for the dispatcher + traffic manager task time increased from the dispatcher's time shown in Figure 8.1 in absolute terms. The average task times were 35 minutes and 41 seconds (2141 seconds) for the voice and pointing mode and 53 minutes and 22 seconds (3202 seconds) for the text mode. This is a difference of 17 minutes 41 seconds or 50%.

Thus, the task time was less in the voice and pointing mode, even though the performance in terms of generating applicable strategies for improving performance was very similar. This is consistent with the study by Chapanis (1975), Neuwirth (1994), and Daly-Jones et al., (1997) who all found efficiency benefits from voice or voice and pointing over text based communications.

Context referencing summary

The design of C-SLANT included three features primarily intended to help provide context to the comments message creators would make: (1) the context framing images forming the slide shows; (2) the voice and pointing recorder which can capture context for the comments by connecting them dynamically to individual locations or sequences of locations in the image; and (3) the static annotations using the arrow marker and pen to help focus attention as the comments are interpreted. Analysis of the messages created revealed two other important dimensions of context: (1) the context created through the comments made themselves; and (2) the temporal context as comments placed on the slides in a particular order during the dispatchers' problem solving / message creation process and, comments were made by traffic managers in a particular order during their interpretation of dispatcher comments / problem solving / message creation process.

• Dispatchers normally did not introduce the flights with some basic classifying data defining the situation

Category	Dispatcher Text Mode	Dispatcher Voice & Pointing Mode		
City pair	5/18	11/18		
Time of day	1/18	4/18		
Time of year	0/18	5/18		
Number of flights involved	1/18	4/18		

Table 7.4 Number of dispatchers mentioning certain flight classifying data

Some basic data defining the situation (the city pair; time of day; time of year; and number of flights involved) was not normally introduced by dispatchers as was shown in Table 7.4. This suggests that in most cases the combination of the information on the slides and their mental model of how the traffic managers would proceed through the task caused them to predict the traffic managers would make note of the information without any verbal prompts from the dispatcher. However, there were instances were not noticing this information caused the participants to indicate it was difficult to determine what was happening to the flights under consideration, suggesting its salience is an important issue. There was also a difference between the two modes in this regard, with more dispatchers mentioning each of these categories in the voice and pointing mode (although only the difference between the number mentioning the city pair and time of year were statistically significant at $\alpha < 0.05$). The reason for this difference could simply be the relative ease with which comments can be made with speech or possibly that a different schema is activated and more of a "story-telling" process is used which begins by "setting the scene".

Almost all voice and pointing mode participants made deictic references while speaking

	Dispatchers	Traffic managers
Participants making deictic gestures	18/18	17/18

Table 7.19 Count of voice and pointing mode participants making deictic gestures

As Table 7.19 showed, all 18 dispatchers working in the voice and pointing mode made deictic gestures, and 17 of the 18 traffic managers working in the voice and pointing mode made deictic gestures. These participants didn't point continuously, but the locations of various objects verbally described were made explicit with synchronized pointing. This reduced ambiguity in some cases because of ambiguity in the language (such as "a northern route") and in others because the recipient would not have the knowledge to know where a named location is (e.g. holding at Scupp). This synchronized pointing also helped identify verbal slips and errors. In some cases participants created negative effects in this mode, by not keeping the pointer motionless when speaking about topics that were not related directly to information on the slides, or by producing noticeably slowed speech as they used other annotation tools while speaking.

(although this should be qualified with the fact they were novice users of C-SLANT). Thus the value of pointing is certainly a function of (1) the comments made; (2) the synchronization; and (3) the knowledge of the recipient.

Seconaria	Disp	atcher	Traffic manager		
Scenario	Text V&P		Text	V & P	
ZNY	5, 21	0,6	0, 10	0,0	
ZOB	5, 12	0,5	0,0	0, 19	
ZID	3, 5	0, 11	0,0	3,1	
ZTL	2, 14	30, 12	0,1	4, 17	
ZFW	0, 1	17, 1	0,0	0, 10	
ZBW	9, 33	3, 2	0,0	0,0	
ZDC	2,3	7,7	2,0	0, 31	
ZMP1	6, 17	15, 1	3,0	10, 0	
ZMP2	9, 14	7,14	0, 1	8,0	
Average	8.94	7.67	0.94	5.72	

• Participants working in the voice and pointing mode still created static annotations

Table 7.25 Number of pen or arrow annotations per slide show

As Table 7.25 showed, 17 dispatchers working in the text mode and 15 dispatchers working in the voice and pointing mode created pen or arrow marks. The difference between traffic managers was greater, with 5 traffic managers working in the text mode creating pen or arrow marks and 9 traffic managers working in the voice and pointing mode creating pen or arrow marks . This suggests the voice and pointing mode participants felt it valuable to leave a static annotation on the slide and they found it valuable to capture the pointing/drawing dynamics as these static annotations were created (although these would be limited because the static annotations are not recreated dynamically during playback – just the movement of the pointer is), or they forgot how limited the capturing of the dynamics was.

• Voice and pointing mode participants created static annotations while speaking

	Dispatchers	Traffic Managers	Dispatchers and Traffic Managers
Voice and pointing mode	9/15	9/9	6/9

Table 7.24 Participants that made arrow or pen mark annotations while speaking

At Table 7.24 showed, 9 of the 15 dispatchers who created pen or arrow marks created them while recording their speech and all 9 traffic managers who created pen or arrow marks created them while recording their speech. This suggests they tended to feel the simultaneous tasks manageable and the results acceptable for the recipient (because both the effect of annotating while speaking and not were included in the training, and none changed to not performing the tasks simultaneously after starting simultaneously).

• The slide image provided a context for comments, but the slide show structure created context problems

Two types of problem were caused by the slide show structure used in C-SLANT: (1) temporal context problems; (2) multiple-slide perspective problems.

A loss of temporal context occurred either because the participant created comments out of order (relative to the slides) or because, in the case of the traffic managers, responses were created on one slide before reading the comments made by the dispatcher on a later slide. This sometimes created inefficient commenting as the traffic manager repeated or corrected previous comments when reaching the later comments. To avoid this situation the respondent needs more encouragement to parse the slide in two stages: a high-level parse first to learn the main points made, followed by a low-level one to work with the details.

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

Participants sometimes refer to multiple slides at once with a comment when, for instance, addressing a common issue or performing juxtaposition, but because of the single slide perspective of the versions of C-SLANT used in the experiment these comments were only made on the slide in a one-to-one relationship and not the one-to-many relationship that was needed. Thus, the limited 'parcel post model' (Tarter et al., 1991), mentioned in Chapter 2, was actually implemented again here.

• . • • •	: · · ·	X
Play 📭	Stop	Rewind Respond Delete Omns O secs
Auto	matically	play next recording Length : 0 mins 45 secs
Author	Slid	S Date Subject
Roger	1	06/18/00 18:25 High fuelburns
Phil	ં 'ના ં	06/18/00 18:37 Note military airspace
Roger	2	06/18/00 18:26 Impacted on times
Roger	3	06/18/00 18:27 Other traffic flows
Phil		06/18/00 18:40 Center interactions
		 Methods and the descent states of the second states of the

Figure 7.20 C-SLANT's recordings window

Perhaps the most significant redesign of C-SLANT that occurred as this research was conducted was the creation of a "recordings window" as was shown in Figure 7.20. This feature was actually implemented and provides an overarching structure for showing where the recordings took place and what the relationship is between recordings. Further modification of the implementation could support recording across multiple slides to obtain the one-to-many relationships needed and also support a textbox as a format that could be used as an alternative to a recording in the table of linked comments.

• The comments made provide a vocabulary of syntax and semantics for a more task specific version of the system

The use of POET images and C-SLANT combine a context specific application that represents information with graphics and tables with an image based asynchronous

communications tool. C-SLANT is not an aviation specific application although its uses with POET was the priority as it was developed. As demonstrated in Table 7.29, the data collected here revealed a vocabulary of syntax and semantics that could be used to help design more task specific annotation tools for C-SLANT and/or which could be used to modify the interface of POET. If changes are made in C-SLANT to achieve this, they could be implemented using a template/configuration file to maintain the generic integrity of the system. The structured interview responses made by the ZNY TMO who used C-SLANT for several months after the study to evaluate it demonstrated that value could be seen by using it in combination with POET, but also in combination with other applications such as the CCFP (Collaborative Convective Forecast Product). In that case a different set of annotations could be developed after analysis of communications using a generic set first as was done in the research reported here.

Conclusions

In this section the research context for the results reported here is described and the application of these results considered.

Research context summary

While considering the results presented here it is important that the context of this research be noted to provide some control and guidance for extrapolation, given the design choices that were made. The participants used a particular software application and dataset and although the study was relatively authentic, within practical constraints and the goal of having control of certain parameters for experimental design reasons, it was not a naturalistic study.

Task	Two person collaborative problem solving to identify strategies
	for improving performance
Data format	Maps, tables, and statistical graphs detailing past performance
Communications format	Images annotated with pen marks, and arrow marks, plus voice
	and pointing or text for comments
Navigation structure	Slide show based
Communications time-space	Different time (asynchronous) with 2 iterations, different place
Poor performance indicators	Tabular data values, patterns in analog images, and patterns in statistical data
Knowledge distribution	Heterogeneous and homogeneous
Authority distribution	Heterogeneous

Table 8.1 Research context summary

As stated in Table 8.1, the task was a two person collaborative problem solving task. As the comments made indicated another Center was sometimes referred to by traffic managers as being appropriate for the dispatcher to contact for further discussion of an idea for improving performance. There might also be situations where the next stage for the pair would be to contact representatives from all airlines involved in a traffic problem or several neighboring Centers to check the feasibility of an idea. Such an idea might be, for instance, changing a filed route to one that passed through a different set of Centers or changing the schedule of flights during a rush period. Including more people in the same context would mean the identification of each person's comments and annotations would be more important, as would the need for clear temporal frames of reference. Content summaries would also be more valuable to allow scanning to refresh memory and preview topics. The redesign ideas presented in the last chapter address these issues. The fact there were 2 "rounds" of message sending also should be noted. Supporting more iterations is somewhat similar to supporting more people, but the temporal context becomes particularly important. The number of iterations that would occur in reality would be impacted by the fact that the collaborative problem solving doesn't necessarily have to occur with C-SLANT as the only communications tool. Variations on this might be, for instance, to have one or more exchanges via C-SLANT followed by a telephone call or other synchronous

communication to complete the process. Another approach might be in initial synchronous communication followed by asynchronous communications as communicators follow up by gathering information and discussing it. The goal would be to support both types of communication well and also the transitions between them.

The task of identifying strategies for improving performance in this case involved participants predicting what would happen to the NAS by changing some actions one or more agents performs, based on the data displayed and their own knowledge. In a more natural environment they would have supplemented this information with images or comments representing knowledge they acquired during the problem solving from other sources. They might even use some other planning application to make predictions and include the results in their responses. This would make the time slides were added and removed important information.

In terms of the results presented here it may mean that a significant change in the image type means some of the results are less applicable. Largely text-based data, versus largely map based data would change the relationship between the comments made and the image in that more accurate reference to geographical locations, aviation specific events, and sequences of locations forming a flight path, would not be the primary benefit from pointing, but literal text and interpretations of text would be. However, there are still common "primitive" costs and benefits of voice and pointing based comments, such as controlling attention to specific positions as comments are made and the fact the comments themselves are "invisible" unless repeated or summarized in a text form.

The distribution of authority/control is heterogeneous in that they each have control over different agents and processes in the system. However, when speaking of possible changes to the route filed or flown the traffic manager can speak to what is realistically likely to be approved and so it is perhaps not surprising most of the questions were asked by dispatchers and the traffic manager tended to be giving "advice" for what was possible and not possible. A different

254

relationship between participants, for instance, two traffic managers at neighboring Centers would perhaps involve a more balanced number of questions, but there is no obvious reason to think this change would have no impact on the difference between the voice and pointing and text modes.

The distribution of knowledge is shown as both homogeneous and heterogeneous in Table 8.1. These groups of participants were both professionals in the aviation industry with a similar distribution of years experience as shown in the questionnaire results, and both groups work with flights to coordinate/manage them and therefore there is a basic knowledge they will have in common (e.g. the names of fixes, the general areas of very high traffic density, such as the New York area and the potential impact of weather on individual flights). However, their perspective and experience is also different with the dispatcher having less knowledge of the cumulative effect from all airlines sending traffic through a particular Center as scheduled and the events that trigger certain actions by the traffic manager, or which are automated in the system through technologies, and the traffic manager does not fully understand the reasoning the dispatcher uses when flight planning. These knowledge similarities and differences affects the language that can be used and understood without explanation, and also the value of pointing to connect comments to locations if the listener can't do that without that information being made explicit. The knowledge gap affects the value of pointing and it seems reasonable to conclude the perceived knowledge gap impacts the language used and the pointing. In a synchronized communications situation the feedback for comments made would occur more quickly, so this perception could more easily be modified as the communication continues in a process to achieve common ground (Clark and Brennan, 1991).

255

Application of this research

A number of factors contribute to the successful introduction and continued use of a system as described, for instance, by Neilson (1993) and summarized in Figure 8.3. In this case the contribution of certain factors was measured for two versions of the system to learn more about their effect in each mode for this system and more generally for systems of this kind. The majority of the data collected in this research was performance data, but a questionnaire was also used to gather information to supplement this data.



Figure 8.3 Factors influencing system acceptability (Neilson, 1993)

Table 7.37 and Table 7.38 summarized the participants ratings regarding usability and utility of the version of the system they used and in general the ratings were positive for both modes by all participants.

Question: On the following scale, pl	ease indicate how eas	y/difficult you found this software
to use?		-

Results	Very easy	Easy	Neutral	Difficult	Very difficult
Text mode dispatchers	8	7	2	0	0
Voice & pointing dispatchers	2	11	4	0	0
Text mode traffic managers	8	8	1	0	0
Voice & pointing traffic managers	12	6	0	0	0

Table 7.37 Perceived usability of C-SLANT

Question: On the following scale, please indicate how useful you think this kind of software might be in your job?

Results	Not useful at all	Not very useful	Neutral	Fairly useful	Very useful
Text mode dispatchers	0	0	2	9	6
Voice & pointing dispatchers	0	3	3	9	2
Text mode traffic managers	0	1	4	8	4
Voice & pointing traffic managers	0	0	2	4	12

Table 7.38 Perceived utility of C-SLANT

Social acceptability might be an issue for the voice and pointing mode in some work situations where the users could not record or listen to the recordings because of auditory demands in that workplace. In this case audio communications, and further the recording of audio communications, is socially commonplace. The utility of the system in each mode was ultimately measured by the number of performance improving ideas that were generated and evaluated. The performance was the same in each mode in this regard. It took approximately the same amount of time to train the participants how to interpret the POET screen captures and use each version of C-SLANT and users in both modes asked very few questions, if any, with regard to the operation of the system. Some users did not keep the pointer motionless while speaking about something not on a slide and users often left windows covering data. These "habits" could be reduced or removed with experience using the system or formal training. In addition the redesign ideas presented in the last chapter may reduce or eliminate these problems. The quality of the comments in terms of their efficiency was not evaluated in detail, but there were incidents of redundancy and inconsistencies across slides, which was due to the largely single-slide perspective given to users and the sequential processing of data performed by users. Some redesign ideas to address this problem were also presented in the previous chapter. The problem of not being able to scan audio comments at a semantic level and edit audio comments easily also reduced the quality of some of the comments made as participants sometimes created redundant or inconsistent comments within the same slide. It is difficult to determine when the participants were aware of these problems, and when they were it may be they decided the points were still made sufficiently clearly and that this was more analogous to a relatively casual recording on an answering machine than a formal presentation of ideas that may ultimately be distributed widely. Even so, redesign ideas were presented in the last chapter to address the problem of making it easier to edit recordings and to make the semantics more visible. The efficiency was measured carefully in terms of the time taken to complete the task and the task time was shorter for the voice and pointing mode by 55% for dispatchers and 50% for dispatcher-traffic manager pairs.

What is an error is open to interpretation, but as has been discussed there were incidents of verbal slips during deictic gesturing, unnecessary covering of important data with windows before sending a message, misreading and non-reading of relevant POET data, incorrect reasoning that weather data was necessary to make deductions, and redundancy or inconsistency in the comments made. These errors provided feedback for redesigning C-SLANT or POET, and for training of participants who might be involved in communications like this. A number of redesign ideas were presented in the last chapter which assumed both the voice and pointing mode and text mode should be supported. The question may be asked, however, as to whether it is necessary to support the voice and pointing mode given that the same number and type of ideas were generated in each mode by dispatchers and the responses by traffic managers were similar in terms of the number of ideas responded to and added by traffic managers. As shown in Table 7.37 and Table 7.38 the "average" usability and utility ratings were higher by dispatchers

258
working with the scenarios in the text mode than the voice and pointing mode. Further, the storage requirements are much higher for the voice and pointing mode, and scanning and editing or audio comments is more difficult. However, the results here show that the task time was shorter for participants working in the voice and pointing mode and the "average" usability and utility ratings were higher by traffic managers working in the voice and pointing mode. While redesign of the system to add more structure and visual summaries might eliminate the difference, using the version described here, dispatchers introduced more basic flight summary data on the first slide in the voice and pointing mode.

Further, other research reported in Chapter 3 provides evidence of advantages of voice or voice and pointing based communications over text-based communications. Chapanis (1975) conducted a much-referenced early study involving human cooperative problem solving and found that speech based communications involved many more words per minute than text based and that tasks were solved faster when speech was used. Faraday and Sutcliffe (1997) investigated attention and comprehension by users interacting with multimedia, and from this study a set of guidelines for controlling attention in multimedia presentations was produced, including: (1) the guideline to use object motion to control attention and viewing order, because participants' attention was drawn to motion and fixations tracked the moving objects path; and (2) the guideline that speech information should reinforce image, because propositions given only in the image or animation without speech cueing were poorly recalled. In this study they used a system that had popup annotations to link text to animation in the primary image, but this was created with a traditional authoring/programming system rather than a communications system, and that environment would be too complicated for the average communicator. Neuwirth et al. (1994) compared the nature and quantity of voice and written comments produced in each mode, and when reviewers gave feedback to writers they found: (1) reviewers used more words in voice than text mode during the same time period, but that the same number of annotations were made on average. The additional words were attributed in part to providing more reasons why the

reviewers thought something was a problem and for polite language that mitigated the problem; and (2) evaluations of reviewers were less positive when reviewers produced written annotations than spoken. Daly-Jones et al. (1997) conducted a study where 'manager-secretary' pairs were asked to complete an asynchronous appointment-scheduling task and an equipment-booking task in three conditions : Fax-only involved using Microsoft Paintbrush ; voicefax involved using Lotus Screencam (an application that continuously records what is displayed on the user's monitor, including the mouse pointer, while synchronizing it to any speech input to create a 'movie' that includes any voice and pointing over the potentially dynamic images) with a Paintbrush image; and voice-only that involved just audio. For both sending and receiving, voicefax was rated most useful, then fax-only, then voice-only. Subject took the same amount of time to complete the tasks in each condition, but fewer messages were sent with voicefax.

To estimate how often the modes would be used when users are given a choice, eight dispatchers not involved in the main experiment described here were given slide shows to annotate using C-SLANT with no modes disabled. Four chose to use voice and pointing alone, three chose to use text alone, and one used both modes. This in combination with the results reported by Bargeron et al., (1999) and discussed in Chapter 3, who found that participants were as likely to use text to reply to audio annotations, as they were to use audio, supports the redesign idea for C-SLANT of not only allowing either text or voice and pointing mode comments to be made, but to allow one mode to be used as a response for the other.

Hardware and compression algorithm developments for computers and communications make voice and pointing based communications more practical today than it has ever been. However, while it is valuable to study differences between voice and pointing and text-based modes to understand them better, and support them effectively, it is not necessary to choose between voice and pointing and text when developing a system if they complement each other. In the context studied here it was determined there were some advantages with voice and pointing, but there is evidence some users would prefer to use a text-based mode. There were

incidents of some undesirable pointing, and there was also some inefficiency due to the lack of visibility with the voice and pointing mode. One redesign idea presented to improve pointing was to support more recording and annotation options through mouse clicks. This however may be complex for the novice user. Another redesign idea was to use "text" to complement the voice and pointing, by supporting text labels and summaries to make critical information visible on individual slides and message structuring information visible across slides. Text comments can still be supported and included in the recordings window as an alternative to voice and pointing, and the corresponding text message would simply appear when the "play/display" button is pressed.

Supporting both modes means the user has the option to be sensitive to storage requirements, consider if audio is practical in the working environment, and depending upon how much pointing is involved, select that feature or not. Even if voice and pointing is mostly used, it may be beneficial to at least type important details that might be misheard, such as a series of fix names to describe a route, or numbers that require a lot of words when spoken. Certainly training could help provide guidance as to what the advantages and disadvantage there are for each mode and the kinds of errors that can occur, and the research presented here could be used to design such training. Users would also likely adapt with experience as they receive both positive and negative feedback, but training could ensure coverage of the important issues without relying on them occurring naturally and with cost when the feedback is negative.

The slide show structure used in C-SLANT has the advantage that it allows multiple context framing images to be structured in a conceptually simple aggregation, but navigation issues demonstrated here suggest it is important to provide an overarching structure that shows how comments have been placed across the slides with a clear temporal frame of reference. This would appear to be an important consideration for all systems involving slide shows, or for that matter hyperlink structures, as the underlying message structure for asynchronous communications. Freestyle (Francik and Akagi, 1989; Francik et al., 1991; Francik 1996) developed in the late 1980's was perhaps the "grandfather" of deictic gesturing communications systems. It was innovative at that time with direct manipulation (Zielgler and Fahnrick, 1988; Shneiderman, 1998) combined with the direct referencing of deictic gesturing, but it did not survive to become part of today's popular software applications. The reasons are ultimately that the system was not "acceptable" at that time, in the targeted workplaces. Research on asynchronous deictic gesturing systems is very limited and such systems are also rarely used. For such systems to emerge and work well in the future it is essential to learn more about designing interfaces for these systems and to do so considering the context carefully. This research was intended to contribute to the specific goals in this domain, but also to contribute to the general application of systems such as C-SLANT.

LIST OF REFERENCES

- Aviation Weather Center. (2001). Collaborative Convective Forecast Product [Online]. Available at: <u>http://www.awc-kc.noaa.gov/ccfp/</u> [2001, August 1].
- Baeker, R. (1993). Readings in groupware and computer supported cooperative work. San Mateo, CA: Morgan Kaufmann.
- Balzer, R. (1989). Process programming: Passing into a new phase. Proceedings of the Fourth International Software Process Workshop, Devon, UK, 43-45. Boltimore, MD: ACM Press, published as Software Engineering Notes, 14, 4.
- Bannon, L., and Schmidt, K. (1991). CSCW: Four characters in search of a context. In J. M. Bowers and D.D. Benford (Eds.), Studies in computer supported cooperative work: Theory, practice, and design. Proceedings of the First European Conference on Computer Supported Cooperative Work, pp 3-16. Held in 1989. North Holland.
- Bargeron, D., Gupta, A., Grudin J.& Sanocki, E. (1999). Annotations for streaming video on the web: System design and usage studies. WWW8/Computer Networks, 31(11-16), 1139-1153.
- Billings, C. E. (1996). Aviation automation: The search for a human-centered approach. Hillsdale, NJ: Lawrence Erlbaum Associates.
- Bly, S. A. (1988). A use of drawing surfaces in different collaborative settings. Proceedings of the Conference on Computer Supported Cooperative Work (CSCW88), Portland, OR, 250-256.
- Bonner, P. (1990, January). Most valuable products of the year/Freestyle. *PC/Computing* 3(1), 78-79.
- Bourguet, M. & Ando A. (1998). Synchronization of speech and hand gestures during multimodal human-computer interaction. *Proceedings of the conference on CHI 98*, 241-242.
- Box, Hunter & Hunter (1978). Statistics for experimenters. New York: John Wiley & Sons.

Chapanis, A., (1975). Interactive human communication. Scientific American, 232, 34-42.

- Chapman, R. J., Smith, P.J., Klopfenstein, M., Jezerinac, J., Obradovich, J., and McCoy, C.E. (2000a). CSLANT: An asynchronous communication tool to support distributed work in the National Airspace System. Proceedings of the 2000 Annual Meeting of the IEEE Society onSystems, Man and Cybernetics, Nashville, TN, 1069-1074.
- Chapman, R. J., Smith, P.J., Klopfenstein, M., Jezerinac, J., Obradovich, J., McCoy, C.E., and Orasanu, J. (2000b). Supporting collaboration in the National Airspace System with multimodal, asynchronous communications. Proceedings of the 2000 Conference on Human Performance, Situation Awareness and Automation: User-Centered Design for the New Millennium, Savannah GA, 324-329.
- Chapman, R. J. (1987). A comprehensive mail system for interpersonal messaging. Unpublished master's thesis, University of Edinburgh, Edinburgh, Scotland.
- Churchill, E. F. and Bly, S. (1999). It's all in the words: Supporting work activities with lightweight tools. *GROUP* 99, Phoenix, AZ, 40-49.
- Clark, H. H. (1996). Arranging to do things with others. Computer Human Interaction, 165-167.
- Clark, H. H, and Brennan, S. (1991). Grounding in communication. In L. B. Resnick, J. M. Levine, and S. D. Teasley (Eds.), *Perspectives on socially shared cognition* (pp. 127-149). Washington, DC: American Psychological Association.
- Clark, H. H. and Schaefer, E. F. (1989). Contributing to discourse. Cognitive Science 13, 259-294.
- Clark, H. H. and Wilkes-Gibbs, D. (1986). Referring as a collaborative processs. Cognition 22, 1-39.
- Daly-Jones, O., Monk, A., Frohlich, D.M., Geelhoed E. & Loughran S. (1997). Multimodal messages: The pen and voice opportunity. *Interacting with Computers*, 9, 1-25.
- Davis, R. C., Landay, J. A., Chen, V., Huang, J., Lee, R. B., Li, F. C., et al. (1999). NotePals: Lightweight note sharing by the group, for the group. *CHI '99*, 338-345.
- Decker, K. S. (1987). Distributed problem-solving techniques: A survey. IEEE Transactions on Systems, Man, and Cybernetics, 17(5), 729-740.
- Degen, L., Mander, R. and Salomon G. (1992). Working with audio: integrating personal tape recorders and desktop computers. *Proceedings on CHI '92 Conference on Human factors in computer systems*, 413-418.
- DeSanctis, G., and Gallupe, R. B. (1985, Winter). Group decision support systems: A new frontier. Database, 3-10.
- Duncan, S. D. (1973). Toward a grammar for dyadic conversation. Semiotica, 9, 29-47.
- Ellis, C., Gibbs, S., and Rein, G., (1991). Groupware: Some issues and experiences. Communications of the ACM, 34(1), 38-58.
- Faraday, P. M. and Sutcliffe, A. G. (1997). Designing effective multimedia presentations. Proceedings of CHI '97, 272-279.

- Faraday, P.M. & Sutcliffe, A.G. (1998). Making contact points between text and images. Proceedings ACM Multimedia 98, Bristol, UK, 29-37.
- Federal Aviation Administration. (2000). Collaborative Decision Making [Online]. Available at: <u>http://ffp1.faa.gov/CDM.htm [2000, June 12]</u>.
- Federal Aviation Administration. (2001). Air Traffic Control System Command Center [Online]. Available at: http://www.fly.faa.gov/flyFAA/index.html [2001, August 5].
- Fish, R., Kraut, R., Leland, M., and Cohen, M. (1988). Quilt: A collaborative tool for cooperative writing. *Proceedings of the Conference on Office Information Systems*, Palo Alto, CA, 30-37.
- Flores, F., Graves, M., Hartfield, B., and Winograd, T. (1988). Computer systems and the design of organizational interaction. ACM Transactions on Office Information Systems. 6(2), 153-172.
- Francik, E. and Akagi, K. (1989). Designing a computer pencil and tablet for handwriting. Proceedings of the 33rd Human Factors Society Annual Meeting, Denver, CO, 445-449.
- Francik, E., Rudman, S. E., Cooper, D., & Levine S. (1991). Putting innovation to work: Adoption strategies for multimedia communications systems. *Communications of the ACM*, 34(12), 53-63.
- Francik, E. (1996). Rapid, integrated design of a multimedia communication system. In Rudisill, M., Lewis, C., Polson, P. G., and McKay T. D. (Eds.). *Human-computer interface design: success* stories, emerging methods, and real-world context. San Francisco, CA: Morgan Kaufmann Publishers, Inc.
- Galegher, J., Kraut, R., and Egido, C. (Eds.) (1990). Intellectual teamwork: Social and technological foundations of cooperative work. Mahwah, NJ: Lawrence Erlbaum Associates.
- Galegher, J. and Kraut, R. E. (1992). Computer-Mediated Communication and Collaborative Writing: Media Influence and Adaptation to Communication Constrains. *Proceedings of the conference on Computer –Supported Cooperative Work* (CSCW'92), Toronto, Canada, 155-162.
- Goodman, Nelson. (1976). Language of art: An approach to a theory of symbols. Indianapolis, IN: Hackett Publishing Company.
- Greenberg, S. (Ed.). (1991). Computer-support cooperative work and groupware. San Diego, CA: Academic Press.
- Greenberg, S., Roseman, M., Webster, D. and Bohnet, R. (1992). Issues and experiences designing and implementing two group drawing tools. *Proceedings of the 25th Annual Hawaii International Conference on the System Sciences*, 139-150.
- Grief, I. (Ed.). (1988). Computer supported cooperative work: A book of readings. San Francisco, CA: Morgan Kaufmann.
- Harmon, R., Patterson, W., Ribarsky, W. and Bolter, J. (1996). The virtual annotation system. Proceedings of VRAIS '96, 239-245.

- Hardock, G., Kurtenbach G., and Buxton W. (1993). A marking based interface for collaborative writing. Proceedings of the Sixth Annual ACM Symposium on User Interface Software and Technology, 259-266.
- Holt A. W. (1088). Diplans: A new language for the study and implementation of coordination. ACM Transactions on Office Information Systems, 6(2), 109-125.
- Huber, G. P. (1982). Group decision support systems as aids in the use of structured group management techniques. In, G. Dickson, Ed., *Proceedings of 2nd International Conference on Decision Support Systems Transactions*, San Francisco, CA, 96-108.
- Huber, G. P. (1984). Issues in the design of group decision support systems. MIS Q. 8, 3, 195-204.
- Koons, D. B., and Sparrell, C. J. (1994). Iconic: Speech and depictive gestures at the humanmachine interface. Proceedings of the CHI '94 Conference Companion on Human Factors in Computers Systems, 453-454.
- Kotovsky, K., Hayes, J. R., and Simon, H. A. (1985). Why are some problems hard? Evidence from Tower of Hanoi. Cognitive Psychology, 17, 248-294.
- Kraemer, K, L., and King, J. L. (1988). Computer-based systems for cooperative work and group decision making. ACM computing survey, 20(2), 115-146.
- Kruse, R. B. (1994). Data Structures and Program Design (3rd ed.). Upper Saddle River, NJ: Prentice Hall.
- Kurose, J. F., and Ross, K. W. (2000). Computer networking: A top-down approach featuring the internet. Reading, MA: Addison-Wesley.
- Kullberg, R. L. (1995). Mark your calendar!; Learning personalized annotation from integrated sketch and speech. Conference companion on human factors in computing systems, 302-303.
- Lewis, P. H. (1988, November 20). Exotic gizmos for aiding workers. The New York Times, F11.
- Lochovsky, F. H., Hogg, J. S., Weiser, S.P., Mendelzon, A.O. (1988). OTM: Specifying office tasks. Proceedings of the Conference on Office Information Systems, Palo Alto, Ca, 46-53.
- Marshall, C. C. (1997). Annotation: From paper books to the digital library. Proceedings of the ACM Digital Libraries '97, 131-140.
- McCoy, C. E., Smith, P. J., Billings, C. E., Chapman, R. J., and Obradovich, J. (2001). Collaboration in air traffic management: Use of coded departure routes. *Proceedings of the Eleventh International Symposium on Aviation Psychology*, Columbus OH.
- McNeill, D. (1002). Hand and Mind: What Gestures Reveal about Thought. Chicago, IL: University of Chicago Press.
- Metron, Inc. Collaborative Decision Making. [Online] <u>http://www.metsci.com/cdm/</u> [August 5, 2001].

- Minneman, S. L. and Bly, S. A. (1991). Managing a trios: A study of a multi-user drawing tool in distributed design work. Proceedings of the 3rd Guelph Symposium on Computer Mediated Communication, 154-167.
- Neuwirth, C. M., Chandhok, R., Charney, D., Wojahn P., & Kim L. (1994). Distributed collaborative writing: A comparison of spoken and written modalities for reviewing and revising documents. Proceedings on CHI '94 Conference on Human factors in computer systems: "celebrating interdependence", Boston, MA, 51-57.
- Neuwirth, C. M., Kaufer, D. S., Chandhok, R. & Morris, J. H. (1990). Issues in the design of computer support for co-authoring and commenting. *Proceedings of the Third Conference on Computer Supported Cooperative Work (CSCW'90)*, 183-195.
- Neuwirth, C. M., Morris, J. H., Regli, S. H., Chandhok, R. & Wenger, G. C. (1998). Envisioning communication: Task-tailorable representations of communications in asynchronous work. CSCW 98, Seattle, WA, 265-273.
- Nielsen, J. (1993). Usability engineering. Chestnut Hill, MA: AP Professional.
- Norman, D. A. (1988). The psychology of everyday things. New York: Basic Books.
- Obradovich, J. H., Smith, P. J., Denning, R., Chapman, R. J., Billings, C., McCoy, E. and Woods, D. (1998). Cooperative problem-solving challenges for the movement of aircraft on the ground. *Proceedings of the Human Factors and Ergonomics Society* 42 *Annual Meeting*, 57-61.
- Oviatt, S. (1999). Ten myths of multimodal interaction. Communications of the ACM. 42(11), 74-81
- Oviatt, S., DeAngeli, A., & Kuhn K. (1997). Integration and synchronization of input modes during multimodal human-computer interaction. *Proceedings of the Human Factors in Computing Systems*, 415-422.
- Paivio, A. (1986). Mental representations. New York: Oxford University.
- Rasmussen, J. Brehmer, B. and Leplat, J. (Eds). (1991). Distributed decision making: cognitive models for cooperative work. West Sussex, England: John Wiley & Sons Ltd.
- Sacks, H. & Schegloff, E. A. (1979). Two preferences in the organization of reference to persons in conversation and their interaction. In G. Psathas, Ed., *Everyday Language: Studies in Ethnomethodology*. New York: Irvington.
- Searle, J. R. (1969). Speech acts: An essay in the philosophy of language. New York: Cambridge University Press.
- Shneiderman, B. (1997). Designing the user interface: strategies for effective human-computer interaction (3rd ed.). Reading, MA: Addison Wesley Longman
- Slack, J. M. (2000). Programming and problem solving with Java. Pacific Grove, CA: Brooks/Cole.
- Smith, P.J., Billings, C. E., McCoy, C.E., & Orasanu, J. (1999). Alternative architectures for distributed cooperative problem-solving in the National Airspace System. (Tech. Rep. CSEL-1999-24). The Ohio State University, Institute for Ergonomics, Cognitive Systems Engineering Laboratory.

- Smith, P. J., Billings, C. E., Chapman, R. J., Obradovich, J., McCoy, C. E., & Orasanu, J. (in press). Alternative "rules of the game" for national airspace systems. *Proceedings of the Fifth Conference on Naturalistic Decision Making*, Tammsvik, Sweden.
- Smith, P. J., McCoy, C. E., Orasanu, J., Billings, C., Denning, R., Rodvold, M., Van Horn, A., & Gee, T. (1995). Cooperative problem-solving activities in flight planning and constraints for commercial aircraft. Proceedings of the 1995 Annual Meeting of the IEEE Society for Systems, Man and Cybernetics. Vancouver, British Columbia, Canada, 4563-4568.
- Sproull, L. & Kiesler, S. (1991). Connections: New ways of working in the networked organization. Cambridge, MA: The MIT Press.
- Stefik, M., Foster, G., Bobrow, D.G., Kahn, K., Lanning, S., and Suchman, L. (1987a). Beyond the chalkboard: Computer support for collaboration and problem solving in meetings. *Communications of the ACM*, 30(1), 32-47.
- Stefik, M., Bobrow, D. G., Foster, G., Lanning, S. and Tatar, D. (1987b). WYSIWIS revised: Early experiences with multiuser interfaces. Proceedings of the Conference on Computer-supported Cooperative Work, 276-290.
- Stern, J. M. (1998). Microsoft flight simulator handbook. Indianapolis, IN: Brady Publishing.
- Streeter, L. A. (1988.) Applying speech synthesis to user interfaces. In M. Helander (Ed.), Handbook of human-computer Interaction (pp. 321-343). North-Holland: Elsewier Science
- Tang, J. C. (1991). Findings from observational studies of collaborative work. Int J Man Machine Studies, 34(2), 143-160.
- Tang, J. C. and Leifer, L. J. (1988). A framework for understanding the workspace activity of design teams. Proceedings of the Conference on Computer Supported Cooperative Work (CSCW88), 244-249.
- Tatar, D. G., Foster, G., & Bobrow, D. G. (1991). Design for conversation: Lessons from cognoter. International Journal of Man-Machine Studies, 34, 185-209.
- Tufte, E. R. (1990). Envisioning information. Chesire, CT: Graphics Press.
- Ware, C. (2000). Information visualization: Perception for design. San Francisco, CA: Morgan Kaufmann Publishers.
- Wittaker, S and Sidner, C. (1996). Email overload: Exploring personal information management of email. *Proceedings of CHI '96 Human Factors in Computing Systems*, 276-283.
- Wojahn, P.G., Neuwirth, C. M., & Bullock, B. (1998). Effects of interfaces for annotation on communication in a collaborative task. Proceeding of CHI '98 Human Factors in Computing Systems, 456-463
- Woo, C. C. (1990). SACT: A tool for automating semi-structured organizational communication. Proceedings of the Conference on Office Information Systems (pp. 89-98). New York: ACM.

- Zhang, J., & Norman, D. A. (1994). Representations in distributed cognitive tasks. *Cognitive Science*, 18, 87-122.
- Zielgler, J.E. and Fahnrick, K. P. (1988). Direct manipulation. In M. Helander (Ed.), Handbook of human-computer interaction (pp. 123-133). The Netherlands: Elsevier Science.

APPENDIX A. SLIDE SHOW SCENARIOS

diwida sch. Summary F	hightal Edeni Houses i			the state of the	• your ave Elight	Cole di
ention: durds sole Summer	Fight: DTW RDU 1000					
weit: Adding and San D.				Ochen	All States	
DL/01/198 To 07/01	(12); Depend Time: (000)	e 2400; Buthut D	W:Amfat:RDU		TAA Ted Junt	
Section of the section of the	L. Sandara Statistics	7.13.2	1121.25.252		And Date 1	
Explore Date Mining		2-2-01-0	AND THE PAS			
	2 P. 11	Ser Startes				
Chel Dauta	ातां स्वेश्वर देव दिवाई स्थिति। जन्म		C Diference	-75		
S No Halding	35 Ar Fuellum Uncort. (Ibel 8,620.0 9,26	11 643.1	7.52		Y-5-21
06/19/1998	Ar FueBurn Uncorr. ((be) 9,600.0 13,86	5.265.0	61.22		
05/14/1998	Air FuelBurn Uncorr. (lbs) 0.000.0 10,64	10 2,540.0	33.02 355		
06/24/1998	Air FuelBurn Uncorr. ((bs) 8,400.0 10,98	50 2,585.0	30.82		
05/22/1988	Air FuelBurn Uncorr. ((bs) 8,500.0 10,23	10 1,730.0	20.42		
	Air FuelBurn Uncorr. ((bs) 0,20010 9,610	10 1,510.0	1962	in the second se	1.25
	All Fueldum Uncom. (201 8,300,010,54	10 1,540,0			10 mil
	An Fuelburn Uncon. (105 0,700.010,220 10-1 0,500,010,000	10 1,320,0	1.25		
	Ar Fueldure Lincore	10-1) 7,400,0 9,463	10 1,385.0	144		
C6/29/1988	Air FuelBurn Uncort. (be) 8500.0 9.44	10 9400	11 12 - 32-24		
05/17/1958	Ait FuelBurn Uncort, [bai 8,100.0 8,97	10 875.0	10.82		Sec.
2	Air FueBurn Uncorr. (bel 8,400.0 9,11	50 7150	852		
<pre>stakth restrict</pre>		TAP? (CONSIG	all worked		Ser Low House	ا د ا
	ىغاد <u>ە</u> ئەكتىر جىررارىيە بەر		Satis and the second states			

Appendix A.1 ZID: DTW to RDU Slide 1

$\mathcal{R}_{i} \geq -1$				
Ele Elt Your Se Addies Y				
The theorem is the Country of high t	Constitute 1		X	iu reft burnning Füchtel filed
Lonation: double ash: Summar. Fig	NH: DTV ROU 1008 D05			
Security Malage and FAA Date	Records Versland: Minde			Strange Strange Strange
04/01/199 Te 07/01/19		Distant Distant	CIOU	TAA Rod Roots
Explore Data Mining	Careful Careford Tra			
	A CARLES AND AND A SALES		27 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	
Find Reads		ALL AND TARKED AND		
DTW_CW/S_VW 3	Air FuelBurn Uncon. (bs)	8,620.0 9,263.1	6431 7.5%	
No Holding 35	5 Air FuelBurn Uncorr. (bs)	8.620.0 9.263.1	<u>643.1 7.5% ▲ 5</u>	
	Ar FuelBurn Lincorr, (lbs)	80000105400	26400 7302	
06/24/1998	Air FuelBurn Uncorr. (bs)	8,400.0 10,995.0	2.585.0 30.8%	
05/22/1988	Air FuelBurn Uncorr. (bs)	8,500.0 10,230.0	1,730.0 20.42	
	At FuelBurn Uncorr. (bs)	0,200,0 9,010,0 0,900,0 10,540,0	1,610.0 19.6%	
05/12/1598	Air FuelBurn Uncort. (bs)	8,700.0 10,220.0	1,520.0 17.5%	The second state
05/07/1958	Ar FuelBurn Uncorr. (bs)	9,500.0 10,885.0	1,395.0 14.6%	
	Air FuelBurn Uncont. (bs)	7,400.0 8,465.0	1,055.0 14.4%	and the second
05/17/1958	All FuelBurn Uncorr. (bs)	8,100.0 8,975.0	675.0 10.62 C	
05/13/1988	Air FuelBurn Uncon. (Ibs)	8,400.0 9,115.0	7150 8,5%	
	AN CONTRACTOR		<u> 1977-1988</u>	स इ. र. न द रहेल
	• · · .			
• ***** 3.3*****************************		A 19 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	PLAT A CONTRACT	

Appendix A.2 ZID: DTW to RDU Slide 2

en de la					
Edt Yow Go Add	ins: Window Holp		المعلادة المستنزي فيحفظ والتحاط		
Itwendig to countermany	Shipter Fried Friday 1			🛃 ju zahi. Summ	ary Elights I Filed
paties: deaduacts: Suma	ay: Fight: DTW ROU 1000 (Stinut 1720 Find Nava			Second and
ander fahlen and bly					
04/01/199 To 07	GI/ISL DesetTes - DOD.T		Ant:RDU	TAA The	
المتنادية ومستعملات المراج			Street for the state	Sala - Anna Bu	
Embra Data Mining		C. Galactic Strategy	1. S. M. S. M. S.	See 19	and a state of the
Ciptor Velenserg.					
1 N	abar of		Difference	ile est es	
Fied Route In	dances - Parlamanca Metric	Planned Actual Ma		11 200 - SQ	
DTW.CANS.W	35 Air FuelBurn Uncorr. (1	a) 8,620.0 9,263.1	643.1 7.5%		CONTRACTOR OF STREET
No Holding	35 Air FuelBurn Uncort. (I	(m) 8.620.0 9.263.1	643.1 7.5%	▲副院	
	All Fueldum Uncorr. (I	56 8.600.0 13.865.0	5.2650 61_2%		
	Ait FuelBurn Uncorr. (as) 8.400.0 10.995.0	2,585.0 30.8%		and the second second
- 05/22/1998	Air FuelBurn Uncorr. (1	as] 8,500.0 10,230.0	1,730.0 20.4%		
06/05/1998	Air FueBurn Uncor. (bs) 0,200.0 9,810.0	1,610.0 19.6%		
	Air FuelBurn Uncorr. []	(as) 8,900.0 10,540.0	1,640.0 18.4%		
NK/N7/1999	Air Fueldum Uncont, (i	Net 9 500 0 10,2200	1,320,0 17.34		ATTEND TH
06/03/1998	Air FueBurn Uncorr. (I	as) 7,400.0 8,465.0	1.065.0 14.4%		A CONTRACTOR OF A CONTRACT
05/28/1998	Air FuelBurn Uncorr. (I	a) 8,500.0 9,440.0	940.0 11.1%		and a second of the second
05/17/1998	Air FuelBurn Uncorr. (es) 0,100.0 8,975.0	875.0 10.8%	13.6	APRICE CONTRACTOR
	Air Fuelburn Uncort. (I	as) 0,400.0 9,115.0	7150 85%		
		······································			
				•••	
The second second second second	1 YO M 100 10 10 100 100 100 100 100			CONTRACTOR DAY	
Production Charles		And the second	2176,240,767,46,12	CONTRACTOR AND AND THE CO	

Appendix A.3 ZID: DTW to RDU Slide 3

$\sigma_{A_{1}} = 1.41$					
Be Ed You Go Addine W		11. X			6)-32 A Pr
3 thready in the optimizer, in high t	Stand Lander			E u.ch Summary Light.	I Eded
Longium doublest: Summer Figh	AL DIWINDU YOU DO	NVA172 Real Protect		The second second	THE 1770
Estate man and the	and the second				
Contraction of the second				Annual Banks Toris	12,35,1 1
Enters Data Mirina		Sec. 17 Sec. 18			
Number of	1977 - T & 1979 - T & 1	555 BES 0	Meneroe Salata		
Find Rode	* Patranen Matter	Percel Actual Lines	Contraction of the second second		
No Ma Malan 30	Air Fuelsum Uncorr. (Ds) Air FuelSum Uncorr. (Ds)	86200 92631	543.1 7.55 A		
06/19/1900	Air FuelBurn Uncorr. (be)	8,600.0 13,865.0	5.265.0 61.22		
05/14/1998	Air FuelBurn Uncon. (bs)	8,000.0 10,640.0	2,640.0 33.0%		
	Air Fuellum Lincon (the)	8500.0 10 230.0	1 730 0 20 43		
05/05/1998	Air FuelBurn Uncorr. (bs)	8,200.0 9,810.0	1,6100 1963		N.S.
05/01/1988	Air FuelBurn Uncorr. (bs)	8,900.0 10,540.0	1,640.0 18.4%		
205/12/1988	Air FuelBurn Uncon. (bs)	8,700.0 10,220.0 9,500.0 10,995.0	1,520.0 17.52		
15/02/1508	Air FuelBurn Uncon. (bs)	7.400.0 8.465.0	1.065.0 14.42		
05/25/1588	Air FuelBurn Uncorr. (bs)	8,500.0 9,440.0	940.0 11.12		
06/17/1998	Air FuelBurn Uncon. (bs)	8,100.0 8,975.0	875.0 10.8%		
	All Fueldum Uncon. (DS)	1.0000 3.1120	/15.0 8.5% State: 1.0 1.0 1.0 1.0 1.0 1.0 1.0 1.0 1.0 1.0		
		a l'anne ann a gailtean an a	a de la cria en la cria		
· · · ·					
•					
		CONTRACTOR AND		and the second	<u>, ,,,,,,,</u>

Appendix A.4 ZID: DTW to RDU Slide 4

Nwolo s bi gronovi i k						Fill sen Sur	more Floatst	1 Filed
allant chondu schr. Summapr.	Finit: DTW RDU-10	BOUGHNAN,	20 Find Ray		- Artes	1		و المراجع
ي د المار و المار و المار و المدينة و الماري . الماري الماري الماري الماري الماري الماري الماري الماري الماري ال	an an tao an tao ang	والمحافظ والمحافظ المحافظ المح			1.010	数字を決	Scale C	يل ج
DLOI / ST CO/OL		1000					12	a zika ≤
Fig and a second second second	Second	a start and a start and	0-00-044	ACTONICAL IN	وببود بيوي	1. S.		1 ave 1
ferditinis, is sufficient a subscription (in Romania Roma Marina)	1	$z \neq z$	Gently .			N 37 4 81	234 F.S.S.	
	1.2.2.1.1.5	1.25	E STER	443 N.A.	153.0	A CARLE		5-1-1
مان المراجع ال المحاصة المراجع		778 8 21		Dimense	carle			A
ind Route instance	Performance Ma	ics Plane	Actual 10	H	1997 - S			1.00
TW.CANS.WW	35 Air FuelBurn Uncor	(bs) 8,620.0	9,263.1	643.1	7.5% ▲		7	- (i) (ii) (ii)
No Holding	35 Ar FueBurn Uncon	(bs) 8,620.0	9,253,1	643.1	7.5≿ ▲	副野人		National State
05/19/1998	Air FuelBurn Uncor	(bs) 8,600.0	13,865.0	5,265.0	61.22			
2 US/14/1338	Ar Fuelsum Uncon	(105) 8.4001	10,640.0	2,540.0	33.02		River	
00201200	At rosean onco		J. 10.3040	2.0000				
··· 06/05/1998	Air FuelBurn Uncor	(bs) 8,200.1	0 9,910.0	1,610.0	19.6%			
1 05/01/1998	Air FueiBurn Uncor	. (bs) 8.900.	0 10,540.0	1.640.0	18.43	: 63165		
. (5/12/1338	Air FuelBurn Uncor	. (ibs) 8,700.	0 10 220 0	1,520.0	17.5%		MALLEN SUT	
	Air Fueldum Uncor	(DS) 9,5000 (Ds) 7,400	10,005.0	1,365.0	14 64			
× 15/23/1998 ·	Air FuelBurn Uncor	. (De) 7,400.(. (De) 8500.(9.440.0	940.0	11.12	المحكمة ال		
66/17/1958	Air FuelBurn Uncor	(ibs) 8,100.	0 8,975.0	875.0	10.9%	1000	Particule & the	
ST 05/13/1990	Air FuelBurn Uncor	[be] 0.400.	9,115.0	715.0	8,5%	비초않습	S	
Million St. 14	Contraction of the second s	1.10-1-1-5-11	A Chevron	a star agent		a la		E . E
منه د	ى د اين ەر قام د. بر يې د تېرىخە د		Contraction of the second					

Appendix A.5 ZID: DTW to RDU Slide 5

	_ (#1 ×
File: Edit: View, Go Add Ins. Window, Help: 1997 A State Party State Party State Party State Party State Party	and the second second
	3
	2
Search Refine and FAA DatafOLD DATAL Minister of E-10 Datas 04/01/1928 (a0//01/1928 - c)	R. C.
Depart Time (000 To 200): Option: DTV: Antier ATL: Base of the State o	ie i
	25
Entrore Data Mining.	
Departure Annvel Departure Aircraft Munder of Water States and Aircraft Air	
Airport Airport Time (2) Type Instances Performance Metrics Planned Actual (Actual Planned)	-11
18. Ar Sunfluent Jacob 10 572 2, 11 651 1 1 229 9, 11 52	
DTW 412 150 105 0 46 Ar Fuelkum Uncon (ta) 10.4831 11.4301 941.0 903	
DTW ATL 1630 DSS 14 Ar FunBurs Uncon. (bs) 9,335.7 10,150.0 814.3 8.7%	2
DTW2 ALC 1055 725 20 39 Ar Fuelbun Uncon. (bs) 13,666.7 14,745.1 1,078.5 7.93	긴
	<u></u>
and a second	
	4
XXA	
33000	
2,500.0	
20000	
1500.	
5 10000	
	77.4
Construction of the second sec	

Appendix A.6 ZTL: DTW to ATL Slide 1

طومية يعتر متنصي الحاجات الحاجر الأ	r to and i	Essats 1.4				×	witch2.petr Summity Flu	972
time: Sourch2 sch: Summar: Fig ch:: 07/01/1998: DepartTime Trate: Source	h::1:DTw/ (D DATA) :000 To 2	NTC1105725 Fied F Minimutar auf TC11 00: Departue DTV7	cules: Deles: 04 setet ATE	01/1 358 1				
in a state of the second s Second second	Number of	Parlamente Mal	10 5.0°.	Actual	Diference			違い
TW: CAVVS: VWV: APE. HVCO Holding Tw: CAVVS: VWV: R0D. J43:0 Holding	1 1 13 6	Air FuelBurn Uncorr. Air FuelBurn Uncorr. Air FuelBurn Uncorr. Air FuelBurn Uncorr.	(bs) 15.900 (bs) 15.900 (bs) 13.776 (bs) 13.466	0 20.020.0 0 20.020.0 9 15.518. 7 16.153.3	4,120.0 4,120.0 1,741.5 2,686.7	25.9X 25.9X 12.6X 20.0X		
D4/05/1998	7	Air Fuellum Uncon. Air Fuellum Uncon. Air Fuellum Uncon. Air Fuellum Uncon. Air Fuellum Uncon. Air Fuellum Uncon.	(bs) 12,600 (bs) 13,600 (bs) 14,100 (bs) 13,600 (bs) 13,600 (bs) 13,000 (bs) 14,042	0 15.580.0 0 16.560.0 0 17.120.0 0 15.580.0 0 14,760.0 9 14,974.3	2,990.0 2,960.0 3,020.0 1,990.0 1,750.0 3,31.4	23.7% 21.8% 21.4% 14.6% 13.5% 6.6%		NIN IN IN
an a	- 14 - 544 S	an Aligenta (12 Sec.) Lista da Santa angles (12 Sec.)	an a	2017 A. 17	ಕ್ಷಣ () ನಿರ್ದಾರ ಎಂ. ವರ್ಷದಲ್ಲಿ 1 ಕ್ಷಣವರ್ಷ	. 		- REGER
ing a second							· · · · · · · · · · · · · · · · · · ·	

Appendix A.7 ZTL: DTW to ATL Slide 2

File Edit Year Go Adding Window Help		()
a second contract contract contract to the discourse of the	E wrold sch Summary Film	
Landsteen Search2 sets Summary Fights1: DTW AT Stills 7.5 Find Rouse 5.1.5 Fight Rouse 5.1.5	Man	l's à
Search: Justice and FAA Data(OLD DATA); Mindowith of 1:10: Datas: 04/01/1998 10		
U/UI/1998 Depart Time: COUD To 2400: Departs DTW, Assert ATL: 314 - 14 - 14	TAA Padama	1. Z
		i∎ ĝ
Englone Data Mining		
S Number of State		
Fiel Route: Planet Alexandre Planet Planet Actual Actual Planet		1
Halding	292	
DTW: CAWS. WW. R0D. 43.0 13 Air FueBurn Uncor. (bs) 13,776.9 15.518.5 1,741.5 1	126x	
04/22/1998 Air FuelBurn Uncorr, (bs) 13,465.7 16,153.3 2,665.7 2	24.62	4
04/05/1998 Air FuelBurn Uncon. [bs] 12,600.0 15,590.0 2,990.0 2	2172	р 19.11
Air Fuelsum Uncon, (bs) 13,5000 16,5600 2,5600 2	21.42	r ž
04/15/1998 Air FuelBurn Uncorr. (bs) 13.600.0 15.580.0 1.980.0 1	14.62	注意
No Holdingt State 7 Ar Funßum Uncon. Ibst 14.042.9 14.974.3 931.4	6.5X	12
		1
		1
al elementaria en la companya de la		
		<u> </u>
		ŝ
		100
		-
		- 1 2

Appendix A.8 ZTL: DTW to ATL Slide 3

Search2.veh: Su	nunary: Flight=1				- - -
ation: Search2.sch	Summer: Fights1		Addition and in		
rok: Astern en	FAA Data(OLD DATA): N	interior artist	04/01/1998 To 07/01/	1998.	
Explore Data Mi	ing.				
Anval	Departure Aircraft Nu	mber of		Diference	T
ost Airport V ATL	Time [2] 7 Type 7 / / In 1105 725 0	tences Performance M 14 Ar FuelBurn Unco	nice: <u> Planned Act</u> nr. (Ibs) 13,928.6 15,8	ud <u>(Actuel Plenn</u> 40.0 1,91	edi - 11 1.4 13.7% -
ATI	1027	45 Ar SupPres Hore	w (lbs) 10,4891 11.4	201 94	10 902
ATL	1630 D95	14 Ar FueBum Unco	nr. (bs) 9,335.7 10,1	50.0 81	4.3 8.72 4
AIL	1055 725 0	39 Ar Fueldum Unco	w. (Ibs) 13,666.7 14,7	45.1 1.1//	15 /3× •
neh seh Summ	ary Bapht - de paays				
Scatter Chart	Ber Chert				
X/Y Avis		Constant of the second			
- Sinnon		AREAS PLAN AND AND A			77.24
5					
E 60000-	1 1 1 1 1 1 1				
4.000.0				14-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-	在车员
2,000.0					10 P. 10
5.570.0			Č S	•	
			with the first state in the	Party and School	

Appendix A.9 ZTL: DTW to ATL Slide 4



Appendix A.10 ZTL: DTW to ATL Slide 5



Appendix A.11 ZTL: DTW to ATL Slide 6

Ele Edt - Yew Go Addine Window	Her	111 C		
🔊 a color an gaas an Englat 1	Chiefe Principles		X	which is such Susanaty Fagblics
Location: Search2sch Summay: Fights1:	DTWATLING DS5 Fiel Round		YZ-W2.5	Contraction and and and and
1.2.1.1.4. 17/01/1998 D 11 11		ATT DE LA CARA		TAA Find Benty
15 1 at a later a state of the state			1.2.2	Lanus Bran
Fundame Data Minima				Charles Constant
			3 G 2 Y	
Num	ber al.	Diferenc		
Fiel Roule	nces - Performence Metrice	Planned Actual Actual - Plan	+++	
DTW-EAVYS WW APE HVE	1 Ar FuelBurn Uncorr. (bs)		750 15.6%	
DTW CAWS WW ROD 4330	A Ar Fuelum Uncon. (05)	12,000,0 13,875,0 12	150 1564 167 902	
05/27/1998	Air Fueßum Uncon. (bs)	11,000.0 15,075.0 4,0	75.0 37.0%	
06/10/1998	Air FuelBurn Uncon. (bs)		30.0 21.97	
04/19/1998	Ar Funibum Uncort, (bs)	12,600,0 15,475,0 2,6	750 22.8%	
04/07/1998	Air Fuellum Uncorr. (bs)	10.000.0 12.020.0 1.2	20.0 11.3%	
No Haking	39 Air FuelBurn Uncon. (bs)	10,348.7 11,038.1 6	89.4 6.7%	
II OTW. CAWS WW. ROD BW	1 Air Fuellum Uncort. (bs)		95.0 1.8%	
Active No. Policing restance - manufacture	1 Ar Fueldum Uncan. (bit)		30.0 1.64	
8				
ii ii				
				March 199
	and the second secon	in the second	لك يول	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
A THE CONTRACTOR AND A CONTRACTOR OF A DESCRIPTION OF				
[4] DU PA THAT A V. LAN 21 ID AND MARK		A CONTRACTOR OF	12 - 14 C - 14 C - 14	

Appendix A.12 ZTL: DTW to ATL Slide 7



Appendix A.13 ZFW: MSP to DFW Slide 1



Appendix A.14 ZFW: MSP to DFW Slide 2



Appendix A.15 ZBW: DTW to BOS Slide 1

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.



Appendix A.16 ZBW: DTW to BOS Slide 2



Appendix A.17 ZBW: DTW to BOS Slide 3

Bit Sam So Address Solution Second Stress Provide Instance Second Stress Address Second Stress <t< th=""></t<>
Alight Instance:221000001 Alight Instance:12100000000000000000000000000000000000
Alter Alter Alter Alter Alter Alter Alter Alter Alter Planner Alter Older Alter Alter Planner Alter Blanner Alter Alter Fulfor Stott Alter Alter Alter Alter Blanner
Browner Argent Browner (Construction) Browner (Construction) Browner (Construction) Browner (Construction) Browner (Construction) Browner (Construction) Browner (Construction) Browner (Construction) Browner (Construction) Browner (Construction) Browner (Construction) Provide (Construction) Browner (Construction) Browner (Construction) Provide (Construction) Browner (Construction) Browner (Construction) Provide (Construction) Browner (Construction) Browner (Construction) Arifume Construction) Browner (Construction) Browner (Construction) Out Trand(2) Browner (Construction) Browner (Construction) Out Trand(2) Browner (Construction) Browner (Construction)
Margin Margin Margin Reserved August Reserved Margin Reserved Margin Reserved Margin Performance Performance Prime Scott Scott Scott Tail Scott Asiline Scott Scott Scott Asiline Scott Cott Scott Out Test Scott Scott Out Test Scott Scott <t< th=""></t<>
Amined Angenet Balance Manager Animate Manager Balance Manager Out Time(2) Balance Manager Out Time(2) Balance Manager <td< th=""></td<>
Mathematic Mathematic Mathematic Mathematic Performance Performance Performance Performance Performance Performance Performance Performance Performance Performance Performance Ariane Constitution 9.0 16.0 4.0 Performance Performance Performance Out Transl 9.00 16.0 4.0 Performance Performance Performance Performance Performance Performance Performance Performance Performance Performac
Add Function Bandres Inducation Dimension 7 Parlowments Parlowments Parlowments 7 Fundioum Uncorn (Ibit) 9:5000 15:5000.0 6:3000 6:532 Add Fundioum Data 9:5000 15:5000.0 6:3000 6:532 Total Fundioum Data 9:000 16:5000.0 6:3000 6:332 Add Fundioum Data 9:000 16:5000.0 6:3000 6:332 Add Fundioum Data 9:000 16:5000.0 6:3000 6:332 Add Fundioum Data 9:000 16:5000.0 6:200 7:332 Add Time Cont (Imm) 80.0 14:9.0 6:20 7:332 Add Time Cont (Imm) 80.0 14:9.0 6:20 7:332 Add Time Cont (Imm) 80.0 14:9.0 6:20 7:332 Tawin (Imform) 9:00 16:0 4:0 2:000 Tawin (Imm) 10:0 10:11:12 10:0 10:0 Out Time(2) 19:00 18:15 10:0 10:0 Out Time(2) 19:47 21:00 7:10 2:1 Unit Mathematical State State State State Mage
Partonennest Personal Actual Research Fuelkum Uncor (bs) 9.500.0 15,800.0 6.300.0 66.332 a Fuelkum Uncor (bs) 9.500.0 15,800.0 6.300.0 66.332 a Total Fuelkum (bs) 9.600.0 15,800.0 6.300.0 66.332 a Arime Concentrining 9.600.0 149.0 62.00.0 66.332 a Arime Concentrining 80.0 149.0 62.0 71.32 Taxi Information 9.00 180.0 10.0 1.0 11.12 Taxi Information 9.00 181.5 15.0 a a Out Trand(2) 1800 181.5 15.0 a a Out Trand(2) 1947 2100 7.30 a Mage a a a a a Mage a a a a a Taxi Inductions 22.1555/2011 a a a a
2717 Process 2 Active (Mercel C Proces 2 Active (Mercel C Process 2 Active (Mercel
FusBun Uncor (bi) 9,500,0 15,500,0 6,300,0 66,32 a Air FusBun Can. (bi) 9,500,0 15,500,0 6,300,0 66,32 a Total FusBun Can. (bi) 9,500,0 15,100,0 6,200,0 66,32 a Air Tane Concentral 80,0 149,0 65,0 79,52 a Air Tane Concentral 90,0 149,0 65,0 79,52 a Tane Informal 90,0 149,0 65,0 79,52 a Tane Informal 90,0 10,0 1,0 11,12 a Tane Uncorn (min) 9,0 10,0 1,0 11,12 a Tane Uncorn (min) 9,0 10,0 1,0 11,12 a Tane Uncorn (min) 9,0 10,0 1,0 11,12 a Un Tane(2) 1900 1815 15,0 Out Tane(2) 1900 1831 11,0 a Un Tane(2) 1947 21,00 7,10 a Information (2) 1555511 a Tane Uncorn (1) 10 7,10 a Information (1) 10 7,10
Ad Fuedban Coll (c) 93000 16:000 6:000 6:02 Tataf Fuedban (c) 93000 16:000 6:00 6:02 AdTime Uncon (min) 800 16:000 6:00 73:52 AdTime Contenting 8:00 100 100 100 11.12 Tata Outemat 2:00 16:0 4:0 -20:02 Out Time(2) 18:00 18:15 11:0 Off Time(2) 18:20 18:31 11:0 Off Time(2) 19:47 2:100 7:30 Interference (c) 19:47 2:100 7:30 Interference (
AsTme Uncon (mm) 83.0 143.0 66.0 73.52 AsTme Conc(mm) 87.0 143.0 62.0 71.32 Tamil (mms) 3.0 10.0 1.0 11.12 Tamil (mms) 2.0.0 16.0 -4.0 -2.0.02 Out Tmm621 18000 1815 15.0 - Off Tmm621 1820 183.1 11.0 - Off Tmm621 1947 21000 7.20 - Imple Instants (22165557) - - - - Mage - - - - -
A#Trme Cont(mina) 67.0 149.0 52.0 71.32 Tawin(mina) 9.0 10.0 1.0 11.12 Tawin(mina) 9.0 10.0 1.0 11.12 Out Time(2) 1800 1815 15.0 4.0 Out Time(2) 1820 1831 11.0 0 Oin Time(2) 1947 2100 7.10 2 Map Tawin(the first state in the first state in
Teac Information 3.0 10.0 1.0 11.12 Teac Information 3.0 10.0 1.0 11.12 Teac Information 3.0 10.0 4.0 -20.02 Second and the se
Out Time(2) 1800 1815 15.0 Out Time(2) 1820 1831 11.0 Out Time(2) 1947 2100 73.0 Imphilications [2155:521] Standard State
Off Transf2 1820 1831 11.0 On Transf2 1947 2100 72.0
0n Tme(2) 1947 2100 73.0 21 hybrid last notice (221655-26) Mage: 10 - 10 - 10 - 10 - 10 - 10 - 10 - 10
TAA Ned here / C

Appendix A.18 ZBW: DTW to BOS Slide 4



Appendix A.19 ZBW: DTW to BOS Slide 5

% POE 7											
Ele Ed. View Ge	Califine >	Station ?!		C126 3		C-12:48-	317-1 - 35-74)		erne ver	and Friday	14
>> Flight Instance(1)	7966774				X						42.
Dependente Da	AX 57			A Type	2 <u></u>						
											2.7
Annual Aire		101000	21 A 1								
		(1991) (1991)		21.4	2456						
Parlamence Matrice	Reutina ia										prm1.fr
Pelenace-	2.0		Dillounce	nine al c	STREET, STREET						i Anti
FindBurn Lincor Det	Parent	Actual (Ac	A 450.0	42.02	State State						
Air FuelBurn Corr.(bs)	10,600.0	15,050.0	4,450.0	42.0%	읡						
Total FueBun(bc)	10,750.0 1	5,500.0	4,750.0	44.22	2						
AirTime Con.(mins)	92.0	121.0	29.0	31.5%	Š.						
Tate In(mine)	7.0	10.0	30	42.92	5.						
Out Time(2)	150	1947	47.0	inner	· · · · · · · ·					j.	
Off Time[2]	1915	2017	62.0								
un ime(2)	2.47	2218	al.n		<u> </u>			·			
	2 1 1		18 /171						- IOLX		
Map	-2748			14.34		The second					
			-1-1	11 M	Charles and	Patri		€ 566	4		
		10				1919	1-4-52.7				
			20.00	- (C. 4		5-14 H.		Sec. 2	2 G		
The second se						1. A.					
				0.4	(1 Sector 1 P)	este lear		in the second			1
		ي د اين اين مين د بر سري است		1.1.1.1				$T \rightarrow T$	97 - I		
				Sa di	1		63 - X		- -		
	÷.	:	्र - स्व			<u> </u>			171/2		ال د ا

Appendix A.20 ZBW: DTW to BOS Slide 6



Appendix A.21 ZDC: MEM to EWR Slide 1



Appendix A.22 ZDC: MEM to EWR Slide 2



Appendix A.23 ZDC: MEM to EWR Slide 3



Appendix A.24 ZDC: MEM to EWR Slide 4



Appendix A.25 ZDC: MEM to EWR Slide 5



Appendix A.26 ZDC: MEM to EWR Slide 6



Appendix A.27 ZMP: MDW to MSP Slide 1

Ele Edt Yam Go Addins Window		1			
a construction to the property property in the pro-			57	Firsht famores, Educt Rendert, Derefore	
Burning and any South stand	n and the new star			ringur droept fac ringular triggiay	
the second second second second	"+Composition	C-T-TOS ALL STA	1.1		
Soundary Delends (VI-224); Minimum Co		E Cincel	2231(322)		and a second
And a second sec			1.1		
State of the second		NA Designation			
Se 1: Emine Date Minnet 14: 1	and some for the second			And the December of the second s	
	21. S				
A Statistic Statistics of Statistics					
Find Route			25-2-12		
HOW BAE EAUGHER 16 A	rTime (mms) 62.6	80.7 18.1	29.02	i	
	Un ime (2) 2125	2148 21.7			
	1 mm (mm () 61.5	30.6 21.3	47.62 -	ilight titospit iliand ttootn3 Gi≤pLay. 	•
		2133 17.6		The second second second second	17-18-18-1
			100	and the second second and the second second	and the second se
			344		1.500
			121		
			3		
	Time (mins) 610	800 190	31 12 8		
	Off Time (2) 2115	2121 60	ž.		
The Name	Time (mins) 62.9	77.3 14.4	22.93 2		
	Off Time (2) 2130	2153 23.0	2	· · · · · · · · · · · · · · · · · · ·	Der .
	118118 (Rend) 55.0	35.0 23.0	43.9조 월		
	Time (mine) 628	A90 270	4352	Hight Group: Filed Goute1, Display	:
A CONTRACT OF THE OWNER	Off Time (2) 2115	2123 8.0		NA STATES AND	St. Carter
01/30/1918 NWA10/3	Time (mins) 59.0	84.0 25.0	4242 *	A DESCRIPTION OF THE PARTY OF THE PARTY OF THE	Carrier Carrier
	Ulf Time (2) 2115	2122 7.0		5372 JUNE	TASUAS
	11 mmet (mans) 54.0 04 Time (72) 2115	91.0 27.0	42.22 -	Service State Stat	
07/08/1998 364(A1075	Time immal 650	87.0 22.0	33.82 5		
A CONTRACTOR OF STREET	Off Time (Z) 2115	2135 200		and the second se	1-
05/10/1938 NWA1256 Ai	Time (mens) 62.0	73.0 11.0	17.7%		
A Contraction of the second states of the second st	Off Time (7) 2115	7176 11.0	الم ۽		14 A
	Territoria and a second se		. 북왕		1.53

Appendix A.28 ZMP: MDW to MSP Slide 2



Appendix A.29 ZMP: MDW to MSP Slide 3

				5-1/m-1	ALLI	a card a tour			
		V		251 X.		FULL AND			
AU		ia/ :≡	NWA		and to the s T a				
	Denedura	- L.	And Time Die		North Party		*****	97 - 19 - 20 - 20 - 20 - 20 - 20 - 20 - 20 - 2	w7:
	AZO	MCO		TUMNIT	Number •	Planned Airlime	ACTUBI AITTIME	Dimerence	Percent Airlin
NWA	MKE	MSD	100	TAINT	13	70.4 ED 7	103.1	34.7	40.4076
- NWA	A DECEMBER	TTO TO				327		319	00,00%
DINWA	PHL	MSP	100	TWINZ	10	3 (M) E	170.2	70'5	21 10%
TINWA	BOI	MSP	2300	SHONN	10	1394	167.9	20.0	22 10%
SINWA	OMA	MSP	2300	ZIBBY	16	45.5	74.7	29.2	64.10%
SINWA	DTW	MSP	100	TWINZ	17	85	114.1	291	34 20%
6 NWA	BWI	MSP	200	TWINZ	11	131.4	159.9	28.5	21.70%
ZINWA	MCI	MSP	200	ZIBBY	10	59.5	87.1	27.6	46.40%
INWA	BOS	MSP	200	TWINZ	14	153.1	180.1	27	17.60%
DINWA	ABO	MSP	2200	SHONN	14	121.4	148.1	26.6	21.90%
E NWA	PHL	MSP	200	TWINZ	14	139	165.3	26.3	18.90%
2 NWA	IAH	MSP	100	ZBBY	17	139	164.6	25.6	18.50%
AWNE	MKE	MSP	200	TWINZ	17	51.9	77.4	25.5	49.20%
INWA	STL	MSP	1300	ZIBBY	14	70.6	95.9	25.4	35.90%
5 NWA	TPA	MSP	0	Z188Y	10	170.2	195.5	25.3	14.90%
<u>S</u> NWA	FCA	MSP	2200	OLLEE	20	123	147.7	24.7	20.00%
Z'NWA	PDX	MSP	200	OLLEE	15	159.8	164.1	24.3	15.20%
BINWA	MKE	MSP	2300	TWINZ	15	52.7	76.7	24.1	45.70%
3 NWA	LAS	MSP	200	SHONN	15	153.7	177 8	24.1	15.70%
DINWA	MDW	MSP	200	TWINZ	15	52.4	86.4	24	38.50%
1-INWA	SEA	MSP	1700	OLLEE	16	157.7	181.4	23.8	15.10%
AWN	PDX	MSP	2200	SHONN	. 11	160.7	184.5	23.8	14.60%
4 NWA	STL	MSP	2100	Z188Y		71.7	_ 95.5	23.7	33.10%
5 NWA	SNA	MSP	200	SHONN	. 15	183.4	206.9	23.5	12.60%
7 NWA	PDX	MSP	1700	SHONN	11	158	181.4	23.4	14.80%

Appendix A.30 ZMP: MDW to MSP Slide 4



Appendix A.31 ZMP: OMA to MSP Slide 1

De ER Yew Go Addies Y	Index Line	1	244 - P.		
State of the state of the state	and a state of the state of the			17	Souch4 sch Elight Grouns
Constant Second and Finds Const					New of the Party of the local division of the local division of the local division of the local division of the
		South the	3.2.40		Sound D
Stander Britan (71-24) Film	THE REPORT OF THE	DO: WA	100	A S WOLLS	The state of the s
States in 2200 To 2	100: Aldina: Math: Bythat: Child		14 C	1.20.20	
A CONTRACTOR OF A CONTRACT	7.5 27 - 6.4		20.00	15 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	to the state of the state
And an and an and the second s				10 10 10 10	
Alson Martin Provide States and the		CARGE COM			1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
Find Room	and the second		And and a little		States of product
TONG FOR KASHIZINGP	24 ArTmelans)	452	69 A	216 52 32	
A subgrade state of the second second	Off Time (2)	2145	2204	18.3	
					and the second second second second
				·	To a sector in the sector
101/14/1980 NWA1082	AirTime (mins)	46.0	96.0	40.0 87.0%	
	Olf Time (2)	2140	2150	10.0	
	Artine (mms)	45.0	63.0	380 84.4%	
		- 2140	2152	200 0257	
	Off Time (2)	2140	2159	190	
07/21/1900 Maria 1082	Air Time (mins)	430	76.0	330 7672	
	Off Time (2)	2140	2147	70	
07/22/1980 NWA1052	AirTime (mins)	45.0	78.0	330 73.3%	
13. 15 M 10 M	Off Time (2)	2140	2157	17.0	
07/13/1980 MWA:082	AirTime (mins)	47.0	74.0	27.0 57.4%	
	Ulf Tene (2)	2140	2150	10.0	
		430	540	16.0 37.26	
	Uninne (c.)	45 3	2143	30	
Barned March 1997 - Charles	CM Time (21	2147	2208	201 44.34	
and the same to provide a strain of the stra			1000	STATISTICS IN AL	
A MARKANING TANK ASSAULT PLAN AND AND			C. Browner C. Albert		The second rest of the second state of the
A REAL PROPERTY AND A REAL		CONTRACTOR STATES			Control of the second

Appendix A.32 ZMP: OMA to MSP Slide 2

						arich1 soft Flight Grou
Jan Barton Barton States To Ma	nter of Performance	10175120		Defenence	E H	Ne
				20.0		
	All the (nets)	410	2145	3510	90.74 **	177 C
17/15/1980 MALASTOR2-5	AirTime (mine)	450	78.0	710	71 72	1
	Off Time (Z)	2140	2149	0 <u>e</u>	ير مندر	
07/01/1998 10:401092	AirTime Iminal	45.0	74.0	29.0	64.43 🗟	TAV STA
	Off Time (2)	2140	2151	11.0	5	
07/03/1500 NWA1002	AirTime (mins)	44.0	71.0	27.0	61.4 % 🗿	10000
	Off Time (2)	2140	2208	28.0	2	
CT/20/1999 NWA1092	AirTime (mins)	47.0	74.0	27.0	- 57.4% 🗿	
	Off Time (2)	2140	2237	57.0	10	A lease lease the
07/31/1999 NWA1082	AirTime (mins)	44.0	65.0	21.0	47.7%	The second second
	Off Time (2)	2140	2150	10.0		
07/06/1989 NWA1092	AirTime (mins)	45.0	66.0	21.0	46.7%	
	Off Time (2)	2257	2342	45.0		
	ArTime (mins)	45.0	65.0	20.0	44.43 🗑	
	Off Time (2)	2140	2209	29.0		
	All Time (mins)	46.0	66.0	20.0	43.5% _3	
	Off Time (2)	2140	2154	14.8	2	
07/07/1998 NWA1082	AirTime (mins)	46.0	64.0	18.0	39.1%	1
	Ult fime (2)	2211	2305	54.0	ř	
	Al i me (mins)	45.0	620	17.0	37.84 🖉	
		2140	2152	120	···	
	All i me (mine) Oli Timo (7)	48.0	2162	120	37.36	
		46.0	2133	150	24.05	
	Off Time (2)	2167	2213	0.01	<u>-</u>	
CT /1 7 /1 600 10 10 10 10 27	AirTime (aire)	460	59.0	130	29.32	
	Off Time (Z)	2140	2147	20		
	AirTime (mine)	44.0	53.0	90	20 52 🛡	4
1	CONTRACTOR OF		178 4 3 T 7	Service Letters	a 🛛 🕅 🕅	3
and a second	1. Contraction of the second	1.4. The second state		الجهد ورور والمطبقا والر		

Appendix A.33 ZMP: OMA to MSP Slide 3



Appendix A.34 ZNY: DTW to EWR Slide 1



Appendix A.35 ZNY: DTW to EWR Slide 2

	RAR
En Ed I Yow Go Adding Window Hop - James and the State of State of State	ALT MAR AND ALT
Statistics of the point and Fraght 3 of party states	X
Learning Search 2 set: Surgary 2 Fight 1: DTW 2 WRI Find Reasons 2: 25 st 2 s	25.5
Nillee and TAA Detail D.D. DATA: Maintail of 1: 10. Detail: U/07/1998 To 17/07/1998	腔
	100 m
Fiel Roue	
•>DTV/PSTN.DX0020.ADRIE: 1 Air FuilBum Uncon. (be) 9.000.0 11.660.0 2.660.0 21.6%	
DTW: TYCDB: Y015098.001.FN 0 14 Ar Fuelkum Uncorr. (bs) 8,450.0 10,215.8 1,755.8 20.93. ▲	3.6
[222] 04/09/1998 · · · · · · · · · · · · · · · · · ·	5
Air FuelBurn Uncorr. (be) 7,800.0 11,460.0 3,660.0 46.9%	
Ar Fuellum Uncorn. (be) 7,700.0 10,780.0 3,080.0 40.03	1 () () () () () () () () () (
Air Fueldam licent (test / Julio 10, 20010 30, 34	
7 205/02/1998 22 4 Ar FuelBurn Uncorr. (bs) 7,700.0 10,390.0 2,650.0 34.93	- 2
Air FuelBurn Uncon. (bs) 11.200.0 14.380.0 3,180.0 28.43	
Constant Con	. 19
	2
	•`
	F
	-1
	E .
	, ¹
	5 m
<u>[]] • · · · · · · · · · · · · · · · · · ·</u>	1.5.

Appendix A.36 ZNY: DTW to EWR Slide 3

> POET				AFE
Bo Edt Your Go Addies	diales deb	ويور مردون ساوي ويوم	كلية مستلق المناهم وتشريهم والمتعاصية فسيقتط	
S Clipht Instance (15159099	-	يع بقد التعاري الا		80.8
Dependent Dame, Silv Antonia Alexant Antonia Alexant Company Tanget 200 200				
Performance Matrice Rowing in Performance Matrice Plann Air FuelBurn Uncarr (bs) 11,900	Actual (Act	7,780.0 65.43		
Air Fueldsum Corr.[les] 11,500 Total Fueldsum(lbs) 12,180 Air Time Uncorr.[min] 63 Air Time Corr.[mins] 70 Taxis In(mins) 7	0 19,580.0 0 20,700.0 0 125.0 0 125.0 0 17.0	7,7800 65,4% 8,5200 70,0% 57.0 62,6% 56.0 80,0% 10.0 142,9%	144-12-5-12-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1	
Tasi Out(mins) 22 Out Time(2) 210 Off Time(2) 211 Off Time(2) 212 On Time(2) 212	0 33.0 10 2239 22 2312 32 0118	11.0 50.0% 99.0 110.0 166.0		

Appendix A.37 ZNY: DTW to EWR Slide 4

288

V POST		X
Br La You Bo Addan Made Hab		
S Eggit Instance(21906360)	F(-) X	
		Ŕ
Allow Made and a second s		÷.
	(1)	1
	1.00	
		1.fr
Performance Mathins - Personal - April - Mathins - Personal - April - Control - Contro		
Total FuelBurgber 7,780.0 11,300.0 3,520.0 45.22		
ArTime Uncorr (min) 63.0 105.0 35.0 52.2% 51		
Tavildfmins 7.0 130 6.0 6572		
TamiOutimumasi 14.0 15.0 1.0 7.1% state		
0 ut ima(2) 1500 1504 1519 5.0 ⊐		
On Time(2) 2040 2104 24.0 💌		
and a second		
Table 1 and		
Annual Restore and the second s		
		. ا
		Ľ

Appendix A.38 ZNY: DTW to EWR Slide 5

- POET					/						F IF
e Edt You Go Ad	des - M		ويتي زمان منهم	9- 5-5 -6	Ser Ser S		N. 172.		- 1 -1-1-		Markelan risering
& Flight Instance(230)	55964)		🖕 Grande I.a								
Departure Date Aidea Departure Aisput Animit Aisput				10 - 17 20 - 17							
Personal Meters Ro	r (1900) Ning Inter					i X				1	
Pulamence Metrics	Planned	Actual LAc	Difference	7.7.02	E.S.						
Air FuelBurn Uncorr.(lbs)	7,800.0 1	0.680.0	2,990.0	36.9%	i	3					
Ar FueBurn Corr.(bs)	7,800.0 1	0,680.0	2,990.0	36.9%		- 1					
Total FueBum(lbs)	9,680.01	1,200.0	1,520.0	15.7%	591						
ArTime Uncon (min)	65.0	970	32.0	49.2%	120						
ArTime Con.(mins)	86.0	97.0	11.0	12.64	ş	4					
i an inimis) Tari Ordenasi	14.0	130	50	80.7%							
n ge ougmensj Out Time 77	1900	1918	190	u.u.s							
Off Time(2)	1914	1972	180		-						
On Tree[2]	2040	2109	29.0								
and the second		e serie de	سر مبدق می	27/19/19		-					
Ne	<u> </u>	n <u>o</u> s y 7	FACTO		114-23	1		Sec. 1		13-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-	
			Strates and	1000 100 100 100	1.			1 7 6 14		11-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1	î l
TAA Fiel here	1 250	×	Card to	2.2	100	$\mathcal{D}_{i} = \mathcal{D}_{i}$	11 in 1			×	-
and the second lines	- -		Section She	200		200-5	1. 1 a	X.4.)	1.1	1.00	
	- <u>-</u> - 5-	677	and the second second		2 d 1 m	1.00	10 30	_ 6	C. Bert	6 R	
Poi subo di			- A.H. 1		2000	10.27	7 . C	20.00	- Carlo		
Categorie Page	No.	1947-1942 1		-	1.1.7	2/	5 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 -	and the		E-16-2	
Part Martin	1.200		1000			1 C .		2.9 5	Sec. 1.	C 5 5 7 7	
		-	2 / Lane		Stat. Co		12.5	53,55	28-3°,	2.57	
Statistics and	100	1.1.5	the state of	Sec States		-	2002	Tank Here	1.4.2	10.20-10	
	and the second second	4	1	3			7	8 2	5		
171										1 . 14	<u> </u>

Appendix A.39 ZNY: DTW to EWR Slide 6



Appendix A.40 ZOB: MSP to EWR Slide 1

			<i>4</i> 97					FIC
Red Rove		Perform			Pland		C Differences (
ISP DLL J34 BAE J34 CRL J584 SLT FUMT EWR	8	AitTime	Uncarr	(mm)	115.1	135.6	20.9	5 176
•								
16/18/1998 3/07 FEET AND THE		Artine	Uncarr		124.0	1820		J_46.8
06/04/1998		Autime	Uncan	[min]	105.0	131.0		20.0
- (6/15/1988		Allme	Uncon	[mm]	120	154.0		232
Noticing	·····	Alline	Uncan	(mn)	1114	123.6		2 30
MSP. DLLJ34.8AE.J70.PMM. CHLJ584.SLT.FOMT.EWH O	<u> </u>	Alline	Uncan	. (men)	121.1	[3].3		0 30
		A 1.7 1	Denes	()	110.0	152.0	24	200
B/COTISSE			Unicon	(min)	119.0	1200		11 3
		Alline	Uncon	. (min)	124.0	1350.0	7	7 64
Noticiand		AFINE	Uncan	(1101)		127.3		
SP SNINE LINE DZZ YYZ DST. BUE MAK SHOP JEWITS			Uncon		120.0			n a7
No holono	·····	ANTIN	Uncon	. (mm)	120.0	137.0		0 79
DIATIZ DEE STAPLIEWI		AirTime	Uncon	(1997)	177.0	137.0	10	79
			Uncar		127.0	137.0		
			Uncer		1250	120.0	č	
			Uncon	. Unanij	1220	1,30,0	يې 1940 - 1940 - 1940 - 1940 - 1940 - 1940 - 1940 - 1940 - 1940 - 1940 - 1940 - 1940 - 1940 - 1940 - 1940 - 1940	
					و غرب _و ستيده د دو کر . که			201
e Talendar en la secola de la construction de la construction de la construction de la construction de la const	- See, 1	.		1				
	12 12	51 0 1	205	705	1. A.	21222		1645
	694		100.2	-	136786			
	1.	1			Description -	Jast	-11 A	200
	200.00		1.3.1.15	. T.	THE OTHER			1.05
		171 153	5.4	-	*****	Jon in	49-14 Marca H	21.2
		0.000	2 cm.	-		-	1.0	- 1. A.
	-			-	1. 7. 7 M		1. 1. 1. 1. 1.	
		AT	· · · ·			the protocol is a	Seal State of the	5.0
	Sheet	JA WAY	1.1					
	C 373	-		- 14	See. Sugar	3000	and the second second	
CONTRACTOR STATES		1		Pines.	1996 - C			5 - V - 5
			10	-		ST 10.3		5.5.7
		1	1.0	-	57, Solid (are lines		
	N. 2. T.	THE REAL	201	1.50				2.14
	200	S 24	- un fait	北西				1.1
	18 A C	N 800 4	the start		TIC STATE	115		1
And a state of the second state		Strap of St	2. B.	1.00	A Destroyed in the second	A 26.47	Sugar	5 A - 7

Appendix A.41 ZOB: MSP to EWR Slide 2

Ed View Go 7/	Addine: Window)	Harris and State		221-2-2				
	ALL THE CONCEPTION			Line and			hanni yarata	S Official Three
			a S parame	Charles and a second	al and and	ALC: ST.C.		
CD CNINCT COOLE	22 42 6 21 6 16	LINK CHARGE CHAR	0.01	1	Time (2)	1959	2011	12.0
	10 - 10 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1		5.	1 04	Time (2)	1959	2011	120
SP/N7 GESH	1552 5 10 10 10 10 10 10 10 10 10 10 10 10 10				Time (2)	1959	2011	12.0
	1. S	N 200 25		1 0#	Tere (2)	1955	2011	120
ASP. DLL J34 BAE J7	70 PMM. CRL J584	SLT.FUNT.EWR	ିଂତ	8 0#	Time (2)	1955	2041	41.4
			•					
C6/17/1998	Server States of			Olf	Time (2)	1955	2119	90.0
1-06/25/1998	a start and a start of the star			ON	Time (2)	1955	1957	-20
No Holding		A State State of the		60#	Time (2)	1955	2042	422
(SP: DLL J34 BAE J	34.CRL.J584.SLT.F	OMT.EWR*	- Q	<u>8 </u> 011	Time [2	195	2044	40.3
					1	1050	2111	72.0
06/18/1506						190	2029	240
COLUM 1330	1	wind a weat for	201		Time (2)	196	1957	.20
We Walder	يعتر فالفرجي سنيشو معا	2016 - R	2.7 ml	s0#	Time (7	199	2052	526
ASP DI IT 124 RAF	POL HOA ST T FOM	TRUE IN STATE	57		Tree (2		2007	80
1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		1244		1 08	Time (2	1955	2007	8.0
Contract of the local difference of the	ين مينيا (يا المنظمي مي مراجعه) مراجع و در من و رو معمد اليو		anna ann an thairte an	nine and an and a		in an interest		and and a strength of the
1	· · · · · · · · · · · · · · · · · · ·	10 2 10 10 10 10 10 10 10 10 10 10 10 10 10						A
		1						
		A local set the set of the set of the set	All and the second					
and the second state of the second states		and the second sec	CT			in the fact		1000 - 10 - 10 - 10 - 10 - 10 - 10 - 10
TAA Fied be	510	Staron Barton	2			3-1-1-1		
FAA Fled In.								
Actual Danto							~	
Actual Parts								
FAA fiel be Actual Basts								
FAA Fiel In Actual Instein								八

Appendix A.42 ZOB: MSP to EWR Slide 3

Edt Winn - Go - Add						19-1-19			26-24
Fied Route			umber of .	Performan	С.	Planned	Actual	Difference - 1 [Actual - Planned	zik i
MSP: DLL-J34 BAE J70 Ph	M. CRI-JSRASLEFTM	LEWH-SANO	32	ArTime Unco	r. (mm)	123.8	138.8	14.9	12.04
-		-121-20	5	ArTime Unco	r (mm)	122.8	164.4	41.6	319
15/25/1998	Contraction of the			AirTime Unco	r. (mm)	1130	162.0	49.0	43.4
04/26/1998	1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.			AirTime Unco	r. (min)	116.0	159.0	43.0	37.1
05/01/1998	a state of the second secon	12 19 19 19		ArTime Unco	u land	131.0	177.0	46.0	Ē 35.1
04/09/1998	مار بالارجام ، تعالى المحروم مار "ماروا محرو المراجع المراجع المحروم مار "ماروا			ArTme Unco	n. (min)	129.0	169.0	40.0	31.0
05/09/1998				ArTime Unco	r. (con)	125.0	155.0	30.0	24.0
No Holdon			27	AirTime Unco	n (mm)	124.0	134.0	10.0	6.0
MSP DILLING RAF CRI	SHA SI T FOMT FUR			ArTime Unco	n (min)	125.0	136.0	11.0	6.8
- No Holding			3	AirTime Unco	u. (min)	125.0	136.0	11.0	8.8
MSP MKC 124 STI JIA MI	ISOS BRING FAIL BILL	APATRINTIO	ī	AirTime Unco	r (com)	178.0	193.0	15.0	0.4
Linking 7		765 20		ArTme I Inco	r (mm)	178.0	193.0	15.0	E 84
HER DIT 124 PAE 124 CE	DI IEM CI T SOMI EU/	A	15	AirTime Linco	n (mini	1239	1335	9.5	77
MOF-DLLJJ4.0ALJJ4.CF	NL JOB DEL FUMILE OF	<u> </u>		ALC: NO CITES					
The Maldan Street		AT AND IN THE CASE	14	AnTimolines	· (min)	123.9	1290	52	12
andream a surger and a second seco	an tan dagi sebagai ding keter Ketara dagi sebagai ding keter sebagai sebagai sebagai sebagai sebagai sebagai s	ANARASI (MARANA)	en de la composition de la composition Composition de la composition de la comp	د مربور است. مربور است. اندر ا	م مرجعة ما مرجعة مرجعة			గా 1 5 గర్రంగా కి.200 పెట్టాలో జీల్ నిరియ	در از کار در از کار
							9-9-9-2-5 -219-5-9-9-9 -219-5-9-9-9-9-9-9-9-9-9-9-9-9-9-9-9-9-9-9		
							J. Com		

Appendix A.43 ZOB: MSP to EWR Slide 4

Field Routh Control of the second secon	Edit View . Go : Adding & Window Hen 7.47		the Contract	Contraction of the	10.00	100		2000 A 1963	27.00
Field Float Partner Partne Partner Partner		Carlo and a la	ALC: No.	Contraction of the second		COMPLETING THE			10.2
BC DLL 2A BAE / DLPANLOR CALASIS I FOMILEVING (a) 32 As Tree Uncom (nm) 1228 138.8 14.5 12 Hoding 3 As Tree Uncom (nm) 132.0 164.4 41.6 20 JOSZEY/1988 As Tree Uncom (nm) 113.0 152.0 430.0 43.0 10.0 43.0 13.0 11.0 40.0 10.0 43.0 11.0 43.0 11.0 43.0 11.0 43.0 11.0 43.0 11.0 43.0 11.0 43.0 13.0 13.0 15.0	FindRow	1.4	i i san sa	S. S. States		Perred	Actual	A second second	3:30
Heding Source Artime Uncom Imm 1228 I64.4 41.6 30.0 06/22/1998 Artime Uncom Imm 113.0 152.0 480.4 40.6 30.0 10.0 80.0 11.0 80.0 11.0 80.0 11.0 80.0 11.0 80.0 11.0 80.0 11.0 80.0 11.0 80.0 11.0 80.0 11.0 80.0 11.0 80.0 10.0 10.0 10.0 10.0 80.0 10.0 10.0	SPEDIES BASING PARTY IS A SUBJECT OF	I AVI EVEDEC	> 32	ArTime Unco	a. jaun	123.0	138.8	14.9	1204
AP Time Uncost. (min) 113.0 162.0 49.0 43.0 43.0 113.0 112.0 43.0 43.0 37.0 43.0 37.0 43.0 37.0 43.0 37.0 43.0 37.0 43.0 37.0 43.0 37.0 43.0 37.0 43.0 37.0 43.0 37.0 43.0 37.0 43.0 37.0 43.0 37.0 43.0 37.0 43.0 37.0 43.0 37.0 43.0 37.0 43.0 37.0 43.0 10.0 37.0 43.0 10.0 37.0 13.0 17.0 43.0 10.0 90.0 37.0 13.0 17.0 13.0 11.0 80.0 10.0 80.0 10.0 80.0 10.0 80.0 10.0 80.0 10.0 80.0 10.0 80.0 10.0 80.0 10.0 80.0 10.0 80.0 10.0 80.0 10.0 80.0 10.0 80.0 10.0 80.0 10.0 10.0 10.0		SP P AND	> 5	ArTime Unco	it. (min	122.8	164.4	41.6	33.97
Akiras Uncar, (min) 116.0 193.0 43.0 37.0 C05/07/1988 Akiras Uncar, (min) 12.0 183.0 40.0 31.0 C05/07/1988 Akiras Uncar, (min) 12.0 185.0 11.0 0.0 C05/07/1988 Akiras Uncar, (min) 12.0 136.0 11.0 0.0 C05/07/1988 Akiras Uncar, (min) 12.0 136.0 11.0 0.0 C05/07/1988 Akiras Uncar, (min) 12.0 133.0 15.0 8.0 C05/07/1988 Akiras Uncar, (min) 178.0 133.0 15.0 8.0 C05/07/1988 Akiras Uncar, (min) 178.0 133.0 15.0 8.0 C05/07/1988 Akiras Uncar, (min) 12.0 136.0 <t< td=""><td></td><td>-</td><td></td><td>AirTime Unco</td><td>u. (min</td><td>1130</td><td>162.0</td><td>49.0</td><td>43.43</td></t<>		-		AirTime Unco	u. (min	1130	162.0	49.0	43.43
AiTree Uncort, (me) 131.0 177.0 460.35 Oxf001/1998 AiTree Uncort, (me) 123.0 155.0 30.0 24.0 No Hodrigg 3 AiTree Uncort, (me) 123.0 155.0 30.0 24.0 No Hodrigg 3 AiTree Uncort, (me) 123.0 155.0 30.0 24.0 No Hodrigg 3 AiTree Uncort, (me) 123.0 100.0 0 PCDL1234, BAC Chill States ST FRMI Event 3 AiTree Uncort, (me) 123.0 130.0 110.0 0 No Hodrigg 3 AiTree Uncort, (me) 123.0 130.0 110.0 0 SP DUL124, STL 2001, US25, DKW, FAC FRICTLENT, FRV1 P 1 AiTree Uncort, (me) 123.0 150.0 0 No Hodrigg 1 AiTree Uncort, (me) 123.0 150.0 0 0 133.0 150.0 0 SP DUL134, BAE ISACPHLERAR ST FRMI EVEN 0 1 AiTree Uncort, (me) 123.0 150.0 70.0 55.0 70.0 55.0 70.0 55.0 70.0 55.0 70.0 55.0 70.0 55.0 70.0 55.0 70.0 55.0 70.0 55.0 <t< td=""><td>D4/26/1998</td><td></td><td></td><td>AirTime Unco</td><td>n. (min</td><td>116.0</td><td>159.0</td><td>43.0</td><td>37.15</td></t<>	D4/26/1998			AirTime Unco	n. (min	116.0	159.0	43.0	37.15
Ad Time Uncost, fine) 123.0 155.0 30.0 240 No Hotsing, 27 Ad Time Uncost, fine) 123.0 155.0 10.0 8 SP - MCL /2451L 1548L 1525 (BKW, FAX FILL (1447) 189 SP - MCL /2451L 1548L 1525 (BKW, FAX FILL (1447) 189 SP - MCL /2451L 1548L 1525 (BKW, FAX FILL (1447) 189 SP - MCL /2451L 1548L 1525 (BKW, FAX FILL (1447) 189 SP - MCL /2451L 1548L 1525 (BKW, FAX FILL (1447) 189 SP - MCL /2451L 1548L 1525 (BKW, FAX FILL (1447) 189 SP - MCL /2451L 1548L 1525 (BKW, FAX FILL (1447) 189 SP - MCL /2451L 1548L 1525 (BKW, FAX FILL (1447) 189 SP - MCL /2451L 1548L 1525 (BKW, FAX FILL (1447) 189 SP - MCL /2451L 1548L 1525 (BKW, FAX FILL (1447) 189 SP - MCL /2451L 1548L 1525 (BKW, FAX FILL (1447) 189 SP - MCL /2451L 1548L 1525 (BKW, FAX FILL (1447) 189 SP - MCL /2451L 1548L 1525 (BKW, FAX FILL (1447) 189 SP - MCL /2451L 1548L 1525 (BKW, FAX FILL (1447) 189 SP - MCL /2451L 1548L 1525 (BKW, FAX FILL (1447) 189 SP - MCL /2451L 1548L 1558 (BKW, FAX FILL (1447) 189 SP - MCL /2451L 1548L 1558 (BKW, FAX FILL (1447) 189 SP - MCL /2451L 1548L 1558 (BKW, FAX FILL (1447) 189 SP - MCL /2451L 1548L 1558 (BKW, FAX FILL (1447) 189 SP - MCL /2451L 1548L 1548 (BKW, FAX FILL (1447) 189 SP - MCL /2451L 1548 (BKW, FAX FILL (1447) 189 SP - MCL /2451L 1548 (BKW, FAX FILL (1447) 189 SP - MCL /2451L 1548 (BKW, FAX FILL (1447) 189 SP - MCL /2451L 1548 (BKW, FAX FILL (1447) 189 SP - MCL /2451L 1548 (BKW, FAX FILL (1447) 189 SP - MCL /2451L	05/01/1998			AirTime Unca	x. (min	131.0	177.0	46.0	35.12
A Group (1995) A Group (1995) SP DLL 12A BAC DEL (59A ST.) FOMT EVANDON SP DL 12A ST.) FOM	04/09/1998	-11/10 Cr2+5		AirTime Unco	n. (min	129.0	169.0	40.0	31.0
No.1106/fmg. 27 Ar Time Uncost. (mm) 124.0 10.0 0 <td>05/09/1998</td> <td></td> <td>···· · · · · · · · · · · · · · · · · ·</td> <td>AisTime Unco</td> <td>n. (min)</td> <td> 125.0</td> <td>155.0</td> <td>30.0</td> <td>24.0</td>	05/09/1998		···· · · · · · · · · · · · · · · · · ·	AisTime Unco	n. (min)	125.0	155.0	30.0	24.0
SP: DLL 138 BAE: FOLL (SPALS); FOLMT EVMS: 3 AFT me Uncost, (mm) 125.0 136.0 11.0 8 SP: MUCL 225 SIC VIEW (AK, BIC, 164.00, 180.0); (b) 1 AFT me Uncost, (mm) 172.0 133.0 15.0 8 SP: DLL 138 BAE: 134 CPL 258 SUL FOLM (EVM) 0 1 AFT me Uncost, (mm) 172.0 133.0 15.0 8 SP: DLL 138 BAE: 134 CPL 258 SUL FOLM (EVM) 0 1 AFT me Uncost, (mm) 172.0 133.0 15.0 8 SP: DLL 138 BAE: 134 CPL 258 SUL FOLM (EVM) 0 1 AFT me Uncost, (mm) 172.0 133.0 15.0 8 SP: DLL 138 BAE: 134 CPL 258 SUL FOLM (EVM) 0 1 AFT me Uncost, (mm) 172.0 135.0 70.0 55 Holding 0 1 AFT me Uncost, (mm) 125.0 136.0 70.0 55 FT 150 1	Na Holding		27	ArTime Unco	H. (MIN	124.0	134.0	100	0.0
Me Hedding: 3 Ar Tree Uncost. (me) 12.0 18.0 11.0 8 SP: MCL 12625 [LidblaL 525 [Riv, FAX [RL []1527] [BV1 [0]] 1 Ar Tree Uncost. (me) 172.0 133.0 15.0 8. Holding: 0 1 Ar Tree Uncost. (me) 172.0 133.0 15.0 8. SP: DLL /38 BAE /34.CPL /994 SLT FOM! EVP 0 1 Ar Tree Uncost. (me) 123.0 135.0 5.5 7.5 7.0 7.0 7.0 7.0 7.0 7.0 7.0 7.0 7.0 7.0 7.0 7.0 7.0 7.0 7.0 7.0 7.0 8. 7.0 <td>SP. DILLISKEAE. CHLISEKSTI FOMTEWA</td> <td>Ser Ser</td> <td>3</td> <td>ArTime Unco</td> <td>n. (min</td> <td>125.0</td> <td>136.0</td> <td>11.0</td> <td></td>	SP. DILLISKEAE. CHLISEKSTI FOMTEWA	Ser Ser	3	ArTime Unco	n. (min	125.0	136.0	11.0	
SP-INCL/26 STLCHUL252 BWW FAX HIC (162X) FBV/ IO 1 AT Time Uncost, [min] 1780 1330 150 8 SP-DLL/34 BAE (34 CPL / S94 SET, FOM1 E W/R 0 15 Air Time Uncost, [min] 1780 1330 150 8 SP-DLL/34 BAE (34 CPL / S94 SET, FOM1 E W/R 0 15 Air Time Uncost, [min] 1280 1380 700 55 Hoding 0 1 Ar Time Uncost, [min] 1280 1380 700 55 Hoding 0 1 Ar Time Uncost, [min] 1280 1360 700 55 Hoding 0 1 Ar Time Uncost, [min] 1280 1380 700 55 Hoding 0 1 Ar Time Uncost, [min] 1280 1380 700 700 700 700 700 700 700 700 700 7	No Holding		3	AsTime Unco	н. (тал	12.0	136.0		8.6
Holding 1 Ar Time Uncorr. (min) 172.0 153.0 15.0 8.5 7.0 5.5 7.0 7.0 7.0 7.0 7.0 7.0 7.0 7.0 7.0 7.0 7.0 7.0 7.0 7.0 7.0	SP_MKCJ24STLIRIUJSZEBKW, FAK_HICIT4	SEXT. BEVI. K	<u> </u>	AirTime Unico	K. (MA)	178.0	193.0	150	
Bir DLL JA BAC TALDE JABA SLT FORM SLT	Holdry		? <u></u>	AirTime Unco	n (min	1780	1930		
	SP: DILLISA BAE ISA CRIJSBA SLT. FUMT.EWH		S 15	Artime Unco	π. (πε	1 123.9	1345	3.	
	Holding	100 m	21	Ari me Unco	n (me) u	120	196.0	741	306
						1. 13. 14. 5 17. 19. 19. 19. 19. 19. 19. 19. 19. 19. 19			/
					5				
		3. <i>19</i>							
							Ź		

Appendix A.44 ZOB: MSP to EWR Slide 5

APPENDIX B. POET WINDOWS USED IN THE SCENARIOS

Slide	Search Summary	Flight Table	Scatter Chart	Filed Route Table	Flight Route Map	Expanded Filed Route Table	Expanded Filed Route Table with Instance Highlighted	Flight Instance Window	Other
ZNY1	x	x	x						
ZNY 2	X				X	x			
ZNY 3	X				X	X			
ZNY4					X			X	
ZNY 5				ĺ	X			X	
ZNY 6		_			X			X	
ZOB 1		X	X						
ZOB 2			_		X	x			
ZOB 3					X	X			
ZOB 4					X		X	_	
ZOB 5					X	X			
ZID 1	<u>x</u>				X	<u>X</u>			
ZID 2	X				X		x		
ZID 3	X				X		<u> </u>		
ZID 4	X				X		<u> </u>		
ZID 5	<u>x</u>				X		x		
ZTL 1	X	X	<u> </u>						
ZTL 2	X				X		X		
ZTL 3	X				X		X		
ZTL 4	X	X	<u>x</u>						
ZTL 5	<u>x</u>	L		L	X	<u> </u>		<u> </u>	
ZTL 6	X	X	<u>x</u>				ļ	 	
ZTL 7	X	L	L	L	X	<u>X</u>			
ZFW1	<u>x</u>			X	<u>X1</u>				
ZFW 2	<u>x</u>			<u>x</u>	<u>X1</u>				
							l		L

78W1	Y	Y	Y		T			1
ZDWI		<u> </u>						
ZDW Z	<u> </u>	ļ	<u>↓ </u>	<u> </u>	<u> </u>			_
ZBW 3	<u>X</u>			<u> </u>	X			
ZBW 4				<u> </u>			X	
ZBW 5				X			X	Τ
ZBW 6				X			X	
ZDC 1	X	X	X					
ZDC 2				X	X	_		
ZDC 3				x		x		
ZDC 4				X		x		
ZDC 5				X	X			
ZDC 6				X		x		
ZMP1-1	X			X ¹	X			
ZMP1-2	X			X1		x		
ZMP1-3	X			X ¹		x		
ZMP1-4								X2
ZMP2-1	X			X ¹		X		
ZMP2-2	X			X	x			
ZMP2-3				X	X			1
							1	1
		<u> </u>	<u>.</u>			and the second se		-

1: The POET screen capture was modified to include several flight instances in one window,

which POET 1.0 does not support.

2: A Microsoft Excel spreadsheet screen capture was included in this slide to show the delays sorted by arrival fix, which POET 1.0 does not support.
APPENDIX C. THE TRAINING SLIDE SHOW



Appendix C.1 Training Slide 1



Appendix C.2 Training Slide 2



Appendix C.3 Training Slide 3

296



Appendix C.4 Training Slide 4



Appendix C.5 Training Slide 5

297

CSLANT	[Eddor]								
									ED 0
Fiel Ed.	iew Go Add	ne" Window							10
< flight (intance[14753	3511							
Constant Ba	entine Date:	070.941.9		3	100712		and the second	Manual constraints	
1.100.00	Autom:	NWALSHE			(FORTE			2	
Chief Dee	etter Airpert	NSP STOR	PAR PE		WAES.		Service Street		3.74
	inivel Aupent-		1272 - 40		an item				523
		23306-2444	1	1.00					1.05
Performant	Metrice Rau	na triamation							1000
	10 C 10 C 10 C	भट्टा दुरुष		Offician	3.20 k	Constraint Committee with the	alian Alian da alian di seraini		
Perform	Nance Malacs	~Planned) /	Actual Ac		ななよ				
Ar Fu	Bum Uncos (bs)	10,000.0	10,600.0	600.0	6.0%	Nap.			500
• • • • • • • • • • • • • • • • • • •	usitium Con.(Ds) Iotal Fuelflum(Ibs)	10,000,0	10,500,0		0.02				
- Ai	Time Uncon.(min)	82.0	81.0	-1.0	-1.2%	TAA Red In			3 7 11
	wTime Con.(mms)	67.0	81.0	-60	-6.9%	1 All and a lot of the second se			2-II
• • • • • • • • • • • • • • • • • • •	Tasi In(mins)	50	5.0		0.02	AND DECK OF BRIDE	State of the second	1.	3×
	I and Uutiminsj	220	2229	-11.0	-50.02		the state for the	54. 4	26
1	Off Time[2]	2762	2340	.120		West Trace of the second second	1. A . A . A . A . A . A . A . A . A . A	ine State	84 H
1	On Time(2)	0119	0101	-180		2050 00 500 000			2. HF
	In Time(2)	0124	0106	-18.0			Edm. Charles	N	87. H
					4				3. IF
						A Contraction of the	Contraction of the second		2
						The same from the second	Section 20	1	e -
						185° an 1960 (m. 1960)			202
· .									

Appendix C.6 Training Slide 6

ESLANT - [Editor]	
S 11 0 1	
Fie Fill Yow Go Addins Window Hep	Fler Delete Vew Volume Help
Location Smith sch. Summer, Fight 1-1717 - 1717 - 1717 - 1717 - 1717	
Contract of the second se	/1998 To 07/01/1998. DepartTence 1
Emisse Data Mining	
Depart Anvel Departure Annual Elight	Difference
MSP LGA 1220 DSS NWA502 O 20 funiture (her 13140.0.1	4.7530 1.5130 1232 ▲
MSP_LGA 1945 M60 NWA556 🗢 17 FuelBum Uncorr. (bs) 15,664.7 1	6,911.8 1,247.1 8.0X ▲
MSP LGA 1425 D95 NWA504 O 20 FunBurn Uncorr. (bs) 14,945.0 1	5.082.8 1.137 8 7.6x 4
MSP LIGA 1425 DSS NWA504 30 FuelBun Lincor, Red 13 290.0 1	3.976.0 596.0 52% ▲
MSP LGA 2150 320 NWA510 🔷 22 FueBun Uncorr. (be) 13,459.1 1	4.055.9 596.8 4.4% ▲
MSP LGA 1235 320 NWA502 Q 27 FuelBun Uncor. [bs] 12,796.3 1	3,347.4551.1 4.33 ▲
MSP LGA 1905 72S NWA512 C 26 FullBum Uncon, (bs) 20,969,2	
	ananan na ana ana minana ao amin' ao amin' a
THE REAL POST AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY.	
Scient Charles Bar Charles	
XX Am	
	06/16/1500 - 06/21/1908 - 06/24/1908 - 06/27/1908 - 06/20/1904

Appendix C.7 Training Slide 7

298

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

ESLANT - [Editor]					A DIA CONT
			ale des		- HILE
Ele Edi: Yew Eo Addins Window	THOSE AND		Eine Deinte 2V	W Volume Links	調査が三
3 court of an way mught					X
Lincolon: Search Lock: Summary: Eightel:	MSP LGA 1000 725 MWA1265 F	ad Poune/ Diff.			
1977 1 DOOL TO 2400 Designe MSP		میں میں اللہ اور این کہ میں سالہ اس میں میں اللہ اور این کہ میں اللہ اور اس		N. 1997 Co. 1997	
			C6	Anna and a family in the	
Explore Data Mining		Section of the	7.2	al ph. Jacks of the	
NUMBER OF STREET, STRE					
Fied Rode:		and #Actual Actual=	Annual State		A TALK MAN
MSP. DLL J34 BAE J34 CRL JO	10 FuelBurn Uncorr. (bs) 21.03	0.0 22,976.0	1,946.0 8.8% A	·····	
+ No Haiding	8 FuelBurn Uncost. (bs) 21,00	0 0 22,285 0	1,295.0 6.1%		
* MSP.DLLJ34.BAE.J34.CRL.C	7 FuelBurn Uncorr. (bs) 19,81 4 FuelBurn Uncorr. (bs) 21,42	4.3 20,914.3	1,100.0 5.6%		
MSP.SNINE1.GRB.J3B.ECKJ1	2 FuelBurn Uncoll. (bs) 22,20	0.0 21,940.0	250.0 1.2%		·
+ MSP.SNINE1.GRB.J38.ECIC.(1	1 FuelBurn Uncar. (bs) 21,60	0.0 20,800.0	800.0 -1.7%		
	<u>in the state of the second seco</u>	al an con	e de la companya de l	in the second	
	100000 1000 1000 1000 1000 1000 1000 1	CARE AND STATES	1	Sacht Street	
Sector Contraction of the		عمر فتعمد عد المر	and a state of the	Established Anti-	<u> </u>
TAA Tied Breas			a	Contraction of the	- - -
		CAR DE			将这一个
1929 - S. C. Martin	A REAL PROPERTY OF	and the second s	A COLOR	578 W28	
100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100	The second second second		1.1.1		
		T. The second			
				2.4-5-5	
		* * * *	<u></u>	*_*_*_*[*_* <u>}</u> [N

Appendix C.8 Training Slide 8

APPENDIX D. DISPATHCHERS MENTIONING BASIC FLIGHT CLASSIFYING DATA

	DTI	DT2	DVp1	DVp2
ZNY				
City pair	×	~	~	×
Time of day	×	×	×	×
Time of year	×	×	~	×
Number of flights involved	×	×	~	×
ZOB				
City pair	-	X	X	-
Time of day	×	×	×	v
Time of year	×	X	X	-
Number of flights involved	×	×	X	v
ZID				
City pair	×	X	X	•
Time of day	×	×	×	×
Time of year	×	×	X	×
Number of flights involved	×	×	X	×
ZIL				
City pair	×	×	X	×
Time of day	×	×	X	×
Time of year	×	×	X	×
Number of flights involved	×	X	X	×
ZDX				
City pair	•	X	X	•
Time of day	×	×	×	-
Time of year	×	×	×	~
Number of flights involved	×	×	×	~
ZBW				
City pair	×	×	~	•
Time of day	×	X	~	×
Time of year	×	X	×	-
Number of flights involved	•	×	×	×
Number of flights involved	X	×	v	×
ZMPL				
City pair	×	×	v	v
Time of day	×	v	×	×

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

	DT1	DT2	DVp1	DVp2
ZDC				
City pair	•	•	•	v
Time of day	×	×	•	×
Time of year	×	×	~	×
Number of flights involved	×	×	•	×
ZMPL				
City pair	×	×	•	•
Time of day	×		×	X
Time of year	×	×	×	×
Number of flights involved	×	×	X	×
ZMP2				
City pair	×	×	×	-
Time of day	X	×	X	×
Time of year	×	×	X	X
Number of flights involved	×	×	×	X

APPENDIX E. DISPATHCHER QUESTIONS

ZNN Scenario - DTW to EWR	1999 B
Text Mode Pair #1	Code
1. Is filing through ZBW the answer to avoid holds?	C
2. Is the traffic balanced coming into this center's airspace from 3 neighboring centers?	C
3. Would a reroute be an option for flights looking at holding close to DTW?	C
Text Mode Pair #2	Code
1. Could we get a reroute one of the northern alternate routes to avoid holding close to	с
	0.1
voice and Pointing Mode Pair #1	Coae
1. Why were these flights held?	A
2. Why were they held so far from the [destination] airport?	A
3. Could we have filed a different route to the north or south to void this? [a number of holds and some vectoring]	С
4. Was it weather, ATC congestion, or an aircraft incident at the airport?	В
5. Couldn't you hold us on the ground rather than shortly after departure?	C
6. Couldn't we have been given a different routing [to avoid this instance of holding]?	C
Voice and Pointing Mode Pair #2	Code
No direct questions asked	-

ZOB Scenario - MSP to EWR	C. Anna Marca
Text Mode Pair #1	Code
1. Would a departure delay be better?	C
2. Is the holding at SLT due to sector saturation? If so, would alternative flight planning be in order?	В
3. Could we hold closer to DTW rather than SLT to give NWA a better scenario for planning in the event the remaining fuel onboard becomes an issue?	С
Text Mode Pair #2	Code
1. Why are so many over filed air-time?	A
2. Routes over crl slt produce holds that are more significant than grb-yyz routes. Would flying a 757 at a higher altitude help the holding situation?	С
3. Flights over grb-yyz seem to get off ground with less delay. Is it better to fly a longer route to reduce ground delays?	С
4. Why is crl slt producing bottleneck? Is traffic coming from other airports (ord-dtw- cvg) producing the problem and can we avoid it?	В
5. Let's consider change of preferred routing to grb-buf or yyz. Maybe we can eliminate the bottleneck over crl slt and reduce ground delays out of msp. What do you think?	С

Voice and Pointing Mode Pair #1	Code
No direct questions asked	-
Voice and Pointing Mode Pair #2	Code
No direct questions asked	-

ZID Scenario - DTW to RDU	
Text Mode Pair #1	Code
1. The majority were diverted to the south - please explain. [Considered equivalent to "would you please explain why?"]	A
Text Mode Pair #2	Code
1. Are the most extreme cases due to weather ?	В
2. Are these offloads because of arrivals into DFW?	В
3. Would a lower altitude help?	С
Voice and Pointing Mode Pair #1	Code
No direct questions asked	-
Voice and Pointing Mode Pair #2	Code
1. Why taken so far south?	A
2. How can we file a route we are more likely to use?	C

TTL Scenario - DTW to ATL	
Text Mode Pair #1	Code
1. Was the reroute predictable before the flight departed?	D
2. Is it traffic saturation?	В
3. Problems with weather?	В
4. Pilot requested?	В
5. Again, what is the reason for the reroutes and was this information known prior to	C
departure?	
Text Mode Pair #2	Code
No direct questions asked	-
Voice and Pointing Mode Pair #1	Code
1. This flight suggests we are doing everything right as far as the route is concerned	
and the predicted fuel burn, and the issue is what can we do for flights at other times of	Α
day?	
2. Are the choke points here and here, or is it strictly a volume issue in Atlanta?	С
Voice and Pointing Mode Pair #2	Code
No direct questions asked	-

Text Mode Pair #1	Code
No direct questions asked	-
Text Mode Pair #2	Code
Are flights getting rerouted because of Center transitions ?	B
Voice and Pointing Mode Pair #1	Code
No direct questions asked	-
Voice and Pointing Mode Pair #2	Code
1. It looks like Kansas is the pointer where flights get moved to the east. Why is that?	A

ZBW Scenario - DTW to BOS	
Text Mode Pair #1	Code
1. What is the reason for the gap in data ? [In the scatter chart]	D
2. For the most part a 9.2% increase in burn for 103 flights will add up. How can we bring this number down?	A
3. What is the reason this flight took such an extensive reroute? [This was one instance.]	A
4. It seems as if there is to much deviation from filed route on good number of flights. What is the reason for this?	A
5. It appears that there are a few different holding fixes. Why the difference?	A
6. I have noticed that this aircraft has been put into a hold at 2 different locations. What is the reason for this?	A
7. Are we checking to see if aircraft are over water equipped?	D
8. Long ground time. What caused this? [For this instance.]	A
9. Could a ground stop have avoided the hold? [For this instance.]	С
Text Mode Pair #2	Code
No direct questions asked	-
Voice and Pointing Mode Pair #1	Code
No direct questions asked	-
Voice and Pointing Mode Pair #2	Code
No direct questions asked	-

ZDC Scenario - MEM to EWR	
Text Mode Pair #1	Code
1. Would it be possible to move some of the traffic to a more northerly route and out of the DCA area?	С
2. Some flights had several thousand pounds over burn. Was that really necessary?	A
Text Mode Pair #2	Code
1. Is it traffic levels or ATC staffing levels that cause traffic from the south to have less holds?	В
2. Can we plan for the holding? (i.e. plan around it based on the time of day and traffic on that day)	С
3. Is there any airspace available around the DCA area where holding occurs?	D
4. Is the holding due to handoff problems?	В
Voice and Pointing Mode Pair #1	Code
No direct questions asked	-
Voice and Pointing Mode Pair #2	Code
No direct questions asked	-

ZMP Scenario MDW to MSP	
Text Mode Pair #1	Code
1. Why is it that flights departing closer to scheduled departure time have longer delays?	A
2. Would a different time mean less delays?	C
Text Mode Pair #2	Code
1. Is this a specific sector in MSP Center airspace where there are so many off-loadings to the north?	В
2. Is sector staffing or combined traffic an issue?	B
3. Would a more southerly route help your sector controllers?	С
Voice and Pointing Mode Pair #1	Code
No direct questions asked	-
Voice and Pointing Mode Pair #2	Code
No direct questions asked	-

ZMP Scenario - OMA to MSP	
Text Mode Pair #1	Code
No direct questions asked	-
Text Mode Pair #2	Code
No direct questions asked	-
Voice and Pointing Mode Pair #1	Code
No direct questions asked	-
Voice and Pointing Mode Pair #2	Code
No direct questions asked	-

APPENDIX F. IDEAS FOR IMPROVING PERFORMANCE

Code	Category	Dispa	tchers	Traf Mana	fic gers
		T	VP	Τ	VP
Α	New preflight route	11	9	9	10
B	Enroute reroute	2	3	2	0
C	Change schedule	1	1	7	3
D	Ground delay/stop	3	4	1	1
E	Same route but miles in trail	0	1	0	0
F	Change Center staffs' workload / traffic volume	2	0	0	0
G	Share information to enable planning (of the fuel burn, filed route, etc.)	3	3	0	1
H	Hold at a different location	1	1	0	1
I	Change altitude	2	2	0	0
J	Fix balancing	0	1	0	0
K	Other	0	1	0	0
Totals:		26	26	19	16

										ZN	IY											
	ŀ	ł		B		C	I)	I	Ε]	F	(3	I	ł		I		<u>ا</u>	J	<u>s</u>
	D	T	D	T	D	T	D	T	D	Τ	D	Т	D	T	D	Τ	D	Τ	D	Τ	D	T
Text 1	X	X	X	X	· · · · ·						X		X									
Text 2	X	X	X	X																		
VP1	X		X			1	X															
VP 2						<u> </u>									X							

										ZC	B											_
	ŀ	ł	1	B	0	()	1	<u> </u>]]	Ξ]]	F	(G	Ē	1		[l –	I	<u> </u>
	D	Т	D	T	D	Τ	D	T	D	T	D	Т	D	T	D	Τ	D	T	D	Τ	D	Т
Text 1							X								X	X						
Text 2	X	X		_				1					1				X					
VP1	X	X				X											X					
VP 2	X	X												1								

,

										ZI	D											<u> </u>
		1		3		2)	1	Ē		F		G	F	ł		I		ſ	1	C
	D	T	D	T	D	Τ	D	T	D	T	D	Τ	D	T	D	Τ	D	T	D	Τ	D	T
Text 1																						
Text 2													X				X					
VP1		X																				
VP 2	X																			[
										ZT	Ľ											
	I	ł	1	B		2	I)	1	Ē		F		G	I	I		I		J	I	<
	D	Τ	D	Τ	D	Τ	D	T	D	Τ	D	T	D	T	D	T	D	T	D	T	D	Τ
Text 1						X			X				X									
Text 2	X																					
VP1		X	I										X								X	
VP 2	X	X			X	X			X				X				X		X			

										ZF	W										_	
		ł		8		2)	1	E		F		G	I	ł		ľ		J	F	ĸ
	D	T	D	Τ	D	T	D	T	D	Τ	D	T	D	T	D	T	D	Τ	D	T	D	Т
Text 1	X	X																				
Text 2	X	X				X		-		[<u> </u>								
VP1	X	X								—			Î		1					[
VP 2	X	X	X		<u> </u>	1		I	t —				1	X	t		1					

										ZB	W											
	F	ł	1	3		C	1	<u> </u>	1	E		F		G	I	H		[[ŀ	<
	D	Τ	D	Τ	D	T	D	T	D	T	D	T	D	T	D	Τ	D	T	D	Τ	D	Т
Text 1							X				-				1							
Text 2						X					1											
VP 1	X							1				1										
VP 2		[†		X	[1		<u> </u>	1							

										ZD	C											
	A	ł	I	B		2	Ī)	Ĩ	Ē		F		3	H	ł	1	[I	I	$\overline{\mathbf{C}}$
	D	Τ	D	T	D	Τ	D	Τ	D	T	D	Τ	D	T	D	Τ	D	T	D	T	D	T
Text 1	X	X				X																
Text 2	X	X																				
VP1		X								[
VP 2	X	X	[X													

	_									ZM	P-1											
	A	ł	1	B		2	Ι	D		E]	F		3	F	Ŧ		[ſ	F	<
	D	Т	D	T	D	T	D	Τ	D	T	D	T	D	Τ	D	Τ	D	T	D	Τ	D	Τ
Text 1	X	X			X	X		Γ														
Text 2	X										X											
VP1		X				X	X	X	[
VP 2			X																			

	_									ZM	P-2	· · · · ·										
	ŀ	1	1	B	(2		2	1	Ē		F		3	F	I		[J	F	<u><</u>
	D	T	D	T	D	Т	D	Τ	D	T	D	T	D	T	D	T	D	Τ	D	T	D	T
Text 1	X	X				X																
Text 2						X		X														
VP1													X									
VP 2													Γ								Ţ	

-	Text Mode Pair #1				
Text M					
Code	Dispatcher	Code	Traffic Manager		
Α	1. Is filing through ZBW the answer to avoid holds?	A	(Response) The traffic manager indicates holding is less likely through ZBW and that the approach would be more direct.		
F	2. Is the traffic balanced coming into this Center's airspace from 3 neighboring centers?		(Response) "There is a geographical balance." The traffic manager also indicates handoff holds probably due to needing to space in a complex sector.		
В	3. Would a reroute be an option for flights looking at holding close to DTW?		(Response) The traffic manager referred the dispatcher to ZBW to answer the question as to whether a reroute would have been possible for this flight instance.		
G	4. Knowing when holding is occurring or going to occur would help the airline file routes.		(Response) The traffic manager indicated that backing up occurs almost immediately.		
		В	(Suggestion - refinement of reroute idea) Suggests if data indicates holding is likely to be at least 15 minutes it would be better to reroute to the north.		

Text Mode Pair #2				
Code	Dispatcher	Code	Traffic Manager	
A	 It does not appear that flights that were filed on an alternate route suffered extensive holding or vectoring. Dispatchers can accommodate this better. Could we get a reroute to one of the 	А, В	(Response) The traffic manager agrees with the idea of filing a northern route and also rerouting onto a northern route when holding close to DTW. He indicates rerouting before reaching the ZOB/ZNY border would be	
В	to DTW?		preferable.	
			(None city pair specific suggestion) The traffic manager indicated that if the Center understood more about fuel burns it would help them evaluate the impact on the airlines when experimenting with strategies such as tunneling and reroutes.	
Voice a	nd Pointing Mode Pair #1			
Code	Dispatcher	Code	Traffic Manager	
Α	1. Could we have filed a different route to the north or south to void this? [a number of holds and some vectoring]		(Response) The traffic manager answers this question by indicating more information is needed to determine the reason for holding on each day.	
D	2. Couldn't you hold us on the ground rather than shortly after departure?		(Response) The traffic manager indicates that the particular hold was probably due to problems getting traffic into EWR.	
В	3. Couldn't we have been given a different routing [to avoid this instance of holding]?		(Response) The traffic manager indicates that in this case the holding was so extreme it probably would not have made a significant difference to have flown a northerly route.	

Voice a	Voice and Pointing Mode Pair #2					
Code	Dispatcher	Code	Traffic Manager			
Н	1. Seems better to hold in Cleveland than the New York Center		(Response) The traffic manager responded that more information would be needed to determine if it is better to hold in Cleveland than the New York Center, in terms of air-time, and that for specific instances it depends on the miles in trail applied as well as the varying skill level of the particular controller taking people out of a hold. (Response) The traffic manager indicated that in the New York Center the holding altitude is lower than in the Cleveland Center so more fuel would be burned in New York airspace when holding.			

ZOB Scenario - MSP to EWR					
Text Mode Pair #1					
Code	Dispatcher	Code	Traffic Manager		
D	1. Would a departure delay be better?		(Response) "ZNY uses SLT as an out		
н	2. Could we hold closer to DTW rather than SLT to give NWA a better scenario for planning in the event the remaining fuel onboard becomes an	(ref) H	all the time, maybe they can hold closer to the airport."		
Tert M	issue:	l			
Code	Dispatcher	Code	Traffic Manager		
I	1. Routes over crl slt produce holds that are more significant than grb-yyz routes. Would flying a 757 at a higher altitude help the holding situation?	A	(Response) "TYPE AIRCRAFT AND ALTITUDE WILL NOT HAVE AN IMPACT ON YOUR DELAYS HERE. ROUTING OVER YYZ THROUGH ZBW VS ZNY WILL REDUCE YOUR DELAYS. THE QUESTION IS, ARE YOU ILLING TO ACCEPT THE COST OF FILING THIS ROUTE FOR ALL FLIGHTS SURING TIS TIME."		
A	2. Flights over grb-yyz seem to get off ground with less delay. Is it better to fly a longer route to reduce ground delays?	A	(Response) "The longer segment will reduce the delay, bcause the flight over GRB does not fly through busy areas until over CRL this may reduce enroute delays."		

r		·	
	3. Let's consider change of preferred		(Response) "THIS ROUTING WILL
	routing to grb-but or yyz. Maybe we		KEDUCE YOUR ENROUTE DELAYS
	can eliminate the bottleneck over crl		(CRL-SLT BOTTLENEK), BUT NOT
A	slt and reduce ground delays out of	A	GROUND DELAYS BECAUSE OF
	msp. What do you think?		THE ARRIVAL RATE AT EWR FOR
			YOUR PROPOSED ETA,
			REGARDLESS OF ROUTE."
Voice an	nd Pointing Mode Pair #1		
Code	Dispatcher	Code	Traffic Manager
	1. If weather isn't the issue, the		(Response) The traffic manager agrees
	northerly route looks like it will run		this would be a better route as long as
	into less traffic.		the arrival times are scheduled to
n		^	avoid international flights
			approaching New York from the
			north-east.
T T	2. Another solution might be to fly at		The traffic manager does not respond
	different altitudes.		to this suggestion directly.
	3. A different preferred route on bad		(Response) The traffic manager
	weather days might help.		considers the northern route to
		A	actually be one that should be the
			preferred route.
			(Suggestion) The traffic manager
1]	suggested changing the schedule, in
			cooperation with the other airlines, to
			avoid exceeding the airport's capacity
			excessively and to avoid flight time
			"padding".
Voice an	nd Pointing Mode Pair #2	•	
Code	Dispatcher	Code	Traffic Manager
	1. If we filed on a northern route it		(Response) The traffic manager agreed
	might be better.	1	a northern route through the Toronto
	_	;	and Boston Center should be
			selectively used each day depending
			on the runway configuration and
			Contract in the contract of the second
1 I			Continental Airlines arrival traffic, so
			that the long holding on the current

	ZID Scenario	DIW	ORDU
Text M	lode Pair #1		
Code	Dispatcher	Code	Traffic Manager
	No performance improving		(Suggestion) The traffic manager's
	ideas/questions		advice was to stay on the APE route
			because recent re-stratification may
			help, and while there are altitude
			restrictions, there is mit over ROD and
			that route would be longer.
Text M	lode Pair #2		
Code	Dispatcher	Code	Traffic Manager
	1. If you believe this will continue, let		(Response) The traffic manager wrote,
G	me know so we can try and plan for		"As far as filing lower altitudes and if
	this situation.	(***	you should reroute those flights over
	2. Would a lower altitude help?		ROD VORTAC on a permanent basis,
I	_		you would be better served to direct
			those questions to Cleveland Center."
Voice a	and Pointing Mode Pair #1		
Code	Dispatcher	Code	Traffic Manager
	No performance improving		(Suggestion) The traffic manager's
	ideas/questions	(ref)	advice was to leave it as it is "unless
		A	you can help us get more control of the
			military airspace".
Voice a	and Pointing Mode Pair #2		
Code	Dispatcher	Code	Traffic Manager
	1. How can we file a route we are		(Response) The traffic manager
Δ	more likely to use?		responded that ZID concurs with ZOB
л			that the offloading when APE is
			congested is the best strategy.

Text M	Text Mode Pair #1					
Code	Dispatcher	Code	Traffic Manager			
D	1. Use of short ground delay for departures would help solve the wasted fuel enroute.	С	(Response) "typically at atl ground delay prgms are not utilized unless wx, & aar rates are low.(ex in the 68 - 72 range or lower.) the normal process is to utilize expanded mit restrictions & perhaps short term ground stops to facillitate traffic flows. most of the time the issue is one of compacted demand where a large quantity of a/c are schedule in a short period of time. these aircraft typically will have to spead out over the whole hour . it is possible that an earlier or later dept time could reduce the amount of holding & fuel burn for you."			
G	2. Again, what is the reason for the reroutes and was this information known prior to departure. If this information is know prior to the departure it is paramount that this reroute information be passed to the dispatcher in order to comply with necessary regulations.		(Response) "during this time frame the ne arvl has significant vol & the mw arvl has limited vol. as a general the idea is to try to allow as many as possible flts to fly on their filed route. normally a decision is made in the vicinity of cvg. at this time a decision will be made as to what route works best for vol & tfc flows."			
		A	(Suggestion) "most specialist prefer to have the a/c fly his filed route and will normally wait to see if they can allow that to occur. if not the a/c is routed via the nw arrivaf fix. this process may not work well for your operation, but is normally done. the reason most specialist do this is because they dont want to move the a/c unless it is necessary. plese advise if your preference is to be moved on the ground. if so you can expect that to happen most of the time due to the vol over the ne arvl fix."			
I	1. Perhaps a lower altitude restriction.		(Response) The traffic manager commented there is an altitude restriction for the ChooChoo arrival, but not Rome, or Macey.			

Text M	Text Mode Pair #2					
Code	Dispatcher	Code	Traffic Manager			
	1. "Would it be possible keep our		(Response) "We do tend to favor the			
	flights away from the rome arrival,		Macey arrival fixby this I mean we			
	during hub times at ATL, and avoid		will impact the other arrival fixes			
	some of these delys we are		earlier with increased in-trail or			
	experiencing?"		holding allowing the northeast fix to			
			continue inboundthis allows the ATL			
			airport to perform at optimum			
			capacityEarly impact or degradation			
		Ì	of the flow over Macey will always			
			result in extended periods of holding,			
A			much longer than those over the other			
			3 fixes. It becomes nearly impossible to			
			recover from holding at Macey and			
l		ļ	increase the reed rate into A1Lrull			
			airport capacity is never realized after			
			allow Massy to pup as long as possible			
			abow Macey to run as long as possible			
			duramically vorsus static same flight			
			resources in order to not impact one			
			user any more than another "			
Voice a	nd Pointing Mode Pair #1	L	user any more char anothers			
Code	Dispatcher	Code	Traffic Manager			
	1. This flight suggests we are doing		(Response) The traffic manager			
	everything right as far as the route is		commented that 8am. 3pm. and 5pm			
	concerned and the predicted fuel		are rush periods, with more traffic			
К	burn, and the issue is what can we do		over Macev and Rome, so the filed			
1	for flights at other times of day?		route is normally not possible during			
			this time, due to fix balancing, but			
ļ		ļ	possible during other times.			
	2. We should be advised of changes to	[(Response) The traffic manager			
	the route so we can plan for fuel		indicated that if the airline filed the			
	burns. If we can get information from		Nashville-Rome arrival rate during			
	you that flights will hold here that	ł	heavy pushes they would get it and			
6	would be very helpful.	Δ	could plan accordingly, but it would			
			be longer than the Macey route [which			
			they sometimes flew]. [Thus, he didn't			
			acknowledge flexibility at the flight			
			instance level, but the flight level for			
L	l	L	this city pair.]			

Voice a	Voice and Pointing Mode Pair #2				
Code	Dispatcher	Code	Traffic Manager		
I	1. Perhaps a lower altitude restriction.		(Response) The traffic manager commented there is an altitude restriction for the ChooChoo arrival, but not Rome, or Macey.		
A	2. Perhaps a different preferred route out of DTW, coordinated with Cleveland and Chicago Centers	A	(Response) The traffic manager suggested exploring a route over Charleston, J186, and then J145, because it would not involve as much holding or altitude restrictions.		
С	3. Looking at departure times, it look s like we should talk to the airport to see what arrival rate they can handle and consider spacing our departures differently.	с	(Response) The traffic manager responded on slide 2 that the 8 o'clock arrival push into Hartsville was shown and on slide 6 that it was the 2 to 3 o'clock rush period.		
D	4. A delay program could have helped to improve spacing.		[The traffic manager did not respond to the idea of a delay program directly.]		
G	5. Perhaps you could publish routes that we could file the flights on and plan the fuel burn, and spread them out.	A	(Response) The traffic manager explained when the three currently flown routes from DTW to ATL would be expected to be flown and proposed		
A	6. Perhaps we should file the route most often actually flown		a new route as worth exploring.		
J	7. Perhaps you could balance them on one arrival fix.		(Response) The traffic manager commented that the holding that took place is to balance workload and the arrival feed. He also explained in more detail the traffic flow into Rome and Macey at different times of the day.		

Text Mode Pair #1					
Code	Dispatcher	Code	Traffic Manager		
A	1. More often than not our flights are getting rerouted. We need to know if we should file a different route to help planning.	A	(Response) Our recommendation is to file arrivals over the arrival fix (UKW/BYP) that is working the least arrival load at that time. [It isn't clear if he means based on historic or real- time data.]		

Text M	lode Pair #2		
Code	Dispatcher	Code	Traffic Manager
A	1. Perhaps we should work a new route or file the east arrival.	A, C	(Response) Based on the historical data presented it would appear better to change the filed route to file over BYP rather than UKW. In conducting further analysis it would be useful to also consider which side of the airport the gates for the airline are on; what the taxi-in and airborne fuel burns are for the two cornerpost approaches; and changing the arrival time by 10 or 15 minutes.
Voice a	nd Pointing Mode Pair #1		
Code	Dispatcher	Code	Traffic Manager
A	1. We would prefer a route to Kansas City and then somewhere near Fort Smith for a Dallas arrival		(Response) Fort Smith would be a problem because you will run into heavy traffic coming from the northeast there.
A	Of the generic routes the one to the east over Kansas City to Tulsa would probably be the best for us. [This generic route is not the route most often filed.]	A	(Response) Routes through Tulsa are best, because that allows us to fix balance on the northwest and northeast corners of DFW and thus get you in as soon as possible that way. If we have a choice we will bring you in on the UKW (northwest) side because you park there, but there are times when there is a lot of traffic coming into DFW from the southwest through UKW.
Voice a	nd Pointing Mode Pair #2		
Code	Dispatcher	Code	Traffic Manager
В	1. We should annotate on the flight plan when a flight is fuel critical because of the dogleg from Tulsa to DFW, so that perhaps we could get a more direct route into DFW.		(Response) A more direct route from Tulsa is not possible because that is a departure sector.
A	2. If this is going to happen on a regular basis, perhaps I need to establish that this is the preferred route.	A, (ref) G	(Response) There are problems with the western route because of traffic from the west and a more easterly route would run into traffic problems at Fort Smith. You might trying contacting the Kansas Center each day and try to get the heads up as to which way the switch has been thrown for traffic filed over Tulsa and then file accordingly.

	ZBW Scenario	EDIN	to BOS
Text M	lode Pair #1		
Code	Dispatcher	Code	Traffic Manager
ĺ	1. Could a ground stop have avoided		(Response) "based on the data
	the hold? [For this instance.]		provided if the ground delay was not
			airline initiated it probably was an faa
			ground delay program. the lengthy
			airborn hold may have been a result of
ł			a reduced acceptance rate that
			occurred while the aircraft was
			enroute" "without all the data in front
			of me the only thing i can say is based
			on your data there was a ground delay
			program in effect which explains the
}			ground and airborn holding. normally
			when airborn delays exceed 20 minutes
			we will implement a ground stop"
Text M	lode Pair #2		······································
Code	Dispatcher	Code	Traffic Manager
	No performance improving		(Response) "Boston is unique because
	ideas/questions		of the meter program filing different
			routes will not necessarily be
			beneficial because delay are
			programed by a computer to make
			them equitable." "Ground Delay
			Programs are necessary to avoid
			overloading airports. The idea is to
		C	give the aircraft most of the delay on
			the ground to avoid costly airborne
			holding." "There are many holding
			patterns along a specified route. A
			possible reason for this holding is
			airlines usually schedule many flight
1			to arrive at a certain airport at a
			certain time due to there fix & hub
			technique."
Voice a	and Pointing Mode Pair #1		
Code	Dispatcher	Code	Traffic Manager
	1. Perhaps a different preferred route		(Response) The traffic manager
	to the north or south might cause us to		recommended staying on the current
	run into less traffic problems		filed route because arriving via Scupp
A			or Providence would require crossing
			departing traffic out of BOS, and the
			metering that is used should mean the
			arrivals operate on a first-come, first-
		1	served basis anyway.

Voice a	Voice and Pointing Mode Pair #2				
Code	Dispatcher	Code	Traffic Manager		
D	We probably should hold on the ground in these situations where we are experiencing extensive holds.		(Response) My understanding is that the airlines would prefer the holding in the air than on the ground in case the weather has moved on when the flight gets there. "We operate on the basis that if the weather is already there we hold on the ground".		

	ZDC Scenario	MEM	DEWR		
Text Mode Pair #1					
Code	Dispatcher	Code	Traffic Manager		
A	1. Would it be possible to move some of the traffic to a more northerly route and out of the DCA area?	A	(Response) It is viable to move traffic through ZOB into ZNY, but a more northerly route through ZDC is not viable because of pref routes or departures out of ZNY.		
		С	(Suggestion) Consider changing the departure time or the aircraft type to reduce the number of flights needed.		
Text M	lode Pair #2				
Code	Dispatcher		Traffic Manager		
А	1. Can we plan for the holding? (i.e. plan around it based on the time of day and traffic on that day)	A	(Response) "One could always expect to hold if your route is files via the northern arrival fixes when the direction of landing is on RWY22 and the southern arrival fixes when the arrival RWY04. Times to expect enroute and arrival fix holding at these fixes is historical data that can be readily accessed from the ATCSCC or the arrival center's TMO."		
A	2. Is there airspace available around this obvious bottleneck?		(Response) "unfortunately, HPW/PXT/RBV is the corridor for all arrivals to EWR from ZID/ZTL/ZJX/ZDC/ZMA/ZSA."		
		A	(Suggestion) "you might want to check with ZTL to see if they would entertain a more direct route that would perhaps overfly GSO J14 RIC WARRD arvl to EWR."		

Voice a	Voice and Pointing Mode Pair #1					
Code	Dispatcher	Code	Traffic Manager			
	No performance improving ideas/questions	A	(Suggestion) "The southern and northern routing [relative to Beckley] would be less likely to hold."			
Voice a	and Pointing Mode Pair #2	<u> </u>				
Code	Dispatcher	Code	Traffic Manager			
E	1. I wonder if we could slow down the flight and have a smoother transition.		(Response) The traffic manager responded that the route itself was the problem as traffic from that route must be merged with an already established flow and would therefore be held for spacing.			
A	2. I wonder if we could get in without a delay by taking a north west approach.		(Response) "Generally, if you come up the east side the chance of coming in with delay are less than if you come in off the west side." [The traffic manager meant the south rather than the east, which was clear because of his synchronized speech and pointing.]			
		A	(Suggestion) The traffic manager indicated the best route would be over Liberty to the south of the currently filed route because it would be much easier to merge with other traffic moving up the northeast corridor.			
		Α	(Suggestion) The traffic manager proposed a new route coming off the currently filed route to join the route through ZOB when runway 4 is in use rather than 22 when the route through the corridor should be used.			

	ZMR Scenario - MDW to MSP					
Text Mode Pair #1						
Code	Dispatcher	Code	Traffic Manager			
С	1. Would a different schedule mean less delays?		(Response) "Most likely, but I don't have that data with me."			
A	2. Is it a time problem only and not dependent on the arrival fix?	с	(Response) "I would like to see the evening arrival schedule spread out over a 3.5 hour period starting at least .5hr earlier. The airport can only handle 60/hr. Put the problem back to marketing."			
		A	(Suggestion) Consider filing MDWALO.KASPER.MSP This route has reduced traffic. Also, consider MDWRST.KASPER.MSP.			
Text M	lode Pair #2					
Code	Dispatcher	Code	Traffic Manager			
F	1. Is sector staffing or combined traffic an issue?		(Response) " This doesn't look like accurate information. I very rarely see delays of more than 14 minutes in metering and most of these are over			
A	2. Would a more southerly route help your sector controllers?		15. I don't experience this without weather or some other unforeseen event." "These must be weather situations. Standard traffic and metering wouldn't create delays of 25-40 minute."			

Voice and Pointing Mode Pair #1					
Code	Dispatcher	Code	Traffic Manager		
П	1. Maybe we should look at a ground	D	(Response) The traffic manager		
	delay if its all the time.	U	agreed ground delays could be used.		
			(Suggestion) The traffic manager		
			commented that if the airline didn't		
}			want to see ground delays another		
		A	approach would be to offload some		
			of the other Northwest flights		
			arriving at this time from the east		
ļ			coast onto a northern approach		
		}	(Suggestion) The traffic manager first		
			commented, "I agree with you this is		
		C	a problem, but it's a Northwest		
			problem. If you change your		
		ļ	schedule by a few minutes you		
			would miss the rush."		
		1	(Suggestion) "Most of the time the		
			flights are taking this busy route over		
			TWINZ. If we offloaded the flights		
			and go up to a northern arrival over		
			Branard or Edward Fields, or a south		
			arrival, we could avoid the busy		
			route at this arrival over TWINZ with		
			fix balancing, because the airport is		
			only busy on this side of the airport at		
			that time of day."		
			(Suggestion) "These routes at this		
			time of day, and the kind of delays		
		Į	we are seeing nere, are a part of		
			marketing - trying to get all the		
		A	rughts here at the same time in a hub		
			and spoke system. If now control		
			Could have more control or these		
		ļ	nights and onload at will we could		
Vaia	d Deintine Mede Dein #2	<u>i</u>	probably take care of these delays.		
Code	nu rointing Nioue rair #2	Code	Traffic Managar		
Loue	1 Partial suggestions would be to	Loue	(Pernonse) The traffic manager		
	offload some of these to the aclars?		responded by stating that if his		
1	arrival to reduce encourts and holding		memory served him correctly the		
R	time		time period involved was one where		
	ullic.		MSP was experiencing runway		
			construction which reduced the		
		l	arrival rate by 10 to 16 an hour		
			arrivariate by to to an noul.		

	ZMP Scenario-OMA to MSP					
Text M	Text Mode Pair #1					
Code	Dispatcher	Code	Traffic Manager			
Α	1. Perhaps we need a new preferred route to avoid this rerouting.	A	(Response) "if on a runway 12 config the flight could be reouted over rwf to avoid sequencing and yes there would be a delay but it would be minimal because we could make that your only delay"			
		с	(Suggestion) "if this flight were probably schedule a little later in the rush the delay would be greatly reduced""			
Text M	lode Pair #2					
Code	Dispatcher	Code	Traffic Manager			
	No performance improving ideas/questions.	С	(Suggestion) "from the data you present thes effected flights are departing during a busy inbound rush on the kaspr arrival. have you considered moving these departure times either ahead or back to miss the major part of the rush."			
		D	(Suggestion) "would it be helpful to consider more ground delays at oma during periods we know there is extensive delays. it is currently my understanding that nwa wants us to get the flights off the ground if there are expected delays of 30 minutes or less."			
Voice a	and Pointing Mode Pair #1					
Code	Dispatcher	Code	Traffic Manager			
G	1. My contention is that not letting the airline know it needed to change the route earlier lead to a high cost of the extra air time. [He was referring to a single flight instance which was rerouted.]		(Response) The traffic manager agreed this shouldn't happen, but pointed out it was a single instance and there must have been special circumstances, such as a weather event.			
Voice a	Voice and Pointing Mode Pair #2					
Code	Dispatcher	Code	Traffic Manager			
	No performance improving ideas/questions.		(Response) The traffic manager commented that the delays were due to volume. She added, "we normally would ground stop aircraft that are going to be held more than 30 minutes, so I suspect some of the delay was for the approach".			

APPENDIX G. DISPATCHERS' PROPOSED ROUTE DESCRIPTIONS

Dispatcher	Description of proposed route in comments	Pen/arrow marks for the route	Direct reference
ZNY Text 1	A filing through ZBW to avoid	Yellow pen	
	holding on the current route		
ZNY Text 1	A reroute to avoid holding close to DTW	None	
ZNY Text 2	An alternate route [sometimes	2 red arrows [on	
	used] where there was not	one of the	
	holding or vectoring	alternate routes]	
ZNY Text 2	A reroute to one of the northern routes to avoid holding at DTW	None	
ZNY V&P 1	A route to the north or south of this area where holding occurs	None	Traced a path through a northerly route while speaking
ZOB Text 2	A longer route than the current route, such as grb-yyz, to reduce ground delays	None	FB
ZOB V&P 1	This northerly route because it appears to run into less traffic	None	Did not point at the route
ZOB V&P 2	A northern route because it didn't appears to have holding	None	Did not point at the route
ZID V&P 2	A filed route that can be flown without being rerouted consistently	None	Did not point at the route [but he probably didn't have one in mind or was being sarcastic about not being able to fly the filed route.]
ZTL Text 2	A route not involving the rome arrival because of the holding there	None	

ZTL V&P 2	A different preferred route on a	None	Traced a route out
	lower altitude out of DTW.		of DTW to the east
	coordinated with Cleveland		[which was not
	and Chicago Centers		mentioned
			verbally.]
ZTL V&P 2	The route most often actually	None	Traced an average
	flown		route to the west
			for the routes
			actually often
			flown
ZFW Text 1	A different route because we	None	
	are normally rerouted		
ZFW Text 2	A new route or the east arrival	None	
ZFW V&P1	Ideally a route through Kansas	Yellow pen and	Drew the route
	City to Fort Smith	4 black arrows	with the pen while
			speaking
ZFW V&P 1	From the routes currently	Red pen and 5	Drew the route
	flown, Kansas City to Tulsa.	black arrows	with the pen while
			speaking
ZFW V&P 2	A more direct route than the	Red pen for the	Drew the change
	dogleg from Tulsa to DFW	approach into	in route with the
		DFW1	pen while
			speaking
ZFW V&P 2	This route we are actually	None	Traced the route
	flying after being deviated from		of the average
	Kansas City		path within the
			collection of
			rerouted flight
			instances
ZBW V&P 1	A different preferred route to	None	Traced a route to
	the north or south to avoid		the north and to
	holding over Gardner		the south quickly,
			perhaps
			suggesting they
			were very
			approximate
ZDC Text 1	A more northerly route out of	None	
	the DCA area		
ZDC Text 2	A flight plan to avoid	None	
	predictable holding		
ZDC V&P 2	A north west approach on a	A red arrow	Pointed at the
	route currently used with no		filed northern
	shown holding, rather than one		route, but didn't
	from the south west		trace it
ZMP-1 Text 1	A route through a different	None	
	arrival fix		
ZMP-1 Text 2	A southerly route to help sector	None	
	controllers		
ZMP-1 V&P 2	A route through the oclare3	None	Did not point at
	arrival to reduce the enroute		the oclare4 route,

	and holding time associated with oclare4		but was busy placing a yellow arrow on the oclare3 route while making this comment
ZMP-2 Text 1	A new preferred route to avoid this rerouting	None	